

# QUALITY MONITOR HEALTH QUALITY ONTARIO

2011 REPORT ON ONTARIO'S HEALTH SYSTEM

The quality of our health system is the responsibility of every Ontarian. We hope this report will help you understand the publicly funded health system better, and give you the information you need to keep up pressure for improvement.

After all, it's your health and your health system.

This report is prepared in partnership with



Institute for Clinical Evaluative Sciences

# QUALITY MONITOR: TABLE OF CONTENTS

ality improvement summary ributes framework6ummariessummariesspital sector summary9mary care summary12me care summary14ng-term care summary16summariessummariesrdiovascular disease summary18betes summary19ncer summary20ntal health summary21ta advocacy22SIBLE24it times in emergency departments24cess to primary care28atment wait times and access to specialists30cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40cronic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care53spital infections56systal infections56systal infections56ga afety in long-term care64ciding harm in home care67IT-CENTRED70tient experience in acute care hospitals and67tient experience in acute care hospitals and67	1	INTE	RODUCTION AND SUMMARIES	2
ributes framework 8 ummaries spital sector summary 9 mary care summary 12 me care summary 14 ng-term care summary 16 summaries rdiovascular disease summary 18 betes summary 20 ncer summary 20 ntal health summary 21 ta advocacy 22 SIBLE 24 it times in emergency departments 24 cess to primary care 28 atment wait times and access to specialists 30 cess to long-term care and home care 36 TIVE 40 ceiving the right treatments in hospital 40 ronic disease management 44 tentially avoidable hospitalizations 48 eping people healthy in long-term care 52 eping people healthy in long-term care 53 spital infections 56 verse events in acute care hospitals 60 rtality in hospitals 62 ug safety in long-term care 64 biding harm in home care 77 IT-CENTRED 70 tient experience in acute care hospitals and ergency departments 70		1.1	Executive summary	2
spital sector summary 9 mary care summary 12 me care summary 14 ng-term care summary 16 summaries rdiovascular disease summary 18 betes summary 19 ncer summary 20 ntal health summary 21 ta advocacy 22 SIBLE 24 it times in emergency departments 24 cess to primary care 28 atment wait times and access to specialists 30 cess to long-term care and home care 36 TIVE 40 ceiving the right treatments in hospital 40 ronic disease management 44 tentially avoidable hospitalizations 48 eping people healthy in long-term care 52 eping people healthy in long-term care 53 spital infections 56 spital infections 56 cress events in acute care hospitals 60 rtality in hospitals 62 us safety in long-term care 64 piding harm in long-term care 67 <b>TI-CENTRED 70</b> tient experience in acute care hospitals and ergency departments 70		1.2	Quality improvement summary	6
spital sector summary 9 mary care summary 12 me care summary 14 ng-term care summary 16 summaries rdiovascular disease summary 18 ubetes summary 19 ncer summary 20 ntal health summary 21 ta advocacy 22 SIBLE 24 it times in emergency departments 24 cress to primary care 28 atment wait times and access to specialists 30 cress to long-term care and home care 36 TIVE 40 ceiving the right treatments in hospital 40 ronic disease management 44 tentially avoidable hospitalizations 48 eping people healthy in long-term care 52 eping people healthy in home care 53 spital infections 56 spital infections 56 spital infections 56 rtality in hospitals 62 rtality in long-term care 64 piding harm in long-term care 67 TI-CENTRED 70 tient experience in acute care hospitals and ergency departments 70		1.3	Attributes framework	8
mary care summary12me care summary14ng-term care summary14ng-term care summary16summaries18ibetes summary19ncer summary20ntal health summary21ta advocacy22SIBLE24it times in emergency departments24cess to primary care28atment wait times and access to specialists30cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40conic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care56spital infections56verse events in acute care hospitals60rtaity in hospitals62ug safety in long-term care64piding harm in long-term care67TC-CENTRED70tient experience in acute care hospitals and ergency departments70		Secto	or summaries	
Image care summary14ng-term care summary16summaries18rdiovascular disease summary19ncer summary20ntal health summary21ta advocacy22SIBLE24it times in emergency departments24cress to primary care28atment wait times and access to specialists30cress to long-term care and home care36TIVE40cress to long-term care and home care36cress to long-term care and home care56spital infections48eping people healthy in long-term care56spital infections56verse events in acute care hospitals60rtaity in hospitals62ug safety in long-term care64olding harm in long-term care66olding harm in home care70tient experience in acute care hospitals and70		1.4	Hospital sector summary	9
ng-term care summary16summariesrdiovascular disease summary18abetes summary19ncer summary20ntal health summary21ta advocacy22SIBLE24it times in emergency departments24cess to primary care28atment wait times and access to specialists30cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40ronic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care56spital infections56verse events in acute care hospitals60rtaity in hospitals62ug safety in long-term care64oiding harm in long-term care67TCCENTRED70tient experience in acute care hospitals and70		1.5	Primary care summary	12
summaries rdiovascular disease summary 18 betes summary 19 nacer summary 20 ntal health summary 21 ta advocacy 22 SIBLE 24 it times in emergency departments 24 cess to primary care 28 atment wait times and access to specialists 30 cess to long-term care and home care 36 TIVE 40 ceiving the right treatments in hospital 40 ronic disease management 44 tentially avoidable hospitalizations 48 eping people healthy in long-term care 52 eping people healthy in long-term care 53 spital infections 56 spital infections 56 uerse events in acute care hospitals 60 rtality in hospitals 62 ug safety in long-term care 64 biding harm in long-term care 67 <b>IT-CENTRED 70</b> tient experience in acute care hospitals and ergency departments 70		1.6	Home care summary	14
rdiovascular disease summary 18 betes summary 19 ncer summary 20 ntal health summary 21 ta advocacy 22 SIBLE 24 it times in emergency departments 24 cess to primary care 28 atment wait times and access to specialists 30 cess to long-term care and home care 36 TIVE 40 ceiving the right treatments in hospital 40 ronic disease management 44 tentially avoidable hospitalizations 48 eping people healthy in long-term care 52 eping people healthy in home care 53 spital infections 56 spital infections 56 uerse events in acute care hospitals 60 rtality in hospitals 62 ug safety in long-term care 64 biding harm in home care 67 <b>IT-CENTRED 70</b> tient experience in acute care hospitals and ergency departments 70		1.7	Long-term care summary	16
Interfactorial interfactorial interfactorial product community19ibetes summary19incer summary20ntal health summary21ta advocacy22SIBLE24it times in emergency departments24cess to primary care28atment wait times and access to specialists30cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40ceiving the right treatments52eping people healthy in long-term care56spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and70			ase summaries	
Incer summary 20 Intal health summary 21 Ita advocacy 22 SIBLE 24 It times in emergency departments 24 cess to primary care 28 atment wait times and access to specialists 30 cess to long-term care and home care 36 TIVE 40 ceiving the right treatments in hospital 40 ronic disease management 44 tentially avoidable hospitalizations 48 eping people healthy in long-term care 52 eping people healthy in long-term care 53 spital infections 56 verse events in acute care hospitals 60 rtality in hospitals 62 ug safety in long-term care 64 biding harm in long-term care 64 biding harm in home care 70 TI-CENTRED 70 tient experience in acute care hospitals and ergency departments 70		1.8	Cardiovascular disease summary	
Intal health summary21ta advocacy22SIBLE24it times in emergency departments24cess to primary care28atment wait times and access to specialists30cess to long-term care and home care36TIVE40cess to long-term care and home care36criving the right treatments in hospital40ceiving people healthy in long-term care52eping people healthy in long-term care56spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67TI-CENTRED70tient experience in acute care hospitals and70		1.9	Diabetes summary	
ta advocacy22SIBLE24it times in emergency departments24cess to primary care28atment wait times and access to specialists30cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40ronic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care56spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care67IT-CENTRED70tient experience in acute care hospitals and70			Cancer summary	
SIBLE24it times in emergency departments24cess to primary care28atment wait times and access to specialists30cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40coiving people healthy in long-term care52eping people healthy in home care56spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67TI-CENTRED70tient experience in acute care hospitals and70			-	
it times in emergency departments 24 cess to primary care 28 atment wait times and access to specialists 30 cess to long-term care and home care 36 TIVE 40 ceiving the right treatments in hospital 40 ronic disease management 44 tentially avoidable hospitalizations 48 eping people healthy in long-term care 52 eping people healthy in home care 53 spital infections 56 verse events in acute care hospitals 60 rtality in hospitals 62 ug safety in long-term care 64 piding harm in long-term care 67 tIT-CENTRED 70 tient experience in acute care hospitals and ergency departments 70				
cess to primary care28atment wait times and access to specialists30cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40ronic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care56spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and70	2	ACC	ESSIBLE	24
atment wait times and access to specialists30cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40ronic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care53spital infections56verse events in acute care hospitals60rtaity in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67TI-CENTRED70tient experience in acute care hospitals and70		2.1	Wait times in emergency departments	24
cess to long-term care and home care36TIVE40ceiving the right treatments in hospital40ronic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care53spital infections56systal infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and70		2.2	Access to primary care	28
TIVE40ceiving the right treatments in hospital40ronic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care53spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and70		2.3	Treatment wait times and access to specialists	30
ceiving the right treatments in hospital40ronic disease management44tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care53spital infections56systal infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and70		2.4	Access to long-term care and home care	36
ronic disease management 44 tentially avoidable hospitalizations 48 eping people healthy in long-term care 52 eping people healthy in home care 53 spital infections 56 verse events in acute care hospitals 60 rtality in hospitals 62 ug safety in long-term care 64 biding harm in long-term care 66 biding harm in home care 67 <b>IT-CENTRED 70</b> tient experience in acute care hospitals and ergency departments 70	3	EFF	ECTIVE	40
tentially avoidable hospitalizations48eping people healthy in long-term care52eping people healthy in home care53spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and70		3.1	Receiving the right treatments in hospital	40
eping people healthy in long-term care52eping people healthy in home care53spital infections56spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67 <b>IT-CENTRED</b> 70tient experience in acute care hospitals and70		3.2	Chronic disease management	44
eping people healthy in home care53spital infections56spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and70		3.3	Potentially avoidable hospitalizations	48
56spital infections56verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67 <b>IT-CENTRED</b> 70tient experience in acute care hospitals and70		3.4	Keeping people healthy in long-term care	52
spital infections 56 verse events in acute care hospitals 60 rtality in hospitals 62 ug safety in long-term care 64 biding harm in long-term care 66 biding harm in home care 67 IT-CENTRED 70 tient experience in acute care hospitals and ergency departments 70		3.5	Keeping people healthy in home care	53
verse events in acute care hospitals60rtality in hospitals62ug safety in long-term care64biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and70	4	SAF	E	56
rtality in hospitals 62 ug safety in long-term care 64 biding harm in long-term care 66 biding harm in home care 67 IT-CENTRED 70 tient experience in acute care hospitals and ergency departments 70		4.1	Hospital infections	56
ug safety in long-term care64biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and ergency departments70		4.2	Adverse events in acute care hospitals	60
biding harm in long-term care66biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and ergency departments70		4.3	Mortality in hospitals	62
biding harm in home care67IT-CENTRED70tient experience in acute care hospitals and ergency departments70		4.4	Drug safety in long-term care	64
IT-CENTRED     70       tient experience in acute care hospitals and ergency departments     70		4.5	Avoiding harm in long-term care	66
tient experience in acute care hospitals and ergency departments <b>70</b>		4.6	Avoiding harm in home care	67
ergency departments <b>70</b>	5	PAT	ENT-CENTRED	70
		5.1	Patient experience in acute care hospitals and	
tient experience in primary care <b>74</b>		_	emergency departments	
		5.2	Patient experience in primary care	74
		5.1 5.2		

6	EFF	ICIENT	76	
	6.1	Cost of service delivery	76	
	6.2	Right service in the right place	78	
	6.3	Avoidable emergency department visits	80	
	6.4	Avoiding unnecessary drugs and tests	82	
7	APP	ROPRIATELY RESOURCED	84	
	7.1	Overall spending and value for money	84	
	7.2	Information technology	86	
	7.3	Healthy work environments	90	
	7.4	Health human resources	92	
8	INT	EGRATED	94	
	8.1	Discharge/transitions from hospital and primary care	94	
9	FOC	USED ON POPULATION HEALTH	98	
	9.1	Unhealthy behaviour	98	
	9.2	Maternal and infant health	102	
	9.3	Sexual health	104	
	9.4	Preventive measures	106	
	9.5	Deaths and harm that could be avoided by prevention	108	
10	EQU	ITABLE	112	
	10.1	Primary care – access and effectiveness	112	
	10.2	Unhealthy behaviour	113	
		Preventive measures	114	
	10.4	Diseases that could be avoided with a population health focus	115	
11	LHI	N ANALYSES	118	
			118	
10		analyses		
12	EXA	MPLES OF SUCCESS	138	
	12.1	Reducing ED waits	138	
		Primary care	140	
		Surgical wait times	141	
		MRI wait times	142	
		Chronic disease management	143 144	
		Congestive heart failure readmissions	144	
	12.7	Ventilator-associated pneumonia, central line infection and <i>c. difficile</i> infection	145	
	12.8	Alternative level of care	146	
13	End	notes	147	
14	Ack	nowledgements	159	
15	Mer	nbers of Health Quality Ontario	160	

# **1.1** Executive summary

Ontarians are fortunate to have a publicly funded healthcare system that provides a comprehensive range of services for all. To help ensure the system is working properly, the provincial government has expanded the mandate of the Ontario Health Quality Council, and to mark this transition, the organization has registered the name "Health Quality Ontario" (HQO) under the Business Names Act and now conducts its affairs under this name. In addition to monitoring all aspects of the system and reporting to the people of Ontario on its quality, HQO has been charged with supporting efforts to improve quality across all healthcare settings, making recommendations on standards of care based on clinical practice guidelines and protocols, and making recommendations on how healthcare services are funded. All of these activities aim to accelerate the adoption of best available scientific evidence in the healthcare system.

Our sixth annual report — Quality Monitor — examines Ontario's healthcare system with our most critical eye. We note changes for the better or for the worse and report them to you. More importantly, we compare how we are doing to the best results elsewhere and provide an opinion about whether quality is good or needs improvement.

#### How we completed this report

HQO routinely monitors indicators and data sources used throughout Ontario, Canada and internationally, and works with its Performance Measurement Advisory Board to select indicators for this report. Data is drawn from sources that include Ministry of Health and Long-term Care (MOHLTC) databases, Census Canada, international surveys from the Commonwealth Fund and many others. The Institute for Clinical Evaluative Sciences (ICES) helped us conduct many of the data analyses. Researchers, clinical experts and healthcare executives reviewed our findings for accuracy and validity.

### Key features of this year's report Broader coverage of the nine attributes of quality

The nine attributes that Ontarians tell us reflect a high performing health system include: accessible, effective, safe, patient-centred, equitable, efficient, appropriately resourced, integrated and focused on population health. This year, we have added new indicators to increase our insight into these nine attributes across all sectors of healthcare. They include:

- Additional international comparisons on patient experience in primary care
- Expanded analysis of hospital infections and adverse events
- Expanded coverage of mental health, including suicide, intentional harm and depression

#### **Compact format**

We describe the entire healthcare system in 35 themes, with approximately two pages per theme. Most indicators have a mini-graph to indicate progress or lack of improvement over time and a three-to-five-sentence description of our interpretation of the data. Information on how indicators were defined and calculated and more detailed tables of actual data points can be found in the technical appendix to this report.

# Mini-summaries for leaders and staff in different sectors and people with different conditions

There are two-page summaries of key findings for hospitals, LTC, home care and primary care, as well as brief summaries for cardiovascular disease, diabetes, mental health and cancer. Each summary includes questions to ask of leaders or caregivers for self-reflection. Summaries of key differences for each local health integration network (LHIN) have also been expanded upon this year.

#### Root cause analyses and change ideas

Traditional public reporting simply gives indicator results and whether they are above or below average. Readers, however, are left wondering why these problems exist and what can be done about them. To counter this, we have included a root cause analysis with each theme, as well as ideas for improvement related to the root causes, as identified in the literature.

#### **Best practice stories**

We have reported on local stories of improvement, selecting those that had substantial improvement over the past year and had a clear aim, measures, change ideas and run charts. These success stories are closely linked to the key findings of the report, demonstrating that improvement is possible.

#### Health Quality Ontario's key findings

This year, Health Quality Ontario (HQO) identified both significant achievements and significant challenges in Ontario's healthcare system. Achievements include reductions in wait times in the emergency department, for many surgeries and for CT scans, as well as improvements in primary care access, patient outcomes for coronary artery disease and smoking rates. However, many of these rates are still far from optimal. Challenges include a continuing problem with patients occupying alternative level of care (ALC) beds and wait times that are still too long for long-term care. In addition, Ontario has long waits for urgent cancer surgeries and specialist appointments and has not improved patient outcomes for congestive heart failure and chronic obstructive pulmonary disorder (COPD). In many instances, progress is linked to a clear strategy to improve results, and lack of progress can be attributed at least in part to the lack of such a strategy.

Overall, there are several success factors for system-wide quality improvement, including leadership and accountability, measuring and reporting on performance, following evidence-based practices, innovative models and processes for strategic system improvement, ensuring providers have access to up-to-date training and resources, change management, patient and family engagement and appropriate capacity throughout the healthcare system.

Broadly speaking, this report identifies three system areas that need to be addressed to improve the quality of Ontario's healthcare system: access to healthcare, chronic disease management and keeping the population healthy. It will be critical to have progress across the different success factors and all three system areas to significantly improve health outcomes and mitigate healthcare costs. HQO has summarized the province's achievements and challenges and indicated specific strategies for improvement for each of the three system areas below.



#### Access to healthcare

HQO measures access to healthcare by looking at how long people have to wait to get the medical attention they need. This year, some wait time indicators improved dramatically — but big problems remain.

In the **emergency department**, the 90<sup>th</sup> percentile length of stay (the maximum amount of time nine in 10 patients spend in the emergency department) decreased by two hours for high complexity patients and by 30 minutes for low complexity patients in the past two years. Length of stay is still well above target for high complexity patients — 12 hours versus eight hours — but progress has been made. Provincial initiatives that have contributed to these results include the ED Pay for Results Program and public reporting of emergency department wait times on the Ontario Wait Times website. In addition, many hospitals have adopted Lean process improvements to enhance efficiency and flow, encouraged by the ED Process Improvement Program (ED PIP).

The average wait time to **long-term care** (LTC) placement is 3.5 months a length of time that is far too high. When people wait in their own homes for a bed in an LTC home, they may not be receiving all of the care they need and, as a result, may experience declines in health and quality of life. When people wait for admission to an LTC home from hospital, that has ripple effects through the health system. These individuals may be occupying ALC beds in hospital (i.e. are in hospital even though they are better served elsewhere), which delays new hospital admissions and can contribute to increased emergency department wait times. Indeed, one in six people in hospital are ALC. This problem has not been improving as of 2009/10, and represents a major inefficiency in the system.

Although LTC wait times are still far too high, they have stopped increasing for the first time since 2005. This is a promising trend, but it is everyone's expectation that these rates now must start to decrease. Going forward, it will be important to address all aspects of this complex problem and to ensure adequate resources exist to support providers and patients at every level of the healthcare system — in hospital, at home and in LTC. It will also be important for Ontario to evaluate the Aging at Home Strategy, in place since 2007, which makes a wider range of home care and community support services available to enable people to continue leading healthy and independent lives in their own homes.

When it comes to **primary care**, Ontario has seen steady improvement in the percentage of people with a family doctor since 2005 and now ranks among the best in Canada. This is likely due in part to investments in medical schools that increased the number of spots available, the introduction of health professionals such as nurse practitioners who can take on some physician responsibilities, the establishment of family health teams and the expansion of community health centres. However, 6.5% of adult Ontarians don't have a family doctor and 3.3% are actively seeking one. These percentages are even higher for people with lower incomes and those who live in northern Ontario. The province also needs to do a better job of making sure people can schedule an appointment quickly. Fewer than half of Ontarians who were sick were able to book time with their doctor on the same or next day. There have been no changes in recent years, and both Canada and Ontario rank poorly on this indicator compared to other countries.

The situation is even more serious for wait times to see a **specialist**. Half the people who are referred to a specialist wait four weeks or longer for an appointment. Canada's and Ontario's standings are the worst among 10 major developed countries and need to improve.

Ontario is getting better at completing a range of surgeries and other procedures within provincially mandated targets, despite increasing demand. The 90th percentile wait times for hip replacements, knee replacements and cataract surgeries have been cut by more than half since 2005. Meanwhile, the 90<sup>th</sup> percentile wait time for CT scans has decreased from 2.5 months to about one month over the same time period. Ontario's improvements in wait times for surgeries and other procedures can be attributed to public reporting on the Ontario Wait Times website, creation of a provincial information system to track wait times, funding for increased volumes of surgeries and investments in new CT equipment. That said, there is still room for improvement in two important areas. The 90<sup>th</sup> percentile wait time for cancer surgeries decreased from 2005 to 2008, but hasn't improved in the past two years and currently sits at 51 days. The 90<sup>th</sup> percentile wait time for MRI scans has remained at about four months since 2005 (113 days in 2010), despite huge increases in the number of scans performed. In the future, it will be important to ensure CT and MRI scans are done only when needed.

#### Strategies for improvement

In the **emergency department**, change ideas include developing better care coordination and moving patients to the right place as soon as possible so they are not occupying beds unnecessarily. In addition, hospitals can make specific process improvements within the emergency department, including creating a fast-track area for less serious cases and changing the location of supplies and equipment to maximize efficiency. They can divert non-urgent cases away from the emergency department to other alternatives. And the system as a whole can focus on improving primary care services so patients are less likely to require emergency care. For more information, see section 2.1.

To shorten wait times for **long-term care** homes, the province needs to ensure that enough supportive housing and assisted living facilities are available as an alternative to LTC for people who are able to live more independently — especially in high-demand regions. Home care needs to be more available as well, with flexibility on the hours allotted to families based on their needs. In addition, when patients enter hospital, it's important to avoid jumping to the conclusion that they need LTC until they have been given time to recover, and to keep them as active as possible to prevent a decline. For more information, see section 2.4 and 6.2.

To maximize **primary care** and **specialist** capacity, providers can take many measures, including adopting advanced access scheduling, improving office efficiency and implementing a well-functioning electronic medical record (EMR) system. For more information, see section 2.2.

Part of the solution to the wait time challenge for **surgeries and other procedures** is prevention. If Ontario can slow down deterioration of health so that the need for surgery is reduced, this can have a positive impact on wait times. There are also major opportunities to streamline the scheduling and organization of services to reduce wait times, as has been done successfully in many centres. Other change ideas include implementing appropriate criteria to ensure that patients truly require surgery or a specific test, and ensuring at least some excess capacity for services to help handle fluctuations in demand. For more information, see section 2.3 and 6.3. 1.1

### Chronic disease management

Ontario has had mixed results when it comes to chronic disease management. The province has experienced steady improvements in the management of coronary artery disease and can be cautiously optimistic about diabetes, thanks to reductions in hospital admissions and complications. However, congestive heart failure and COPD remain significant challenges. For more information about chronic diseases in Ontario, see the Disease Summaries sections in this chapter.

In the area of **coronary artery disease**, heart attack mortality within 30 days and between 30 days and one year have decreased over the past seven years. About half as many people per capita are readmitted to hospital following a heart attack. One-third as many people per capita are admitted to hospital for angina. There is strong scientific evidence that access to revascularization (cardiac bypass or stent) can reduce mortality and disability for people who need this type of care, and Ontario has a highly structured regional system for delivering these services. The appropriate use of drugs such as statins may also be contributing to these improvements.

Use of statins and angiotensin-converting enzyme inhibitors (ACEIs)/ angiotensin-receptor blockers (ARBs) has also improved among elderly people with **diabetes**, with the application of both at once nearly doubling over the past seven years. This may be a factor in the one-quarter decrease in hospital admission rates and nearly one-third decrease in serious complication rates (within a year) people with diabetes have experienced over the past seven years. On the other hand, when it comes to on-going management of diabetes, there has been no improvement in the percentage of people with diabetes who had an eye exam in the past 12 months. This number is stagnant at one in two.

Furthermore, there has been only a slight decline in hospitalizations for **congestive heart failure** in the past four years (down to 52 from 54 per 100,000 people). About one in five people are readmitted for congestive heart failure, and this number has not changed at all over the same time period. Finally, more than one in three patients admitted to hospital for congestive heart failure die within a year, and this indicator has not improved in seven years.

Similarly, there has been just a slight improvement in **COPD** hospital admission rates in the past four years (down to 84 from 92 per 100,000 people), and COPD is now the most common ambulatory care sensitive condition — where appropriate care in the community may prevent hospitalization. As with congestive heart failure, there has been little change in readmission rates in the past four years. About one in five people are readmitted for COPD.

A coordinated strategy, such as the one in place for revascularization (bypass or stent), could be beneficial in tackling other areas, including congestive heart failure and COPD.

### Strategies for improvement

Disease registries can help practitioners better monitor how well patients with chronic diseases are doing and provide timely follow-up. Such a registry is an important part of the province's Diabetes Strategy. In addition, flow sheets in patient charts, reminder systems and well-functioning EMRs can help healthcare providers deliver more consistent care to improve chronic disease management.

People with chronic conditions can benefit from access to professionals with special skills, including pharmacists to review their medications, dietitians to provide advice on nutrition and primary care providers to monitor key health indicators. The best case is when people have access to a team that meets all their health-related needs and engages them in self-management by helping them develop their own action plans and set personal goals. For more information, see section 3.2.

Meanwhile, making sure that patients receive the right medications while they are in hospital and after they are discharged can, as already discussed, help with the management of coronary artery disease and diabetes. Hospitals can increase the availability and use of standardized admission orders, discharge checklists, order sets or electronic health records (EHRs) that guide clinicians' decisions and generate clinical reminders. Also critical to a smooth transition from hospital back into the community is ensuring the seamless transmission of patient information to the primary care provider — ideally directly into the patient's EMR. For more information, see section 3.1 and 8.1.

These are elements of a comprehensive chronic care model. For more information about Ontario's framework for preventing and managing chronic disease, see www.health.gov.on.ca/english/providers/program/cdpm/index.html.

# HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

### Keeping the population healthy

One of the best ways to manage healthcare costs is to implement effective programs that help keep the population healthy. Ontario has had major success in reducing the smoking rate by 25% since 2001, but continues to struggle to improve physical activity and reduce obesity.

Policy changes such as the 2006 *Smoke-Free Ontario Act* have likely contributed to a steady decline in smoking rates and exposure to second-hand smoke over the past six years, and Ontario is now among the best-performing provinces in Canada on these indicators. The current provincial smoking rate is 19%, however, meaning that nearly one in five people across Ontario continue to smoke. Furthermore, people with low incomes or less education are twice as likely to smoke as people with high incomes or a post-secondary education. More needs to be done to address these specific population groups.

The rate of obesity has increased from 16% to 18% over the past eight years. Just one in two Ontarians get enough exercise and this number has remained the same since 2007. Finally, fewer than half get the recommended five or more servings of fruits or vegetables every day, with no major improvement since 2003. A coordinated obesity strategy may be part of the solution.

Early screening for illness is another important component of population health. Ontario's aggressive ColonCancerCheck campaign targeted screening for one disease that can be much more successfully treated when caught early. Screening for colon cancer has increased by 68% in the last four years, with nearly one in three people aged 50 to 74 taking a fecal occult blood test in 2009. However, many other types of screening can help the province detect additional diseases sooner, treat them earlier and improve people's survival rates.

As already suggested in relation to smoking, there are still huge inequalities in the province. For example, income levels affect the likelihood of having a heart attack, with the poorest Ontarians 36% more likely to experience an acute myocardial infarction than the richest Ontarians. Meanwhile, the lowest-income Ontarians have a 32% higher rate of injury-related hospitalization compared to the highest-income Ontarians. Ontario is still a long way from achieving equitable health outcomes.

#### Strategies for improvement

Toronto stepped up ahead of the rest of the province to prohibit smoking in most public places and workplaces (1999), then in restaurants, dinner theatres and bowling centres (2001), and finally in bars, billiard halls, bingo halls, casinos and racetracks (2004). Toronto currently has the province's lowest smoking rate, at 13%, so it appears these types of measures are effective.

Other change ideas include improving access to smoking cessation programs, making it easier for people to use nicotine replacement therapies, "de-normalizing" unhealthy behaviours and addressing the range of constraining and enabling factors that influence peoples' health and access to care. It will also be important to close knowledge and awareness gaps and make sure healthy choices are obvious and accessible. The Ministry of Health Promotion and Sport will have an important role to play in these types of initiatives. For more information, see section 9.1.

To improve prevention, it may be helpful to develop provincial registries and EMRs/EHRs to remind patients when they are due for screening tests, to encourage adherence through public awareness campaigns and culturally sensitive screening programs, and to reach out to vulnerable and hard-to-reach populations through mobile care units and access to vaccinations outside primary care offices. For more information, see section 9.4.

Other provinces sometimes have ideas that may translate well to Ontario. In British Columbia, as just one example, it has been suggested that the province limit marketing of unhealthy food and beverages to children, add nutritional labeling on menus, tax unhealthy choices, expand wellness programs in the workplace and revitalize recreation facilities, among other measures<sup>i</sup>.

Finally, as recommended last year, strategies tailored to the most vulnerable populations may speed up progress on chronic diseases and help to improve equity in the province's healthcare system. Making it easier for lower-income individuals to access healthy food, offering free physical activity programs and implementing healthy community initiatives can contribute to a healthier province and lower healthcare costs over the long term.

# **1.2** Quality improvement summary

# KEY SUCCESS FACTORS FOR SYSTEM-WIDE QUALITY IMPROVEMENT

The Quality Monitor identifies not only problems with quality, but also the root causes of these problems and ideas for improvement. For each individual indicator, specific ideas have been identified from the scientific literature, best practice stories and successful quality improvement campaigns around the world. These ideas fall into several broad categories of success factors for system-wide improvement identified below. Ensuring that all of these success factors are adopted in a coordinated fashion is essential.

### 1. Leadership and accountability

"The strategic plan? Isn't that something to do with head office?"

"We had great ideas for improvement, but no one listened to us!"

Change is hard. Change can generate fear of the unknown. Changes that result in a shift in roles can generate resistance, if one group perceives it as a loss. Successful change requires strong leadership that involves:

- Creating a compelling vision for quality, and specific targets and timeframes for completion that describe that vision in tangible terms. Leaders can communicate the vision and inspire people to achieve it.
- Describing a clear strategy. This involves identifying those activities or evidence-based best practices that have the biggest impact. Leaders set clear expectations and clear accountability for all levels of the organization for implementing these activities, where each individual understands his or her role in achieving the vision.
- Creating a culture of quality improvement, with an emphasis on learning, trying new ideas, and being encouraged to report quality problems instead of being punished. Leaders develop conflict management skills in the organization to manage resistance. Leaders engage people at the front line in the change process. Leaders find champions to promote change among their peers (especially important for physicians). Lastly, leaders celebrate successes and facilitate the sharing of best practices.

### 2. Well designed processes, systems and teamwork for delivering healthcare

"I can't find those lab results - just run the test again."

"Why do I have to give the same information to five different people?"

Healthcare is a complex system that involves many tasks carried out by different people in different organizations. Every time one provider hands off care to the next, errors or miscommunication can occur. Over time, some steps become redundant. Often, sequences of tasks differ among different organizations, individuals, or even different floors or shifts within one organization. This creates confusion or delay, and leads to inconvenience, waste, poor care or harm. A recurring theme in this report is the need to streamline processes. Specific examples include eliminating redundant steps, standardizing processes, doing things in parallel instead of in sequence, making processes error-proof, making defects easy to spot and fixing them early before they are passed on. Teamwork can be strengthened by establishing good processes for handing off care, consistent routines for communication and role clarity.<sup>i</sup>

Many healthcare organizations are now using process improvement methods such as Lean (elimination of waste) and Six Sigma (elimination of defects). It is important to ensure that organizations have staff with deep expertise in the use of quality improvement methods who can mentor or lead process improvement activities.

### 3. Measurement and reporting

"I thought we were doing a great job until I saw our data."

"You can't manage what you can't measure."

Quality measurement helps identify areas that need improvement, make the case for change, identify the root cause of problems, and monitor whether an intervention to improve quality is working. Ontario has a wealth of data on quality, but there are still huge gaps to be filled (see the data advocacy section, see 1.12). Reporting this information to the public is important, because public comparisons between institutions can spur leaders to outdo their peers. Although Ontario already reports on many wait times and safety indicators by institution, there are other opportunities to expand public reporting further.

Ontario, however, is weak in its ability to provide instant feedback on quality performance to individual health care providers or teams. This feedback is essential to getting people engaged about quality and to allow improvement teams to change their strategy if it is not working. Developing instant feedback will require that electronic health records are carefully designed for this purpose.

### 4. Tools to address information overload

"Trying to keep up with the literature is like drinking from a fire hose."

In 2009, 1.5 million articles were published in the area of science, technology and medicine."

Today's healthcare providers are bombarded with information. Best evidence often isn't followed, resulting in poorer outcomes and wasted resources. This can be because people forget (the brain has natural limits on memory) or there are distractions (too many competing demands for attention), no time to read all the evidence, and difficulty applying different guidelines simultaneously to complicated patients with different conditions.<sup>III</sup> Practice guidelines tell providers what to do, but cannot change practice on their own.

iii Grol R, Wensing M. What drives change? Barriers to and incentives for achieving evidence-based practice. Med J Aust. 2004;180(6 Suppl):S57-60.

<sup>&</sup>lt;sup>1</sup> Ontario Health Quality Council. Improvement guide. 2009. www.ohqc.ca/pdfs/qi\_guide.pdf; accessed February 25, 2011.

<sup>&</sup>lt;sup>ii</sup> Ware M, Mabe M. The STM report. An overview of scientific and scholarly journal publishing. International Association of Scientific, Technical and Medical Publishers, 2009.

# HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

Practice tools can help address these problems. They include checklists, standardized care orders, flow sheets, appropriateness criteria, reminder systems and other decision support tools. These tools can be verbal, paper or electronic-based, and can prompt providers to perform or avoid a particular action, help decide on a drug or test, or document if a best practice was followed.

### 5. Verifying and upgrading providers' clinical skills

"When was the last time you did that procedure?"

### "I've lost my touch ... "

Clinical skills are the ability of health care providers to evaluate a clinical problem or execute a task that may require knowledge, critical thinking, experience or technical skills. There may be gaps in skills if: the original training was inadequate; the task is done infrequently and skills deteriorate over time ("use it or lose it"); the technique changes but there is no opportunity to upgrade skills; or, there is no mechanism to spot and fix problems with clinical skills. Traditional activities like lectures do not address well these problems. Other approaches include: formal recertification programs; direct observation of providers performing specified tasks when the technique is taught, when staff are hired, and at regular intervals); and patient simulation labs. Planners can also organizing care so that certain services are delivered only by institutions or individuals that maintain a minimum volume.

### 6. Patient and family engagement

#### "No one ever told me about my options!"

Quality improvement depends on patients being engaged in their care. This means that they understand the importance of healthy living and the nature of their illnesses. They know their treatment options and make informed choices which fit their preferences and lifestyle. They feel encouraged to ask the right questions about their care — what is happening to them and why, and are not afraid to point out problems with their care (e.g. when someone forgets to wash their hands). Patient education and health promotion are helpful but even more powerful are patient self-management skills.<sup>w</sup> When individuals feel they are in control of their health, they are more likely to continue taking the right drugs or adopt healthy living habits.

Health and health behaviours are generally worse for people with low income or education, or in certain communities. It is important to ensure that patient engagement activities are tailored to these groups as well as those from different language or cultural groups, and that these activities are coordinated with other activities that address broader determinants of health (e.g. housing, employment, healthy communities).

#### 7. The right policies and incentives for quality

"Is there any reason why we should do things differently? Why should I change?"

Incentives and policies can motivate change. People want to improve quality if they are recognized and appreciated for their efforts; can see tangible results; and feel supported. Providers embrace quality improvement initiatives when the outcome is not only better clinical care but also reduced wasted time or administrative hassle. "Herd behaviour" is important for both patients and providers. Seeing a peer choose the right treatment can influence a provider to do the same. For patients, policies like smoking bans which "de-normalize" unhealthy behavior appear to be effective. Policies on how services are funded represent another lever for change. Many believe that funding should be tied to quality. For example, extra costs resulting from poor quality should be borne by the organization that provided and was funded for that care.<sup>v</sup>

### 8. The right capacity throughout the healthcare system

"Why do we have wait lists for some services and but not for others?"

Supply and demand for services need to be matched to reduce both waste and unnecessary wait times. The health care system needs to be constantly estimating what the future demand is for different type of services in different communities and then making sure it has the right amount of equipment, programs or healthcare providers to meet this demand. When managing wait lists, it is best to have at least some excess capacity to allow the system to handle random surges in demand. Estimates of demand should also consider opportunities to reduce demand, through better prevention and reduction of inappropriate use of services.

<sup>iv</sup> Coleman MT, Newton KS. Supporting self-management in patients with chronic illness. Am Fam Physician. October 15, 2005;72(8):1503–1510.

v www.leapfroggroup.org/for\_hospitals/leapfrog\_hospital\_quality\_and\_safety\_survey\_copy/never\_events. Accessed April 14, 2011.

# 1.3 Attributes framework

The attributes of a high-performing health system

### ONTARIANS WANT THEIR HEALTH SYSTEM TO BE:

### ACCESSIBLE

People should be able to get timely and appropriate healthcare services to achieve the best possible health outcomes. For example, when a special test is needed, you should receive it when needed and without causing you extra strain and upset. If you have a chronic illness such as diabetes and asthma, you should be able to find help to manage your disease and avoid more serious problems.

### EFFECTIVE

People should receive care that works and is based on the best available scientific information. For example, your doctor (or healthcare provider) should know what the proven treatments are for your particular needs including best ways of coordinating care, preventing disease or using technology.

### SAFE

**People should not be harmed by an accident or mistakes when they receive care.** For example, steps should be taken so that elderly people are less likely to fall in long-term care homes. There should be systems in place so you are not given the wrong drug, or the wrong dose of a drug.

### PATIENT-CENTRED

Healthcare providers should offer services in a way that is sensitive to an individual's needs and preferences. For example, you should receive care that respects your dignity and privacy. You should be able to find care that respects your religious, cultural and language needs and your life's circumstances.

### EQUITABLE

People should get the same quality of care regardless of who they are and where they live. For example, if you don't speak English or French it can be hard to find out about the health services you need and to get to those services. The same can be true for people who are poor or less educated, or for those who live in small or far-off communities. Extra help is sometimes needed to make sure everyone gets the care they need.

### **EFFICIENT**

The health system should continually look for ways to reduce waste, including waste of supplies, equipment, time, ideas and information. For example, to avoid the need to repeat tests or wait for reports to be sent from one doctor to another, your health information should be available to all of your doctors through a secure computer system.

### APPROPRIATELY RESOURCED

The health system should have enough qualified providers, funding, information, equipment, supplies and facilities to look after people's health needs. For example, as people age they develop more health problems. This means there will be more need for specialized machines, doctors, nurses and others to provide good care. A high-performing health system will plan and prepare for this.

### **INTEGRATED**

All parts of the health system should be organized, connected and work with one another to provide high quality care. For example, if you need major surgery, your care should be managed so that you move smoothly from hospital to rehabilitation and into the care you need after you go home.

### FOCUSED ON POPULATION HEALTH

The health system should work to prevent sickness and improve the health of the people of Ontario.



# Hospital sector summary

ACUTE CARE HOSPITALS: summary for boards, CEOs, senior management and clinical leaders

HOSPITAL
LONG-TERM CARE
HOME CARE
PRIMARY CARE

Topic area	Key facts	Questions to consider
1. ALC (section 6.2)	<ul> <li>Approximately one in six hospital beds in Ontario are filled with patients who should be cared for somewhere else. This problem has not improved in the last year (2009/10). This represents a major inefficiency in the health care system.</li> </ul>	<ul> <li>Are we working with community care access centres (CCACs) to apply the Home First approach, where frail individuals admitted to hospital go home with the necessary home care support, where they can then make a decision about whether they need to be in a LTC home?</li> <li>Can we identify people at risk for being ALC earlier (e.g. when they visit the emergency department for the first time), so that home care services can be arranged before their health deteriorates?</li> <li>Are we using rehabilitation resources to help patients regain their function and return home to live as independently as possible?</li> <li>Are we working with CCACs and local health integration networks (LHINs) to promote supportive housing models or similar options for frail individuals?</li> <li>What are the characteristics of ALC patients that are difficult to place into LTC homes? What services need to be developed in the community or in LTC homes to serve them?</li> </ul>
2. Emergency department (ED) wait times (section 2.1)	<ul> <li>ED wait times have improved but are still well short of provincial targets. It is the admitted patients who are least likely to be admitted within the recommended timeframe. Specifically:</li> <li>The maximum time nine in 10 patients spend in an ED decreased in the last two years by two hours for high complexity patients (from 14 to 12 hours), and by half an hour for low complexity patients (from 4.8 to 4.3 hours). The target is 8 and 4 hours respectively.</li> <li>One in 20 Ontarians who visit the emergency department leave without being seen by a physician, likely because they were tired of waiting. This figure has been improving.</li> <li>Among patients admitted to hospital from the ED, half of them waited more than three hours for a bed, and this has not improved in the past two years. Also, only 41% of these patients are served within the recommended timeframe, compared to 81% overall.</li> <li>Ontario and Canada fare poorly on international comparisons of ED waits.</li> </ul>	<ul> <li>Are we moving patients who do not need to be receiving care in the hospital to the right place as quickly as possible (see above)?</li> <li>Have we considered all the different ideas for improving patient flow within the ED (e.g., fast-track area for less serious cases, improved layout by changing the location of supplies and equipment to maximize efficiency, chairs for acute patients, flexible human resources scheduling and information systems to track patients and results)?</li> <li>Do we have a organization-wide contingency plan when the ED reaches critical gridlock?</li> <li>Have we considered diverting non-urgent cases away from the ED to alternatives that are most appropriate for their conditions? Are we redirecting people who are using the ED for their main source of care to family doctors who are taking new patients?</li> <li>Can we work with other stakeholders to improve primary care services so patients are less likely to need ED care?</li> </ul>
3. Surgical and CT/MRI wait times (section 2.3)	<ul> <li>Wait times have decreased for many procedures overall, but there is still room to improve, especially for high-priority cases for all surgeries (particularly cancer) and MRI. Specifically:</li> <li>The 90<sup>th</sup> percentile wait for hip or knee replacement is just under seven months; for cataracts, four months; and for CT scans, one month. Waits are less than half of what they were in 2005.</li> <li>The 90<sup>th</sup> percentile wait for MRI is four months — same as it was in 2005, despite a nearly three-fold increase in the number of scans being done.</li> <li>In the case of general surgery and surgery for cancer, cataracts, coronary artery bypass, angiography and percutaneous coronary intervention (e.g. cardiac stents), 93% to 99% of elective cases are done within the recommended time frame. However, for urgent cases, the percentage done within the recommended time is much lower: for cancer,67%; cataract, 66%; general surgery, 83%; coronary bypass, 86%; angiography, 85%; percutaneous coronary intervention, 85%.</li> <li>CT and MRI scans are exceptions to the above trend; urgent cases are more likely to be done within the recommended time frame than elective cases.</li> </ul>	<ul> <li>Are there poor hand-offs, poor communication or lack of standardized processes that contribute to delays?</li> <li>Do we have appropriate criteria to ensure that patients truly require surgery or tests?</li> <li>Can we ensure at least some excess capacity for services associated with surgery (such as space in the intensive care unit)?</li> <li>Have we considered centralized booking systems that could connect patients to places with the shortest wait time?</li> <li>Do we monitor key process metrics (e.g., on-time case starts and downtime) so we can maximize our efficiency and increase our capacity?</li> <li>Do we measure demand and supply and do we know if they are in balance? Have we ever done queue-clearing blitzes — for example, temporarily increasing the rate of procedures done until backlog is eliminated?</li> </ul>

# 1.4

Topic area	Key facts	Questions to consider
<ul> <li>4. Safety — hospital infections and other areas (sections 4.1, 4.2 and 4.3)</li> </ul>	<ul> <li>Hospital-acquired infections have improved but there is major room for improvement.</li> <li>Infection rates for <i>C. difficile</i> remained stable in the last year after gains in the previous year.</li> <li>Ontario hospitals decreased the number of cases of ventilator-associated pneumonia and central line infection from January 2009 to September 2010, but have not yet reached the desired level of zero.</li> <li>Two in three Ontario healthcare providers wash their hands before seeing patients, while almost eight in 10 wash their hands after patient contact. There have been improvements, but these rates are still too low.</li> <li>Last year, almost seven in 10 reportable hospitals experienced a decrease in their hospital standardized mortality ratio (HSMR) score, which is encouraging.</li> </ul>	<ul> <li>How well are we following standard protocols for preventing ventilator-associated pneumonia and central line infections? Is anyone deviating from the protocol, and why?</li> <li>What are we doing to ensure that all staff and physicians use proper hand washing techniques? Are we auditing hand hygiene frequently (more than just once a year) and providing feedback to staff on compliance (e.g., monthly or quarterly, by ward or by provider group)? Are hand washing stations conveniently located, never empty, and being used? Are we encouraging patients to ask providers if they've washed their hands?</li> <li>Are we regularly using checklists, standardized order sets or protocols to minimize reliance on memory? If so, are we actually tracking how we're using them and if they are having an effect?</li> <li>Are we promoting a culture of safety, where people feel free to speak up if they see a safety issue?</li> <li>Do we have physician champions for safety?</li> <li>Are we considering the use of rapid response teams, with clear guidelines for when they are to be used?</li> </ul>
5. Effective- ness/ evidence- based practices (section 3.1)	<ul> <li>Although there has been some increase in the percentage of patients filling prescriptions upon discharge for heart attack and congestive heart failure, there is still room to improve. Hospitals who do not do this well tend to have higher readmission rates.</li> <li>Although there have been gains over the past six years, only one in eight patients who has a stroke and could benefit from clot-busting drugs receives those drugs within one hour of arriving in the emergency department.</li> </ul>	<ul> <li>Do we have information technology systems in place to remind doctors of standard protocols and treatment plans or to track compliance with guidelines?</li> <li>Are we using checklists or standardized order sets to make sure people with different conditions (e.g. heart attack, heart failure, stroke) get the right drugs and tests?</li> <li>Are we educating patients on the importance of these medications and how to take them?</li> <li>Do we have a standardized process for handling time-sensitive situations like a new stroke? If so, how well are we following it?</li> </ul>
6. Readmissions (section 3.3)	<ul> <li>About one in five patients with congestive heart failure or COPD are readmitted within a month, for any cause. There is huge room to improve these rates.</li> <li>Readmissions over the past seven years have decreased by almost half for heart attack, which is good news.</li> </ul>	<ul> <li>Do we know why our patients are being readmitted?</li> <li>To reduce readmission, are we making sure patients have all the information they need when they are sent home (see below)?</li> <li>Do patients at high risk for readmission have a follow-up home care visit immediately after discharge and an appointment with their doctor soon after discharge (e.g. a week)?</li> <li>How quickly are we transferring discharge summaries to family physicians?</li> </ul>
7. Discharge hand-offs (section 8.1) and communica- tion (section 5.1)	<ul> <li>Many patients leave hospital without the information they need, which can lead to readmissions. For example:</li> <li>Only half of hospital patients know when to resume normal activities</li> <li>Only half of ED patients know what danger signs to look out for at home</li> <li>About one in three patients do not know what side effects of medications to look for</li> <li>There are general problems with communication:</li> <li>About one in three patients felt that staff did not provide explanations to questions that were understandable</li> </ul>	<ul> <li>Are written discharge instructions routine for all of our patients (including warning signs, whom to call, etc.)?</li> <li>Do we use methods like "teach-back" to verify patients understand all the information they have been given?</li> <li>What are we doing to simplify communication to people with low literacy? Do we have translators for people who speak different languages?</li> </ul>

1.4 Hospital sector summary



Topic area	Key facts	Questions to consider
8. Patient experience (section 5.1)	<ul> <li>Pain control and responsiveness can be improved.</li> <li>About one in two patients felt staff did not do everything they could to control pain</li> <li>About one in three expressed concern about timeliness of requests for help (e.g. assistance to go to the bathroom, response to a call button).</li> </ul>	• Are we following standard procedures and best practices for pain control (e.g. monitoring pain like a vital sign; using visual analogue scales; standard pain protocols; patient-controlled anesthesia in certain settings)?
9. Hospital finances (section 6.1)	<ul> <li>27% of hospitals reported a deficit in FY 2009/10. Among large community hospitals, 36% reported a deficit.</li> <li>Some hospitals (especially large community and teaching hospitals) continue to struggle to manage their current ability to pay bills without having to borrow.</li> </ul>	• While the first instinct when facing a deficit is to cut services, has hospital management carefully thought of all the different sources of waste in the system and made aggressive plans to eliminate them (e.g., unnecessary tests or services and waste of staffing, space, inventory and supplies)?

# Primary care summary PRIMARY CARE: summary for primary care practitioners 1.5

Topic area	Key facts	Questions to consider
1. Access to primary care (see section 2.2)	<ul> <li>There has been important progress in ensuring Ontarians have a regular doctor, but wait times to get an appointment are too long.</li> <li>About one in 16 Ontarians are without a regular doctor, and almost half of those individuals are seeking one. Over the past three years, the percentage of adults without a family doctor has dropped from 8.2% to 6.5%.</li> <li>Based on survey data, Ontarians are more likely than people in Quebec and Western Canada to have a regular primary care provider, and as likely as those in Atlantic Canada to have one. Internationally, Ontario is on par with other countries that had the best results among 11 countries surveyed.</li> <li>Among Ontarians who are sick, fewer than 50% are able to see their doctor's office. This indicator has not improved in recent years, and Ontario and Canada are behind other countries.</li> </ul>	<ul> <li>Are we using advanced access, the system of scheduling appointments and managing patient flow to reduce or eliminate wait times for appointments?</li> <li>Could we reduce unnecessary repeat visits to free up more time to serve people better (e.g., by giving lab results over the phone instead of requiring a visit)?</li> <li>Are our processes as efficient as they could be? For example, is each exam room set up exactly the same way? Can things be relocated to reduce walking around?</li> <li>Do we have a well-functioning electronic medical record (EMR) system?</li> <li>Are we working in a team? If yes, are we using each team member to his/her fullest capacity? Are each team member's roles and responsibilities clear? What tasks could be shifted from one team member to another?</li> </ul>
2. Access to surgery, CT/ MRI scans and specialist care (see section 2.3)	<ul> <li>Wait times have decreased for many surgeries like hip and knee replacements and cataracts, and for CT scans. There is still room to improve, especially for high-priority cases for all surgeries (particularly cancer) and MRI.</li> <li>Half the people who are referred to a specialist wait four weeks or longer for an appointment. Canada's and Ontario's standings are the worst among 10 major developed countries.</li> </ul>	<ul> <li>For CT/MRI scans, are all the tests we are ordering necessary? Do we find ourselves pressured into ordering tests that are not needed? What could we do about that?</li> <li>Do we use the Ontario Wait Times website to find places that can do a surgery sooner if the patient wants this?</li> <li>Have we done all we can to encourage patients to adopt good health behaviours to avoid surgery (e.g. lose weight to avoid knee replacement)?</li> </ul>
3. Wait times from the community to long-term care (LTC) (see section 2.4)	• The median wait time for people placed into a LTC home from the community is well over 5 months, which is too long.	• Have we explored alternatives to LTC placement with patients and families (e.g., assisted living or supportive housing)?
4. Chronic disease Management (see section 3.2 and 3.3 on avoidable hospitaliza- tions)	<ul> <li>There are signs of improvement in chronic disease management — complications rates of diabetes are decreasing and heart attack mortality is dropping.</li> <li>However, there is still room to do better:</li> <li>Only about half of people with diabetes received an eye exam in the past 12 months.</li> <li>The use of ACEIs/ARBs and statins is increasing among elderly people with diabetes but there is still room to improve, since only half of people are getting both of these drugs.</li> <li>Heavy drinking and obesity have been on the rise over the past nine years among people with chronic diseases. Rates of physical inactivity, smoking and inadequate fruit and vegetable consumption have seen minor improvements in this timeframe but are still too high.</li> <li>Over one-third of patients admitted to hospital for congestive heart failure die within the next years.</li> <li>In FY 2009/10, roughly 34,100 people were admitted to hospitals in Ontario for complications from chronic diseases that could have been prevented with good primary care. COPD and congestive heart failure are the two most common conditions associated with these hospital admissions.</li> </ul>	<ul> <li>Are we using methods such as flow sheets to remind us of all the best practices?</li> <li>If we have an EMR, does it provide us with data on the percentage of our patients with diabetes who are on the right drugs (e.g., a statin, ACEI/ARB and acetylsalicylic acid) and who have received a recent A1C test or eye exam?</li> <li>Have we set the EMR up so it reminds us when patients need tests or follow-up?</li> <li>Do all of our patients know what their targets are for good disease control (e.g., blood pressure &lt; 130/80 for diabetes or A1C &lt; 7)?</li> <li>Have patients identified their own goals for improving their health (e.g., personal targets for weight reduction)? Have they all been connected with a chronic disease self-management program?</li> <li>Do we know the community supports available for our patients to help them sustain healthy living?</li> <li>Do we refer the most complex patients to specialized clinics for care of conditions such as diabetes, COPD and congestive heart failure?</li> </ul>

1.5 Primary care summary



HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

Topic area	Key facts	Questions to consider
5. Efficiency	Only one in six elderly patients with uncomplicated hypertension receive treatment with diuretics such as thiazides. The rest are taking medications that are more expensive and no more effective.	<ul> <li>Do we have protocols or decision tools to guide the choice of medications to use? Do they take into account best quality at lowest cost?</li> <li>Are we allowing drug companies to influence our prescribing decisions?</li> </ul>
6. EMR adoption (see section 7.2)	<ul> <li>Good progress is being made in introducing electronic medical records (EMRs) to primary care. It's important, however, that EMRs are fully used as a tool to manage quality, rather than just a place to store information.</li> <li>As of September 2010, more than 4,300 physicians have enrolled in provincial EMR adoption programs that support physicians to use EMRs. The target is to have 9,000 primary care and specialist physicians enrolled by March 2012.</li> <li>Only 3.4% of adults in Ontario have e-mailed their family doctor with a medical question. Both Ontario and Canada lag behind a number of other countries.</li> </ul>	<ul> <li>If we are considering buying an EMR system, are we getting answers from potential vendors on these questions:</li> <li>Can it give me a list of all patients with certain chronic diseases?</li> <li>Can it track key indicators, such as percentage of diabetes patients with A1C under control?</li> <li>Will it send reminders when patients need follow-up or tests?</li> <li>Can it connect to pharmacies, labs, hospitals and other providers?</li> <li>Do we have a plan to minimize disruptions to workflow and productivity when implementing an EMR system?</li> </ul>
7. Health human resources (see section 7.4)	<ul> <li>From 2005 to 2009, there has been an increase in the supply of family doctors (3.4%) and specialists (6.4%) per 100,000 people. Large regional variations are present across the province in the supply of healthcare providers.</li> <li>There has been an 83% increase in the supply of nurse practitioners, but there is still only one nurse practitioner for every eight family physicians in Ontario.</li> </ul>	<ul> <li>Are we using a team approach in our practice? If not, what are the reasons for not using one?</li> <li>Could our current practice be more efficient and/or effective (e.g., do we have good communication and are we using everyone's role to its full potential)?</li> <li>What are the roles and responsibilities of the various health professionals? How can using other professionals reduce the workload in our practice?</li> </ul>
8. Population- based health (see chapter 9)	<ul> <li>Smoking has seen a major decrease, but there has been little change in other health behaviours. There is still major room to improve population screening for preventable diseases.</li> <li>Smoking rates have decreased by 25% over the past eight years. However, one in five Ontarians aged 12 and over still smoke, and rates are highest among those with low income and education and in rural areas. Also, about one in eight pregnant women still smoke.</li> <li>18% of Ontarians are obese, half the population does not get enough physical activity, and more than half do not eat enough fruits and vegetables.</li> <li>More than one in four seniors did not receive a flu shot. About one-third of women aged 50 to 69 did not have a mammogram in the past two years. Nearly one in four adult women did not have a pap test in the last three years. About one in five elderly women did not get screened for osteoporosis.</li> <li>Only about one in three people aged 50 to 74 were screened for colorectal cancer with a fecal occult blood test in 2009, but this indicator is improving quickly.</li> </ul>	<ul> <li>Do we ask our patients about their smoking cessation at each visit? Do we have a list of all smoking cessation supports in our community for our patients? Have we considered ways to improve access to smoking cessation programs and make it easier for people to use nicotine replacement therapies?</li> <li>Do we have outreach programs for people in high-risk groups? Have we made sure they know how to access them?</li> <li>Do we use flow sheets to remind us of all the health prevention interventions that need to be done during periodic health exams?</li> <li>If we have an EMR, does it generate reminders when people are due for their next health prevention service?</li> <li>Are we offering patients screening procedures performed by either a male or female provider, depending on patient preference, as well as culturally sensitive screening programs?</li> <li>Can we provide access to vaccinations outside primary care offices?</li> </ul>

# **1.6** Home care summary

HOME CARE: summary for home and community care leaders, staff and clients

Topic area	Key facts	Questions to consider
1. LTC wait times/ALC (see section 2.4 and 6.2)	<ul> <li>Despite a major increase in the number of LTC beds several years ago, wait times to get into an LTC home are still too high. The median wait time is 3.5 months (103 days), which is nearly three times higher than in the spring of 2005. For those waiting in the community, the wait is over five months; for those waiting in hospital, it is just under two months. The latter contributes to the serious problem of ALC beds in hospitals — approximately one in six hospital beds in Ontario are filled with people who could best be cared for elsewhere.</li> <li>In the past year, the number of people placed into LTC from hospital dropped by 19%, while the number placed from home rose by 15%. This could be because many communities are adopting a "Home First" approach, where hospital patients who might have been referred straight to LTC in the past are instead offered additional home care services to help them return home and allow them to make their decision about future LTC placement at home.</li> <li>Over one in five people placed in LTC homes do not have high or very high care needs. These people could potentially be cared for in other settings in the community (e.g., with more home care or in supportive housing arrangements).</li> </ul>	<ul> <li>Is home care involved early during the hospital stay for vulnerable clients?</li> <li>What additional home care services are needed to keep people out of LTC?</li> <li>What alternatives to LTC need to be considered for those who do not need the full range of LTC services? More home care services? Assisted living or supportive housing options?</li> <li>Are clients and caregivers aware of all the options and services available to them?</li> <li>Are decisions to apply for LTC placement being made prematurely for hospital patients, before they have had a chance to recover?</li> <li>What are the characteristics of people who are the most difficult to place into a LTC home, and do more specialized services in LTC homes need to be developed for such individuals?</li> </ul>
2. Falls (see section 4.6)	<ul> <li>One in four home care clients report falling in the last 90 days. There has been no major improvement in the past three years.</li> </ul>	<ul> <li>Do we do routine safety assessments? Are we checking for clutter or poor lighting in the home? Are there safety bars?</li> <li>Are we encouraging the use of mobility aids (e.g., walkers) and checking for proper use?</li> <li>Are we conducting risk assessment for falls and identifying those clients at highest risk of falling?</li> <li>Do high-risk clients receive rehabilitation to improve strength and balance?</li> <li>Are any clients on a drug with side effects that might cause a fall? If so, have we discussed safer alternatives with the doctor?</li> </ul>
3. Pressure ulcers (see section 4.6)	• Among long-stay home care clients, 1.6% have developed a new pressure ulcer (stages 2 to 4) identified over the previous six months. There has been no improvement in the past three years.	<ul> <li>Do vulnerable clients get risk assessments for ulcers? Are they up to date? Are staff regularly monitoring for early signs of ulcers? Are high-risk clients getting special padding to avoid ulcers on pressure points?</li> <li>Are home care workers well trained in identifying clients at risk for pressure ulcers and aware of appropriate prevention and treatment strategies?</li> </ul>
4. Injuries (see section 4.6)	• About one in 15 home care clients report unexplained injuries assessed over the past 90 days.	• Are we checking for safety hazards in the home (e.g., hot water temperature, electrical outlets and clutter)?
5. Bladder incontinence (see section 3.5)	• Half of long-stay home care clients have had a decrease in bladder function, or no improvement of a past bladder control problem since their previous assessment.	<ul> <li>Are there home care staff who can teach "prompted voiding" protocols or bladder strengthening exercises to clients and their informal care providers to prevent deteriorating bladder control?</li> <li>Are clients advised to stop certain foods (e.g., caffeine)?</li> </ul>

1.6 Home care summary



Topic area	Key facts	Questions to consider
6. Activities of daily living (see section 3.5)	• Close to half of long-stay home care clients experienced a new problem with activities of daily living or have an old problem that is not getting better.	<ul> <li>Are home care clients being offered physiotherapy or rehabilitation services to keep them mobile?</li> </ul>
7. Mental health (see section 3.5)	• Among all home care clients, one in 10 exhibit a sad mood and at least two depressive symptoms. There has been no improvement in the last three years.	<ul> <li>Is home care making arrangements for social activities or coordinating treatment of depression with the family doctor?</li> </ul>
8. Pain control (see section 3.5)	• Almost one-quarter of home care clients experiencing pain are not having their pain appropriately managed. This indicator has not improved in the last three years.	<ul> <li>Are home care clients getting frequent assessments of pain?</li> <li>Are home care workers communicating information about pain to the doctor so treatment plans can be adjusted?</li> </ul>
9. Readmissions (see section 3.3)	<ul> <li>About one in five patients with congestive heart failure or COPD are readmitted within a month, for any cause. There is huge room to improve these rates.</li> <li>Readmissions over the past seven years have decreased by almost half for heart attack, which is good news.</li> </ul>	<ul> <li>Are we making sure clients at high risk for readmission are seen at home right after discharge?</li> <li>Are we making sure clients leave hospital on the right medications and know what warning signs to look out for and who to call for help?</li> <li>Are we screening and monitoring high-risk clients who are at risk of readmission?</li> <li>Are clients getting the right monitoring at home (e.g., daily weight checks for congestive heart failure)?</li> <li>Do we have a process to ensure medication reviews are done routinely (e.g., MedsCheck)?</li> <li>Do we have standard discharge follow-up protocols?</li> <li>Are we promoting client self-management?</li> </ul>

### 1.7 Long-term care summary LONG-TERM CARE: summary for LTC leaders, staff, residents and family members

Topic area	Key facts	Questions to consider
1. LTC wait times (see section 2.4)	<ul> <li>Despite a major increase in the number of LTC beds several years ago, wait times to get into an LTC home are still too high. The median wait time is 3.5 months (103 days), which is nearly three times higher than in the spring of 2005. For those waiting in the community, the wait is over five months; for those waiting in hospital, it is just under two months. This contributes to the problem of ALC beds in hospitals.</li> <li>Only four in 10 people waiting for LTC placement get their first choice when placed for the first time.</li> <li>Over one in five people placed in LTC homes do not have high or very high care needs. These people could potentially be cared for in other settings in the community (e.g., with more home care or in supportive housing arrangements).</li> </ul>	<ul> <li>What alternatives to LTC need to be considered or developed for those who do not need the full range of LTC services? More home care services? Assisted living or supportive housing options?</li> <li>Are there bottlenecks that delay the admission of residents to a home? How can the admission intake process be redesigned to make it more efficient?</li> <li>Are there certain LTC homes with long waiting lists that are frequent first choices because they serve a particular cultural, ethnic or linguistic group, or because they specialize in handling particular types of illnesses? If so, can we redeploy LTC resources in our region to better serve these groups?</li> </ul>
2. Falls (see section 4.5)	<ul> <li>Falls are common and have changed little in recent years:</li> <li>One in seven LTC residents have fallen in the past month.</li> <li>There has been no major change in the rate of emergency department visits or hospitalizations as a result of falls. There are about 3 hospitalizations for falls per 100 residents each year.</li> </ul>	<ul> <li>Are we evaluating the cause of each fall? Are we doing risk assessments for falls consistently? Are they up to date?</li> <li>Are we checking for clutter, poor lighting or other hazards? Are we avoiding physical restraints, which can cause falls?</li> <li>Are we offering and encouraging the use of assistive devices (e.g., walkers), hip protectors for those at high risk and exercise programs to maintain strength and balance?</li> <li>Are we avoiding drugs that make residents dizzy or confused (see 6 below)? If a resident is on such drugs, have we considered safer alternatives?</li> <li>Are residents getting prompt help when they want to go to the washroom?</li> </ul>
3. Pressure ulcers (see section 4.5)	• One in 36 residents develop a new, serious pressure ulcer over a period of three months; that's about one in nine residents each year. Ontario can strive to achieve a value closer to zero.	<ul> <li>Are we doing risk scoring for ulcers consistently for all residents?</li> <li>Do we provide training for all staff in protocols for prevention (e.g., early detection, turning immobile residents regularly and proper techniques to avoid tearing the skin when moving a resident)?</li> <li>Do we have proper padding or special mattresses for high-risk residents?</li> <li>Do we have standard protocols agreed to by all doctors for treating pressure ulcers?</li> </ul>
4. Bladder incontinence (see section 3.4)	• 21% of residents find that their bladder control has gotten worse over the past three months.	<ul> <li>Are residents getting help with either exercises to strengthen bladder muscles or learning "prompted voiding" protocols that can help avoid incontinence?</li> <li>Are residents getting prompt help when they want to go to the washroom?</li> <li>Do residents know that some food items (e.g., drinks with caffeine) can worsen incontinence?</li> </ul>
5. Avoidable emergency department visits (see section 6.3)	<ul> <li>Potentially avoidable emergency department visits are common among LTC residents. There has been no major change in the past seven years in this area.</li> <li>Over the past seven years, there has been continuous improvement in the rate of low acuity emergency department visits by LTC residents. This is good news, but there is still be room to improve.</li> </ul>	<ul> <li>What training or support do staff need to increase their skills in handling minor emergencies without needing to transfer to the emergency department?</li> <li>Have we considered using nurse practitioners, telemedicine or better organized call schedules for physicians to improve the availability of people to assess minor emergencies within the home?</li> <li>Are family members aware of the potential risks of emergency department visits (e.g., confusion and hospital-acquired infections)?</li> </ul>



Topic area	Key facts	Questions to consider
6. Drug safety (see section 4.4)	<ul> <li>The use of drugs that should be avoided in the elderly is gradually decreasing. However, about one in five LTC residents is still prescribed these drugs.</li> <li>Shortly after entering an LTC home, one in six residents are given an antipsychotic drug that they were not receiving before (i.e., the LTC home physician — not the previous family doctor — started the drug). One in four are given a new drug for anxiety or sleep. There has been no major change in the past three years. These drugs have many risks should be avoided where possible.</li> </ul>	<ul> <li>Why are people started on sleeping pills once they enter an LTC home? What non-drug options are being tried to reduce insomnia (e.g., avoiding caffeine, reducing noise, adopting a regular sleep routine, avoiding long naps and managing underlying depression)?</li> <li>Are physicians and staff familiar with drugs to avoid in the elderly? Should some drugs be removed from the formulary?</li> <li>Does a pharmacist do regular, detailed reviews of medications, with the involvement of family and staff?</li> <li>Have we tried non-drug approaches for behavioural issues such as aggression (see 8 below)?</li> <li>Do we have an EHR to better track medication and prevent adverse drug events?</li> </ul>
7. Restraint use (see section 4.5)	• Almost one in six LTC residents were physically restrained in the previous three months. Many LTC homes are adopting zero-restraint policies, and some countries have rates lower than Ontario's.	<ul> <li>Are we educating staff and family members who ask for restraints about their hazards (e.g., falls, pressure ulcers and asphyxiation)?</li> <li>For people who wander, have we considered alternatives to restraints, such as bed and door alarms?</li> </ul>
8. Behavioural issues (see section 4.5)	• 14% of LTC residents were noted to have worsening behaviour (e.g., aggression or wandering) in the past three months.	<ul> <li>Are staff trained in communication and conflict de-escalation techniques to avoid making residents frustrated (e.g., good eye contact and one-sentence questions)?</li> <li>Can we communicate in the various languages of our residents?</li> <li>Have staff considered if this behaviour is due to an existing or new health problem, discomfort or fear? When causes of disruptive behaviour are identified, are solutions incorporated into care plans?</li> </ul>

# **1.8** Cardiovascular disease summary

Overall, there has been great improvement in the management of cardiovascular disease, but there are still areas where Ontario can do better — particularly with congestive heart failure.

- 1. Wait times for cardiovascular surgeries are within target for most patients, but there is still room to improve. For coronary artery bypass, angiography and percutaneous coronary intervention, the vast majority of patients have their procedure within the target time. There is room to improve for semi-urgent angiographies (70% completed within the target time) and semi-urgent percutaneous coronary interventions (79% completed within the target time).
- 2. More patients are on the right medications after a heart attack but, again, there is room to improve. Use of statins, beta-blockers and ACEIs/ARBs are at 87%, 79% and 79%, respectively, but experts suggest these rates should be closer to 90%.
- 3. Heart attack incidence, mortality and readmissions are declining. Over the last seven years, there has been a 66% reduction in hospitalizations for angina, with a 13% decrease in the last year.
- 4. Congestive heart failure is still associated with high mortality and readmissions rates. Over one-third of patients admitted with congestive heart failure for the first time die within the following year, and this indicator has not improved over the last six years. About 22% congestive heart failure patients are readmitted within 30 days, and this has not improved recently.
- 5. Mortality has improved for stroke but there are opportunities to do better. Stroke mortality rates are declining and almost all patients are on acetylsalicylic acid or other anti-thrombotic drugs, as guidelines suggest. However, only 13% of stroke patients arriving in an emergency department get within one hour a clot-busting drug that can reduce disability from stroke. Also, only 29% of stroke patients get in-hospital rehabilitation to help them regain their function, while experts suggest this figure should be closer to 40%.
- 6. There have been great reductions in smoking, but progress on other unhealthy behaviours related to heart disease has stalled. Smoking rates have decreased from 25% in 2001 to 19% in 2009. However, currently 18% of Ontarians are obese and half are physically inactive, and these rates have changed little.
- 7. Those with low incomes and poor education continue to be at greatest risk for unhealthy behaviours. The poorest Ontarians are 36% more likely to experience a heart attack than the richest Ontarians. Ontarians with low incomes or poor education are twice as likely to smoke as people with higher incomes or a better education.

Questions for healthcare leaders and staff to consider:

- How well are recommended practices being followed? Are primary care
  providers using tools like flowsheets for heart failure or coronary artery
  disease to monitor if the right drugs and tests are being done? Can
  electronic medical records flag which patients are due for follow-up or
  require closer surveillance because they have problems managing their
  condition well?
- Are hospitals using tools like standardized orders at admission or checklists at discharge to ensure patients are getting all the recommended drugs and tests? Are individual providers getting feedback on how well they are following recommended practices?
- Are patients engaged in their care? Have we verified they understand the information given to them? Are they encouraged to make their own decisions regarding their care? Are they regularly monitoring key measures, like blood pressure or weight (for heart failure)? Do they have an action plan that tells them what to do or who to call if they get worse?
- Can we target interventions for reducing unhealthy behaviours towards those with the most to gain?

Key questions for patients with heart disease to ask themselves or discuss with their healthcare provider:

- Am I on all the right medications? Ask about acetylsalicylic acid (Aspirin), a cholesterol-lowering drug (e.g., a statin), an ACEI/ARB and a beta-blocker for past heart attacks, blocked arteries and congestive heart failure.
- Am I getting all the right monitoring? This includes blood pressure checks, periodic cholesterol tests and, for congestive heart failure patients, an echocardiogram and daily weight monitoring.
- Do I know the early signs of a stroke (such as weakness, numbness, inability speak or write, double vision) so I know when to go to the hospital immediately?
- Do I know my targets for blood pressure and cholesterol? Typical blood pressure targets are 140/90, or 130/80 for those who also have diabetes; the target for LDL ("bad cholesterol") is two or less.
- What am I doing to eliminate smoking, improve my physical activity and achieve or maintain my ideal weight? What personal goals would I like to set for myself? What help do I need e.g., nutrition counselling, exercise groups, smoking cessation aids and support from friends or family?

# **1.9** Diabetes summary

# HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

Overall, there are cautious signs of improvement in managing diabetes, but there is still a lot of room to do better — particularly in monitoring and screening patients' conditions and risk factors and filling prescriptions. If Ontario is to see further progress, it will also be important for patients to be engaged in managing their own care and setting their own targets and plans for improving their lifestyle choices.

- 1. The incidence of serious complications and hospitalizations from diabetes has decreased in the last five years, but there is still room for improvement. Slightly more than one in 25 diabetes patients will experience a major complication (death, heart attack, stroke, amputation or kidney failure) in a year.
- More patients are on the right medications for diabetes, but Ontario is still far from ideal. Only 60% regularly fill their prescriptions for a cholesterol-lowering drug (e.g., a statin), 67% for an ACEI/ARB, and 48% for both. Experts suggest that most elderly diabetes patients should be on these drugs.
- 3. **Monitoring diabetes conditions is poor.** Although all diabetes patients should get regular eye exams, only about half do.
- 4. Smoking has decreased sharply in the past decade, but progress on other unhealthy behaviours that lead to or worsen diabetes has stalled. Currently 19% of Ontarians smoke, 18% are obese and half are physically inactive.
- 5. Those with low education or income level are at greatest risk of unhealthy behaviours that lead to or worsen diabetes. Smoking and physical inactivity is worse among those with low income or education. Obesity is worse among those with low education.

Key questions for healthcare leaders and staff to consider:

- Are we using methods such as diabetes flow sheets to remind us of all the best practices?
- If we have an EMR, does it provide us with data on the percentage of our diabetes patients who are on the right drugs (e.g., a statin and ACEI/ARB) or who have had a recent A1C test or eye exam? Have we set up the EMR so that it reminds us when diabetes patients need testing or follow-up?
- Have our patients identified goals for improving their health? Have they been connected with a chronic disease self-management program?
- Do we use a monofilament in the office to do proper diabetes foot exams?
- What are we doing to reach out to the most vulnerable populations to ensure they are getting services targeted to their education level, culture or language?

Key questions for patients with diabetes to ask themselves or discuss with their healthcare provider:

- Am I on all the right medications? Ask about a statin, ACEI/ARB and acetylsalicylic acid (aspirin), in addition to medications to control blood sugar.
- Am I getting all the right monitoring? This includes eye checks, foot exams and urine tests, as well as regular blood tests for cholesterol and A1C (a three-month average of blood sugar).
- Am I doing my own monitoring of blood sugar and blood pressure? Do I keep a log of my measurements at home?
- Have I set targets for blood sugar, blood pressure (ideally 130/80) and weight with my doctor?
- Am I eating properly and staying physically active? What personal goals do I
  want to set for improving my health? What support do I need to achieve my
  goals e.g., nutrition counselling, exercise groups, smoking cessation
  aids and support from friends or family?

# 1.10 Cancer summary

There has been a steady decline in death rates from some major cancers in Ontario, likely due to better treatments and decreased smoking. However, wait times for urgent cancer surgery and systemic treatments (chemotherapy) need to be shortened, and more progress is needed in reducing unhealthy behaviours and improving cancer screening.

- 1. Rates of lung cancer and mortality from breast cancer have improved over the last 10 years. This is good news and may be due to reduced smoking in previous decades and better treatments over time.
- 2. Wait times for cancer care can be improved, especially for urgent cancer surgeries and systemic treatments (e.g., chemotherapy). Only 67% of urgent (priority 2) patients have their surgery within the recommended timeframe. Some hospitals, including North York General Hospital, have achieved 100% through well-designed and efficient scheduling processes. Other hospitals could do the same. Wait times for radiation therapy have improved, with nearly four in five patients treated within the target timeframe, but there is still room to do better. Nearly two-thirds of patients needing systemic treatments are seen by a specialist within the 14-day target. After being seen by the specialist, only about half of the patients get treatment within the target timeframe.
- 3. Screening rates for breast and cervical cancer can still improve. Approximately one-third of the women who need a mammogram, and one-quarter of women who need a Pap test still do not get them. Screening rates for colon cancer are increasing but, at 35%, are still too low.
- 4. Smoking has improved significantly, but progress in reducing other unhealthy behaviours, such as inadequate consumption of fruits and vegetables, obesity, physical inactivity and heavy drinking, has stalled recently. At present, the rates of these behaviours in the population are 19%, 56%, 18%, 49% and 22%, respectively. These unhealthy behaviours have been linked to breast, colon, lung, liver, kidney and other cancers.
- 5. People with low incomes and poor education levels continue to be at greatest risk for unhealthy behaviours and for not receiving preventive screening. For example, smoking rates are 35% for those without a high school diploma and 15% for those with post-secondary education. Rates for mammography screening are 59% among low-income women, compared to 71% for those with higher incomes. Future plans to battle cancer need to consider strategies that target the most vulnerable in the population.

Key questions for healthcare leaders and staff to consider:

- What targets are we setting for wait times? If some places have achieved major improvements (e.g., North York General Hospital), why can we not do the same thing?
- Have we mapped out the processes involved in arranging cancer surgery, radiation or chemotherapy? Where are the areas of waste, duplication, error or missed hand-offs? What are we doing to make our processes more timely and reliable?
- Do we have information systems to ensure that everyone due for cancer screening is reminded?
- What are we doing to reach out to the most vulnerable populations?

Key questions for people to ask themselves or discuss with their healthcare provider:

- Which screening tests do I need for my age and gender and how often? When am I due for each of these?
- What am I doing to eliminate smoking, become more physically active and achieve or maintain my ideal weight? What personal goals would I like to set for myself? What help do I need e.g., nutrition counselling, exercise groups, smoking cessation aids and support from friends or family?

# 1.11 Mental health summary

# HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

Suicide, depression and poor handoffs of care remain important areas of concern in mental health. The description of mental health for Ontarians, however, is seriously limited because there are huge gaps in the health care system's ability to measure the quality of services provided.

- 1. The rate of intentional self-harm presenting in emergency departments has dropped in recent years, but suicide rates remain constant. There is still room for improvement. At present, there are 88 emergency department visits for intentional self-harm per 100,000 people. Women and in those in lower income brackets are at greatest risk. In 2007, 1,000 people in Ontario committed suicide. There has been no change in the past seven years. The tracking of suicide is poor, and there is concern that suicides may be underreported.
- 2. Many people with mental health problems do not get primary care follow-up after they are discharged from hospital. Almost four in 10 patients discharged from hospital after treatment for a mental illness do not have a primary care physician visit within 30 days, and this has not improved in the last three years.
- 3. Depression is a significant problem among frail or elderly individuals. Among people receiving home care, 9.2% show serious signs of depression, such as profound sadness or withdrawal from normal activities. Currently, 26% of those living in LTC showed increasing symptoms of depression or anxiety in the preceding three months.
- 4. Inappropriate behaviour, such as aggression, agitation or wandering, is common among LTC residents. Since their last assessment (three months ago), 14% of residents exhibited worsening behaviour. These behaviours are particularly common among those with Alzheimer's or other dementias.
- 5. Many people entering a LTC home for the first time are placed on an anti-psychotic or anti-anxiety drug that they weren't on before. One in six residents are given a new antipsychotic drug, and one in four are given a new drug for anxiety (often used as sleeping pills). These drugs have many risks should be avoided where possible.

Key questions for healthcare leaders and staff to consider:

- Do we screen for warning signs of depression?
- Are we overprescribing antipsychotic and anti-anxiety drugs? Are we using non-drug methods to deal with agitation, insomnia or anxiety? Are we offering people in home care or LTC social activities or counselling? To avoid frustration among LTC residents, are we using strategies such as one-sentence communication, maintaining good eye contact and conflict de-escalation techniques?
- Are we ensuring regular medication reviews by a pharmacist, with input from the client/resident, the family and staff?
- If we have an EMR, does it monitor drug utilization patterns?
- What are we doing to reach out to the most vulnerable populations to
  ensure they are getting the counselling they need to reduce the incidence
  of self-harm? Are we making sure the services we provide take into account
  people's culture, financial and family situation?

Key questions for family members of patients experiencing symptoms of mental illness to ask themselves or discuss with their healthcare provider:

- Is my family member showing signs of depression? What is being done to treat these symptoms? If my family member is in LTC, is there anything in the surroundings that could be contributing? What could be done to improve participation in activities or social networks?
- For people who have just been discharged from hospital for a mental health problem, has follow-up been arranged with a family doctor and other mental health services?
- Is my family member being given antipsychotic or anti-anxiety/hypnotic drugs or sleeping pills (such as Valium or Ativan)? Have I discussed with his or her doctor if these medications are necessary and if there are alternative methods to deal with agitation, sleeplessness or anxiety?

# 1.12 Data advocacy

HQO has an ongoing commitment to report on the performance of Ontario's publicly funded healthcare system. To fulfill that commitment, the organization needs access to up-to-date, high-quality data on the delivery and impact of the full range of healthcare services provided to Ontarians. Ontario already has some of the best data in Canada, collected by a range of different organizations. However, there are still gaps; in some cases, the data exist but are inaccurate or impossible to access, while in other cases, there are no data at all.

HQO believes it is important to advocate for improved data. Better data means improved reporting, and improved reporting is essential to better care. This year, HQO has worked with stakeholders and experts to identify the questions about quality that we cannot answer without better data, why the questions are important, and ideas on how data can be obtained.

Торіс	The questions Ontario needs to answer	Why this is important	Ideas on how to get data
Chronic disease management	<ul> <li>Are best practices being followed? Are patients getting the right drugs, monitoring?</li> <li>Is chronic disease well controlled? Are physical measures (e.g. blood pressure, weight) in a desirable range?</li> </ul>	Chronic diseases affect a large number of Ontarians. Billions of dollars are being spent on treatment, yet we do not know if we are doing a good job.	A Diabetes Registry is in progress which will soon fill some of these gaps. EMRs in primary care could be used to extract this data for other conditions in the future.
Appropri- ateness of services	<ul> <li>Are people receiving unnecessary tests, surgeries or other procedures?</li> </ul>	Sustainability of the health care system depends on being able to reduce expenditures for unnecessary care while preserving dollars for when it's needed.	Collect data on at least a sample of services about whether appropriateness criteria were met. Consider developing electronic order entry systems where this data could be entered.
Mental health outcomes and access	<ul> <li>Do people feel any better or function in society any better after receiving mental health services?</li> <li>Are people getting access to the services they need?</li> </ul>	There are thousands of mental health service providers in Ontario, but little information about the impact of their programs.	A mental health service user experience survey could be used to measure outcomes and access.
Patient report- ed outcomes	- Do people function better after treatment or surgery? Has their ability to move, see, hear, function normally or live pain-free improved?	Ontario has spent billions of dollars in reducing wait times for surgeries, but we do not know whether or not people are truly better off.	Consider patient surveys after receiving a procedure. Consider adding questions to existing surveys.
Drug utilization and safety	<ul> <li>What drugs are people on?</li> <li>How frequently is harm from avoidable adverse events from drugs occurring?</li> </ul>	Data on drug use is needed to track if people with certain conditions are on the right drug; currently, this is available only for people aged 65 and over or on social assistance. There is very little data on drug errors (e.g. wrong dose, wrong drug, wrong site) and the level of harm caused.	Ontario could learn from Alberta and Saskatchewan*, which already have data systems to track all drugs and allergies for all persons. Ontario currently has a database where hospitals report drug errors and harm voluntarily <sup>†</sup> ; this process could be expanded and made mandatory.
Hospital safety — broader measures	<ul> <li>What is the overall probability of being harmed during a hospital stay?</li> <li>How frequently do these events occur: missed diagnoses, surgical complications, "never events" (e.g. wrong-site surgery), death from sepsis, or avoidable harm during a transfer between small and large hospitals?</li> </ul>	Ontario has invested heavily in measuring hospital-acquired infections, but relatively little attention has been paid to measuring other serious patient safety issues in hospital.	Consider enhancing the existing process where chart abstractors code data into the Discharge Abstract Database, or using "trigger tools" where physicians and nurses review charts that are flagged as being anomalous (e.g. length of stay for a condition unusually long).

\* Alberta's Pharmacy Information Network and Saskatchewan's Pharmacy Information Program offer this functionality.

<sup>†</sup> These data are reported to the Institute for Safe Medication Practices — Canada.

# HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

Торіс	The questions Ontario needs to answer	Why this is important	Ideas on how to get data
Health equity	<ul> <li>What is the gap in outcomes or access to services between different groups in society?</li> </ul>	Information is available on income, education, ethnicity and immigration status on national survey data, but not on other data sources. This limits the ability to monitor societal inequalities. Also, Ontario's databases currently do not identify First Nations status.	Consider adding more questions on income or ethnicity on other surveys. Consider data linkages with federal government data on "Registered Indian" status. Consider ways of adding this data to chronic disease registries as they are developed.
Cost of services	- How much does a health care service cost? Who can provide a service at the highest quality and lowest cost?	It is important to identify which models of care achieve the best outcomes at lowest cost, so that others can achieve the same. Currently, there is very little data on costs per services.	Ontario has limited experience with "case- costing" in hospitals that could either be expanded or leveraged to other sectors.
Patient experience (beyond hospitals)	- Do people who use the health care services feel the system is designed around their needs? How good is communication, courtesy, involvement in decisions, or responsiveness to concerns?	Patient-centredness is a core attribute of quality. It is measured routinely in hospitals but not in other sectors. The Excellent Care for All Act will eventually make patient surveying mandatory across the system.	Surveys are already in development for home care and long-term care. Consider patient / user experience surveys in primary care and mental health.
Accurate reporting of mortality	- What is the infant mortality rate? Suicide rate? Mortality rate for other conditions? These are currently tracked, but there are concerns about accuracy.	Experts are concerned that infant deaths are undercounted due to failure to complete forms. Suicide may be underestimated because information on cause of death may only become available after the original death certificate is complete.	Consider re-designing the process used to track deaths or determining cause of death.

### 2.1 Wait times in emergency departments

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
promptly and efficiently in EDs.	Long ED wait times are inconvenient and, in some cases, negatively affect a patient's health. <sup>1</sup> Spending a long time in the waiting room, or on hallway stretchers waiting for admission, can also compromise comfort and privacy.	Ontario's one in five residents who visit an ED at least once a year. $^{\rm 2,3}$

Indicator	Value	Time trends & comparisons	Bottom line
90 <sup>th</sup> percentile wait time for ED patients: → High complexity <sup>†</sup> → Low complexity <sup>††</sup>	12 h* 4.3 h	16 8 0 Apr 08 5ep 10	The maximum time nine in 10 patients spend in an ED decreased in the last two years by two hours for high complexity patients (from 14 to 12 hours), and by half an hour for low complexity patients (from 4.8 to 4.3 hours). Policies, investments and quality improvement activities likely contributed to this decrease (see page 26). However, there is room to improve, as the targets for high and low complexity patients are 8 and 4 hours respectively. <sup>4</sup>
Percentage of patients who left without being seen	4.7%**	10 5 0 Apr-Jun 08 BETTER Jan-Mar 10	One in every 20 Ontarians who visit the ED leave without being seen by a physician, likely because they were tired of waiting. <sup>5</sup> This indicator has improved over the last two years, probably because wait times have decreased. There is still room to do better.
Percentage of ED care completed within recommended time:	81%***	100 50 0 Apr-Jun 08 BETTER BETTER Apr-Jun 10	Over the past two years, the percentage of people who were treated in the ED within the recommended time frame has increased modestly from 78% to 81%. While this improve- ment is encouraging, more than half of those patients who are admitted to hospital are still spending more time in the ED than is desirable. There is room for improvement.
<ul> <li>Admitted patients<sup>‡</sup></li> <li>Non-admitted high complexity patients<sup>†‡</sup></li> <li>Non-admitted low complexity patients<sup>††</sup><sup>‡‡</sup></li> </ul>	41%§ 90%§ 85%§	BETTER BETTER 0 50 100 Percent	

Data sources: \* National Ambulatory Care Reporting System Database (NACRS), September 2010, provided by MOHLTC. \*\* NACRS, January to March 2010, provided by MOHLTC \*\*\*Emergency Department Reporting System (EDRS), April to June 2010, provided by Cancer Care Ontario (CCO). § EDRS, FY 2009/10, provided by CCO. † Includes patients classified as triage level 1 (resuscitation), 2 (emergent), or 3 (urgent). † Includes patients classified as 4 (semi-urgent) or 5 (not urgent). ‡ Target eight hours. \*\* Target four hours.

2.1 Wait times in emergency departments



Indicator	Value	Time trends & comparisons	Bottom line
Median time to physician assessment	1.2 h**	State of the second sec	People are waiting too long to see a doctor in the ED, according to national guidelines <sup>6</sup> . The median time from when a patient is registered in the ED to when they see a doctor is 1.2 hours, and this has not changed over the past two years. The current wait time is 18 minutes worse than what it was in 2004 (see last year's report <sup>7</sup> ). There is obvious room for improvement.
Median time from admission to transfer to bed	3.4 h**	5.0 Sector 2.5 O Apr - Jun 08 BETTER Jan - Mar 10	Half of patients waited more than three hours for a bed after being admitted to hospital from the ED. There has been no improvement in the past two years. This is consistent with the above finding that it is the admitted patients who are least likely to get their ED care within the recommended time. Future efforts need to focus on this group.
Percentage of Ontarians who visited the ED and waited two hours or more for treatment after arrival	50% <sup>\$\$</sup>	100 - Generation of the second	Half of Ontarians who visited the ED said they waited two hours or more for treatment after arrival. <sup>†††</sup> Compared to 10 other countries surveyed, Canada and Ontario are doing poorly. Thus, although the recent reductions in time spent in the ED described above are welcome, Ontario is still far from where it needs to be.

Data sources: \*\* NACRS, January to March 2010, provided by MOHLTC Se Commonwealth Fund International Survey of Adults, 2010. \*\*\* This indicator measures 'treatment', which typically occurs after the physician assessment which has a mean wait time of 1.2 hours.

# 2.1

# **Root Cause**

**Inefficient processes for discharging patients.** Patients are occupying beds in the ED when they should have been discharged (e.g., to home, home care, longterm care).<sup>8</sup>

### Backlogs elsewhere in the

**hospital** prevent patients admitted in the ED from being sent to a bed on the ward in a timely manner.<sup>9,10</sup> One study found a 10% increase in hospital occupancy was associated with an 18-minute increase in ED length of stay.<sup>11</sup>

### Inefficiencies within the ED,

such as staffing shift change handoffs, accessing equipment and supplies, or delays in the admission process, which can slow processes.

### **Ideas for Improvement**

Develop better care coordination and move patients to the right place as soon as possible so they are not occupying beds unnecessarily.

- Provide alternatives for alternative level of care (ALC) patients (see section 6.2).
- Start discharge planning early and request home care assessments at appropriate times to avoid delays.
- Use bed tracking systems to reduce the time between when a patient leaves a bed and when a new patient can use it.<sup>12</sup>
- Smooth the inflow and outflow of admitted patients. For example, Monday admissions tend to stay longer, as do people discharged on Mondays. When admissions outnumber discharges in one day, ED waits are longer the following day.<sup>13</sup> Better discharge planning and more even access to certain services throughout the week could smooth out flow.
- Spread elective surgery cases more evenly throughout the week.<sup>14</sup> This allows for more flexibility to accept surgical cases from the ED as they arise.

#### Make specific process improvements within the ED,<sup>15, 16</sup> such as:

- Create a fast-track area for less serious cases.
- Create special units for patients who need to be under observation for several hours.
- Make simple changes in location of supplies and equipment in the ED, and use consistent locations for common supplies to minimize wasted staff time walking back and forth.
- Use process management software to analyze data and processes to reduce non-productive work and improve use of resources.<sup>17</sup>
- Use flexible staff scheduling. Work with staff to create arrangements where staff can be brought in for a sudden surge in visits or stay at home when it's quiet.<sup>18</sup>

Many hospitals have already begun adopting these ideas; future progress may depend on ensuring these ideas are used more consistently in the province.

HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

### **Root Cause**

Avoidable demand from

**non-urgent cases**, <sup>19</sup> which could be addressed in settings other than the ED.

See section 6.3 for more on avoidable emergency department visits.

### **Ideas for Improvement**

Divert non-urgent cases away from the ED to other alternatives:<sup>20</sup>

- Improve access to after-hours primary care.<sup>21, 22</sup>
- Encourage people to call Telehealth Ontario for advice on whether an ED visit is needed.
- Implement targeted programs for specific health conditions that do not necessarily require hospital admission.
- Create "urgent care" clinics that have instant access to imaging and labs, where lower acuity patients can be seen. For example, Alberta implemented five urgent care centres that reduced ED use by 22% to 25% during their hours of operation.<sup>23</sup>
- Divert patients to the most appropriate service for their condition. For example, North York General
  Hospital developed a successful program that diverts mental health patients from the ED to more appropriate,
  community-based care services. More than half of patients who were referred to the Emergency Department
  Diversion Program may have been admitted to an in-patient bed if the program's services had not been available.<sup>24</sup>

Avoidable demand due to poorly controlled conditions that could have been addressed in primary care before they escalated to emergency care.

**Improve primary care services**. Improve access to primary care (see section 2.2), including after-hours primary care and management of chronic diseases (see section 3.2), so patients are less likely to require emergency care.

# What is Ontario doing?

- In 2008, Ontario announced \$109 million in funding to reduce ED waits, by adding more home care support; dedicated nurses to care for patients
  who arrive at EDs by ambulance to ease ambulance offload delays; and nurse-led outreach teams to provide care in long-term care homes and
  avoid transfers to ED.<sup>25,26</sup> Other announcements have committed to adding 27 physician assistants in EDs<sup>27</sup> and 90 ED nursing positions.<sup>28</sup>
- The ED Pay-For-Results Program provides incentives to hospitals with high ED volumes and long wait times to reduce length of stay. It has helped
  participating hospitals lower overall wait times by 4.7 hours (28%) for patients requiring complex medical care or hospital admission, and by 1.4
  hours (22%) for patients with minor conditions. A further \$100 million investment will expand the program to 25 more hospitals, for a total of 71.<sup>29</sup>
- The ED Process Improvement Program (ED-PIP) has provided 64 hospitals with on-site coaching support, quality improvement training and other resources on how to lower wait times and improve patient experience. The program will support 90 hospitals by FY 2011/12.<sup>30</sup>

### 2.2 Access to primary care

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
All Ontarians have a regular primary care provider — a family doctor or nurse practitioner — and no one has to wait long to see him or her.	Patients who do not get care when they need it may wait and get sicker — or they may resort to an emergency department <sup>31</sup> or different provider who is unfamiliar with their medical history. Both waiting and/or seeing an unfamiliar provider can negatively affect patients' health <sup>32</sup> and waste time and resources.	Ontario's 13 million residents.
Infants see a primary care provider for follow-up soon after birth.	Without follow-up after birth, certain illnesses could be missed and the risk of infant morbidity and mortality increases. $^{\rm 33}$	Ontario's 136,000 babies born every year, and their families. <sup>34</sup>



Data sources: \* Based on the Primary Care Access Survey, a quarterly phone survey of Ontario adults (aged 16 and over). Most recent results represent average data from FY 2009/10 \*\* Ontario Health Insurance Plan, calculated by Institute for Clinical Evaluative Sciences, FY 2009/10. \* This indicator includes "other primary care providers" where as the previous indicator does not. Also, each indicator uses different survey instruments.

Hence, the results for the two are different.

2.2 Access to primary care



HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

### **Root Cause**

**Inconvenient or inefficient patient scheduling.** When wait times are long, office staff waste time responding to requests to change or cancel appointments and triaging patients by urgency.

Unnecessary work is done in primary care offices. Wasted staff time means less time for appointments.

**Inefficient or inconsistent processes exist,** such as wasting time searching for information.

# There is a lack of teamwork and inefficient use of staff.

Doctors may be performing tasks or procedures that could be performed by another healthcare professional. Better teamwork could increase the number of patients in a primary care practice and improve the availability of appointments.

### **Ideas for Improvement**

**Advanced access scheduling**,<sup>38,39</sup> is a method of scheduling appointments that aims to reduce wait times to see a healthcare provider.<sup>40</sup> Basic principles of queue management can bring wait times for appointments close to zero:

- Carefully track incoming requests and actual slots available.
- Match supply and demand<sup>41</sup> in order to make more slots available to accommodate people who arrive with urgent problems (e.g., if Mondays are busiest, schedule more slots then and put optional meetings on slower days).
- Work down the backlog. If supply and demand are matched but there is still a queue, then work extra hours or hire extra help for a limited period of time to eliminate the queue.
- **Reshape the demand**<sup>42</sup> to help ensure supply and demand are matched. Reduce unnecessary demand, for example by eliminating unnecessary follow-up visits or handling minor issues over the phone or by email (e.g., providing lab results, or renewing certain prescriptions).
- Establish contingency plans, to bring the queue back to zero when there is a surge in waits (for example, after someone goes on vacation).

The Athens District Family Health Team, using these ideas, decreased the wait time for appointments from 27 days to between zero to two days.<sup>43</sup> For more information, see www.ohgc.ca/pdfs/access.pdf.

Use **email or online scheduling systems**<sup>44</sup> to schedule or cancel appointments, as telephone lines are often busy or the office may be closed.

**Improve office efficiency.**<sup>45</sup> Simple steps that save minutes or seconds of each clinic visit can add up to days or weeks of saved time over a year. These include setting up every patient room in the same manner, organizing patient records more effectively (see below), and using flow mapping to clarify and standardize complex processes to identify where time can be saved.

**Have a well-functioning electronic medical record (EMR) system.**<sup>46</sup> EMRs can save time by decreasing paperwork and making it easier to access test results or other information in real time. The EMR can also be set up to monitor statistics on wait times or office efficiency.<sup>47</sup>

**Establish interdisciplinary primary care teams and use all team members to the fullest.**<sup>48,49</sup> This can help increase the number of patients served; for example, after the establishment of family health teams in Peterborough, 17,000 without a family doctor gained access to primary care.<sup>56</sup> Ensure roles and responsibilities are clear for each team member.<sup>50</sup> For example, nurses can do preventive health counselling, give immunizations and coordinate patient care;<sup>51</sup> nurse practitioners can diagnose, treat and monitor a wide variety of conditions;<sup>52,53</sup> pharmacists can help manage medication use;<sup>54</sup> and dieticians can provide nutritional counselling and promote behaviour change.<sup>55</sup> Office staff can check height and weight, check blood pressure with an automatic cuff and/or ensure that data is input properly into EMRs.

Consider expanding legislated scopes of practice for capable healthcare professionals.

# What is Ontario doing?

- 170 family health teams (FHTs) provide 2.5 million Ontarians (416,000 of whom didn't have a family doctor) with access to primary healthcare, with 30 more scheduled to open in August 2011.<sup>57</sup>
- The Quality Improvement and Innovation Partnership (QIIP) provides training to FHTs on how to implement advanced access.<sup>58</sup>
- 7,514 physicians cared for 9.35 million patients through primary healthcare models in October 2010, up from 2,034 in January 2004.<sup>59</sup>
- The MOHLTC has funded nurse practitioners in primary care settings and committed to fund up to 500 registered nurses in physician practice groups focusing on aging at home, end of life care, and mental health and addictions.<sup>60</sup> By the end of 2012, MOHLTC aims to have 25 nurse practitioner-led clinics operational, caring for more than 40,000 people.<sup>61</sup>
- Northern and rural communities have access to the Northern and Rural Recruitment and Retention Initiative, including free tuition for return of service, outreach programs, and support for visiting physicians and health centres.<sup>62</sup>

### 2.3 Treatment wait times and access to specialists

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
No one has to wait a long time to see a specialist.	Patients worried about a disturbing symptom may experience unnecessary anxiety waiting to see a specialist for a diagnosis. They may also get sicker while waiting, leading to a more extensive surgical procedure or a more advanced stage of the disease.	Ontarians requiring a specialist referral. Six people are referred to a specialist each year for every 10 people who have a family doctor. <sup>64</sup>
No one has to wait longer for hip and knee replacements than the target wait time.	Patients will endure pain and suffering for a longer period of time. <sup>65</sup> They may also lose their ability to participate in activities and exercise, <sup>66</sup> which could worsen other conditions such as diabetes, hypertension and depression.	Ontario's 35,000 people who have hip or knee replacements each year.
No one has to wait longer for cataract surgery than the target wait time.	People will endure poor vision for a longer period of time and may experience more falls. $^{\rm 67}$	Ontario's 139,000 people who have cataract surgeries each year. <sup>68</sup>
No one has to wait longer for cardiac procedures — angiography, percu- taneous coronary intervention <sup>†</sup> or coronary artery bypass graft — than the target wait time.	Waiting too long for coronary interventions increases risk of death. <sup>69</sup>	Ontario's 64,364 people who have angiographies, 20,598 people who have percutaneous coronary interventions and 8,308 people who have coronary artery bypass grafts each year. <sup>70</sup>
No one has to wait longer for cancer treatments — surgery, radiation or systemic therapy (chemotherapy) — than the target wait time.	Patients anxious about a cancer diagnosis may experience high levels of stress waiting for treatment, which makes wait times undesirable, even if they do not statistically affect survival.	Ontario's more than 62,000 people who will be diagnosed with cancer this year and their families. <sup>71</sup>
No one has to wait longer for CT or MRI scans than the target wait time.	If a CT or MRI scan is not carried out promptly to find cancer, surgery may be delayed <sup>72</sup> and patients may experience unnecessary anxiety waiting for a diagnosis.	Ontario's residents who receive 1.8 million CT and 570,000 MRI scans each year. <sup>73</sup>

Procedure	Priority 1* (immediate)	Priority 2* (high urgency)	Priority 3* (medium urgency)	Priority 4* (low urgency)
MRI/CT scan	Immediate	2 days	10 days	4 weeks
Cataract surgery	Immediate	6 weeks	12 weeks	26 weeks
Hip and knee replacements	Immediate	6 weeks	12 weeks	26 weeks
Cancer surgery	Immediate	2 weeks	4 weeks	12 weeks
General surgery	Immediate	2-4 weeks	12 weeks	26 weeks
Cardiac procedures (coronary artery bypass graft, angiography, percutaneous coronary intervention <sup>†</sup> )	Wait time targets are specific to each patient.			
Cancer — radiation therapy	From 1 to 14 days depending on the patient's condition.			
Cancer — from referral to consulta- tion with specialist for chemotherapy	14 days			
Cancer — from consultation with specialist to start of chemotherapy	14 days			

Data sources:

MOHLTC, Ontario Wait Times Strategy, provided by Cancer Care Ontario and Cardiac Care Network. Note that the wait for surgery is defined as starting the day the surgeon decides to operate and the patient agrees, and ending the day the surgery is performed. Target wait times vary depending on the priority score, indicating the seriousness of the conditions, assigned by the main physician.

Also known as balloon angioplasty, where a catheter with a balloon is threaded into the artery of the heart to open blockages. Usually, a stent is inserted to keep it open.



Indicator — specialist wait times	Value	Time trends & comparisons	Bottom line
Percentage of people who report waiting four weeks or more for an appointment after they were advised to see or decided to see a specialist doctor (or consultant)	51%*	Percent CANADA ONTARIO NORWWY FRANCE AUSTRALIA SWEDEN NELLAND SWITZERLAND SWITZERLAND SWITZERLAND SWITZERLAND SWITZERLAND	Half of Ontarians report waiting four weeks or more for an appointment to see a specialist doctor (or consultant). Compared to 10 other countries surveyed, Canada and Ontario have the worst standing on achieving timely access to specialist doctor/consultant care. There is major room for improvement.

Data sources: \* Based on Commonwealth Fund International Survey of Physician Practices, 2010

Indicator — cancer wait times	Value	Time trends & comparisons	Bottom line
90 <sup>th</sup> percentile wait time for cancer surgeries	51 days*	<sup>5</sup> <sup>6</sup> <sup>6</sup> <sup>6</sup> <sup>6</sup> <sup>6</sup> <sup>6</sup> <sup>6</sup> <sup>6</sup> <sup>6</sup> <sup>6</sup>	The maximum amount of time nine in 10 people will wait for cancer surgery is just under two months. Wait times decreased from 2005 to 2008, but there have been no further improvements in the past two years. There is still room for improvement.
Percentage of cancer surgeries done within target: Priority 2 Priority 3 Priority 4	67%* 78% 93%	100 100 100 100 100 100 100 100	One-third of urgent (priority 2) cancer surgery patients are not getting their surgery within the recommended two weeks. This is still a major problem to be addressed. The results for priority 2 do appear to have improved in the last year, but it is possible that much of this apparent improvement may have been due to changes in coding practices. The number of cases coded priority 2 have dropped by 40%, while priority 3 and 4 cases have increased by 17% and 25%, respectively.
Percentage of patients where radiation therapy started within target (from being ready to treat to getting treatment)	84%**	100 100 50 0 Apr 09 Dec 10 BETTER BETTER	More than eight in 10 patients receive radiation therapy within the targeted time. This has improved in the past two years, but opportunities to improve further need to be explored.
Percentage of systemic treatment (chemotherapy) com- pleted within target: → Referral to consult → Consult to treatment	61%** 47%	100 better 50 Apr 09 Dec 10	Nearly two-thirds of patients needing systemic treatment are seen by a specialist within the 14-day target. This indicator has improved in the last two years. After being seen by the specialist, almost half of the patients receive treatment within 14 days. This component of a patient's wait, however, has not improved. There is still significant room for improvement.

Data sources: \* Cancer Care Ontario and Wait Times Information System; data values represent data for December 2010. \*\* Cancer Care Ontario; data values represent data for December 2010.

31

# 2.3

Indicator — CT/MRI wait times	Value	Time trends & comparisons	Bottom line
90 <sup>th</sup> percentile wait time for: → CT scans → MRI scans	29 days* 113 days	<sup>150</sup> 75 <sub>Aug</sub> Sept 05 Dec 10	The maximum amount of time nine in 10 people wait for a CT scan is about one month. This is a remarkable improvement; back in 2005, the wait was almost three months. Wait times for MRI, however, remain stuck at around four months, the same level seen in 2005. Wait times did decrease in 2008 but have since rebounded upwards to their previous level. The number of CT scans has almost doubled in the last seven years, while the number of MRI scans has almost tripled. <sup>74</sup> These increases in volume appear to have helped reduce waits for CT, but not for MRI. Clearly, there is room for improvement, particularly for MRI wait times.
Percentage of CT scans done within target: → Priority 2 → Priority 3 → Priority 4	90%* 64% 81%	100 50 0 Jan 08 10EAL 4ID	Over the past three years, the percentage of CT scans done within the target time has improved at all priority levels. High priority patients are the most likely to have their CT scan done within the target time, which is reassuring. While the improvement is encouraging, there is still room to do better.
Percentage of MRI scans done within target: → Priority 2 → Priority 3 → Priority 4	75%* 51% 30%	100	Over the past three years, MRI waits have improved for urgent priority (2) cases but gotten worse for low priority (4) cases. At present, three-quarters of urgent priority patients received an MRI within the target time frame, but only 30% of priority 4 patients did so. There is major room for improvement.

Data sources: \*Cancer Care Ontario and Wait Times Information System; data values represent data for December 2010.

# HOSPITAL LONG-TERM CARE **HOME CARE**

**PRIMARY CARE** 

Indicator cardiac wait time		Value	Time trends & comparisons	Bottom line
graft	ary bypass graphy taneous ary	33 days* 17 days 15 days	50 25 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	The 90 <sup>th</sup> percentile wait times for each of these cardiac procedures have remained fairly constant over the past three years. There are peaks in wait times around January and August, coinciding with post-holiday periods.
Percentage coronary ar bypass gra within targe Urgent > Semi-u > Electiv	rtery ifts done et: t urgent	86%* 94% 99%	100 u 50 0 0 0 0 0 0 0 0 0 0 0 0 0	Almost all elective and semi-urgent coronary artery bypass grafts are done within the target time. There is some room for improvement for urgent cases as the rate is 86%.
Percentage angiograph within targe Urgent Semi-u Electiv	nies done et: t urgent	85%* 70% 98%	100 100 100 100 100 100 100 100	Almost all elective and 85% of urgent angiographies are completed within the recommended timeframe. There has been improvement from the previous year for semi-urgent cases; however, there is room to do better.
Percentage percutaneo coronary interventior within targe Urgent > Semi-u > Electiv	ous ns done et: t urgent	85%* 79% 98%	100 50 0 0 0 0 0 0 0 0 0 0 0 0 0	For elective percutaneous coronary intervention, the results look good as the rate of cases done within target continues to be close to 100%. Performance for urgent cases, however has declined. Two years ago, 99% were done within the target time; the current figure is 85%. For semi-urgent cases, there are on-going challenges as the current result (79%) has not improved in the last two years. There is room for improvement.

Data sources: \* Cardiac Care Network, November 2010. Note that some patients move between urgency categories.



Data sources: \* Cancer Care Ontario and Wait Times Information System; data values represent data for December 2010.
## HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

### **Root Cause**

### People are placed on waiting lists for

**procedures they don't need.** One study found 27% of people who got cataract surgery did not get improved vision, and may not have needed it in the first place.<sup>75</sup> Another study suggests 26% of CT and MRI scans for headaches are inappropriate.<sup>76</sup> Overuse of CT is potentially dangerous; each CT scan emits 100 times the radiation of standard X-rays.<sup>77</sup>

Overuse may be due to providers not following best evidence when ordering a test or procedure, or the patients wanting a procedure without understanding risks and benefits.

People are getting a procedure for a problem that could have been prevented with better health habits.

Queues may vary in size between different sites that provide a service. Those who are "stuck" in a particularly slow queue may have very high wait times.

There is natural variation in demand and sometimes surges in demand occur, which can lead to a major increase in wait times.

Even if the demand and supply of services is matched, any queue that may have built up in the past will persist and lead to people waiting unnecessarily.

**There are inefficiencies in arranging surgery.** This may include delays or errors when sending booking information or delays in arranging preoperative tests or consults.

**Demand is greater than the available supply of services,** and in many cases is growing. In some instances, the demand for surgery is not measured and the capacity for the procedures is inadequate.

### **Ideas for Improvement**

**Implement appropriateness criteria** to ensure that patients truly require surgery or a specific test. (See section 6.4 for more about avoiding unnecessary drugs and tests). There are objective criteria for determining urgency for certain procedures (e.g., hip or knee replacement<sup>78,79,80,81</sup>, and cataract surgery<sup>82,83</sup>), but no requirement to use them at this time. Provider education and tools that provide clinical decision support would help ensure that patients receive CT scans and MRIs in the most appropriate circumstances. MOHLTC and University Health Network in partnership with St. Joseph's Healthcare Hamilton have developed an online CT and MRI decision support tool for use by providers.<sup>85</sup>

**Use priority tools.** These are designed to prioritize patients on scheduled waiting lists to ensure that those who are at greatest need receive the procedure first.<sup>86,87</sup>

Adopt strategies to improve health habits (see section 9.1). Tackle obesity, which worsens arthritis and can lead to more hip and knee replacements.<sup>88</sup> Reduce smoking, which can lead to coronary artery disease and more cardiac procedures.

**Pool different queues into a single queue.** At a typical bank, having a single line where each person goes to the first available teller is far more efficient than having a separate line for each teller. Based on this principle of queuing theory<sup>89</sup>, several LHINs have implemented centralized bookings for surgical procedures, specialist appointments and diagnostic tests to reduce wait times.<sup>90,91</sup>

Queue management principles suggest that to accommodate surges in demand, **at least some slack capacity is needed** to keep wait times low.<sup>92</sup> This is true for both the surgery and other services that need to happen before and after. Complex surgeries may require the patient to stay in the intensive care unit (ICU) afterwards. Potential solutions include innovations in surge capacity, evening out surgical procedures and/or a surgical ICU throughput unit for short length of stay post-operative ICU patients.

**Work down the backlog.**<sup>93</sup> Temporarily increase the rate of procedures being performed until this backlog is eliminated, and then return to the previous rate.<sup>94</sup>

**Implement standardized processes for arranging surgery.** This could include on-line booking tools (instead of relying on paper-based referrals that can be illegible or get lost) or standard checklists for patients and providers to ensure all steps are arranged well in advance.

**Measure demand for surgical services and respond accordingly.** This should be done routinely so the capacity for procedures can be adequately matched to the demand. That is, balance supply and demand.<sup>95</sup> Careful monitoring of incoming demand, projections of future demand, and careful planning of the number of procedures needed now and in future years will help to ensure the demand is met.<sup>96</sup> Consider all possible ways to introduce efficiencies before adding additional capacity.

- MOHLTC committed \$53 million towards capital project costs and \$8.5 million towards annual funding for the Stronach Regional Cancer Centre. It is expected to provide radiation treatment to more than 2,400 patients annually.<sup>97</sup> It currently delivers 7,500 chemotherapy treatments annually, a number expected to increase by 10% to 15% annually until 2013. By 2012, it should accommodate 100,000 outpatient visits annually.<sup>98</sup>
- Through the Wait Times Website, Ontarians have access to the most comprehensive surgical wait time information in Canada.<sup>99</sup>
- The MRI Process Improvement Program has provided 17 hospitals with on-site coaching and training to improve efficiency and access to diagnostic imaging and lower MRI wait times. It will support all Ontario hospitals with an MRI scanner by FY 2011/2012.<sup>100</sup>
- The Peri-Operative Coaching Program has provided 61 hospitals with on-site and remote peer coaching to improve the efficiency and quality of
  pre- and post-operative care and reduce surgery wait times.<sup>101</sup>

## 2.4 Access to long-term care and home care

A central role of home care is to help people who have difficulty caring for themselves continue to live independently. In Ontario, community care access centres (CCACs) arrange services such as nursing, personal support, physiotherapy, occupational therapy, speech-language therapy, social work, nutritional counselling, medical supplies and equipment.<sup>102</sup> When people can no longer stay at home, even with home care, CCACs arrange placements to long-term care (LTC) homes.<sup>103</sup> Normally, the person and/or his or her loved ones select up to three homes, in order of preference. Once it is determined that someone needs LTC, it is vital to place them as quickly as possible or they may not get the care they require and, if their condition deteriorates, those caring for them at home may struggle to cope.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
No one has to wait a long time to get into an LTC home.	People waiting at home for LTC may be at risk of further decline and may place a heavy burden on loved ones who are caring for them. People waiting in hospital for LTC are occupying hospital beds unnecessarily, and this can lead to overcrowding in emergency departments and wasted resources.	Ontario's 21,500 seniors who are on the waiting list for placement into an LTC home each year, and their families and caregivers. <sup>104</sup>
As many people as possible get their first choice of LTC home. <sup>105</sup>	When people are placed in their second or third choice for an LTC home, they may end up further from loved ones or in a home that doesn't specialize in meeting their ethnic, cultural or medical needs. It is possible for LTC residents to move to a higher-ranked choice later, but this may be inconvenient and disrupt continuity of care.	
No one has to wait a long time to get home care services.	People waiting for home care services may experience a decline in their condi- tion, and this may result in hospitalizations that could have been avoided.	Ontario's more than 585,000 <sup>106</sup> home care clients who receive services from CCACs each year.

Indicator	Value	Time trends & comparisons	Bottom line
Median number of days to LTC home placement:		250	Wait times to get into an LTC home are too high. The median wait time is 3.5 months. The wait time for those placed from home is over five months, and for those placed from hospital, it is almost two months.
Overall placements Those placed from hospital Those placed from home	103 days* 56 days 161 days	Se 125 0 Apr-Jun 03 BETTER Apr-Jun 10	Wait times increased sharply between 2005/06 and 08/09. In the most recent 12 months, however, overall wait times have stopped increasing. This suggests that the problem is starting to stabilize. The reasons for this remain to be determined, but it is possible that the province's Aging at Home strategy (see below) may have helped stabilize waits. However, the problem is still far from being fixed. The current overall wait is still nearly three times higher than in the spring of 2005.
LTC home place- ment volumes: Overall Those placed from hospital	3,348 1,179 1,860	7,000 biggs 3,500	The way people enter LTC has shifted recently. In the past year, the number of people placed into LTC from hospital dropped by 19%, while the number placed from home rose by 15%. <sup>107</sup> This could be because many communities are adopting a "Home First" approach, <sup>108</sup> where hospital patients who might have in the past been referred straight to LTC are instead offered additional home care services to help them return home and allow them to make their decision about future LTC placement at home.
from home	1,000	0 Apr-Jun 03 Apr-Jun 10	The wait time for those placed from hospital has increased by 10 days in the past year even though fewer people were admitted to LTC from hospital, possibly because less complex patients are better candidates for the Home First approach, while those remaining take longer to place. The decrease in patients being referred to LTC from hospital is a welcome step to help address the problem of Alternative Level of Care patients occupying beds in hospitals, but it is important to ensure that that excessive burden is not put on informal caregivers at home in the process.

Data sources: \* MOHLTC Long-term Care Client Profile Database, April to June 2010.





Data sources: \*\* MOHLTC Long-term Care Client Profile Database, FY 2009/10. \*\*\*Home Care Database , Discharge Abstract Database, Registered Persons Database, FY 2009/10.

## 2.4

## **Root Cause**

## Ideas for Improvement

Issue: LTC waits and people in LTC who don't necessarily require it

#### Services needed exceed what caregivers can provide and what home care can offer. Reasons may include:

- Family caregivers, who provide 80% of home care for individuals with chronic conditions<sup>110</sup>, may burn out or have health problems themselves, and then can no longer provide care.
- The person is socially isolated and has no family caregivers.
- The person cannot afford to pay for extra care (e.g. more home care or retirement home).
- The person and/or family are not aware of options.
- A CCAC can only provide a level of service that its budget allows.

There were **missed opportunities to prevent a decline in health** that led to admission to a LTC home. These could include:

- Preventable falls or injuries
- Deterioration of a chronic condition
- · Poor health habits
- "Deconditioning" among hospitalized patients; being in an unfamiliar environment like a hospital can accelerate one's loss of physical function.

**Premature labelling of hospital patients as needing LTC**. When people go to hospital with a sudden worsening of their condition, they may be told to go to LTC before they have had a chance to recover.

People who fear long waits may want to get themselves on the list "just in case." Recent data indicates that 22% of clients placed in LTC Homes have MaPLE scores that are not high or very high indicating they could potentially be placed elsewhere (see section 6.2). The following are alternatives to long-term care placement:

**Respite care** for family caregivers<sup>11,112</sup> (e.g. allow placement of a person in a hospital for a limited time to give caregivers a break).

### Community support or day programs.

**Extended home care hours or services.** This could include nursing, rehabilitation, home making, meals on wheels, and personal support services that include dressing, personal hygiene, assisting with mobility, monitoring of medicine use and other routine activities of living.<sup>113</sup>

**Supportive housing/assisted living arrangements.** These are publicly funded or subsidized housing options where people live in their own apartments but there are caregivers available on site to provide more frequent services. Planners in Lethbridge, Alberta using this lower-cost model have kept LTC waits to less than a month, while using one-third fewer LTC beds per capita than in Ontario.<sup>114</sup> At present, there are some such arrangements in Ontario<sup>115,116</sup>, often managed by non-profit organizations. Rent subsidies are available to eligible seniors, but the criteria are stringent and waiting lists for subsidized units can be long. <sup>117</sup>Overall, Ontario does not have a coordinated province-wide plan for supportive housing. This is now the third year that Quality Monitor has raised this issue.

**Retirement homes** can also provide some types of additional support, but are available only to those who can afford them.

Opportunities to preserve health include:

- Falls prevention (see section 4.5).
- Better chronic disease management through better monitoring or more consistent use of the right drugs or treatments. (see section 3.2).
- Promotion of a healthy lifestyle (see section 9.1).
- Rehabilitation and other services for the frail elderly to combat de-conditioning in hospitals.<sup>118</sup> Ontario's Senior Friendly Hospital Strategy has tips on making the physical environment more pleasant, increasing the use of assistive devices (e.g. canes and walkers), improved nutrition and better communication.<sup>119</sup>

The **"Home First**" approach, as noted above, aims to return hospitalized patients to their home first and avoid making a decision about LTC placement in the hospital. This approach gives the patient time to recover from the worsening of his or her condition, and allows home care workers to make a better assessment of an individual's true level of function.

**Apply the MaPLE score**<sup>120</sup> to objectively determine an individual's needs. Encourage those people who do not have high or very high needs to consider alternatives to LTC.

### HOSPITAL

**LONG-TERM CARE** 

**HOME CARE** 

**PRIMARY CARE** 

## Root Cause Ideas for Improvement

Issue: People do not get their first choice for LTC.

There are not enough LTC homes serving specific populations.

The community lacks sufficient LTC capacity to meet local needs. Some people may wish to stay in a community where their family resides or where they have support, but there may be little LTC capacity there. **Consider planning for more capacity,** or shifting existing LTC bed capacity, to serve ethnic or linguistic groups or communities that have particularly long waiting lists. With more options available, individuals might have a better chance of getting their first choice.

**Establish regional plans for LTC.** Residents should not have to move far outside their communities to receive LTC.

**Establish more specialized services** — for example, programs that target individuals with complex behavioural issues or medical complexities.<sup>121, 122</sup>

- The province's Aging at Home strategy<sup>123, 124</sup> is a \$700 million investment over three years (2008/09 to 2010/11) with the aim of enabling seniors and their caregivers to live healthy, independent lives in their own homes and avoid premature admission to LTC homes. LHINs are using this funding to provide enhanced home care and community support services, as well as fund innovative projects with their LHIN.<sup>125</sup>
- MOHLTC is rebuilding 4,183 existing beds and updating facilities at 37 LTC homes.<sup>126</sup> This is part of a provincial plan to redevelop 35,000 older beds over 10 years to help improve access to LTC throughout Ontario.<sup>127</sup>
- Since 2003, MOHLTC has created 8,032 new LTC beds, and it will be opening an additional 1,942 LTC beds over the next few years.<sup>128</sup>
- In 2008, MOHLTC announced the Strengthening Home Care Services in Ontario strategy, which consists of four key deliverables to strengthen home care: improving accountability for the provision of quality home care services through public reporting on quality measures; delivering improved health outcomes for Ontarians through the Integrated Client Care Project; enhancing fairness, transparency and communication in the competitive procurement process; and promoting innovation and flexibility in service provision.<sup>129</sup>

### **Receiving the right treatments in hospital** 3.1

What do Ontarians want?What if it doesn't happen?Who benefits most?Appropriate administration* of a beta-blocker, <sup>131</sup> a stain to lower cholesterol**2 and an angiotensin- converting enzyme inhibitor (ACE) <sup>135</sup> or angiotensin- receptor blocker (ARB)*4 to people who have had a heart attack (acute myocardial infarction — AMI).Patients may experience more strokes, repeat heart attacks and death.Ontario's 20,000 people hospitalized for heart attacks each year.***Appropriate administration* of an ACEI/ARB**Patients may experience more hospitalizations, worse quality of life and death.Ontario's 16,000 people and a beta-blocker***Appropriate administration* of an ACEI/ARB**Patients may experience more hospitalizations, worse quality of life and death.Ontario's 16,000 people adited to hospital with congestive heart failure.Appropriate administration* of a acetylsalicylic acid (ASA, or aspirin) or an anti-thrombotic drug (blood thinner)*** for people who have had a stroke.Patients may experience more repeat strokes.Ontario's 16,000 people who experience a new isolemic stroke each year.**1Appropriate administration* of a clotbusting drug for people who have had a stroke.Patients may experience more disability (e.g., loss of use of arm or leg, or speech) and death.Ontario's over 135,000 women who give birth in ontario's over 135,000 women who give birth in ontario each year.**3Appropriate administration* of a clotbusting drug for people who have had a stroke.Patients may experience more disability (e.g., loss of use of arm or leg, or speech) and death.Ontario's over 135,000 women who give birth in ontario's over 135,000 women who give birth in ontario each year.*			
statin to lower cholesterol132 and an angiotensin- converting enzyme inhibitor (ACEI)133 or angiotensin- receptor blocker (ARE)134 to people who have had a heart attack (acute myocardial infarction — AMI).attacks and death.hospitalized for heart attacks each year.135Appropriate administration* of an ACEL/ARB136 congestive heart failure. Patients with the slightly more common*38 "systolic dysfunction" form of heart failure (weak pumping action) should get these drugs; for those with the "diastolic dysfunction" form (stiff heart wall), these drugs should be considered.139,140Patients may experience more repeat strokes.Ontario's 16,000 people admitted to hospital with congestive heart failure each year.141Appropriate administration* of a cetylsalicylic acid (ASA, or aspirin) or an anti-thrombotic drug (blood thinner)142 for people who have had a stroke.Patients may experience more disability (e.g., loss of use of arm or leg, or speech) and death.Ontario's 16,000 people who experience a new ischemic stroke each year.141Appropriate administration* of a clot-busting drug for people who can get to a major hospital right after symptoms of a stroke begin.144Patients may experience more disability (e.g., loss of use of arm or leg, or speech) and death.Ontario's over 135,000 women who give birth in ontario's over 135,000 women who give birth in ontario's each year.147	What do Ontarians want?	What if it doesn't happen?	Who benefits most?
and a beta-blocker <sup>137</sup> for most people who have congestive heart failure. Patients with the slightly more common <sup>138</sup> "systolic dysfunction" form of heart failure (weak pumping action) should get these drugs; for those with the "diastolic dysfunction" form (stiff heart wall), these drugs should be considered. <sup>139,140</sup> of life and death.admitted to hospital with congestive heart failure each year. <sup>141</sup> Appropriate administration <sup>+</sup> of acetylsalicylic acid (ASA, or aspirin) or an anti-thrombotic drug (blood thinner) <sup>142</sup> for people who have had a stroke.Patients may experience more repeat strokes.Ontario's 16,000 people who experience a new ischemic stroke each year. <sup>143</sup> Appropriate administration <sup>+</sup> of a clotbusting drug for people who can get to a major hospital right after symptoms of a stroke begin. <sup>144</sup> Patients may experience more disability (e.g., loss of use of arm or leg, or speech) and death.Ontario's over 135,000 women who give birth in Ontario's over 135,000 women who give birth in Ontario sectors are performed only when necessary.Ontario's over 135,000 women who give birth in Ontario sector year. <sup>147</sup>	statin to lower cholesterol <sup>132</sup> and an angiotensin- converting enzyme inhibitor (ACEI) <sup>133</sup> or angiotensin- receptor blocker (ARB) <sup>134</sup> to people who have had a		hospitalized for heart
(ASA, or aspirin) or an anti-thrombotic drug (blood thinner)142 for people who have had a stroke.who experience a new ischemic stroke each year.143Appropriate administration† of a clot-busting drug for people who can get to a major hospital right after symptoms of a stroke begin.144Patients may experience more disability (e.g., loss of use of arm or leg, or speech) and death.who experience a new 	and a beta-blocker <sup>137</sup> for most people who have congestive heart failure. Patients with the slightly more common <sup>138</sup> "systolic dysfunction" form of heart failure (weak pumping action) should get these drugs; for those with the "diastolic dysfunction" form (stiff heart		admitted to hospital with congestive heart failure
for people who can get to a major hospital right after symptoms of a stroke begin.144arm or leg, or speech) and death.onlyAn effective method of delivery for women in labour. Caesarean sections are performed only when necessary.Caesarean sections may lead to higher infection rates,145 increased respiratory distress of the newborn,146 a stay in hospital that is twice as long, increased risk of readmission to hospital andOntario's over 135,000 women who give birth in Ontario each year.147	(ASA, or aspirin) or an anti-thrombotic drug (blood	Patients may experience more repeat strokes.	who experience a new ischemic stroke
labour. Caesarean sections are performed only when necessary.respiratory distress of the newborn,146 a stay in hospital that is twice as long, increased risk of readmission to hospital andwomen who give birth in Ontario each year.147	for people who can get to a major hospital right		
	labour. Caesarean sections are performed	respiratory distress of the newborn, <sup>146</sup> a stay in hospital that is twice as long, increased risk of readmission to hospital and	women who give birth in

Indicator	Value	Time trends & comparisons	Bottom line
Percentage of elderly patients with AMI who, within 90 days of discharge, filled a prescription for the recommended drugs: Statin Beta-blocker ACEI/ARB All three at once	87%* 79% 79% 59%	100 50 0 2009/10 BETTER BETTER	Over the last seven years, there has been major improvement in the use of statins after a heart attack (from 67% to 87%). However, there has been little change in the use of beta-blockers and ACEIs/ARBs. Use of all three drugs at once has improved over the last seven years, but the percentage has levelled off in the last two years. There is room for further improvement as guidelines suggest it may be possible to increase the use of these drugs to 90%. <sup>148</sup>
Percentage of elderly patients with congestive heart failure who, within 90 days of dis- charge, filled a prescription for the recommended drugs: ACEI/ARB Beta-blocker Both at once	71%* 66% 50%	100 100 100 100 100 100 100 100	The use of beta-blockers for congestive heart failure patients after discharge has increased over the past seven years, while ACEI/ARB use has decreased over the same time period. There is probably room for improvement in the use of these drugs, although making a definitive conclusion on that will be possible only when Ontario has better data on what type of heart failure patients have (see data advocacy section, 1.12).

Data sources: \* Registered Persons Database, Discharge Abstract Database (DAD), Ontario Drug Benefits Database, FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES). These indicators are calculated only for patients aged 66 and older, as data on drug use was only available for this group. The indicators track prescriptions filled. Some people might fill the prescription but not actually take the drug; hence, the rate of actual use may be lower. <sup>†</sup> These drugs should be used except when contraindicated, such as allergy to the drug.



Indicator	Value	Time trends & comparisons	Bottom line
Percentage of acute stroke patients discharged on ASA or anti-thrombotic therapy	92%**	100 50 0 2003/04 better better 2009/10	About nine in 10 stroke patients are receiving the recommended blood-thinning drug when they are discharged from hospital. This indicator has not changed in the last four years. The typical rate of allergies or sensitivity to these drugs <sup>149,150</sup> is close to the rate of people not on these drugs (8%); hence, there is probably little if any room for improvement.
Percentage of ischemic stroke patients eligible for thromboly- sis (clot-busting drug) who get it within one hour of arriving in the emergency department <sup>††</sup>	13%**	100 50 0 2003/04 BETTER BETTER	Only one in eight patients who had a stroke and could benefit from clot-busting drugs received those drugs within one hour of arriving in the emergency department. Although gains have been made over the past six years, there is still huge room for improvement.
Rate of delivering via Caesarean section (per 100 deliveries)	29***	100 <sup>0</sup> / <sub>2005/06</sub> <b>BETTER</b> <b>D</b> <b>D</b> <b>D</b> <b>D</b> <b>D</b> <b>D</b> <b>D</b> <b>D</b>	The Caesarean section rate in Ontario is 29%. This has not changed in the past four years. Among low-risk mothers, 15% of pregnancies were delivered by Caesarean section. The overall Caesarean section rate is higher than the World Health Organization's recommended rate of 15%. <sup>151</sup> However, the validity of this target is disputed and there is no consensus in Canada
Rate of low-risk <sup>#</sup> first-time moth- ers who deliver a full-term baby via Caesarean section (per 100 deliveries)	15 <sup>§</sup>		on what the ideal rate should be. Nonetheless, these rates should be monitored carefully and opportunities to avoid unnecessary Caesarean sections should be explored.
Data and a			

Data sources:
 \*\* Registry of the Canadian Stroke Network, FY 2009/10, calculated by ICES. This indicator looks at ischemic stroke/transient ischemic stroke patients discharged alive from the emergency department or an acute in-patient setting of a regional stroke centre (note that this analysis does not include hemorrhagic stroke).
 \*\*\* DAD, calculated by ICES, FY 2009/10. Indicator methodology adapted from the POWER Report.<sup>152</sup>
 § Niday Perinatal Database, FY 2009/10, provided by BORN Ontario.
 \*\* This indicator looks at ischemic stroke patients who arrive at the emergency department of a regional stroke centre within 2.5 hours of stroke symptom onset.
 \* Low-risk women are those who do not have many of the common indications for a C-section. They are first time mothers and therefore could not have had a past C-section; the baby's head is pointing downwards; the baby is not premature; the labour was spontaneous; and it is a single pregnancy (i.e. not twins). They may still have a C-section if the labour is not progressing.

## 3.1

## **Root Cause**

### Issue: Drug management.

Healthcare providers may not order the right drugs or treatments because of **information overload** — they are busy, distracted by other patient issues or there are too many things to remember.

Poor communication when care is transferred between providers could lead to information about the right treatments not being passed on. For example, there may be a good reason to initially delay giving a drug (e.g., beta-blockers for a heart attack), but whoever is handling the care at a later point must know that the drug should be started when safe to do so.

Providers are not aware of how closely they are following guidelines.

**Patients do not fill prescriptions** given to them at discharge. This could be because they fear the side effects, or do not understand when, why, how to, or how long to take the drug. Some drugs have minor side effects that require some getting used to; if the patient does not know that, he or she may discontinue the drug prematurely.

#### Medication discrepancies put patients at risk of medication errors. This happens when someone enters or leaves hospital, and there is confusion about what drugs he or she was on previously.

### Issue: Timely thrombolysis of stroke patients.

**Diagnosis of stroke is delayed.** This may occur if a patient goes to a centre that does not have the right equipment or special skills in handling strokes.

**Poor hand-offs or communication might delay timeliness.** One US study found major delays between writing the order and giving the thrombolysis drug.<sup>171</sup>

## **Ideas for Improvement**

## Increase the availability and use of standardized admission orders, discharge checklists, standardized care plans or algorithms to help decide when to order a drug or test.

**Use clinical decision support systems to guide prescribing choices**. Increase the use of electronic health records (EHRs) that can guide clinicians' decisions and generate clinical reminders.<sup>153</sup> Such systems provide recommendations on when to start, monitor, or stop treatments and can identify safety issues (e.g. dangerous drug combinations). These systems may have a useful influence on provider behaviour<sup>154</sup> and patient outcomes,<sup>155</sup> especially when they deliver timely and relevant information, suit local needs (avoiding a "one size fits all" approach) and are accompanied by ongoing technical support.<sup>156</sup>

**Provide regular feedback to prescribers.** For example, the Enhanced Feedback for Effective Cardiac Treatment (EFFECT) publicly releases hospital performance data on indicators for AMI and congestive heart failure treatments.<sup>157</sup> This data should also be provided at an individual provider level.<sup>158</sup>

**Encourage hospital staff to review medications with patients prior to discharge**<sup>159,160</sup> and ensure that patients leave hospital with printed information, including an easy-to-follow "meds list" with key changes noted. Providers can use the **"teach back" method** to ensure information has been understood by patients and caregivers.<sup>161,162</sup> Encouraging patients to keep meds lists also keeps them engaged in their own care.<sup>163</sup>

**Follow-up by telephone and interactive voice response systems** may help identify drug interactions and encourage newly discharged patients to take medications as prescribed.<sup>164,165</sup>

In one study, almost half of all patients had at least one medical error when discharged from hospital to the community.<sup>166</sup> **Encourage primary care providers to spend time with patients reviewing drugs after discharge**<sup>167</sup> — and ensure the provider has received hospital discharge information by the time this visit takes place.

**Ensure stroke cases are sent by ambulance directly to designated stroke centres** that have the most experience in handling stroke. This resulted in faster arrival times to a stroke centre in Toronto and a four-fold increase in getting thrombolysis to patients who need it.<sup>168</sup> This is also a recommendation from national stroke guidelines.<sup>169</sup>

**Enhance telemedicine capabilities** so that in rural hospitals, where stoke expertise may be limited, CT scans can be reviewed by neurologists and stroke experts located elsewhere.<sup>170</sup>

**Create standardized processes and order sets for initiation of thrombolysis**.<sup>172</sup> **Consider creating a specialized team to administer thrombolysis**, as one hospital in Calgary has done.<sup>173</sup>



## **Root Cause**

### Issue: High Caesarean section rates.

#### More women may be having planned C-sections with no medical reasons.

There is no data on how often this happens in Ontario, but it is a growing world-wide phenomenon.<sup>174,175</sup> Reasons may include: the convenience of knowing the delivery date in advance; fear of pain; and fear of complications of vaginal delivery (e.g. incontinence or vaginal prolapse). Also, a planned C-section is more convenient for providers, allowing for predictable scheduling.

**Medicolegal concerns**. Obstetricians already have the highest malpractice insurance premiums among specialists in Canada<sup>180</sup>.

**Skill of provider** — physicians may not feel comfortable with procedures that are alternatives to C-sections, such as forceps to rotate the baby into a better position for delivery,<sup>182</sup> trial of labour among women with past C-sections<sup>183</sup>, or vaginal delivery of breech presentations.<sup>184</sup>

**Unnecessary inductions.** When labour is induced, the chance of a C-section is much higher.<sup>186</sup>

Lack of consensus among providers that C-section rates are a problem.

**Rising obesity rates.** Obesity is associated with a two-fold higher rate of C-sections. <sup>189</sup>

## **Ideas for Improvement**

**Inform the public and healthcare providers about the risks of planned C-sections.** Although overall risk of Caesarean section is small, planned Caesarean sections are associated with a three times greater risk of severe complications (e.g., hemorrhage requiring hysterectomy, cardiac arrest, venous thromboembolism, major infection) than planned vaginal birth,<sup>176</sup> as well as higher healthcare costs.<sup>177</sup> Efforts to educate can include individual discussions between patients and providers, as well as **public awareness campaigns**. For example, several New Jersey hospitals have signed onto a "Worst to First" campaign to reduce C-section rates to 15%.<sup>178</sup> Public campaigns can also emphasize that childbirth is a natural physiological process.<sup>179</sup>

Use of **second opinions** and **applying objective criteria for C-sections**<sup>181</sup> were found in one study to reduce C-sections by one-third. It is possible that these approaches could help reassure obstetricians they are making the right decision to hold off on a C-section when they are worried about a potential medicolegal case.

Consider augmenting training standards for obstetricians, family doctors and midwives in training in the use of these skills, or providing more retraining opportunities for existing practitioners who wish to sharpen these skills. Emergency drills and patient simulators allow health care providers to practice handling difficult situations as a team (similar to flight simulators for pilots), and their use in obstetrics could be greatly expanded.<sup>185</sup>

**Set up protocols for appropriate induction.** Doctors at a Pittsburgh hospital developed a protocol where elective inductions could be scheduled only if the baby was at least at 39 weeks and the cervix was "ripe" based on an objective score. Inductions were reduced by one-third and C-sections among first-time mothers being induced decreased from 35% to 14%.<sup>187</sup>

Audit and feedback mechanisms allow individual practitioners to regularly review their own data on measures like C-section rates, complication rates and adherence to guidelines and identify areas for improvement.<sup>188</sup>

Consider **population-based strategies to reduce obesity** overall (e.g. food labelling, healthy communities that promote physical activity, etc.; see section 9.1).

- The Champlain Get with the Guideline Initiative helps ensure that all patients in the Champlain LHIN who are admitted to hospital for a heart attack or symptoms of a heart attack get the best care.<sup>190</sup> Participating hospitals implement a discharge tool that includes care pathways, standard orders, discharge checklists, reminders and a data collection tool that tracks whether patients are getting all of evidence-based guidelines (e.g. the right drugs for heart attack). Hospitals also get several days of coaching time to help them use these tools correctly.
- The Ontario Mother and Infant Study is a multi-site investigation involving 2,500 Ontario women to assess the impact of planned Caesarean vs. planned vaginal delivery on maternal and infant health, service use and the cost of care.<sup>191</sup>

### Chronic disease management 3.2

Chronic diseases, which require management over a long period of time, are widespread.<sup>192</sup> They affect one in three Ontarians and almost four in five Ontarians aged 45 and over. Among Ontarians with a chronic disease, about seven in 10 have two or more chronic conditions.<sup>193</sup> These conditions tend to get gradually worse over time and can result in pain, suffering, disabling complications or premature death.<sup>194</sup> There is also a tremendous economic burden associated with chronic diseases, with the expense amounting to 55% of total direct and indirect health costs in Ontario.<sup>195</sup> Although chronic diseases cannot be completely cured, lifestyle changes, medical treatments and carrely moritoring can reduce the risk of acting them or slow their programming <sup>196</sup> This unarise super fields are not be diseases.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Regular monitoring for people with chronic diseases (e.g., regular eye and foot exams for patients with diabetes). <sup>197</sup>	The consequences of not having regular eye and foot exams may include more blindness, <sup>198</sup> skin ulcers <sup>199</sup> and amputations. <sup>200</sup>	Ontario's 937,000 people with diabetes, <sup>201</sup> over 18,000 people who experienced a heart attack last year, and almost 16,000 people admitted to hospital with congestive heart failure each year. <sup>202</sup>
Healthy lifestyles for people with chronic diseases (e.g. avoidance of smoking, obesity, poor diet, physical inactivity and heavy drinking). <sup>203</sup>	Unhealthy behaviours can lead to worsening of chronic diseases and speed up the development of long-term complications or other conditions. <sup>204,205</sup>	
People with chronic conditions should be taking the right medications and feel confident about how to manage their own conditions.	For diabetes, the consequences may be more deaths and more complica- tions, such as strokes, heart attacks, amputations and other surgeries for poor circulation, kidney failure and dialysis. <sup>206,207,208,209</sup> Other consequences may include avoidable hospitalizations and emergency department visits, which are stressful for patients and waste healthcare resources. <sup>210</sup>	

### Monitoring and management of chronic diseases

Indicator	Value	Time trends & comparisons	Bottom line
Percentage of people with diabetes who had an eye exam in the past 12 months	51%*	100 100 100 100 100 100 100 100	Approximately half of people with diabetes received an eye exam in the past 12 months. This rate has been generally steady over the past 8 years. There is huge room for improvement; nearly all patients with diabetes should get this test regularly.
Unhealthy habits of people with chronic diseases <sup>†</sup> : Heavy drinking Low fruit/vegetable consumption Physical inactivity Obesity Smoking	18%** 57% 56% 25% 19%	80 40 0 2001 better	Smoking rates have improved significantly among people with chronic diseases in the past nine years. Rates of physical inactivity and inadequate fruit/vegetable intake have improved slightly, but heavy drinking and obesity have been on the rise. All of these self-reported rates are still too high, and there is still major room for improvement as healthy behaviours are critical for this population.
Percentage of elderly people with diabetes (aged 66+) who, in the past year, regularly filled prescriptions for: 	67%*** 60% 48%	100 50 0 2003/04	Over time, Ontario has seen slow but steady improvement in the use of ACEIs/ARBs and statins among people with diabetes. However, there is still room to improve, since only half of elderly people with diabetes are getting both of these drugs and experts suggest most elderly people with diabetes should be receiving them. <sup>211,212,213</sup>

Data sources:

Ontario Diabetes Database (ODD), Ontario Health Insurance Plan, Registered Persons Database, FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES).

\*\*\*Canadian Community Health Survey, 2009, calculated by ICES.
 \*\*\*Ontario Drug Benefits Database, ODD, FY 2009/10, calculated by ICES. This indicator tracks prescriptions filled. Some people might fill the prescription but not actually take the drug; hence, the rate of actual use may be lower.

<sup>†</sup> Includes high blood pressure, diabetes, arthritis, heart disease, cancer, asthma, respiratory problems and depression. <sup>††</sup> ACEI=angiotensin-converting enzyme inhibitor. ARB=angiotensin receptor blocker.



HOSPITAL LONG-TERM CARE **HOME CARE PRIMARY CARE** 

### **Complications of chronic diseases**

Indicator	Value	Time trends & comparisons	Bottom line
Percentage of people with diabetes for more than a year who had a serious diabetes complication within a year: → Any serious complication → Surgery for circulation problem (including amputation) → Death → Heart attack → Stroke → Kidney failure	4.30%* 0.15% 2.70% 1.10% 0.51% 0.17%	6 0 0 0 0 0 0 0 0 0 0 0 0 0	Slightly more than one in 25 people with diabetes will experience a major complication of diabetes within a year. Last year, this amounted to 40,000 people in Ontario. The rates of these complications have decreased steadily in Ontario, which may be due to better use of drugs such as statins and ACEIs or decreased smoking (as noted above), or to earlier detection of diabetes. Despite the improvement in complication rates, there is room for even more improvement. As noted above, services like eye visits are not done reliably, and recent one-time studies in Ontario have identified major gaps in blood pressure, blood sugar and cholesterol control. <sup>214</sup> Ontario does not collect this information regularly, and needs to do so in the future (see data advocacy section 1.12).
Adjusted mortality rate (chance of death) in the year after a congestive heart failure hospitalization	35**	50 <sup>50</sup> <sup>20</sup> <sup>20</sup> <sup>20</sup> <sup>2002/03</sup> <sup>2008/09</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup> <sup>50</sup>	Over one-third of patients admitted to hospital for congestive heart failure die within the next year. There has been no improvement in this indicator over the last six years. There is likely room to improve.
Adjusted rate of death per 100 heart attack patients between 30 days and one year after thei first heart attack	9.1**	20 <sup>2</sup> / <sub>2</sub> 10 0 2002/03 BETTER BETTER	One in 11 patients die within one year of having a heart attack. This has improved over the last three years. There may still be some room to improve — for example, by further reducing smoking rates.

Data sources: \*Discharge Abstract Database (DAD), Ontario Health Insurance Plan physician billings database, Registered Persons Database (RPD) and Ontario Diabetes Database (ODD), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES). Complication rates adjusted for age, sex and number of years since diagnosis of diabetes. \*\*DAD and RPD, FY 2008/09, calculated by ICES. Mortality rates adjusted for age and sex. Indicator methodology adapted from the POWER Report.<sup>215</sup>

## 3.2

## **Root Cause**

It is challenging for providers to follow best practice guidelines 100% of the time. This may be due to: 216,217,218,219

- Information overload. It is easy to forget to order a drug or test, or keep track of latest guidelines.
- Organizational barriers (e.g. difficulty accessing information about the patient that could change a clinical decision)
- **Scepticism** about the validity of the evidence.

It is difficult for healthcare providers to manage frail elderly patients with multiple chronic conditions. Managing these patients well may require a more intensive level of monitoring or more specialized expertise.

Patients may not be engaged in their own care. People may not be motivated to change, or may not feel as if they have the power to do so.

### **Ideas for Improvement**

**Use flow sheets in patient charts.**<sup>220</sup> Flow sheets are one-page documents with checkboxes to record compliance with best practices for each patient encounter. The Canadian Diabetes Association has a recommended version for diabetes.<sup>221</sup>

**Reminder systems** prompt a health care provider to perform a particular action (e.g. remember when it is time to schedule another annual eye exam), and can be paper-based or electronic.<sup>222,223</sup>

**Have a well-functioning electronic medical record (EMR)**. EMRs often have flow sheets and reminder utilities built into the software. **Decision support features** in EMRs go beyond reminders and help providers choose among treatment options, such as drugs that are recommended in specific clinical circumstances (e.g., diabetes).<sup>224, 225</sup> One clinical decision support system improved rates of prescription of hypertension drugs by 32%.<sup>226</sup>

**Provide confidential feedback on a clinic or provider's performance** according to best practice guidelines, to give them an idea of how they are performing and to help them identify areas for improvement.<sup>227, 228, 229</sup> **Opinion leaders** can help encourage the adoption of evidence-based practices. Identified by their peers, these

"educational influentials" operate within their communities to teach and facilitate change.<sup>230, 231</sup> For best results, simultaneously employ as many as possible of the change ideas noted above.

Multifaceted interventions are the most likely to be successful in changing provider behaviour.<sup>232</sup>

**Consider establishing specialized clinics for chronic disease(s)**, such as diabetes or congestive heart failure, staffed by a multidisciplinary team. A review of heart failure treatment programs found that patients receiving this type of care had approximately 25% reduced risk of mortality and hospitalization for heart failure.<sup>233</sup> An anticoagulation clinic set up in Peterborough has achieved an 80% rate of patients on anticoagulation medication in the safe dosing range (vs. 55% of patients in usual monitoring).<sup>234</sup>

**Employ telemedicine technology to monitor patients.** Consider using "tele-homecare" to monitor and communicate with patients, <sup>235</sup> especially those who are house-bound or who live in remote locations.<sup>236,237</sup>

Promote **patient self-management** of behavioural risk factors and chronic disease (see section 9.1). Congestive heart failure patients who self-monitor their weight daily can spot warning signs of worsening congestive heart failure quickly and get their medications adjusted before they need to go to hospital.<sup>238</sup> Improve access to **health** education and health promotion activities and materials.<sup>239, 240, 241</sup> (see section 9.1).

- The Ontario Diabetes Strategy has two public targets: ensuring that all people with diabetes have access to a primary health provider and that 80% of people with diabetes, aged 18 and older, have all three diabetes tests (cholesterol, retinal eye exam and A1C) within recommended guidelines for optimal diabetes management.<sup>242</sup> As part of this strategy, MOHLTC is creating 51 new diabetes education teams, expanding chronic kidney condition services, creating up to 14 regional coordination centres and expanding diabetes care and prevention resources.<sup>243</sup>
- Primary and specialty care providers in Peterborough have implemented a population-based care model for vascular disease called the Comprehensive Vascular Disease Prevention and Management Initiative (CVDPMI). This initiative targets the asymptomatic "non-help-seeking" patients and uses the electronic medical record and practice protocols to screen them for conditions like high blood pressure or high cholesterol. Those found to be at high risk for vascular events receive dietary counselling, lifestyle management planning and medications where needed. Initial clinical findings show that CVDPMI patients have an up to 50% reduction in cardiac event risk. <sup>244</sup>
- MOHLTC plans to further reform the prescription drug system by reducing the cost of generic drugs by at least 50%, eliminating excessive
  payments to pharmacy owners from generic drug companies, and offering patients in rural communities wider access to lower-cost drugs.<sup>245</sup>
- The Heart & Stroke Foundation recently launched an online Heart & Stroke Risk Assessment Report for self-evaluation based on a short, simple survey that looks at risk factors for heart disease. <sup>246</sup> It also started the Heart & Stroke Foundation Hypertension Management Initiative and Program helps primary care providers and patients improve hypertension control. It includes flowsheets, reminders, audit and feedback tools, automatic blood pressure cuffs and support in implementing these ideas, which is designed to enhance the management and control of hypertension by primary care providers, including doctors, nurses and pharmacists, and patients.<sup>247</sup>

## HOSPITAL

LONG-TERM CARE

HOME CARE

**PRIMARY CARE** 



### Potentially avoidable hospitalizations 3.3

People who are frail with multiple chronic conditions require hospitalization if their condition deteriorates. Often, however, these hospitalizations could be avoided. Ambulatory care sensitive admissions (ACSCs) is a measure which tracks admissions which took place for a condition which, if carefully managed in a primary care setting, might have resulted in not needing to go to hospital. Another measure is readmissions, which can occur if there is not a smooth handoff from the hospital to primary care providers or home care.<sup>248,249,250</sup> Readmissions are tracked using two methods: one that tracks readmissions for the same condition, and follows a national definition that allows for provincial comparisons, and

What do Ontarians want?		What if it doesn't happen?	Who benefits most?
	The best possible management of chronic diseases, so that people are less likely to be admitted to hospital.	Worsening of a chronic condition that could have been avoided and that results in admission means worse quality of life, increased stress, exposure to risks while in hospital (e.g. hospital acquired infection), increased burden on family and wasted healthcare resources. <sup>251</sup>	The Ontarians who contributed to the over 34,000 potentially avoidable hospital admissions last year. <sup>252</sup>
	Effective treatment in hospital and proper follow-up care arranged in the community to avoid readmission.	When patients are readmitted, their health may worsen and, along with their families, they may experience lost time and economic productivity. In addition, unnecessary readmissions increase the cost of hospital care. <sup>253</sup>	Ontario's residents who visit an emergency department or hospital.

#### Ambulatory care sensitive conditions

Indicator	Value	Time trends & comparisons	Bottom line
Hospital admission rate per 100,000 population for all ambulatory care sensitive conditions (ACSCs)	278*	450 <sup>B</sup> / <sub>E</sub> 225 0 2002/03 BETTER 2009/10 BETTER	In FY 2009/10, roughly 34,100 people were admitted to hospitals in Ontario for complications from chronic diseases that could potentially have been prevented with good primary care. The rate has decreased by 33% over the last seven years, and decreased by 5.2% compared to last year. There is still room for improvement.
Interprovincial comparisons for overall ACSCs	**	1'400 N.W.T. N.W	In FY 2008/09, Ontario had the second lowest rate of ACSCs compared to other provinces. However, there are individual regions within the country that have even lower rates — for example, in Richmond, B.C., the rate is 154 (almost half as low as Ontario's rate), and the Central LHIN rate is 183. This suggests that there is room to improve.
Hospital admission rate per 100,000 population for: Chronic obstruc- tive pulmonary disorder (COPD — e.g., emphysema, chronic bronchitis) Congestive heart failure (CHF) Asthma Angina Diabetes	84* 52* 40* 36* 32*	150 g 75 0 0002/03 EETTER BETTER	<ul> <li>COPD is currently the most common ACSC. Admission rates have not changed since FY 2002/03, and there is major room to improve.</li> <li>CHF is the second-most common ACSC. Admission rates declined from FY 2002/03 to FY 2006/07, and better use of certain drugs (e.g., beta-blockers) might have contributed to this trend. There has been no change in the last three years and there is major room to improve.</li> <li>The rate of asthma admissions improved significantly from FY 2005/06 to FY 2007/08, but there has been no change in the past two years. Continued improvement is necessary.</li> <li>Over the last seven years, there has been a 66% reduction in hospitalizations for angina, with a 13% reduction just in the last year. These improvements could be due to reduced smoking (section 9.1), better use of drugs like statins (section 3.1), and timely use of revascularization procedures (e.g. bypass or stents; section 2.3).<sup>254</sup> This is good news, and Ontario should continue to aim for more reductions.</li> <li>Diabetes hospitalizations have decreased by 26% over the last seven years, with much of the change seen in the last year. Continued emphasis is necessary to sustain this level of improvement.</li> </ul>

Data sources: \*Discharge Abstract Database, FY 2009/10, calculated by Institute for Clinical Evaluative Sciences. \*\*Canadian Institute for Health Information, Health Indicators, 2010. Note that minor variations in methodologies were used to calculate overall ACSC rates when comparing time trending to interprovincial comparisons.



#### Readmissions

Indicator	Value	Time trends & comparisons	Bottom line
Indicator	value	lime trends & comparisons	
Readmissions for any reason:Congestive heart failure (age $\geq$ 45)COPD (age $\geq$ 45)Gastrointestinal conditions (all ages)Diabetes (all ages)Cardiac conditions (age $\geq$ 40)Pneumonia (all ages)Stroke (age $\geq$ 45)	22%* 19% 16% 14% 12% 12% 8.2%*	0 15 30	People who have been discharged from hospital for COPD or congestive heart failure have the highest rates of readmission for any cause in Ontario — higher than those who have been hospitalized for pneumonia or stroke.
Readmission rates for a specific or related condition: 	11%** 7.7% 7.8% 5.4% 4.3% 3.9% 2.6%	14 9 0 14 14 14 14 14 14 14 14 14 14	Readmissions over the past seven years have decreased by almost half for heart attack, which is good news. However, there have been only very slight improvements for COPD, CHF, asthma, diabetes, gastrointestinal conditions and stroke. There is huge room for improvement.
Readmission rates for specific or related mental health conditions: Schizophrenia and bipolar disorder Mental health and addictions Depression	9.6% 6.3% 3.9%**	<sup>14</sup> 7 0 2005/06 better 2009/10	Nearly one in 10 individuals with schizophrenia or bipolar disorder is readmitted within 30 days of being discharged from hospital. This has not improved over the past three years. The complexity of care for psychotic illnesses is higher than for other conditions and non-adherence with treatment is among the biggest risk factors for readmission. <sup>255</sup> Improvement is necessary.
AMI readmission rates for specific or related conditions across Canada	4.6%§	14 NELL/LAB NOVA SCOTIA MANITOBA SASKATCHEWAN B.C. ONTARIO ALBERTA ALBERTA	Ontario has the second lowest rate of readmissions for people who have been hospitalized for an AMI. It is still behind Alberta, suggesting there may still be room to improve.
Relationship between AMI readmission rate and percent of seniors who were on three drugs recommended for AMI after discharge <sup>†</sup>		10 5 -40% 40.49% 50.59% 60.69% >70% Percent of AMI patients (66 and over) who filed prescriptions for three key drugs after discharge	Hospitals whose patients use more evidence-based drugs for heart attack after discharge also tend to have lower readmission rates.

Data sources: \*MOHLTC-LHIN Performance Agreement, FY 2009/10. \*\*Discharge Abstract Database (DAD) and Registered Persons Database (RPD), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES). All figures represent readmission rates to any acute care hospital within 28 days of discharge, for people aged 15 to 84, per 100 discharges. <sup>5</sup>DAD, FY 2008/09, Canadian Institute for Health Information, Health Indicators, 2010. 'The three drugs are beta-blockers, angiotensin converting enzyme inhibitors or angiotensin receptor blockers, and statins. Prescriptions had to be filled within 90 days after discharge. Each dot represents a group of hospitals within each category.

## **Root Cause**

## Patients do not get all the right medications while in hospital.

(See scatterplot on previous page, which shows that low use of the right drugs is associated with higher readmissions.) Healthcare providers may not order the right drugs or treatments because they are busy, distracted by other patient issues or there are too many things to remember.

Administrators and healthcare providers may not be aware of the extent of the readmission problem.<sup>258</sup> In urban communities where multiple hospitals serve a large population base, readmitted patients may end up at a different hospital.

Pressure to free up beds may lead to premature discharge and subsequent readmission.

People are not getting the intensity of follow-up that they require.

Information about the admission and discharge plan is not quickly transferred to the patient's primary care provider or to community service providers. Failure to communicate relevant information may adversely affect patient care.<sup>262</sup>,<sup>263</sup>

#### Patients may not understand

**instructions for their care** following discharge from hospital — how to take their medications, what signs to look for or whom to call if they have concerns.

Poor coordination of transition from hospital to home.

## **Ideas for Improvement**

Use standardized admission orders, order sets for common clinical conditions and discharge checklists, as well as EHRs that generate clinical reminders.<sup>257</sup> These can help remind prescribers to order the right drugs and coordinate integrated care after discharge. See section 3.1.

**Feedback information on readmissions, or data on compliance with guidelines, should go to hospital administrators, providers and staff.**<sup>259</sup> Ensure this data includes readmission to other hospitals. If possible, provide this information at an individual healthcare provider level to help each provider and unit develop their own quality improvement plans.<sup>260</sup>

Implement strategies to reduce avoidable demand for hospital beds. See section 2.1, 2.2 and 2.4

**Do risk-scoring to identify patients at high risk for readmission, and arrange more intensive follow-up for them.** One such tool, the LACE index, takes into account the length of stay ("L"); acuity of the admission ("A"); comorbidity ("C"); and recent emergency department use ("E").<sup>261</sup> High-risk patients could be referred to more intensive follow-up such as a specialized clinic or tele-home care.

**Consider database-generated discharge summaries** and other strategies discussed in section 8.1 to improve patient care during the transition from hospital to home.

Use a structured discharge summary for frail elderly patients, with standardized elements which contain information that healthcare providers in the community receiving the summary need to know to manage the patient well. This is the approach used by one team in Quebec.<sup>264</sup>

**Consider contacting patients by phone 48 hours after discharge** to ensure the planned treatment plan is progressing well.<sup>265</sup> Or, have patients return to hospital for an ambulatory visit soon after discharge. **Improve and expand the information given to patients and caregivers at discharge**, incorporating "best practice" methods of communication,<sup>266</sup> and conduct teaching sessions that include patients and family members post-discharge. Providing patients with **written discharge instructions** has been shown to decrease readmissions, particularly for congestive heart failure.<sup>267</sup> See section 8.1 for details on the **"teach-back" method** and other communication strategies to improve patient understanding of care and discharge instructions.

**Ensure better coordination** among hospital, community service and primary care settings (see section 2.4). **Institute standard discharge follow-up protocols**<sup>268</sup> for frail patients, such as a home visit through a CCAC on the patient's first day post-hospital. To ensure a patient gets a follow-up primary care visit soon after discharge (e.g. one week), have hospital staff request or confirm the booking of the appointment. **Keep in touch with patients**. An Ottawa-based project successfully used interactive voice response systems to monitor patients post-discharge.<sup>269</sup> An automated telephone recording asks patients certain questions about symptoms or use of drugs and patients enter their response. If there are worrisome responses, a nurse follows up with a phone-call.



### **Root Cause**

Patients with chronic conditions don't get the right monitoring.

Patients develop infections while in hospital that may not appear until after the patient has gone home.

Patients don't get the right rehabilitation services after discharge.

## **Ideas for Improvement**

**Promote patient self-management** (see section 9.1). Congestive heart failure (CHF) patients who self-monitor their weight daily can spot warning signs of worsening CHF quickly and get their medications adjusted before they need to go to hospital.<sup>270</sup>

Consider establishing dedicated clinics for chronic diseases, such as heart failure,<sup>271,272</sup> diabetes<sup>273,274</sup> and conditions requiring anticoagulation therapy.<sup>275</sup> See section 3.2 for more on how to improve the management of chronic diseases.

See section 4.1 for ideas on how to improve infection control.

## Set up appropriate home care to provide patients with rehabilitation and monitoring of conditions.

Consider arranging a falls risk assessment through a CCAC soon after the patient has returned home.

- The "Virtual Ward" project,<sup>276</sup> funded by MOHLTC and being piloted at several sites, improves transitions from hospital to home by combining the best aspects of hospital, primary and home care after hospital discharge. Launched in March 2010, the project is an innovative partnership between the Toronto Central CCAC,<sup>277</sup> St. Michael's Hospital and Women's College Hospital. Toronto Central LHIN<sup>278</sup> is also participating. The partners are working together to create an integrated team including an attending physician and nurse practitioner from the hospital and a care coordinator, pharmacist and nurse practitioner from the CCAC to support clients discharged from hospital who are at high risk for readmission. This team meets daily to discuss the care management of clients on the virtual ward. Once clients have safely transitioned home and no longer need ongoing follow-up, they are discharged from the virtual ward. Many clients continue to receive CCAC services.
- MOHLTC recently added financial incentives for primary care practices that provide comprehensive primary healthcare services to patients. Compensation is based on blended payments linked to the number of patients enrolled and includes additional financial incentives for accepting acute care and vulnerable complex patients who were previously without a family physician.<sup>279</sup>

### Keeping people healthy in long-term care 3.4

	What do Ontarians want?	What if it doesn't hap	pen?	Who benefits most?		
	LTC residents preserve their well-being and ability to function as long as they can, through activities to help preserve bladder function and physical mobility, control pain, preserve language, memory and thinking abilities, and avoid depression and weight loss.	of independence, reduced q ulcers. Without appropriate they may endure needless s and feel social isolation. Dep	nd physical mobility, residents may experience loss quality of life and increased risk of pressure e pain control and activities that are stimulating, suffering, be unable to participate in activities epression and a loss of appetite can lead d and mobility, and premature death.		Ontario's 75,000 residents in the over 600 LTC homes across the province. <sup>283</sup>	
	Indicator			Bottom line		
	Percentage of residents with worsening <sup>+</sup> bladder control.			Ontario has just started to report these indicators so it is too early to tell if they are improving, and there are no international benchmarks available yet. Even though there is a tendency for LTC residents to decline over time, effective care can help to slow the rate of decline and research shows that there is still room for improvement in each of these areas.		
	Percentage of residents with increasing <sup>+</sup> difficulty carrying out nor- mal, everyday tasks (getting dressed, eating, personal hygiene)					
	Percentage of residents with pain that got worse recently $\ensuremath{^\dagger}$			Visit http://www.hqontario.ca/en/lt information on individual homes.	c_landing.php for more	
	Percentage of residents with worsening $^{\rm t}$ symptoms of depression or anxiety					
Percentage of residents whose language, memory and thinking abilities have decreased recently $^{\!\dagger}$			13%			
	Percentage of residents with recent $^{\dagger\dagger}$ unintended weight loss					

Data source: \*Continuing Care Reporting System, April 2009 to March 2010, calculated by Canadian Institute for Health Information. Under the system, every resident undergoes a detailed assessment using the Resident Assessment Instrument — Minimum Data Set (RAI-MDS) 2.0 tool of his or her health at least once every three months by a staff member at the home specially trained to collect this information. MOHLTC has implemented RAI-MDS in all LTC homes across the province. Results are based on 626 homes that have enough data to report. <sup>1</sup>From one assessment period to the next — typically, every three months. <sup>11</sup>A 5% loss over three months, or a 10% loss over six months.

Please see section 54 for root causes and ideas for improvement.

### Keeping people healthy in home care 3.5

## HOSPITAL **LONG-TERM CARE HOME CARE** PRIMARY CARE

### What do Ontarians want?

Home care clients preserve their wellbeing and ability to function as long as they can, through activities to preserve bladder function and mobility, control pain, preserve communication ability, memory and thinking abilities, and avoid depression and weight loss.

#### What if it doesn't happen?

Without bladder function and mobility, home care clients may experience loss of independence, reduced quality of life and increased risk of pressure ulcers. Without appropriate pain control and activities that stimulate their minds, they may endure needless suffering, be unable to participate in activities and feel social isolation. Depression and a loss of appetite can lead to decreased energy, mood and mobility, and premature death.

Ontario's 185,000 clients who receive services through CCACs on any given day<sup>328</sup> and the over 600,000<sup>329</sup> clients who received home care services from CCACs in 2009. Many of these individuals are long-stay clients.

Who benefits most?



#### Data source:

Home Care Reporting System, FY 2009/10, calculated by Canadian Institute for Health Information. Under the system, every long-stay home care client undergoes a detailed assessment using the Resident Assessment Instrument — Home Care (RAI-HC) tool of his or her health at least once every six months by someone specially trained to collect this information. The following RAI-HC indicators were adjusted for various risk factors: failure to improve/incidence of bladder incontinence; failure to improve/incidence of bladder incontinence; failure to improve/incidence of activities of daily living (long form) impairment; prevalence of inadequate pain control among those with pain; prevalence of negative mood; failure to improve/incidence of cognitive decline; and prevalence of weight loss

## 3.5

## **Root Cause**

### Issue: Bladder incontinence.

**Staff lack familiarity** with strategies such as prompted voiding to reduce incontinence.<sup>284</sup> **Residents might find it uncomfortable** or unnatural to go to the bathroom on schedule.

### Issue: Decline in mobility.

**Residents underuse mobility aids**, such as canes or walkers, even though they are known to prevent falls.<sup>288,289</sup> This could be due to feelings of embarrassment about using them, because they are uncomfortable, because they don't know how to use them, or because they were never offered.

**Lack of exercise or rehabilitation**, because services are not available, not tailored to the individual's needs or too expensive.

### Issue: Pain.

**Providers have difficulty recognizing pain**, particularly among residents with dementia.<sup>294,295</sup>

#### Residents may be reluctant to take

**pain medications** due to fear of addiction or tolerance or dislike of side effects (e.g., constipation).

### Physicians may be reluctant to prescribe pain medications.

They may worry that residents will become addicted; this is a particular concern with narcotic agents.

Age can affect a drug's effectiveness, sensitivity and toxicity, and it may be difficult to predict optimal dosages and potential side effects.<sup>300</sup>

## **Ideas for Improvement**

**Staff training and standard protocols** for implementing prompted voiding routines have been shown to improve continence in LTC.<sup>285</sup> With this approach, staff remind residents to void at certain times in the day, which helps avoid accidents. Ensure new or short-term staff are familiar with these techniques or partnered with those who are. The Registered Nurses' Association of Ontario has workshops and materials to support these strategies.<sup>286,287</sup>

**Encourage use of mobility aids**. Connect residents with others who have overcome shame of mobility aids and who now live more active lives. Ensure that users are well trained and comfortable with how to use them.

Make assessment of the need for mobility aids routine, and conduct systematic checks of mobility aids to ensure they are appropriate. The RAI-MDS 2.0 tool provides guidance on how and when to assess residents' use of mobility aids.<sup>290,291</sup>

**Ensure residents are capable of using devices and are trained in their proper use**. Some devices require the user to have appropriate cognitive function and strength to use them properly and not increase the risk of falls.

**Offer a variety of different exercise or rehabilitation therapies**.<sup>292</sup> A systematic review determined that exercise interventions reduced the risk of falls in the elderly by 14%.<sup>293</sup>

**Use a validated pain assessment tool**. Several pain assessment tools exist that are designed as checklists and that use visual cues, such as facial expressions and changes in behaviour, to recognize pain.<sup>296,297,298</sup> A review that rates pain recognition tools by validity, reliability and feasibility can be found here: www.biomedcentral.com/content/pdf/1471-2318-6-3.pdf. RAI-MDS 2.0 also provides for regular assessment of pain.<sup>299</sup>

**Make the use of these pain assessment tools a part of routine assessments of residents**. Train all staff in their use.

Guidelines are useful in helping staff manage persistent pain and assess responsible medication use. Develop standardized protocols for pain control, agreed to by all staff, outlining how to use short-acting and long-acting narcotics.

**Maximize use of safe medications,** such as acetaminophen. Consider adding non-addictive drugs for chronic pain (e.g., low-dose nortriptyline or gabapentin).<sup>301</sup> Also consider non-drug alternative therapies for pain control, such as acupuncture,<sup>302</sup> exercise or physical therapy.<sup>303</sup>

**Consider adopting a "universal precautions" approach**,<sup>304</sup> assessing every resident for risk factors associated with addiction using brief screening tools.<sup>305,306,307</sup> Employing these tools can increase confidence in prescribing and monitoring the use of drugs and help identify patients for whom more intensive monitoring is appropriate.



### **Root Cause**

Issue: Depression.

**Residents experience social isolation**, especially with the initial move to LTC.<sup>308</sup>

A decline in general health or physical or psychological capacity may make people feel depressed, especially if it limits their daily activities.<sup>319</sup>

Changes in the brain may make older people more vulnerable to depression.<sup>320</sup>

Deaths of friends or family, which become more common with age, can lead to depression.<sup>321</sup>

**Staff may not recognize depression in older people**, as it frequently differs compared to depression that occurs earlier in life, or it may be considered a "natural" part of aging.<sup>322,323</sup>

## **Ideas for Improvement**

**Offer a range of social activities**, such as exercise classes, field trips, music and art classes, activities that provide opportunities for intergenerational mixing and pet therapy. Group activities address feelings of isolation and loneliness.<sup>313,314</sup> Activities are especially likely to prevent (or alleviate) depression when they engage residents' interests, rather than just occupying their time. Ensuring residents are able to choose activities is especially important.<sup>315</sup>

Offer counselling or antidepressant medications.<sup>316,317,318</sup>

**Screen for signs of depression** using the RAI-MDS 2.0 tool or other validated depression scales.<sup>309,310,311,312</sup> **Better equip staff to identify and manage depression in residents**. Knowledge of the ways in which age may alter factors associated with the onset and persistence of depression is essential for effective treatment of depressed older adults.<sup>324</sup>

- The Residents First Initiative: Advancing Quality in Long-Term Care, now in its second year, supports LTC homes to use quality improvement techniques to improve the delivery of care. Approximately 75% of LTC homes are participating in the Leading Quality Program, which helps leaders incorporate quality as a core organizational strategy.<sup>325</sup> Roughly 20% of LTC homes are completing structured quality improvement training that focuses on improving resident care across a number of clinical areas, such as falls prevention and continence care. It is anticipated that many more homes across the province will take advantage of this training opportunity.
- MOHLTC is implementing a new inspection methodology for LTC homes. LTC homes can replicate the comprehensive Resident Quality Inspection as they work to improve resident care.<sup>326</sup>

### **Hospital infections** 4.1

Infections acquired in a hospital waste healthcare resources. They also cause patients to suffer and sometimes result in death.331 Hospitals have an obligation to do everything they can to prevent hospital-acquired infections.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
As few hospital-acquired infections as possible.	Unless Ontario eliminates hospital-acquired infections, Ontarians will continue to bear the burden of unnecessary deaths, longer hospital stays, more hospital costs, more disability and more psychological effects. <sup>332</sup>	Ontario's hospital patients; each year there are nearly
Hospitals to adopt evidence-based prevention practices, including effective hand washing and protocols to prevent surgical site infections.	Without complete adoption of effective prevention practices, Ontarians will experience more hospital-acquired infections. <sup>334, 335</sup>	1.1 million hospital discharges in Ontario. <sup>333</sup>



Data source:

MOHLTC. See also www.ontario.ca/patientsafety. Most recent values: C. difficile, MRSA, VRE — average over 2010; VAP, CLI, SSI – October to December 2010; hand hygiene – FY 2009/10. VRE cases include only those that result in bacteremia (infection in blood); VAP cases include only those occurring in intensive care units (ICU) after at least 48 hours of mechanical ventilation; CLI cases include only those occurring in the ICU after at least 48 hours of being placed on a central line. Hand hygiene compliance represents percent of instances where proper hand hygiene took place when it should have.





#### Data source:

MOHLTC. See also www.ontario.ca/patientsafety. Most recent values: C. difficile, MRSA, VRE — average over 2010; VAP, CLI, SSI — October to December 2010; hand hygiene — FY 2009/10. VRE cases include only those that result in bacteremia (infection in blood); VAP cases include only those occurring in intensive care units (ICU) after at least 48 hours of mechanical ventilation; CLI cases include only those occurring in the ICU after at least 48 hours of being placed on a central line. Hand hygiene compliance represents percent of instances where proper hand hygiene took place when it should have.

4.1 Hospital infections

41

## **Root Cause**

Lack of urgency or sense of importance in tackling hospital infections within the organization or staff.

Providers may forget to follow all recommended procedures for infection control.

Lack of experience.

Providers are unaware of how poorly they are following infection control guidelines.

**Providers may be reluctant to follow infection control guidelines** if they are skeptical of the evidence or don't think it is a problem for them.

### **Ideas for improvement**

Exert strong leadership to develop of a culture of safety. This may include any of the following:

- Leaders can deliver frequent messages to staff about the importance of safety and the
  organization's commitment to improve. Delivering these messages in person (e.g. through
  walkarounds with staff)<sup>345</sup> can be more powerful than by print.
- Leaders can also reinforce the message about the importance of reporting safety incidents or deficiencies, and that staff will be supported rather than punished when they do so.
- Offer rewards or recognition to units or programs for achieving infection control benchmarks. Promote those successful at improving safety to prominent positions.
- Leaders can use regular safety culture surveys of employees to monitor how strong the safety culture is within the organization.<sup>346</sup>

**Use checklists and flow sheets to prompt action**. For ventilator-associated pneumonia, use checklists to remind everyone to keep the head of the bed at 45 degrees, try daily sedation breaks, and enact other best practices.<sup>347, 348</sup> For surgical site infection prevention, checklists before surgery can ensure the right antibiotics are given at the right time.<sup>349, 350</sup> Give cleaning staff a checklist of items that require daily cleaning or cleaning at discharge.<sup>351</sup>

The Keystone ICU project in Michigan implemented a five-step checklist to reduce catheter-related bloodstream infections, which helped decrease the infection rate from 2.7 infections per 1,000 catheter days to zero.<sup>352</sup>

Ensure that only those trained to do intensive care medicine work in ICUs.<sup>353, 354, 355</sup>

**Ensure that all new or temporary staff are oriented to infection control procedures.** Develop a process to verify their skills or compliance with protocols early in their time with the organization.

**Regularly monitor compliance with protocols and report on performance.**<sup>356, 357</sup> Although the province mandates public reporting on many key measures, leaders can consider posting this information within the hospital more prominently, or breaking down statistics by individual department<sup>358</sup> or type of provider (e.g. hand hygiene compliance among physicians, nurses, etc.). While some measures such as C. difficile must be collected monthly, hand hygiene reporting is mandated in Ontario only once a year. Hospitals that are aiming for faster improvement may consider collecting and reporting this data to staff much more frequently.

**Explore strategies for healthcare provider buy-in.** Identify leaders or champions<sup>359</sup> in different professions in the hospital (e.g., doctors, nurses, administrators) to work with those refusing to comply.

**Review clinical evidence.** Look at the hospital's infection statistics and make the case for change. Use existing evidence-based tools<sup>360</sup> and work with all providers' ideas to make the process as efficient as possible. Work with administration and the hospital board to revoke privileges for staff and professionals who refuse to practice at the standard of care.<sup>361</sup>

HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

## Root Cause Ideas for improvement

### Issue: Poor compliance with hand hygiene protocols.

No clear accountability to the patient.

Healthcare providers may feel that they don't have time to wash their hands.<sup>364</sup>

Hand hygiene stations are not conveniently located.

Hand washing solution and alcohol-based gels may irritate skin and cause chapped hands.<sup>366</sup>

**Encourage patients and families to take a more active role in hand hygiene promotion.** Create tools (e.g., brochures, posters, videos, buttons) that educate them on when, why and where hand hygiene tasks should be performed and give them explicit permission to ask or remind healthcare providers to follow hand hygiene protocol.<sup>362, 363</sup>

Work hand washing into routines, such as washing hands when first meeting patients.

**Put hand washing stations in convenient areas**,<sup>365</sup> such as in patient rooms, at the entrance to rooms, at the entrance to wards and by elevator doors. Provide alcohol-based hand rub at patient bedsides. Regularly inspect stations to ensure they are not empty.

Provide soaps or hand sanitizer products that have moisturizers.  $^{\rm 367}$ 

- Hospitals are required to immediately report *C. difficile* outbreaks to their local public health units. This gives medical officers of health the
  information they need to monitor and respond to emergent outbreaks. Infection control resource teams are available to help hospitals respond
  to outbreaks.<sup>368</sup>
- The hand hygiene program for Ontario hospitals Just Clean Your Hands was launched in March 2008. Hospitals are required to publicly report hand hygiene compliance among healthcare workers.
- There are 14 Regional Infection Control Networks (RICNs) across the province that promote the best approaches to infection prevention and control.<sup>369</sup>
- Since 2008, Ontario has promoted public reporting of hospital-acquired infection indicators by individual hospital.<sup>370</sup>
- Three Ontario hospitals are participating in the Canadian Positive Deviance project as pilot sites.<sup>371</sup> This approach identifies individuals who follow a beneficial practice when most others do not, uncovers the secret behind their behaviour and aims to spread it to others.

## 4.2 Adverse events in acute care hospitals

When a patient experiences an unintended, undesirable change in health caused by healthcare services, it is described as an adverse event.<sup>372, 373</sup> According to a Canadian study, 37% of adverse events are preventable, often because they are due to medical error.<sup>374</sup> Examples of medical errors include forgetting to give a drug or treatment, giving the wrong treatment, doing a procedure with improper technique, not recognizing a warning sign early or making the wrong diagnosis.<sup>375, 376</sup>

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Surgery patients experiencing as few pulmo- nary embolisms (blood clots in the lung) or deep vein thrombosis (DVT; blood clots in the leg) as possible, through the use of blood thinners and encouraging people to be up and about where possible. <sup>377</sup>	Blood clots in the legs can break off and end up in the lungs, leading to a pulmonary embolism. The chance of death is 5% from DVT and 33% from pulmonary embolism. <sup>378</sup> Pulmonary embolism is the most common preventable cause of hospital death, <sup>379</sup> and DVT can lead to long-term problems with blood circulation in the leg. <sup>380</sup> DVT and pulmonary embolism also increase hospital costs. <sup>381</sup>	Ontario's surgery patients; in FY 2009/10, there were 1,207,000 day surgeries <sup>382</sup> and 285,000 acute in-patient surgical discharges <sup>383</sup>
Use of best practices like the surgical check- list, to ensure 26 key tasks are always carried out by operating room teams.	Use of the surgical checklist has been shown to reduce deaths and complications following surgery. $^{\rm 384}$	in Ontario.
No "never events", such as surgery on the wrong site or wrong patient, <sup>385</sup> which are clearly due to a failure to follow standard procedures. This report examines one type of never event: a foreign object left inside the patient after a procedure.	Never events have devastating consequences for patients. In the case of retained foreign objects, they include infections, pain, unnecessary repeat surgery or death. Reactions to these events from patients, families and the community include anger and loss of trust. Hospitals pay for both the costs of additional care and potential lawsuits. Hospital staff may suffer terrible guilt from these events.	Ontario's hospital patients who account for more than 1.1 million hospital discharges each year. <sup>386</sup>
All hospital patients avoiding harm from their medical care. Specific examples include falls, pressure ulcers, bladder infections and pneumonia – events that occur less frequently when the right level of nursing care is provided. <sup>387</sup>	Adverse events can cause pain and suffering, unintended injuries, disability, longer stay in hospital and increased risk of death. <sup>388, 389, 390</sup>	

Indicator	Value	Time trends & comparisons	Bottom line
Adjusted rate of in-hospital pulmonary embolism and DVT per 100 surgical procedures: → DVT → Pulmonary embolism → Sum of DVT and pulmonary embolism	0.20* 0.32* 0.52*	0.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0	Following a surgical procedure, one in 200 patients develop a serious blood clot. In the past two years, these rates have not improved, and in fact the rate of pulmo- nary embolism has increased. This may reflect improved reporting and focus on this issue. There is room for improvement.
Percent of surgeries in which a surgical safety checklist was performed	98%**	100 50 Apr-Jun 10 Jul-Dec 10 BETTER	Use of the surgical checklist is now just under 100%, which is welcome news. It will be important to monitor its impact in reducing deaths and complications in the near future.
Rates of adverse events per 1,000 medical/surgical patients in hospital: Pressure ulcer Fracture Hinary tract infection Pneumonia	2.1*** 0.8 15 10	15.0 <sup>a</sup> / <sub>22</sub> 7.5 0.0 2007/08 2009/10 better	Over the past two years, rates of these adverse events have either increased or remained stable. This increase may be a result of increased reporting of these events. There is room for improvement.
Cases per 100,000 procedures where a foreign object was left in the patient	9.9***	20 <u>BETTER</u> <u>BETTER</u> <u>BETTER</u> <u>BETTER</u> <u>BETTER</u>	In about one in ten thousand procedures in Ontario, a foreign object was left in the patient. This rate has increased in recent years, although this could be due to in- creased reporting of these events. Although these instances are rare, they can cause serious harm and should never happen.

Data sources: \*Discharge Abstract Database (DAD), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences. \*\*Web-enabled Reporting System (WERS) or Surgical Efficiency Target Program (SETp), hospital self-reported data, July to December 2010, provided by MOHLTC. \*\*\*DAD, FY 2009/10, provided by the Canadian Institute for Health Information (CIHI), FY 2009/10.

## HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

## **Root Cause**

Healthcare providers may not follow best

**practices** because they are busy, distracted by other patient issues or there are too many things to remember.

## **Ideas for Improvement**

Use standardized admission orders, checklists, order sets or decision support tools.<sup>391</sup> Examples include:

- Standard risk scoring sheets to identify people at high risk for venous thromboembolism<sup>392, 393</sup> falls or ulcers<sup>394</sup>;
- Checklists to remind staff to follow recommended practices for falls prevention for those at high risk<sup>395</sup> (e.g. keep bed low, make call bell or commode easily accessible, provide no-slip footwear, check for clutter, arrange for physiotherapy or assistive devices for walking; see sections 4.5 and 4.6);
- Checklists of best practices for pressure ulcer prevention (e.g., special mattresses or padding, turning immobile patients regularly, placing labels in chart or by bed to remind staff who is at high risk of an ulcer);<sup>396, 397</sup>
- Visual reminders and cues to help staff spot any evidence of new pressure ulcers in patients on their unit.<sup>398</sup> (See section 4.5.)

**Routine performance monitoring and feedback.**<sup>399</sup> Feed back data to surgeons on their rate of pulmonary embolism and DVT or compliance with the use of blood thinners. Regularly feed back data to nurses on nursing-sensitive adverse events.

**Set limits on on-call hours.** Hospitals can set policies to limit the consecutive hours on call or require rest time after being on call.

**Focus on healthcare provider wellness.**<sup>401</sup> This includes ensuring providers have proper nutrition and hydration throughout their shifts.<sup>402</sup>

**Increase available staff time.** Many adverse events have been shown to occur more frequently when there are fewer nurses available.<sup>403</sup> Although increasing nursing staff time at the bedside can be achieved by adding more nurses, it is also important to first consider how the same staff could do more bedside care by eliminating unnecessary tasks or streamlining their work.<sup>404</sup> Programs such as "Releasing Time to Care" in the UK have redirected 18% of nurses' time away from administrative or non-clinical tasks back to direct patient care in selected pilot sites.<sup>405</sup>

Providers are unaware of how significant a problem adverse events are.

Healthcare providers make judgment errors because of fatigue.<sup>400</sup>

Staff have too little time to do all recommended procedures.

## What is Ontario doing?

The Excellent Care for All Act, enacted in June 2010, requires hospital boards to make certain that the administrator establishes a system that
ensures a disclosed incident is analyzed and a plan is developed with systematic steps to avoid/reduce the risk of further similar incidents.<sup>406</sup>

### **Mortality in hospitals** 4.3

Hospitals try to provide patients with timely service, a positive experience and recovery from their medical condition — but most critical of all is their ability to save lives. Despite complex, challenging situations, hospitals must apply appropriate treatments and avoid medical errors that can lead to needless deaths. There are two main ways to measure mortality: the hospital standardized mortality ratio (HSMR<sup>†</sup>), which compares how many deaths occurred to what might be expected given the types of cases the hospital sees, and mortality rates for specific medical conditions.

What if it doesn't happen?

Higher mortality rates are

clearly undesirable.

Who benefits most?

Ontario's hospital

for more than 1.1

million hospital

year.412

discharges each

patients who account

### What do Ontarians want?

- Heart attacks
- Surgery and procedures such as coronary artery bypass grafts and percutaneous coronary interventions; and,
- Other common conditions, including congestive heart failure, pneumonia, chronic obstructive pulmonary disorder (COPD), septicaemia, lung cancer, stroke, respiratory failure and hip fracture.
- Ontarians want hospitals to do everything possible to prevent deaths, including:
- Ensure patients get the right drugs, tests and treatments;
- Follow hand washing routines, protocols to prevent surgical site infection<sup>407</sup> and blood clots,<sup>408</sup> and use surgical checklists409;
- Do complicated procedures only if they have an abundance of experience;
- Not delay time-sensitive treatments, such as clot-busters for heart attack and stroke<sup>410</sup> and antibiotics for serious infections; and,
- Implement information technology systems and medication reconciliation to prevent drug errors.<sup>411</sup>

Indicator	Value	Time trends & comparisons	Bottom line
Percentage of report- able hospitals whose HSMR has decreased compared to the previ- ous year	69%*	100 100 100 100 100 100 100 100	Last year, almost seven out of 10 reportable hospitals expe- rienced a decrease in their HSMR score. There have been substantial improvements within the last four years; however, there may still be room to improve.
Adjusted in-hospital rate of death within 30 days per 100 patients admitted for stroke	16*	25.0 # 12.5 0.0 2003/04 BETTER BETTER 0.0 2009/10	About one in six stroke patients die within 30 days of a stroke. There has been minor improvement over the past four years. This implementation of the Ontario Stroke Strategy (see next page) may have contributed to this trend. <sup>413</sup> There is still room to improve.
Adjusted rate of death within 30 days per 100 patients admitted for heart attack	9.6**	25.0 BETTER BETTER 0.0 2002/03 2008/09	One in 10 patients die within a month of having a heart attack. Continuous decline in mortality in the past six years may be due to newer treatments (e.g., bypass or stents fol- lowing a heart attack) and increased use of life-saving drugs such as statins (see section 3.1).
Adjusted rate of death within 30 days per 100 ← coronary artery bypass grafts ← percutaneous coronary interventions	2.1** 0.78**	3.0 BETTER 0.0 2002/03 BETTER 0.0 2008/09	Approximately one in 50 people die after having a coronary artery bypass graft one in 135 people die after having a percutaneous coronary intervention. The mortality rates for these services were higher in 2008/09 than in previous years, although it is possible that this increase could be due to more high risk patients undergoing and benefitting from these procedures. <sup>1†</sup> It will be important to further investigate causes for this recent increase and ensure recommended guidelines are followed.

Data sources: \*Canadian Institute for Health Information, FY 2009/10.

Data sources: \*Canadian Institute for Health Information, FY 2009/10. \*\*Discharge Abstract Database, Registered Persons Database, FY 2008/09, calculated by Institute for Clinical Evaluative Sciences. \*The HSMR is the ratio of **actual** (observed) deaths to **expoeted** deaths. It focuses on the diagnosis groups that account for the majority of in-hospital deaths. Using a logistic regression model, it is adjusted for several factors that affect in-hospital mortality, including age, sex, length of stay, admission category, diagnosis group, co-morbidity and transfer from another acute care institution. An HSMR of 100 suggests that a hospital's mortality rate is the same as the national average, given the types of patients cared for. An HSMR greater or less than 100 suggests that a local mortality rate is higher or lower, respectively, than the national experience. See secure.cihi.ca/cliniweb/dispPage.jsp2cw\_page=hsmr\_results\_home\_e. \*\*This indicator has been casemix-adjusted for factors including age, gender and existence of certain conditions (e.g. diabetes, congestive heart failure, past heart attack or kidney failure), but not for factors such as the urgency of the surgery. Future versions of this indicator will aim to include this information.



See sections 3.1, 4.1 and 4.2 for specific change ideas related to heart attack, stroke, congestive heart failure, hospital infections and pulmonary embolism that could affect mortality.

## **Root Cause**

Healthcare providers may find it difficult to consistently follow all life-saving best practices. This could be because they are busy or overwhelmed with too much information and forget, or are not aware of the latest evidence.

**Facilities and providers may be inexperienced at handling certain conditions.** Mortality rates are lower for esophageal, pancreatic and liver cancer surgery,<sup>419</sup> cardiac surgery,<sup>420</sup> pulmonary embolism,<sup>421</sup> abdominal aneurysm repair<sup>422</sup> and carotid endarterectomy<sup>423</sup> when done by doctors and in hospitals that perform more surgeries.<sup>424</sup>

**Delivery of time-sensitive care is delayed.** It is important for heart attack patients to get thrombolysis or surgery quickly,<sup>432</sup> for stroke patients to get thrombolysis<sup>433</sup> and for pneumonia patients to get antibiotics as soon as possible.

**Failure to rescue.** Warning signs of rapid deterioration might not be recognized because the diagnostic cues are subtle and unrelated,<sup>435</sup> or they may not be acted on quickly because of poor communication, shift changes or being too busy or distracted.

### **Ideas for improvement**

**Increase the use of standardized admission orders, discharge checklists, order sets and decision support tools.**<sup>414</sup> In addition to the tools mentioned in sections 4.1 and 4.2 (e.g. checklists for surgery, falls, ulcers, deep vein thrombosis), hospitals can use "care bundles," or checklists of accepted clinical guidelines printed on forms that are made conveniently available to clinicians. Hospitals in the UK have used these bundles for stroke, congestive heart failure and chronic obstructive pulmonary disease to reduce mortality.<sup>415</sup> Different bundles for sepsis exist, and their use is associated with a doubling of the odds of survival.<sup>416, 417</sup> The Canadian Patient Safety Institute and the Safer Healthcare Now! Initiative<sup>418</sup> promotes Canadian care bundles that cover topics such as ventilator-associated pneumonia, central line infections and medication reconciliation.

**Create dedicated centres of excellence.** Canadian stroke guidelines recommend that patients be sent to designated stroke centres whenever possible,<sup>425</sup> as such centres have better outcomes.<sup>426</sup> Ontario's stroke system follows this model (see below). This principle can be followed for other diagnoses.

**Consider performing high-risk surgeries only in facilities that maintain a "threshold volume" of cases**, <sup>427, 428, 429, 430</sup> and where only surgeons with a minimum volume per year do the surgery. Currently, in cardiac care, there are minimum standards for facilities that are being followed. Cancer Care Ontario has released a minimum volume standard for hepatic-pancreatic-biliary tract surgery to improve outcomes for cancer surgery patients. These standards include minimum volumes of surgeries, a minimum number of surgeons and a minimum of necessary physical resources and human resources that must be available.<sup>431</sup>

**Develop standardized processes or put clinical pathways in place.** Improve emergency department triage processes.<sup>434</sup> Identify in advance who does what, when and in what order. For example, at the Windsor Essex District Stroke Centre, there is an acute stroke protocol facilitating thrombolysis, an acute stroke clinical pathway and order set supporting current best practice, multidisciplinary stroke care, a neurology unit providing high-level multidisciplinary expert care, onsite CT, MRI, interventional radiological, neurosurgical and vascular surgery services, and an acute stroke resource nurse who provides case management, as well as professional and patient education.

**Consider the use of rapid response teams**,<sup>436</sup> **with clear guidelines for when they are to be used.** Rapid response teams are clinical teams with critical care expertise who can be called at a moment's notice by anyone to assess and stabilize a patient whose condition is deteriorating.<sup>437, 438</sup>

**Consider addressing critical reasoning skills**<sup>439</sup> and teaching communication techniques, such as "situation-background-assessment-recommendation," which can help nursing staff accurately describe the critical nature of a situation to a physician or nurse practitioner.<sup>440</sup> Also consider teaching the critical language approach of "concerned-uncomfortable-unsafe-scared," which helps providers raise issues related to patient safety.<sup>441</sup>

- Safer Healthcare Now! is a national grassroots campaign aimed at reducing adverse events that cause harm or death. In Ontario, over 500 healthcare teams from 161 organizations have enrolled, representing almost half of all teams enrolled across the country. The greatest uptake is with medication reconciliation in acute care (162 teams) and surgical site infection prevention (159 teams).<sup>442</sup> Other teams are working on reducing hospital acquired infections, creating rapid response teams, preventing blood clots, reducing falls and optimizing care for heart attacks.
- Ontario's stroke system has nine regional and 18 district stroke centres, each with expertise and resources to give specialized stroke care.<sup>443</sup>
   Ambulances bypass other hospitals to go directly to these centres.

### Drug safety in long-term care 4.4

Medications, which can save lives and improve quality of life, may have side effects. Those side effects may be more severe for elderly people, who often have more complex medical problems that interact with the drugs.<sup>444</sup> This makes drug safety a very important issue in long-term care (LTC) homes. Many adverse drug events can be prevented<sup>445</sup> — for example, by steering clear of drugs known to be dangerous for the elderly,<sup>446</sup> by avoiding confusion on which drugs or doses a resident should be taking and by monitoring those drugs once a resident is receiving appropriate therapies.447

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Medications with serious side effects for the elderly are avoided when there are safer alternatives available. LTC staff can consult the "Beers" list of drugs to avoid <sup>448</sup> and the Agency for Healthcare Research and Quality (AHRQ) "never prescribe" list of drugs. <sup>449</sup>	Medications on the Beers and AHRQ lists may increase the risk of falls and cause dizziness, confusion or death. They may also lead to irritating side effects, such as dry mouth.	Ontario's 75,000 residents in 626 LTC homes across the province. <sup>450</sup>
Prescriptions to LTC residents for antipsychotic or anti-anxiety drugs are avoided unless there is a specific reason.	Antipsychotic and anti-anxiety drugs may increase the risk of $falls^{451}$ and cause dizziness, confusion, stroke or death. $^{452}$	



Data sources:

Registered Persons Database (RPD), Ontario Drug Benefits Database (ODBD), Ontario Health Insurance Plan (OHIP) Claims Database, Discharge Abstract Database (DAD), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES). \*\*RPD, ODBD, OHIP Claims Database, DAD, Client Profile Database, FY 2009/10, calculated by ICES.

### 4.4 Drug safety in long-term care



### **Root Cause**

Inappropriate resident behaviours, such as aggressiveness, lead to the prescription of antipsychotics or sedative-hypnotic drugs.

**Difficulty stopping drugs that patients have been on for years, because of an addiction or tolerance to the drug.** Withdrawal symptoms like insomnia or headaches might appear when a drug is stopped.<sup>456</sup>

Some LTC homes may have an internal norm where heavy drug prescribing is viewed as standard care. In one American study, residents entering a LTC home that already had high prescribing rates for anti-psychotics were more likely to be put on one without a clear indication.<sup>460</sup>

Physicians may be ordering high risk drugs because they are unaware of the risks or available alternatives, or are sceptical about the evidence that they can cause harm.

Feelings of loneliness, loss of autonomy or lack of purpose may lead to anxiety or difficult behaviour.

Staff feel they do not have enough time to provide face-toface care that would allow them to help residents address underlying issues behind anxiety or difficult behaviour.

**Residents entering a LTC home for the first time may face a difficult adjustment period.** Unfamiliar surroundings, different routines and new people can worsen confusion or generate anxiety, insomnia, or fear and aggressiveness. These symptoms, in turn, may lead to greater use of antipsychotics or anti-anxiety drugs.

### **Ideas for improvement**

**Encourage non-drug approaches to managing inappropriate behaviour.**<sup>453</sup> These include conflict de-escalation and good communication techniques, such as making good eye contact, using simple sentences and one-step instructions, and avoiding making the resident feel rushed.<sup>454</sup> Cognitive-behavioural therapy may be helpful for certain types of elderly persons<sup>455</sup>, to address the underlying cause of anxiety. Look to the Resident Assessment Protocol in the RAI-MDS 2.0 assessment tool, which is designed to guide care planning for a resident exhibiting aggressive behaviour.

**Gradually wean residents off these drugs** over a period of several weeks. Standard protocols<sup>457</sup> could help providers accomplish this.

**Switch to safer drugs.** For example, some antidepressants are preferable to sedative-hypnotics for anxiety.<sup>458</sup>

Consider treatments for withdrawal side effects, such as carbamazepine for benzodiazepine withdrawal,<sup>459</sup> if slow tapering proves to be difficult.

**Public reporting** of the use of these drugs by individual long-term care home can help motivate homes with unusually high rates of prescribing to address the problem. Provide **anonymous feedback to individual physicians** within a LTC home on their rates of use of high risk medications, for their own improvement purposes.

**Remove the most dangerous drugs from the formulary of LTC homes. Implement a well-designed electronic medical record (EMR),** which can provide warnings about potentially harmful prescriptions.<sup>461</sup> **Use computerized physician** 

**order entry (CPOE) with clinical decision support.**<sup>462</sup> When a particular drug is selected, these systems can ask the prescriber if a checklist of other strategies have been tried first.<sup>463</sup> **Consider academic detailing programs,** where specially trained individuals (typically pharmacists or nurses) visit practitioners to promote evidence-based drug prescribing practices. These have been successful at changing prescribing practices.<sup>464, 465</sup> Unlike drug companies, their information is unbiased.

Conduct regular **medication reviews**. While Ontario's LTC homes are already required to do so,<sup>466</sup> these reviews could be strengthened by having pharmacists use checklists or appropriateness criteria, or having an interdisciplinary team review drugs.<sup>467, 468</sup>

**Offer a range of social and recreational activities** to residents to address feelings of isolation and loneliness<sup>469, 470</sup> and prevent or alleviate depression.<sup>471</sup> (See section 3.4.)

**Help staff spend more time in direct contact with residents.** The Releasing Time to Care program guides clinical teams as they identify and streamline processes in order to free up staff time for activities that add value for residents.<sup>472</sup>

**Consider improvements to orientation programs for new residents.** This may include more intensive time spent up front getting to know residents and families and understanding their concerns, or better communication between staff in LTC and health care workers looked after the resident while he or she was still in the community to understand his or her medical and psychosocial needs.

- The Institute for Safe Medication Practice (ISMP) works with MOHLTC and other agencies to provide tools and support for medication reconciliation, optimize the use of medication incident report information, develop medication safety indicators, and implement the Medication Safety Self-Assessment program.<sup>473</sup>
- Ontario is expanding its MedsCheck program, which offers Ontarians one-on-one consultations with their local pharmacist to ensure that they are using medications safely and effectively. Three new services are now available: MedsCheck for Diabetes, MedsCheck LTC and MedsCheck at Home.<sup>474</sup>
- Ontario Regulation 79/10 of the LTC Home Act was introduced in March 2010, which requires homes to document medication incidents, disclose
  them to residents, and review these incidents regularly with a view to preventing them in the future.<sup>475</sup>

## 4.5

## Avoiding harm in long-term care

loss of cognitive functioning (memory, language and thinking abilities) may also contribute to their responding to situations with fear and aggression, or wandering and sleeplessness. These factors put residents at high risk of unintended harm. Healthcare providers need to do everything they can to minimize the risk of harm to residents and those who provide care.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Avoid falls.	Falls may lead to injuries, fractures or death. They can also increase the number of emergency department visits and hospitalizations. <sup>478</sup>	Ontario's 75,000 residents in 626 LTC homes
Avoid new pressure ulcers.	Pressure ulcers may lead to pain and suffering, worsening infection, risk of amputation or death. $^{\mbox{\tiny 480}}$	across the province. <sup>479</sup>
Avoid physical restraints.	Physical restraints may lead to a loss of control and depression. Paradoxi- cally, they can also increase the risk of falls, and the restraint itself may be a safety hazard (e.g., causing asphyxiation). <sup>481</sup>	
Avoid worsening behaviour (e.g., aggression or wandering).	Worsening behaviour may cause physical or psychological harm to the resident, other residents and staff.	
Avoid bladder infections.	Bladder infections may lead to more serious infections or delirium. <sup>482</sup>	

Indicator	Value	Time trends & comparisons	Bottom line
Percentage of LTC residents with a new <sup>†</sup> pressure ulcer (stage 2 or higher)	2.7%*		One in 36 residents will develop a new, serious pressure ulcer over a period of three months; that's about one in nine residents each year. There is major room for improvement in this indicator. Ontario can strive to achieve a value closer to zero.
Percentage of LTC residents whose behaviour has recently <sup>†</sup> worsened	14%*		HQO has just started reporting these indicators and there are no international benchmarks available yet; however, there is very likely room for improvement. Visit
Percentage of LTC residents with a recent <sup>†</sup> bladder infection	5.4%*		www.ohqc.ca/en/ltc_landing.php for more information.
Percentage of LTC residents who were physically restrained	17%*		Almost one in six LTC residents were physically restrained in the previous three months. Many LTC homes are adopting zero-restraint policies, and some countries have rates lower than Ontario's (e.g. USA, 8%; Switzerland, 6%). <sup>483</sup> Although these comparisons should be interpreted with caution because of different definitions of a restraint, it is reasonable to conclude that there is room for major improvement.
Percentage of LTC residents who had a fall in the last 30 days	14%*		Falls are common; one in seven LTC residents had a fall in the past month. In recent years, there has been no major change in the rate of emergency department
Rate of falls among LTC senior residents (aged 65+) per 100 resident years resulting in: Emergency department visit with fracture Emergency department visit without fracture Hospitalization	1.8** 8.8 3.0	20 BETTER BETTER D D D D D D D D D D D D D	visits or hospitalizations as a result of falls. Substan- tial variation in the rate of falls across LTC homes in Ontario suggests there is likely room for improve- ment. Visit www.ohqc.ca/en/ltc_landing.php for more information.

Data sources: \*Resident Assessment Instrument — Minimum Data Set (RAI-MDS), FY 2009/10, calculated by Canadian Institute for Health Information. Under the system, every resident undergoes a detailed assessment of their health at least once every three months by a staff member at the home specially trained to collect this information. \*\*Registered Persons Database, Ontario Health Insurance Plan Claims Database, Discharge Abstract Database, National Ambulatory Care Reporting System Database, FY 2009/10, calculated by Institute for Clinical Evaluative Sciences. \*Errom one assessment period to the next: twicelly, every three months

<sup>†</sup>From one assessment period to the next; typically, every three months.

## 4.6 Avoiding harm in home care

People with chronic conditions or complex needs who need healthcare services (such as nursing or rehabilitation) or personal support services over a long period of time may become long-stay home care clients. These individuals often have physical disabilities that make them more likely to fall, injure themselves, develop skin ulcers and experience other problems. Home care workers have an important role to play in reducing the risk of harm.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?	
Avoid falls or other injuries.	Falls and other injuries carry with them the risk of temporary or permanent disability, or death. They may also lead to more emergency department visits and hospitalizations.	Ontario's 185,000 residents who receive services through CCACs on any given day and the 603,535 <sup>508</sup> residents who received home care services from CCACs in 2009. Many of these individuals are long-stay clients.	
Avoid skin ulcers.	Skin ulcers may lead to pain and suffering, worsening infection, risk of amputation or death. They can also cause avoidable healthcare costs.		
Avoid neglect or abuse.	Neglect or abuse may cause worsening physical or psychological health.		
Avoid delirium (sudden confusion or decreased alertness).	Delirium may increase the risk of injury and/or rapid deterioration resulting in hospitalization or death.		

Indicator	Value*	Time trends & comparisons	Bottom line
Percentage of home care clients who report that they have fallen in the last 90 days	25%	30 50 15 0 Apr-Jun 07 50 15 0 Jan-Mar 09 50 50 50 50 50 50 50 50 50 50	One in four home care clients report falling in the last 90 days. There has been no ma- jor improvement in the past three years. There is room to improve.
Percentage of home care clients with a new pressure ulcer (stage 2 to 4)	1.6%	10 better be	Among long-stay home care clients, 1.6% have developed a new pressure ulcer (stage 2 to 4) identified over the previous six months. There has been no improve- ment in the past three years. There is likely room to improve.
Percentage of home care clients with unexplained injuries, burns or fractures	12%	25.0 12.5 0.0 Apr-Jun 07 BETTER BETTER	Approximately one in 15 home care clients report unexplained injuries assessed over the past 90 days. There have been modest improvements in the past three years. These events tend to be unreported or un- disclosed, so when they are reported it is important to pay special attention to these cases. There is still room to improve.
Percentage of home care clients showing signs of neglect or abuse	1.2%*	BETTER	Among all home care clients, 1.2% showed signs of neglect or abuse. There has been no change in the last three years. These events also tend to be underreported, so when they are reported it is critical to pay special attention to these cases. There is room for improvement.

Data source:

\*Resident Assessment Instrument — Home Care, FY 2009/10, calculated by Canadian Institute for Health Information. Under this system, every long-stay home care client is supposed to undergo a detailed assessment of his or her health at least once every six months by someone specially trained to collect this information. Home care clients noted above are long-stay clients.

## HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

45

## Root Cause

**Residents at high risk of adverse events are not identified early.** Early identification can ensure that those at highest risk get more vigilant care.

Staff may be unaware of the extent of the problem.

Staff may **forget to carry out all best practices** because they feel overwhelmed with information or distracted by other tasks.

**Lack of skills, training or experience.** Being able to identify early signs of ulcers, or knowing how to move a frail resident without shearing the skin, are examples of tasks that require training.

Staff may feel they have **not enough time** to complete all best practices.

**Lack of certain equipment.** Pressure relieving mattresses can prevent ulcers. High-low beds that can drop to a low height can reduce injuries from falls.

### Issue: Use of restraints.

Family or staff are concerned that residents will wander if not restrained.

### Issue: Falls.

A medical condition leads to frequent falls that are extremely difficult to avoid.

**Medication side effects, such as confusion and dizziness, can lead to falls.** Age can affect a drug's effectiveness, sensitivity and toxicity, and it may be difficult to predict potential side effects.<sup>492</sup> (See section 4.4.)

**Residents have difficulty moving around.** This increases the chance of falls.

Residents fall when rushing to get to the toilet.

### Ideas for improvement

**Conduct risk assessments.** These are typically done at admission to an LTC home but it is important to keep these up to date. Examples include the Braden Scale and the Pressure Ulcer Risk Scale (PURS)<sup>484</sup> embedded in the RAI-MDS dataset now used by all LTC homes. For falls, many risk assessment tools exist,<sup>485</sup> and RAI-MDS also generates Resident Assessment Protocols that can serve as risk assessments.<sup>486</sup>

**Provide real-time feedback to providers.** Ontario is already moving towards public reporting of falls and pressure ulcers by individual LTC home. Administrators can go further and post this information by individual unit within their homes.

Increase the use of **checklists, reminders, standard orders or decision tools.** For example, the "turn clock tool" posted on a resident's door reminds staff of which position a resident should be in at certain times of the day.<sup>487</sup>

**Provide appropriate training.** Consider **mentorship** models, where inexperienced staff are paired with experienced ones. **Verify proper technique by direct observation of staff.** This may be particularly important for new or temporary staff.

**Eliminate other activities that waste time,** such as duplicate documentation, so staff can spend more time on care at the bedside.<sup>488</sup> The Releasing Time to Care program, developed in the UK, guides clinical teams as they identify and streamline processes in order to free up staff time for activities that add value for residents.<sup>489</sup>

**Develop the business case** to show that these investments pay for themselves in the long run.

**Provide education about the hazards of restraints.** Restraints can increase the risk of falls, pressure ulcers and asphyxiation, worsen an injury if a fall occurs and worsen depression.<sup>490</sup>

Use alternatives to track when a potential wanderer gets up, such as bed or door alarms to signal when someone leaves unexpectedly.

**Try using special equipment that protects the resident from harm from a fall,** such as hip protectors, helmets, non-slip footwear, keeping the height of the bed low and having padding on the floor next to the bed in case the resident falls when getting in or out of bed.<sup>491</sup>

**Avoid certain medications.** Avoid drugs on the "Beers" list<sup>493, 494</sup> and use safer substitutes. Have a medication review done to check for drug interactions (see section 4.4).

Use a well-designed EMR to track medication use and reduce the use of psychoactive drugs that can contribute to falls.  $^{\rm 495}$ 

**Introduce mobility aids.** Residents should be fitted appropriately for aids such as canes, walkers and scooters.<sup>496, 497</sup> Trial equipment should also be available for residents to test before investing in an aid. Consider ways to make their use fashionable, especially for those who may feel shame at having to use them.

Consider physiotherapy and rehabilitation. Exercise programs and Tai Chi are also effective in reducing falls.  $^{\rm 498}$ 

**Consider using prompted voiding techniques** where residents to go the washroom on a schedule and avoid incontinence (see section 3.4).

### HOSPITAL

**LONG-TERM CARE** 

**HOME CARE** 

**PRIMARY CARE** 



- The Residents First initiative supports LTC homes to reduce harm to their residents and improve their lived experiences through the implementation of best practices and use of quality improvement tools. Topics of focus to date have included reducing falls and pressure ulcers, and changing processes which waste staff time so that staff can provide more face-to-face care for the resident. The initiative is now in its second year. Almost every Ontario LTC home is participating in the Leading Quality Program, which provides leaders with skills and tools that help them incorporate quality improvement as a core organizational strategy.<sup>504</sup>
- A new inspection method, Resident Quality Inspection, is being implemented in Ontario's LTC homes. It includes interviews with up to 40 residents in every home and screening to understand where more in-depth inspections are required. LTC homes themselves can replicate this comprehensive assessment as they work to improve resident care.<sup>505</sup>
- The North Simcoe Muskoka LHIN is implementing a LHIN-wide wound care project to reduce the prevalence of pressure ulcers in LTC, acute care
  and home care.<sup>506</sup> The LHIN is also partnering with Alzheimer Society of Ontario and the Alzheimer Knowledge Exchange on the Behavioural
  Support System project, which seeks to improve care for older adults with complex and responsive behaviours associated with cognitive
  impairments.<sup>507</sup>
- The Integrated Client Care Project introduces new models of integrated care to support people with complex clinical and psycho-social needs who require prolonged help to live in the community. The model of integrated home care delivery partners and aligns with other sectors to integrate care across the healthcare continuum. In 2010, four CCACs launched early implementation sites focusing on wound care, with the aim of improving client outcomes and increasing value for the healthcare system.<sup>512</sup>
- In 2008, MOHLTC announced the Strengthening Home Care Services in Ontario strategy, which consists of four key deliverables: improving
  accountability for the provision of quality home care services through public reporting on quality measures; delivering improved health outcomes
  for Ontarians through the Integrated Client Care Project (see above); enhancing fairness, transparency and communication in the competitive
  procurement process; and promoting innovation and flexibility in service provision.<sup>513</sup>

# **5.1** Patient experience in acute care hospitals and emergency departments

In 2009 in Ontario, hospitals discharged nearly 1.1 million people from acute care beds and emergency departments (EDs) handled 5.4 million visits.<sup>514</sup> Looking at both hospital and ED stays from the patient's perspective can help to identify strengths and areas for improvement in the healthcare system.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
An overall positive experience for patients in hospitals and EDs.	When the care experience is not good, that fundamentally means the health care system is not meeting the public's expectations. That erodes public confidence in the health care system and can decrease staff morale. <sup>515</sup> Also, people with bad past experiences with care may hesitate to seek care when they need it. <sup>516</sup>	All Ontarians who visit an emer- gency department or hospital.
Patients feel they have all the informa- tion they need and are engaged in decision-making.	When questions are not answered fully, there may be misunderstandings about instructions on how to stay in good health. <sup>517, 518</sup> When people do not feel engaged in decisions, they may be less likely to adhere to recommended drugs, <sup>519</sup> tests or other advice.	
A health care system that is easy to navigate responds quickly to needs and controls pain as best as possible.	Needless waits lead to lower satisfaction with care. Poor pain control leads to unnecessary suffering and in some instances, slower rates of recovery from illness. $^{\rm 520,\;521}$	

Indicator	Value*	Time trends & comparisons	Bottom line
Percentage of patients who would definitely recom- mend their hospital to friends and family → Hospital → ED	74% 58%	100 til 50 0 2005/07 til 50 0 2009/10 til 2009/10	In Ontario, 74% of patients would definitely recom- mend the hospital in which they received care. This is higher than the average in the USA (69%) but leading American hospitals achieve rates of around 85%. <sup>522</sup> Results are much lower for ED patients; only 58% would definitely recommend their ED to others. There has been no major change in these indicators in the last four years, and there is obvious room for improvement.
Percentage of patients who felt they were treated with respect and dignity → Hospital → ED	83% 77%	100 <u>100</u> <u>50</u> 0 <u>2006/07</u> <u>2009/10</u> <b>BETTER</b>	While most patients in hospital and the ED feel they are treated with dignity and respect, about one in five do not. There has been no major change in the past five years, and there is room for improvement.
<ul> <li>Percentage of hospital patients who:</li> <li>Got bathroom help in time</li> <li>Though the wait time after a call button was reasonable</li> </ul>	68% 62%	100 100 50 0 2006/07 100 BETTER BETTER	About one in three hospital patients express concern about timeliness of requests for help — either assistance to go to the bathroom or response to a call button. This has not improved over time. There is room for improvement.
Percentage of ED patients who said they waited too long to see a doctor	48%	100 50 0 2006/07 50 0 2009/10 BETTER BETTER	About half of ED patients reported waiting too long to see a doctor. This has not improved in the last five years. There is room for improvement.

Data source: \*NRC-Picker patient satisfaction surveys, provided by the Ontario Hospital Association, FY 2009/10.
5.1 Patient experience in acute care hospitals and emergency departments



Indicator	Value*	Time trends & comparisons	Bottom line
Percentage of patients who thought the staff did everything they could to help control their pain → Hospital → ED	78% 52%	100 100 50 0 0 0 0 0 0 0 0 0 0 0 0 0	Nearly three in four hospital patients, but only about one in two ED patients thought the staff did everything they could to help control their pain. This indicator has not improved in the last five years. There is major room for improvement.
Percentage of patients who received answers they could understand when they asked important questions to: A nurse in hospital A nurse in ED A doctor in hospital A doctor in ED Percent of patients who were able to understand explanations about test results Hospital ED	70% 66% 73% 70% 69% 65%	0 50 100 Percent 0 50 100 Detter 0 50 100 D 50 100	Staff in hospitals and EDs do not answer questions or provide explanations that are understandable to patients about one-third of the time. This has not improved in the last four years (data not shown). Improvement is absolutely necessary.

Data source: \*NRC-Picker patient satisfaction surveys, provided by the Ontario Hospital Association, FY 2009/10.

# 5.1

## Root Cause Ideas for improvement

#### Issue: Low rating of overall satisfaction in emergency department

	Informe metions about compared wait times and reason for delays 524 Good computing the patients
	Care and Home Care.).
the emergency department.523	placement (see section 2.1 — Wait Times in Emergency Departments, and section 2.4 — Access to Long Term
Patients do not want long waits in	Improve patient flow in the emergency department. Address ALC bed problem and wait times for LTC

**Inform patients about expected wait times and reason for delays.**<sup>524</sup> Good communication with patients, particularly on arrival at the ED, helps manage expectations. Patients' perceptions that wait times are appropriate or are shorter than expected has been shown to be more important than the actual wait time.<sup>525</sup>

When patients don't feel like they're treated with respect or courtesy, or staff are unresponsive to their needs, they rate their overall satisfaction lower.<sup>526, 527, 528</sup> **Improve staff and provider customer service and interpersonal skills** through clinically focused customer service training.<sup>529</sup> Examples of ways customer service can be improved include: acknowledging people when they arrive, making eye contact, being polite despite stressful situations, showing a positive attitude, and telling people in advance what they can expect and when.

**Provide cultural competence training** to help staff understand the role culture plays in interactions between health care staff and patients and their families.<sup>530</sup>

#### Issue: Low rating on pain control

Pain is not adequately recognized.Monitor pain as if it were the "fifth vital sign"<br/>fifth vital sign"<br/>531 Mandate the routine use of visual analog scales or numeric<br/>rating scales for assessing pain.<br/>532The physician order for pain relief<br/>is delayed.Consider techniques such as patient-controlled anaesthesia, where the patient determines, within limits, how much<br/>pain relief he or she needs.<br/>533Physicians are afraid of drug-<br/>seeking behaviour or creating<br/>addiction among patients.<br/>534Educate prescribers on appropriate pain management,<br/>535 so they better understand drug-seeking behaviours,<br/>and ensure they don't overestimate the risk of addiction.<br/>Use standard protocols for pain control. One protocol, for example, helps providers identify different types of pain<br/>(e.g., acute vs. chronic; somatic, visceral or neuropathic) and recommends the best drugs for each scenario.<br/>536

#### Issue: Patients do not get the information they require in a way they can understand

Patients may forget verbal explanations, especially if they are stressed from their illness.

Patients may not understand instructions.

Patients may have poor language skills, particularly if English is not their first language.

Providers may use medical terms that patients do not understand.

Patients and family members may feel uncomfortable asking questions to clarify instructions.

This may be due to not wanting to look foolish or uneducated, being uncomfortable with questioning authority or sensing the provider is rushed and not wanting to bother him/her. **Provide written discharge instructions for all hospital and emergency department patients.**<sup>537, 538</sup> Written discharge instructions should be simple and easy to understand and translated where necessary (see below).

**Provide patients and families with simplified instructions, using plain language**<sup>539</sup>; written material could also use pictures to reinforce instructions. Patient information tools can be standardized province-wide for common conditions.

Have interpreter services available with interpreters who are trained in medical terminology<sup>540</sup> for commonly spoken languages in the community. Also have information available in multiple languages that is written in plain language.

Use different media (e.g., patient videos) to explain complex information to patients.

Use the "teach-back" method to ensure that patients understand instructions. Patients are asked to repeat back any key instructions given.<sup>541</sup>

**Give patients and families a chance to ask questions.** Budget enough time for questions after patients and families have reviewed written material and have had a chance to absorb the information.<sup>542</sup> Providers should not be rushing out the door immediately after asking if there are any questions.

Offer reassurance that there are no "bad" questions.543

Provide Frequently Asked Question (FAQ) sheets to answer the most commonly asked questions.

#### HOSPITAL

**LONG-TERM CARE** 

**HOME CARE** 

**PRIMARY CARE** 



- The Excellent Care for All Act, enacted in June 2010, requires every hospital to have a patient relations process to address patient experience issues.<sup>544</sup> The Act also states that health care organizations must conduct patient experience surveys. Most hospitals already do so voluntarily, and those smaller hospitals who do not yet have a survey will establish one in the coming year. In future years, this requirement will spread to all health care organizations. The Act also requires hospitals to submit annual quality improvement plans, and hospitals have been asked to include at least one performance goal for patient experience in their plans.<sup>545</sup>
- Cancer Care Ontario's Ontario Cancer Plan 2011–2015 outlines how all cancer patients in Ontario will have their interests represented at a Patient Advisory Council, a forum to advise on initiatives to improve the patient experience, as well as having access to tools to help navigate the cancer system and manage their own journey (ocp.cancercare.on.ca).

## **5.2** Patient experience in primary care

Primary care is often the first place Ontarians seek health care. Ensuring patients have a good experience at the primary care level is important to maintaining a positive view of the entire health care system. A good experience, however, is more than just feeling satisfied with care; it also means that the individual feels like care is designed around his or her needs, that communication is clear and that the right to make one's own decisions is respected.

What if it depends her and	
What if it doesn't happen?	Who benefits most?
Having clear explanations to answers is an inherent right, and poor com- munication leads to poorer overall satisfaction with care. Better communica- tion can help people understand why it is important to take certain drugs, treatments or to change health habits, which in turn can lead to better health outcomes. <sup>546, 547</sup>	All residents of Ontario.
Encouraging people to manage their own chronic conditions and to feel they are in control of their health has also been shown to result in better health behaviours and health outcomes. <sup>548</sup>	
Disorganization and wasted time leads to confusion, inconvenience for patients and families, lower overall satisfaction with care and lost economic productivity.	
	<ul> <li>Having clear explanations to answers is an inherent right, and poor communication leads to poorer overall satisfaction with care. Better communication can help people understand why it is important to take certain drugs, treatments or to change health habits, which in turn can lead to better health outcomes.<sup>546, 547</sup></li> <li>Encouraging people to manage their own chronic conditions and to feel they are in control of their health has also been shown to result in better health behaviours and health outcomes.<sup>548</sup></li> <li>Disorganization and wasted time leads to confusion, inconvenience for patients and families, lower overall satisfaction with care and lost</li> </ul>

Indicator	Value*	Time trends & comparisons	Bottom line
Percentage of adults who have a regular doctor or place of care who rate the overall quality of medical care they received in the past 12 months as excellent or very good	77%	Percent Percent BETHER, AND CANADA UK NORMAN INFER, AND CANADA NORMAN INFER, AND CANADA NORMAN INFER, AND CANADA NORMAN INFER, AND O CANADA NORMAN INFER, AND O NORMAN NETHER, AND NORMAN NETHER, AND NORMAN N	Approximately three in four adults in Ontario who have a regular doctor or place of care rated the overall quality of the medical care they received in the past 12 months as excel- lent or very good. Ontario performs relatively well on this indicator. The province is doing better than Quebec and many European countries, and is on par with Western and Atlantic Canada, the US, UK and Australia. Only New Zealand has better results (84%).
Percentage of adults who have a regular doctor or place of care who said this provider always explains things in a way that is easy to understand	74%	Percent MORWAY LK NORWAY LK CANADA US CANADA US CANADA US CANADA US CANADA US SWITZFRLAND SWITZFRLAND SWITZFRLAND	Among adult Ontarians, 74% felt that their regular doctor or place of care always gave easy to understand explanations. Compared to other countries, Ontario and Canada are in the middle of the pack. Ontario could aspire to match the best results achieved by Swit- zerland (82%) and New Zealand (80%).
Percentage of adults who have a regular doctor or place of care who said this provider always tells them about treatment options and involves them in decisions about the best treatment	66%	Percent Percent NORMANY IK NORMAN	Only two in three adults who have a regular doctor or place of care said their provider always tells them about treatment options and involves them in decisions about the best treatment. This is an area where there is poor performance across the globe. Compared to other countries, Ontario and Canada are in the middle of the pack. Ontario could aspire to match New Zealand's result of 80%.
Percentage of adults who have a regular doctor or place of care who said this provider always spends enough time with them	58%	Percent Percent Percent Norway	Only 58% of Ontarians felt their regular doctor always spent enough time with them. There are many other countries with better performance than Ontario and Canada in this area. Again, Ontario could aspire to match New Zealand's result of 80%. Having the best performance on this and the previous two indicators may help explain why that country has the best satisfaction with the care received.

Data source: \*Commonwealth Fund International Survey of adults, 2010.





Data source: \*Commonwealth Fund International Survey of adults, 2010.

## **Root Cause**

## **Ideas for improvement**

#### Issue: Doctors do not have time to spend with patients

Providers waste time searching for things and doing unnecessary paperwork, resulting in less time to spend with patients.

**Use a well-designed EMR** that contains all the patient information and can save time by making it easier to access test results or other information in real-time.<sup>549, 550</sup>

**Improve office efficiency.**<sup>551</sup> Simple steps that save minutes or seconds of each clinic visit can add up to days or weeks of saved time over a year. See section 2.2 for more information.

Physicians may spend time doing tasks that could be performed by other staff.

Continue to **promote team-based models of care**, such as Family Health Teams, where physicians could share tasks with other providers and the practice as a whole can have more time to spend with patients. See section 2.2 for more information.

#### Issue: Explanations not always easy to understand.

Patients may not understand questions or instructions.

Patients and family members may feel uncomfortable asking questions.

**Use the "teach-back" method to ensure that patients understand instructions.** Patients are asked to repeat back any key instructions given.<sup>552</sup>

**Give patients and families a chance to ask questions.**<sup>553</sup> Allow for questions after patients and families have reviewed written material and have had a chance to absorb the information. See section 5.1 for more information.

#### Issue: Patients are not involved in decisions or treatment options

It takes time to list the options for screening or treatment as well as the pros and cons of each, and to identify what's important to patients (eg. side effects of treatment, cost of different options, etc.). **Develop standard tools for providers to help explain options to patients, as well as the benefits and drawbacks of each.** For example, Cancer Care Ontario provides a toolkit for patients that is designed to help them make informed decisions about their cancer screening options. The tools include easy to read information about risk factors, methods of screening, pros and cons of various approaches, and where to go for more information. The material is available from: www.cancercare. on.ca/cms/one.aspx?objectld=44184&contextld=1377

#### Issue: Patients have a negative customer service experience

When patients don't feel like they're treated with respect and courtesy, or that staff are unresponsive to their needs, they rate their overall satisfaction lower.<sup>554, 555</sup>

**Improve staff and provider customer service and interpersonal skills** through programs such as clinically focused customer service and communication training.<sup>556</sup> Examples of ways customer service can be improved include: acknowledging people when they arrive, making eye contact, being polite despite stressful situations, showing a positive attitude, and telling people in advance what they can expect and when.

#### What is Ontario doing?

• The ministry has commissioned a five-year longitudinal study to evaluate the effectiveness of Family Health Teams, including patient experience and patient satisfaction.<sup>557</sup>

## 6.1

## Cost of service delivery

It is important for hospitals and community agencies to operate efficiently by eliminating waste or using the most effective treatment options, so they can offer the best patient care in exchange for the lowest cost. Part of any organization's plan for efficiency must be to manage finances well and avoid incurring deficits or running into circumstances where it becomes hard to meet commitments to pay bills when they are due. Provincial data on cost effectiveness is poor, and it is difficult to measure inefficiencies across the system. Thus, better data capture is necessary (see section on data advocacy).

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Hospitals with sufficient revenue to cover their costs.	When hospitals spend more than they receive, they have to borrow money — and when that happens, taxpayer dollars go towards interest charges instead of towards purchasing equipment or providing patient care.	Ontario's taxpayers, who want to know their tax dollars are being wisely managed.
Hospitals able to pay bills on time with liquid resources (i.e., cash).	Again, when hospitals have to borrow to pay bills, taxpayer dollars go towards interest instead of towards needed goods and services.	
Hospital expenses minimized as much as possible, without compromising high-quality patient care.	When hospitals are compared and have significantly different costs to treat the same type of patient, it may mean that less efficient hospitals are using resources inappropriately.	



Data source: \*MOHLTC, FY 2009/10.

<sup>1</sup>Technically, a "negative budget position." Based on the Ontario Hospital Service Accountability Agreements. (Total Revenues — Facility Grant Amortization) — (Total Expenses — Facility Amortization). <sup>1</sup>Based on the Ontario Hospital Service Accountability Agreements. The numerator includes Current Assets plus debit Current Liability Balances, excluding Deferred Revenues. In plain language, this is the amount of cash or other assets that can be converted quickly into cash. The denominator includes Current Liabilities excluding Deferred Revenues plus credit Current Assets, except Current Asset Contra Accounts. In plain language, this is the amount of short-term debts.

Contra Accounts. In plain language, this is the amount of short-term debts. <sup>111</sup>Resources Utilization Groups III grouping methodology. The in-patient case weight information enables comparisons among hospitals regardless of differences in the severity of illness and complexity of cases served by these facilities. Costs were not adjusted for inflation.





HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

#### **Root Cause**

#### **Ideas for improvement**

Issue: Overall demand for hospital services is increasing beyond what hospitals can provide with current budgets.<sup>558</sup>

Avoidable demand on hospitals.

Avoidable hospital readmissions.

**Patients are occupying hospital beds unnecessarily.** These patients do not require the intensity of resources or degree of services provided in that setting, and could be better served elsewhere.<sup>571</sup>

Hospital services such as diagnostic screening and surgical procedures may be used inappropriately.

Unnecessary duplication of tests due to fragmentation of care. The results of tests ordered in primary care are often not available to hospital staff.

Avoidable complications of hospital care waste resources. Hospital-acquired infections<sup>579</sup> and other complications that develop while in hospital increase length of stay and overall costs.

Inefficient processes within the hospital waste time and resources.

**Improve chronic disease management in the community** (see section 3.2). Acute care in-patient costs for conditions related to chronic disease are high, with acute myocardial infarctions (AMIs) and cerebrovascular diseases representing the two most expensive medical conditions in Canada.<sup>559</sup> Studies have shown that improving the care of people with chronic diseases before they require hospitalization can decrease associated hospital costs.<sup>560, 561, 562, 563, 564</sup>

Improve access to long-term care (LTC) and home care, and ensure safe conditions and appropriate care to prevent avoidable hospitalizations (see sections 2.4, 4.5 and 4.6).

**Improve access to primary care** (see section 2.2). Studies have shown that better access to primary care results in fewer visits to the emergency department and admissions to hospital.<sup>565, 566</sup>

**Coordinate among health services,** such as primary care, specialists and hospitals, to improve health outcomes and use of resources.<sup>567</sup>

Ensure patients leave the hospital on the right medications (see section 3.1) and have all the information they need to function at home. Consider specialized out-patient clinics (e.g., for congestive heart failure<sup>568, 569, 570</sup>) or similar services that have been shown to reduce readmissions (see section 3.3).

**Ensure that patients are being cared for in the most appropriate place** (see sections 2.4 and 6.2).

**Eliminate unnecessary tests or procedures, such as repeat tests or pre-operative tests for minor procedures** (see section 6.3). A study of Ontario doctors revealed that use of CT and MRI screening may not always be for appropriate reasons.<sup>572</sup> Consider using appropriateness criteria for CT and MRI scans,<sup>573</sup> as well as for procedures such as hip and knee replacements<sup>574, 575, 576</sup> and cataract surgery.<sup>577</sup>

**Implement integrated electronic health records (EHRs)**, which provide convenient access to test results ordered by other providers and in different settings.<sup>578</sup> This could help reduce unnecessary repeat tests that are done because test results from elsewhere are unavailable.

Work on areas where there is a strong business case for quality.<sup>580</sup> A strong business case can be made when investments in improving quality are more than offset by reductions in cost for the same organization within a reasonable period of time. Business cases are well established for areas such as ventilator-associated pneumonia<sup>581</sup> and pressure ulcer prevention.<sup>582</sup>

Improve use of staff time and reduce redundancies (see section 2.1).583

**Improve efficiency of discharge processes.**<sup>584</sup> Consider options for improving the discharge rate, such as setting target discharge dates, staggering discharge times during the day, and clearly communicating plans with the patient and family so they can be ready for discharge. Utilization management software may help guide decisions on when it is safe to discharge patients. (See section 8.1 for information on improving communication and collaboration between hospitals and those who will be providing post-discharge care, and section 2.1 for information on improving care coordination to ensure that patients are moved to the right place as soon as possible so they are not occupying beds unnecessarily.)

## What is Ontario doing?

• The MOHLTC is working on a major reform of how hospitals are funded. Currently, hospitals are funded mostly through fixed global budgets which are largely determined by historical factors. In many cases, this funding does not reflect the population served or the types of patients cared for. Hence, a hospital in a high growth area may find it difficult to provide services to the expanded population if its global budget does not increase to reflect the greater demand. The MOHLTC's Patient-Based Payment strategy<sup>585</sup> aims to use a "money follows the patient" approach which links hospitals' funding to the level of services and quality of care they deliver. The strategy draws on the Health-based Allocation Model (HBAM), which determines the expected costs of delivering care, taking into account differences across communities in age, socioeconomic status and existing health conditions.

## 6.2 Right service in the right place

Individuals should be cared for in settings that best meet their needs, rather than settings that are more costly and not designed to accommodate them. The most pressing example is the Alternate Level of Care (ALC) patient.<sup>586</sup> These individuals (often, frail, elderly with multiple chronic conditions) may enter hospital with an acute problem needing immediate attention and then recover but still have difficulty functioning independently. The doctor may not discharge the patient if she is concerned that he cannot get enough home care to live safely at home. The patient may then be referred to a long-term care (LTC) home, but must wait in the hospital, often for weeks or months as an ALC patient, until a bed becomes available (see section 2.4).

Another example is people who are placed into a LTC home who could have been served elsewhere. Individuals who do may not need all the services that a LTC home provides may still end up being referred there because of lack of alternatives. The MAPLe (Method for Assigning Priority Levels) algorithm allows planners to identify these individuals.<sup>587</sup>

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Patients who no longer need hospital services are discharged to a more appropriate setting.	The cost of caring for an ALC patient in hospital is higher than caring for the same person in a more appropriate setting, such as an LTC home. Furthermore, LTC staff are specially trained to take care of people who are frail and need many support services; thus, the resident can be provided with more appropriate care. Another consequence of the ALC challenge is that beds occupied by ALC patients are not available to patients waiting to be admitted from the emergency department, leading to long wait times. Elective surgeries could be delayed as well. Finally, when older people are hospitalized for acute illnesses, they may lose their independence and are at high risk for discharge to LTC homes. <sup>588, 589, 590, 591</sup>	Ontario's 2,800 acute care hospital patients and 1,900 patients in other hospital settings (e.g. mental health, rehabilitation facilities) who, on any given day, are designated as ALC patients. <sup>592</sup>
People whose needs can be met through alternatives to LTC (e.g., home care or supportive housing) can remain living in the community — with LTC beds reserved for those who truly need them.	When less expensive alternatives of the same or better quality are available but not used, healthcare resources are wasted.	Ontario's 170,000 <sup>593</sup> long-stay home care clients who may need more care in the near future.



Data sources:

\* Discharge Abstract Database (DAD), Canadian Institute for Health Information (CIHI), April 2006 to March 2010, provided by Cancer Care Ontario. \*\*Based on the MAPLe score.<sup>597</sup> Client Profile Database, MOHLTC, January to March 2010, provided by the Toronto Central CCAC.

<sup>†</sup>R-squared = 0.86, indicating a strong relationship.



#### **Root Cause**

People's care needs are inadequately assessed.

Patients may decondition while in hospital. Being in a foreign environment may lead some patients to lose their function.

People are prematurely labelled as needing LTC, or go into LTC before they need it.

Alternatives to LTC are not being fully utilized or supported.

Limited availability of home care.

Avoidable demand due to poorly controlled conditions that could have been addressed in a primary care setting before they escalated.

#### **Ideas for improvement**

**Identify early those at risk of being hospitalized and subsequently becoming LTC patients** — for example, frail individuals or those with dementia living in the community with unmet needs.<sup>594, 595, 596</sup> Identifying these people early and providing them with adequate home care support may help to slow down the decline in their health and make it easier for them to go back home should they ever need to be hospitalized.

One program in the Mississauga Halton LHIN aims to ensure that every person aged 75 and over who comes to an emergency department has a home care assessment.<sup>597</sup> Use of the interRAI Community Health Assessment by community support agencies can help to identify more objective levels of risk in this population.

Started in 2007, **transitional care programs** give former hospital patients a period of reconditioning in a retirement home before they return home.<sup>598</sup>

The Champlain LHIN has piloted a 22-bed program that provides alternative care to medically stable hospital patients waiting for an LTC bed. This has reduced bottlenecks that lead to emergency room crowding and cancelled surgeries.<sup>599</sup>

Patients may benefit from acute, in-hospital rehabilitation and intensive, long-term, follow-up rehabilitation to improve the chances that they will recover their pre-hospitalization abilities.<sup>600</sup>

**Use objective criteria to help determine who truly needs LTC.** Carefully screen individuals' healthcare needs to ensure that only those with heavy needs actually get on the waiting list. This may help to address situations where people who fear long waits get themselves on the list "just in case." Tools such as the MAPLe score<sup>601</sup> can help care planners decide whether an individual's needs are heavy enough that they should be put on the list.

Employ the Home First Approach and send patients home from the hospital rather than directly to  $LTC^{602}$  (see section 2.4).

**Ensure there are sufficient alternatives to LTC homes, such as assisted living homes or supportive housing**,<sup>603</sup> where frail individuals can access some degree of ongoing care if their needs are less than what an LTC home would provide (see section 2.4).

Alberta has developed a Continuing Care System that includes supportive housing in its strategy.<sup>604, 605</sup> In 2009, HQO reported on the region around Lethbridge, Alberta, where this strategy helped keep the waiting list for LTC at only 29 days despite using one-third fewer LTC beds than Ontario. This region provides publicly funded options for assisted living and supportive housing that allow people to live in a home-like environment with 24-hour assistance when needed if they require less care than that provided by LTC but more than that offered by home care.<sup>606</sup>

**Consider increases in home care availability.** In the past, there were caps on hours of care for home care clients; these have recently been extended. This change may allow some clients to avoid being put on a waiting list for LTC. However, for people with heavier needs, other options such as assisted living may be more cost-effective than home care.

**Improve primary care services.** Improve access to primary care (see section 2.2), including after-hours primary care and management of chronic diseases (see section 3.2) so patients are less likely to require hospitalization.

#### What is Ontario doing?

• Ontario has invested over \$700 million over the past three years in the Aging at Home strategy,<sup>607</sup> which recognizes the need to provide access to a continuum of services for seniors and their caregivers, help seniors continue to lead healthy and independent lives, and assist those who wish to remain at home to live safely and with dignity and independence. The strategy was expected to relieve pressures on hospitals and LTC homes by helping to find more appropriate settings for patients ready for discharge from hospital.<sup>608</sup> It was also intended to help seniors avoid unnecessary visits to the hospital, reduce emergency department wait times<sup>609</sup> and ultimately reduce ALC. However, the ALC situation has not yet improved, and it is important to evaluate the impact of this strategy (please see the data advocacy section 1.12).

#### 6.3

## Avoidable emergency department visits

Many people end up in an emergency department (ED) because they are not familiar with other available choices. While EDs are intended to handle serious illnesses and injuries that require fast, highly skilled care, people often go to an ED for minor problems that could be treated in a doctor's office or after-hours clinic. This section examines the rate of avoidable ED visits in Ontario.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
People with non-urgent conditions (e.g., colds, sore throats, ear aches, bladder infections or needing prescription renewals) should be treated outside the ED in doctor's offices, after-hours clinics or urgent care centres.	Beyond the high cost of treatment in an ED, patients miss out on the benefits they receive when they are treated by a primary care doctor who knows them and is familiar with their medical history.	Ontario's one in five residents who visit an ED each year <sup>610</sup> (there are approximately 5.5 million ED visits every year in major cities within Ontario <sup>611</sup> ).
Early identification and treatment of long-term care (LTC) residents' worsening medical conditions (e.g., diabetes, pneumonia and congestive heart failure), so that their condi- tion does not deteriorate to the point that they need to be sent to an ED.	Without early identification and treatment, LTC residents may experience an avoidable ED visit <sup>612</sup> and suffer harm because of an avoidable worsen- ing of their medical conditions. <sup>613</sup>	Ontario's 75,000 residents in 626 LTC homes across the province. <sup>614</sup>
Treatment within an LTC home for residents with non-urgent, low acuity conditions, so it doesn't become necessary to send them to an ED. <sup>†</sup>	An LTC resident who experiences an avoidable ED visit is exposed to an unfamiliar place, <sup>615</sup> which may be distressing if he or she has dementia. <sup>616</sup> Other negative consequences include inconvenience for the resident (with long lengths of stay in the ED) and expense for the province (LTC residents usually arrive in the ED by ambulance).	

Indicator	Value	Time trends & comparisons	Bottom line
Percentage of people in Ontario who thought they could have been treated by the doctor or staff at the place where they usually get medical care if care had been avail- able the last time they went to the hospital or ED	48%*	Percent ILS CANADA NOTARIO CANADA CANADA NOTARIO CANADA CANADA NOTARIO CANADA CANADA NOTARIO CA	Almost one in two Ontarians felt they could have been treated by their primary care provider if care had been available the last time they went to the hospital or ED. Ontario is tied with the rest of Canada, the US and Switzerland as having the worst results on this indicator compared with eight other countries. Hospital and ED resources could be better utilized if individuals were seen in more appropriate care settings.
Rate of potentially avoidable ED visits per 100 LTC residents per year	21**	40 20 0 2002/03 2009/10 BETTER	Potentially avoidable ED visits are common among LTC residents. There has been no major change in the past seven years in this area. There is likely room for major improvement.
Rate of low acuity ED visits per 100 LTC residents per year	7.8***	40 <sup>10</sup> / <sub>22</sub> 20 0 0 2002/03 2008/09 BETTER 0 2008/09	Over the past seven years, there has been continuous improvement in the rate of low acuity ED visits by LTC residents. This is good news, but there may still be room for further improvement.

Data sources:

\*Commonwealth Fund International Survey of Adults, 2010. \*National Ambulatory Care Reporting System Database (NACRS) and Ontario Health Insurance Plan (OHIP), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES). \*\*\*Registered Persons Database, OHIP, Discharge Abstract Database, NACRS, FY 2009/10, calculated by ICES

<sup>†</sup> Non-urgent visits are defined as those assessed according to the Canadian Triage and Acuity Scale (CTAS) as level 4 or 5, where the person was subsequently sent back to the LTC home.



#### **Root Cause**

Issue: Non-urgent ED visits.

**People do not understand the purpose of the ED** or may be unaware of alternatives to the ED, such as after-hours or walk-in clinics.<sup>617</sup>

**Poor access to primary care.** People will use the ED if they don't have a primary healthcare provider, cannot get a timely appointment with their provider, or afterhours service is not available.<sup>621, 622</sup>

The frail elderly and people who have poorly controlled chronic diseases are prone to health crises. Patients may not follow healthcare providers' recommendations for self-management or receive all evidencebased treatments from their providers.

## Ideas for improvement

**Consider public education and awareness campaigns about the appropriate use of the ED.** Also consider further promotion of the use of the Telehealth Ontario toll-free number, which gives Ontarians the opportunity to talk with a nurse to help assess when to go to the ED.<sup>618,619</sup> Telephone nursing triage systems such as Telehealth Ontario have been found to reduce inappropriate use of EDs.<sup>620</sup>

**Improve access to primary care.** Consider better organization of primary care offices and management of patient appointments to reduce wait times, and provide after-hours service (see section 2.2).

Patients of family physicians who practice within groups or teams that provide after-hours clinics and access to on-call advice from a clinician are less likely to go to an ED.<sup>623</sup> Establishment of family health teams in Peterborough resulted in 15,000 fewer ED visits.<sup>624</sup>

**Work towards better management of patients with chronic diseases.** Patients with chronic diseases need to be more engaged in the care and management of their conditions. Primary care of these patients should be carefully managed to avoid crises requiring immediate attention (see section 3.2).

#### Issue: Avoidable ED visits by LTC residents.

Staff at LTC homes may be uncomfortable handling relatively minor emergencies and may have a low threshold for sending recently discharged patients back to hospital at the first sign of any complication or regression.

An on-call primary care provider is not available to assess the resident's situation.

A lack of diagnostic equipment in the LTC home (e.g., X-ray machines, urgent lab services) requires residents to go to hospitals.

Family members may exert pressure to send a resident to an ED for assessment. This may occur if the family does not have confidence in the staff's ability to handle the situation. **Increase training of staff at LTC homes so they know how to handle and assess minor emergencies.** Nurse practitioners in LTC homes can provide clinical support for both patients and staff to reduce unnecessary use of EDs.

Consider use of telemedicine to access expert advice with a video link.625

**Consider redesigning on-call schedules** — for example, sharing an on-call physician, nurse practitioner or physician assistant between LTC homes in close proximity to each other.

**Consider using nurse practitioners and registered nurses** either in the on-call schedule or to mentor other staff. For example, the Toronto Western Hospital created a Mobile Emergency Nursing Team of registered nurses who take acute care nursing expertise directly to the bedside of LTC home residents to reduce ED visits. In the first year of operation, the team was able to care for 78% of residents who would otherwise have been transferred to the ED for assessment and care.<sup>626</sup> Similar programs exist in the Central East LHIN<sup>627</sup> and Central West LHIN.

**Reassure families that strategies above have been taken to ensure proper assessment within the home.** Also inform families of potential risks of ED transfers (e.g., hospital-acquired infection, worsening confusion, wandering, falls in an unfamiliar environment).

- Your Healthcare Options (www.health.gov.on.ca/en/public/programs/hco/default.aspx), launched in 2009, provides information about different ways to access healthcare (e.g., walk-in clinics, urgent care centres and family health teams).<sup>628</sup> In addition, public awareness campaigns continue to encourage use of Telehealth Ontario's toll-free services, which advise whether an ED visit is necessary.<sup>629</sup>
- In FY 2010/2011, funding for new community health centres (CHCs) will help provide physician services 24/7, so patients can avoid unnecessary ED visits.<sup>630</sup>
- The \$4.8 million Geriatric Assessment and Intervention Network (GAIN) program is establishing four urgent/emergent clinics for seniors in the Central East LHIN's
  largest community hospitals. Emergency and community-based physicians can refer patients to these clinics, staffed by a highly trained geriatric team and
  supported by a geriatrician, for specialized assessment and intervention. If required, patients are admitted to an in-patient unit specially designed to care for frail
  seniors and/or linked to community support services. The clinics are expected to see nearly 8,000 visits annually after all four open by April 2011.<sup>631</sup>
- Physicians participating in eligible primary healthcare models are offering "After Hours" and "Telephone Health Advisory Service" (THAS) services to enrolled patients to improve access to family health care and divert non-emergency visits from hospital EDs.<sup>632</sup>

#### Avoiding unnecessary drugs and tests 6.4

In healthcare, there are many instances where the choice to administer drugs or tests adds no value and should not be made, or where it's pre-operative testing for cataract surgery<sup>633</sup> and using more expensive alternatives to thiazides to treat high blood pressure.<sup>634</sup>

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
The elimination of tests that do not improve patient safety, including electrocardiograms (ECGs) and chest X-rays before minor procedures (e.g., cataract surgery). <sup>635, 636</sup>	Unnecessary tests waste time and money. <sup>637</sup> In the case of chest X-rays, patients are also exposed to extra radiation.	Ontario's 13 million residents, including the approximate 140,000 people who receive cataract surgery each year. <sup>638</sup>
The use of lower-cost drugs if they are just as effective as newer, more expensive ones — such as thiazides, which are a type of diuretic or "water pill." <sup><math>1639</math></sup>	Using more expensive medications, when cheaper, equally effective ones are available, wastes money. <sup>640</sup>	Ontario's residents who are newly diagnosed with uncomplicated hypertension, including more than 14,000 seniors last year. <sup>641</sup>



Data sources:

\*Registered Persons Database (RPD), Discharge Abstract Database (DAD), Same Day Surgery Database, Ontario Health Insurance Plan (OHIP) Claims Database, FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES).

\*\*RPD, Ontario Diabetes Database, Ontario Drug Benefits Database, DAD, OHIP Claims Database, FY 2009/10, calculated by ICES. \*Uncomplicated high blood pressure is defined as high blood pressure where the patient does not have diabetes, kidney failure, coronary artery disease, stroke, migraine or liver failure, where other drugs for high blood pressure would be more appropriate. Guidelines suggest that in such cases the drug of choice is a thiazide.



#### **Root Cause**

Providers may be unaware of or may not remember all the details of practice guidelines or appropriateness criteria, especially if the criteria are complex with multiple recommendations and exceptions to recommendations.

An outdated practice may persist because of force of habit.

Providers may not know whether or not they are following guidelines for appropriate use, and may not realize there is a problem.

Providers may be sceptical of the validity of practice guidelines or appropriateness criteria.

There is no incentive to stop ordering unnecessary tests or services.

#### Healthcare providers may be influenced by sales representatives from drug

**companies.**<sup>651</sup> Healthcare providers may lack time to critically appraise the literature regarding more expensive alternatives and defer to information provided by sales representatives from drug companies. Often, marketing campaigns promote more expensive drugs over older, less expensive drugs.

Making change is difficult in a complex health care system, where all of the root causes above reinforce each other.

#### **Ideas for improvement**

**Organize information in appropriateness criteria into decision trees or algorithms.** Appropriateness criteria often identify many different clinical scenarios and the tests which are and are not appropriate in each situation.<sup>642, 643</sup> These criteria are often not followed,<sup>644</sup> however, and part of the reason may be that the way they are designed makes it hard to quickly look up the right clinical scenario. A decision-tree format can help the user find the right recommendation with greater ease.

**Embed decision support algorithms into the test ordering process and issue prompts or request confirmation if appropriateness is rated low.** Such algorithms ask the health care provider questions about the patient's symptoms or conditions and match that information against appropriateness criteria. If the test is likely inappropriate, the algorithm may issue a warning, suggest alternatives to the test, or request confirmation that the test is still needed. Such algorithms work best with electronic ordering systems. Massachusetts General Hospital, using such a system, reduced the number of likely unnecessary CT scans by more than half.<sup>645</sup> The Medicare program in the US is currently running broad pilot tests of these systems.<sup>646</sup>

For certain tests that should be ordered not routinely but only in exceptional circumstances, eliminate tick-box options in order forms or standard order sets. When this was done for urea and Erythrocyte Sedimentation Rate (ESR) testing in Ontario in the 1990s, use of these tests decreased 58%.<sup>647</sup>

**Regularly measure rates of appropriate use of tests or other services** to give providers and organizations an idea of how they are performing and to help them identify areas for improvement.<sup>648</sup>

**Opinion leaders can help encourage the adoption of evidence.** Identified by their peers, these "educational influentials" operate within their communities to teach and facilitate change.<sup>649, 650</sup>

**Do not pay for services when they do not meet criteria for appropriateness.** See "What is Ontario Doing?" below for examples. When policies like this are enacted, there are usually mechanisms to allow payment in exceptional circumstances.

**Programs involving trained academic detailers** (often pharmacists or nurses), who visit primary care offices to promote evidence-based drug prescribing practices based on objective appraisal of the literature, have successfully changed prescribing practices.<sup>652, 653, 654, 655</sup>

**Multifaceted interventions are the most likely to change provider behaviour.** For example, a project by the Ontario Guidelines Advisory Committee and the Institute for Clinical Evaluative Sciences reduced the use of pre-operative chest radiography for low-risk surgeries in Ontario hospitals by 13%, and the largest changes were observed in institutions with the highest rates of chest radiography use pre-intervention.<sup>656</sup> This intervention involved workshops to disseminate information, the development of a pre-operative testing checklist, and identification of opinion leaders, who were consulted about barriers to adoption and trained to facilitate change. Hospital-specific feedback on pre-operative testing rates was also provided to all Ontario hospitals.

- On June 3, 2010, the Excellent Care for All Act was passed into law. Health Quality Ontario, under the legislation, will make recommendations on possible changes to the way health care is covered and paid for to ensure that it is consistent with the best clinical evidence.<sup>657</sup>
- Effective July 1, 2010, changes were made to the OHIP fee schedule that stopped payment for certain services judged to be unnecessary, including ECGs and chest X-rays before cataract surgery (unless there is some other reason to do them) and sleep studies that are repeated too soon after a previous study.<sup>658</sup>

#### Overall spending and value for money 7.1

Healthcare is a large and complex system that consumes significant financial resources. Calculating healthcare spending as a percentage of provincial gross domestic product (the total value of all the goods and services Ontario produces) is one way to quantify Ontario's investment. However, a higher or lower number doesn't necessarily mean the province has found the right or best level of healthcare spending. Paying more might be a good decision if Ontarians could be confident they were buying higher-quality healthcare. At the same time, healthcare spending must be balanced against spending in other areas, such as education, social services and infrastructure.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Good value, including high-quality healthcare services, in exchange for their taxpayer dollars.	When Ontarians overpay for healthcare, they have less money to spend on other high-priority areas.	Ontario's 13 million residents.

Indicator	Value*	Time trends & comparisons	Bottom line
Total healthcare spending as a percentage of gross domestic product (GDP) <sup>†</sup> → Ontario → Canada	12.2% 11.7%	Percent Percen	In 2010, Ontario spent 12.2% of its total GDP on health- care, a 38% increase from 2000. Historically, Ontario spent less of its GDP on healthcare compared to Canada as a whole — but now it spends more. The peak seen in 2009 may not be a reflection of increased healthcare spending, but rather the result of a contraction in overall economic activity. Especially during difficult economic times, it is a challenge to sustain spending increases. In 2008, Canada placed fifth in spending when compared to other Organisation for Economic Co-operation and Development (OECD) countries in total healthcare spending." Ontario appears to spend more of its GDP on healthcare than all countries except the United States and France.
Health Expenditure as a Proportion of Total Health Expenditure → Public spending → Private spending	67% 33%	100 100 50 0 1981 2010	Currently, two-thirds of healthcare spending comes from public dollars. This proportion has been stable since 1996. There was a significant shift towards more private spending from 1992 to 1996; prior to that time period, almost three-quarters of health expenditures were publicly funded.
Total healthcare expendi- tures in Ontario — Total spending — Public spending — Private spending	\$75 billion \$50 billion \$24 billion	$\begin{bmatrix} 80\\ 40\\ 0\\ 1975 \end{bmatrix} = \begin{bmatrix} 0\\ 2010 \end{bmatrix}$	In 2010, Ontario's total spending on healthcare was pro- jected to be \$75 billion — a major increase of 91% since 2000. Healthcare expenditures have risen by an average of 9.1% per year since 2000.
Total healthcare expendi- tures in Ontario per capita — Total spending — Public spending — Private spending	\$5,641 \$3,800 \$1,841	6,000 \$\$3,000 0 1975 2010	Ontario's per capita spending on healthcare was pro- jected to be \$5,641 in 2010 — an increase of 69% since 2000. Since 2000, per capita healthcare expenditures have risen by an average of 6.9% per year.

Data sources:

Canadian Institute for Health Information, 2010.<sup>659</sup> Data for calendar year 2009 and 2010 are forecasted. "Organisation for Economic Co-operation and Development, 2008. Note the OECD calculation is different, giving rise to slightly lower numbers.<sup>660</sup>



#### Expenditures versus value for money

Ontario spends more on healthcare than many other industrialized countries — but is the province getting added value for its extra investment? Unfortunately, as the table below indicates. Ontario gets lower scores on several of the selected key quality indicators than eight countries that spend less. When researchers<sup>661</sup> compared Canada's performance to the performance of European countries across a much more detailed list of indicators, they observed that treatment quality in Canada was on par with most European countries, but wait times and patients' rights scored lower. They ranked Canada in last place among 32 nations on the "Bang-for-the-Buck" index.

Indicator	Ontario	UK	Netherlands	Norway	Sweden	Australia	New Zealand	USA	France
Total healthcare expenditure as a percentage of GDP, 2008 <sup>-</sup>	11.2%†	8.7%††	9.9%††	8.5%††	9.4%	8.5%***	9.9%	16%	11%
Total percentage of public health expenditure, 2008	68%	83%	n/a	84%††	82%	n/a	80%	47%	78%
Percentage of family doc- tors using electronic medical records (EMRs), 2009 <sup>**</sup>	43%	96%	99%	97%	94%	95%	97%	46%	68%
Percentage of adults able to see a doctor the same or next day, 2010 <sup></sup>	48%	69%	69%	42%	48%	65%	77%	54%	60%
Percentage of adults who waited four weeks or more to see a specialist, 2010 <sup></sup>	51%	28%	30%	48%	45%	46%	38%	17%	47%
Percentage of adults who rate the overall quality of medical care they received in the past 12 months from their regular doctor's practice or clinic as excellent or very good, 2010 <sup></sup>	77%	79%	54%	59%	43%	76%	84%	74%	67%
Mammography screening rates, 2008 <sup>§</sup>	73%	75%	85%	98%	84%	n/a	n/a	n/a	78%
Life expectancy (age), 2007 <sup>§§</sup>	81	80	80	81	81	82	81††	78	81††
Infant mortality rate per 1,000, 2007 <sup>888</sup>	5.2	4.8	4.1	3.1	2.5	4.2	4.8	n/a	3.8

Data sources:

\*\*Commonwealth Fund International Survey of Physician Practices, 2009. \*\*Commonwealth Fund International Survey of Adults, 2010.

<sup>8</sup> EuroHealth Consumer Index 2008;<sup>663</sup> Ontario rates from Canadian Community Health Survey, 2008.
 <sup>8</sup> World Health Organization;<sup>664</sup> Ontario data from Statistics Canada, Canadian Vital Statistics, Death Database and Demography Division (population estimates), CANSIM table 102-4307, Ontario result is a three-year average of 2005-2007.

556 Organisation for Economic Co-operation and Development, 2010, Ontario numbers from Statistics Canada, CANSIM, table 102-0504 and Catalogue no. 84F0211X.

\*Estimate, adjusting to the Organisation for Economic Co-operation and Development calculation method.

<sup>††</sup> Estimated results. <sup>†††</sup> Data for 2007.

#### 7.2 Information technology

Ontario in 2008 to fulfill its promise of "an electronic health record by 2015." The IT systems of the future should make it possible to easily share information throughout the healthcare system, as well as supporting clinical decision-making. IT in healthcare refers to secure, computerized systems designed to collect, manage and relate healthcare information to healthcare providers. An electronic health record (EHR) is generally defined as a secure and private lifetime record of an individual's health history and care across the healthcare continuum, which is available electronically to authorized healthcare providers. In comparison, electronic medical records (EMRs) are less integrated secure systems, used in one healthcare setting, and are not accessible outside that setting.<sup>665</sup> EMRs often also include higher-level

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Electronic entering and storage of notes and orders.	Illegible handwriting or transcription mistakes may lead to errors, and time may be wasted searching for missing or misfiled information.	Ontario's 13 million residents and healthcare
Electronic error checks and reminders of follow-ups and treatment timing.	Less than optimal drugs may be prescribed, doses may be miscalculated, drug interactions and allergies may not be flagged and providers may forget to schedule follow-up tests and visits.	professionals.
Shared but secure information for providers about medical history and data such as test results.	Teamwork and communication among providers may not be as effective, tests may be unnecessarily repeated, the wrong treatments may be recommended if the provider doesn't have the most current information, and time may be wasted on repeated data entry.	
Access to medical information that helps individuals better manage their own care.	Delayed communication with a primary care provider or patients with limited access to their own medical information may lead to patients being less able to carry out self-management activities, more unnecessary doctor visits and extra time spent scheduling appointments.	



Data sources:

\*\* HIMSS Analytics<sup>™</sup> database, December 2010, provided by MOHLTC. \*\* HIMSS Analytics<sup>™</sup> database, December 2010, provided by the Ontario Hospital Association. \*\* HE lectronic Medical Record Adoption Model<sup>®M</sup> score<sup>570</sup> is proprietary to HIMSS Analytics<sup>™</sup> and includes the following categories: 1 — basic IT in pharmacy, lab, Xray; 2 — data pooled together, doctors can check results on system; 3 — nursing flow sheets documented electronically, system flags errors; 4 — computerized physician order entry, electronic clinical protocols, can send Xray files digitally outside hospital; 5 — advanced tools for drug safety (closed loop medication administration); 6 — doctors enter clinical notes electronically; 7 — paperless hospital.<sup>671</sup>





Data sources: \*\*\*Based on Commonwealth Fund International Survey of Adults, 2010.

<sup>§</sup> 2006–2010 cumulative data as of November 24, 2010, including data from both the Primary Care Program (implemented in 2005) and the New Adopters program (implemented in November 2009), supplied by OntarioMD. April to June 2006 data includes cumulative data from July to September 2005, and the score for each quarter is a rolled-up value of the preceding quarter(s).

7.2 Information technology

# 7.2

## **Root Cause**

**Concerns about the costs of implementing an EMR.**<sup>672</sup> Costs include hardware, software licences, training, maintenance and upgrades.

**Concerns about lost productivity when transitioning to an EMR.**<sup>683, 684</sup> Some physicians report that it takes months or years to fully implement an EMR. Healthcare providers may feel there is no incentive to absorb the cost and hassle.

Lack of comfort with a new system and fear of problems, such as system crashes, data loss or security breaches.<sup>689</sup>

Lack of technical expertise and general computer skills,<sup>692</sup> such as navigating with a mouse and typing proficiency.

Systems are not very user-friendly and may have poor interface design — for example, requiring the user to navigate through too many menus to access information.

Uncertainty about what type of system to use, and concerns that it may become outdated.

Uncertainty over which software will prevail.

Difficulty using the data in EMRs to monitor quality.

#### **Ideas for Improvement**

**Offer incentive programs** to either subsidize healthcare providers' office EMR costs or provide bonuses for delivering higher quality care.<sup>673</sup> In September 2010, Canada Health Infoway pledged \$380 million to speed up implementation of EMRs across Canada.<sup>674</sup> Ontario's EMR Adoption Program provides funding to promote and support EMR use by community-based physicians.<sup>675</sup>

**Develop a business case for an EMR.** For example, at Brigham and Women's Hospital in Boston, the implementation of a computerized order entry system resulted in net savings of \$16.7 million over 10 years.<sup>676</sup> Consider how an EMR can decrease waste — for example, by decreasing test repetition, reducing time spent searching for results or entering information, and reducing the need for clerical personnel.<sup>677, 678</sup>

**Consider the health outcome benefits and cost reductions from better patient care** — for example, reducing health disparities, improving outcomes and providing better continuity of care.<sup>679, 680, 681</sup> Decision support systems can help improve prescribing practices, reduce drug errors and remind providers when screening is due.<sup>682</sup>

**Promote an EMR's capacity to improve productivity.** Once integrated into a practice and its workflow, an EMR can increase productivity by streamlining processes and reducing duplication, such as unnecessarily repeated tests.<sup>685, 686</sup>

**Minimize disruptions** to workflow and productivity when implementing an EMR by establishing appropriate procedures and policies early in the process.<sup>687</sup> OntarioMD's Transition Support Program supports groups through this process.<sup>688</sup>

Standard protocols exist for when and how to back up data, how to prevent system errors and data loss, and how to address security and privacy issues.<sup>690</sup>

**Identify champions or leaders.**<sup>691</sup> Healthcare providers who have experience with EMRs can reassure others that there are standard protocols to protect against computer problems (e.g., back-up systems, firewalls, etc.) or provide tips on how to implement an EMR more smoothly.

**Consider voice recognition software**<sup>693</sup> **or tablet-based systems** to make data entry and navigation more convenient.

**Continue development with input from end users**<sup>694</sup> to improve functionality and usability of the systems for users with varied levels of comfort and expertise with IT.

**Promote common standards** for data exchange at a provincial or national level, so that even if a software company fails, key data could still be transferred to another system. Canada Health Infoway and the Canadian Institute for Health Information established a Standards Collaborative in 2006 to create pan-Canadian standards for EHRs.<sup>695</sup>

Develop standards that specify how EMRs should capture, aggregate and export data related to quality measures.

HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

#### **Root Cause**

Concerns that true benefits will only be realized when other parts of the system are built and an EHR is realized. If other sites, such as lab or X-ray, cannot send data electronically, then staff may have to scan information into each individual EMR, which is inefficient.

#### **Ideas for Improvement**

**Continue investments** in disease registries, lab information systems, electronic prescribing systems and picture archiving and communications systems (PACS) for diagnostic imaging, as many other countries, and provinces such as Alberta, have done.<sup>696, 697</sup> Ensure that providers have high-speed Internet access so they can send and receive files.

## What is Ontario doing?

MOHLTC is focusing on specific EHR-related initiatives, including:

- A diabetes registry that will help patients and healthcare providers control and manage diabetes more effectively to reduce associated complications and costs.
- An online medication management system that will minimize preventable adverse drug reactions.
- A single information system (the Ontario Laboratories Information System) that will allow all laboratory test information to be electronically exchanged among authorized practitioners and lab service providers.

In September 2010, Canada Health Infoway pledged \$380 million to speed up implementation of EMRs.<sup>698</sup>

#### Healthy work environments 7.3

Healthy workplaces make an organizational commitment to safety, emphasizing proper safety training, providing appropriate equipment (e.g., lifts for moving patients) and carrying out regular inspections to maintain high safety standards. A safe, well-run workplace general results in healthier staff, fewer work-related injuries and more satisfied workers — and research shows that satisfied, healthy healthcare workers are more courteous and less likely to make mistakes due to fatigue or stress.<sup>699, 700</sup>

What do Ontarians want?	What if it doesn't happen?	Who benefits most?	
The lowest possible injury rates for healthcare workers.	When staff are injured and can't go to work, those who cover for them have to manage a greater workload and often experience more stress. This, in turn, may boost staff turnover, disrupting continuity of care and increasing recruitment-related costs. It may become more difficult to attract well-qualified workers to the organization. In addition, as Workplace Safety and Insurance Board (WSIB) claims increase, premiums may rise.	Ontario's 709,000 healthcare and social services workers, who represent 9.9% of the province's workforce. <sup>701</sup> Their	
The lowest possible absenteeism and inactivity rates for healthcare workers.	High absenteeism rates are disruptive to proper work scheduling and output, and costly to an organization and the economy as a whole. Absenteeism and inactivity can also be indicative of high worker stress and low job satisfaction. <sup>702</sup>	health matters.	

Indicator	Value	Time trends & comparisons	Bottom line
Lost time and non-lost time injury rates per 100 full-time equivalent workers: LTC homes Hospitals <sup>†</sup> Nursing services (home care and other settings) <sup>††</sup> Treatment clinics <sup>†††</sup> Professional offices and labs <sup>†††††</sup>	8.3* 4.7 4.9 2.9 1.9	10 5 0 2002 0 0 0 0 0 0 0 0 0 0 0 0 0	After several years of little change, lost-time and non- lost-time injury rates dropped in 2009, for healthcare workers in all different settings. From 2008 to 2009, injury rates decreased by 7% in LTC, 5% in hospitals, 11% for nursing services, 14% for treatment clinics and 17% for professional offices and labs. These decreases are comparable to declines seen in injury rates across all industries in Ontario. <sup>703</sup> The reason for the drop in injuries in all workplaces is not clear, but lower claims are often observed during reces- sions. <sup>704</sup> This could occur because workers fearing job loss may avoid claims during recession, or because inexperi- enced workers are often the first to face work reductions and these individuals typically have a higher risk of injury. It will be important to verify whether the reductions in injury claims seen represent true improvements in safety or simply a recession effect.
<ul> <li>Incidence of absences (percent- age of workers who report being absent within the past week)<sup>1</sup></li> <li>Inactivity rate (hours lost as a proportion of the usual weekly hours of all full-time employees)<sup>91</sup></li> </ul>	9.1%**	<sup>9</sup> <sup>9</sup> <sup>9</sup> <sup>9</sup> <sup>9</sup> <sup>9</sup> <sup>9</sup> <sup>9</sup> <sup>9</sup> <sup>9</sup>	Ontario healthcare workers report being absent from work 12 days per year and 9.1% of them had an absence in the past week. This rate is higher than the rate for the overall Ontario workforce (7.8%). <sup>705</sup> These rates for healthcare workers have changed little in the past four years and there is room to improve.
Days lost per worker in year <sup>१११</sup>	12**	<sup>20</sup> 10 0 2005 BETTER 2009	

Data sources:

Workplace Safety and Insurance Board, 2009; this indicator represents the total number of injuries causing time away from work (lost time) or not (non-lost time, which includes cases where the employer provided modified work duties to accommodate a disabled worker) per 100 workers per year. Statistics Canada, special tabulation, unpublished data, Labour Force Survey, 2005 to 2009.

Includes acute care, rehabilitation, psychiatric, paediatric and other specialty hospitals.

<sup>11</sup> Includes acute cate, renalization, psychiatic, pagenatic and other spectral sprate.
<sup>11</sup> Includes agencies that provide nursing, rehabilitation and personal support services (e.g., homemaking) for provincial home care programs, as well as hospitals or other organizations that need short-term staff to fill scheduling gaps.<sup>111</sup> Includes clinics for mental health and addiction, rehabilitation and public health, as well as community health centres.
<sup>111</sup> Includes offices of doctors, dentists, physiotherapists and other healthcare professionals, medical laboratories, radiology suites, and agencies for research, health promotion, worker safety or social service planning. <sup>1</sup>Includes absences of any length (e.g. an hour, day or week).
<sup>111</sup> This takes into account both the incidence and length of absence.<sup>111</sup> Calculated by multiplying the inactivity rate by working days per year (250).



#### **Root Cause**

**Lack of knowledge about safety.** Healthcare workers might not be fully aware of safety hazards in the workplace or ways to avoid them, including infection control standards.

**Lack of "safety culture."**<sup>712</sup> If safety culture is weak, then staff may skip safety procedures when they think they are inconvenient, take too much time, or are not important. Or, a poor culture may lead staff to avoid reporting safety issues due to fear of being punished. Staff then lose the chance to learn from their mistakes.

Unhealthy lifestyles can lead to more illnesses and more absenteeism.

**Excessive workloads may lead to fatigue, stress or burnout,**<sup>716</sup> which can also increase the risk of workplace accidents and absenteeism.

Lack of safety equipment. Staff do not always have access to equipment that could reduce their risk of injury from performing their duties. Environmental hazards innate to healthcare settings (e.g., sharps, infectious diseases) can also cause injury or harm.

**High prevalence of shift work.** Healthcare is a sector where employees frequently work shifts. There is a persistently higher risk of work-related injury for extended duration work shifts, particularly during the night.<sup>722</sup>

#### **Ideas for Improvement**

**Educate staff and supervisors about hazards to their own safety and provide training to reduce their risk.**<sup>706</sup> Educate staff and supervisors to look out for safety hazards (e.g., clutter, poor lighting and slippery areas) and pay closer attention to safe lifting protocols, appropriate use of safety equipment and infection control.

7.3 Healthy work environments

**Conduct risk assessments.** Have staff use standardized checklists to help them identify environmental hazards, repetitive motions that could lead to injury and faulty equipment.<sup>707</sup>

**Prevent abuse towards staff.**<sup>708</sup> Implement standard abuse protocols of zero tolerance. Have security available, provide panic buttons for staff at high risk, use a buddy system and teach conflict de-escalation techniques to staff.<sup>709, 710, 711</sup>

**Provide visible leadership for workplace safety.**<sup>713</sup> Set targets with deadlines for reducing workplace injuries and publicize these widely across the organization. Post frequently updated charts showing progress. CEOs and managers should "walk the shop floor" to talk about safety and listen to concerns.

**Monitor safety statistics,** such as injury rates, at the board level. The Duke Health and Safety Surveillance system was developed to help identify and track health and safety issues, as well as occupational diseases.<sup>714</sup>

**Promote healthy lifestyles within the workplace.**<sup>715</sup> Encourage activities like stretch breaks in meetings, healthy food choices for meeting snacks, vending machines and cafeteria food, pedometer or stair climbing challenges, bike to work campaigns and making bike lockers and/or showers available in the workplace. Offer tobacco cessation programs and universal flu vaccination.

Allow frequent breaks to ease fatigue, and give staff the opportunity to eat and drink to maintain energy.<sup>717</sup>

**Consider how staff can be used more effectively to decrease time pressures on them.** This could involve improving teamwork and work processes, eliminating unnecessary or redundant tasks, and improving role clarity.

Consider increasing staffing levels, but only after all efforts to increase efficiency are exhausted.

**Ensure that adequate safety equipment is available to healthcare providers and staff and that they are trained in this equipment's proper use.** For example, musculoskeletal injuries from heavy lifting (e.g., while moving a patient out of bed) are very common in healthcare.<sup>718</sup> Use mechanical lifts to assist with patient transfers. Needleless IV systems can help reduce needle stick injuries.<sup>719</sup> Consider ergonomic workstations to reduce injuries related to repetitive strain.<sup>720, 721</sup> Make sure all providers and staff have appropriate infection control training and materials (e.g., conveniently located sinks, gloves and masks).

**Implement strategies to mitigate the effects of shift work.**<sup>723</sup> Provide employees with information about sleep hygiene, and encourage strategies that can be used to reduce fatigue, such as napping.<sup>724</sup> Consider altering shifts to make them more compatible with circadian rhythms and provide adequate recovery time between shifts, especially when rotating off night shifts.<sup>725</sup>

- Bill 168, which amended the Occupational Health and Safety Act (OHSA) to prevent violence and harassment in the workplace, came into effect on June 15, 2010.<sup>726</sup> The final phase of the Needle Safety Regulation, which requires all healthcare workplaces to use safety-engineered needles, came into effect on July 1, 2010.<sup>727</sup> And the Excellent Care for All Act, which requires healthcare organizations, starting with hospitals, to develop worker satisfaction surveys, passed on June 3, 2010.<sup>728</sup>
- The Healthy Work Environments Innovation Fund Grant Program dedicated \$2.8 million in FY 2009/10 to funding 18 projects, which developed tools and resources to promote healthy work environments.
- HealthForceOntario partnered with the Ontario Hospital Association in 2010 to launch the Quality Healthcare Workplace Awards, which recognize leadership in creating healthy and productive workplaces in Ontario.<sup>729</sup> HealthForceOntario also launched a social media initiative in the summer of 2010, including a Twitter feed (@HFO\_HWE), to promote healthy work environments.<sup>730</sup>

## 7.4 Health human resources

It's clear that having enough staff is essential to deliver high-quality healthcare, but researchers and planners struggle to define the optimal number, mix and distribution. Variables include the demand for healthcare, model of care selected, worker productivity and number of hours staff are willing to work. As new models are developed and work is structured more efficiently, human resources targets can change. MOHLTC has not yet set any official planning targets in this area, so this report will not comment on whether greater supplies of professionals are positive or negative. It will, however, indicate whether Ontario appears to be moving towards more team-based care in primary care practices — an important factor when it comes to delivering excellent, efficient care.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
The right number and type of healthcare practi- tioners available when Ontarians need them.	Too few healthcare practitioners may lead to increased wait times, more travel to get care and, in some cases, no access to certain services. Providers compensating for the shortfall may experience extra workload and stress.	Ontario's 13 million residents.



Data sources:

\*Entry-level student positions data provided by MOHLTC and MTCU for academic year 2009/10.

"Ontario Physician Human Resources Data Centre, 2009.

"College of Nurses of Ontario, 2010.

<sup>§</sup> College of Physicians and Surgeons of Ontario, Ontario Physician Human Resources Data Centre, 2009, calculated by HQO.

<sup>1</sup>These data reflect only students funded by MOHLTC; MOHLTC has set a target of 200 registered international medical graduates per year, and continues to surpass this target.

#### HOSPITAL

**LONG-TERM CARE** 

**HOME CARE** 

**PRIMARY CARE** 

- MOHLTC has created more than 10,000 nursing positions since 2003. In FY 2009/10, it exceeded its goal of creating 900 nursing positions.<sup>732</sup> Over the past seven years, MOHLTC also invested over \$1 billion in initiatives to enhance education, recruitment and retention of Ontario nurses.<sup>733</sup>
- The Nursing Graduate Guarantee (NGG) program, launched in 2007, gives every Ontario nursing graduate the opportunity to work full-time in the province and 9,600 nurses have participated to date.<sup>734</sup>
- MOHLTC has named six French-language health planning entities to work with LHINs to ensure local health planning reflects Francophone communities' needs.<sup>735</sup>
- Ontario's first class of 21 Physician Assistants (PAs) graduated on November 19, 2010 from McMaster University,<sup>736</sup> and University of Toronto now
  also offers the two year Physician Assistant training program. Working under a medical doctor's supervision, physician assistants are now being
  deployed in family health teams, community health centres, emergency departments and hospitals.<sup>737, 738</sup>

## 8.1

## Discharge/transitions from hospital and primary care

When Ontarians move from one provider or setting to another — for example, from a hospital back to the community — there is a risk that poor communication may lead to errors<sup>739</sup> that adversely affect patient care and health outcomes. To mitigate this risk, healthcare providers must share all key information in a timely fashion, and any necessary follow-up care must be arranged.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
All relevant information provided to patients when they are discharged from a hospital or emergency department (ED) to their homes, including warning signs to watch for, side effects of new drugs and where to go if they have problems.	When patients don't receive or understand the information they need at discharge, they may not get proper follow-up or know what they need to do to take care of themselves. This increases the chance their health may get worse and they may have to return to hospital. <sup>740</sup>	Ontario's 20% of residents who visit an ED each year, <sup>741</sup> and the On- tarians who account for over one million hospital discharges each year. <sup>742</sup>
Follow-up by a physician within 30 days for patients discharged from hospital for mental health services.	After discharge, people with mental health conditions may return to stressful situations at home, or have side effects of their medication, or may stop their medications prematurely. Early follow-up helps identify these problems and address them before they escalate. Without follow-up, these individuals are more likely to end up back at hospital. <sup>743</sup>	The approximately 42,000 people in Ontario who were discharged from hospital for a mental health and ad- dictions condition in FY 2009/10. <sup>744</sup>
Rehabilitation services for most patients after hospitalization for a stroke. <sup>745</sup>	When people recovering from a stroke don't receive rehabilitation services, they may have to live with worse speech or movement disabilities. <sup>746</sup>	Ontario's 90,000 people living with the effects of strokes. <sup>747</sup>



Data sources:

\*NRC-Picker patient satisfaction surveys, FY 2009/10, provided by the Ontario Hospital Association.





Data sources:

Data sources: \*\*Based on Commonwealth Fund International Survey of Adults, 2010. \*\*Discharge Abstract Database (DAD), Ontario Mental Health Reporting System, Ontario Health Insurance Plan (OHIP), Registered Persons Database (RPD), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES). Indicator methodology adapted from the POWER Report.<sup>749</sup> <sup>§</sup>DAD, National Rehab System Database, FY 2009/10, calculated by ICES; includes only patients treated in stroke centres in Ontario.

# 8.1

## Ideas for Improvement

Issue: Patients do not receive or understand discharge instructions.

Healthcare providers may forget to provide all relevant details about discharge care to patients.

**Root Cause** 

Patients may be too stressed with their medical condition and forget discharge instructions when given.

Staff explain instructions at a level of language that patients cannot understand, or patients don't understand English well.<sup>754</sup>

Patients may receive different instructions from different providers.

Patients and family members may feel uncomfortable asking questions to clarify instructions (see section 5.1). **Provide written discharge instructions for all hospital and emergency department patients**.<sup>750, 751</sup> Duplicates should be made, with one copy given to the patient and the other kept in the chart. Use a standard sheet for certain common conditions (e.g., gastroenteritis, head injury), with room to add details unique to the patient. Structure the form to ensure that the key details (e.g., how to take medications, symptoms or side effects of drugs to look out for, or whom to call if things get worse) are always included.

**Use the "teach back" method.**<sup>752, 753</sup> Ask patients to repeat discharge instructions to verify that they understand them. If they don't, clarify errors and try again. Consider requiring staff to record in the chart whether the "teach back" confirmed understanding of written discharge instructions.

Have interpreter services available, with interpreters who are trained in medical terminology<sup>755</sup> for commonly spoken languages in the community. Also have information available in multiple languages that is written in plain language.

Provide patients and families with simplified instructions, using plain language; written material could also use pictures to reinforce instructions.<sup>756</sup>

Consider having **the most responsible physician in the hospital call the patient's family physician** to discuss a discharge plan, especially for complex internal medicine or mental health patients.

**Give patients and families a chance to ask questions.**<sup>757</sup> Clarify misunderstandings and ask if there are any questions or if there are instructions that they will have difficulty following.



## Root Cause Ideas for Improvement

#### Issue: Discharge information is not received by primary care provider, or is incomplete.

Some doctors don't dictate discharge summaries immediately after discharge. This slows the transfer of information to primary care providers, and may also contribute to information about hospital visit being incomplete.

Delays in getting information to the healthcare provider.

**Try database-generated discharge summaries**, where much of the key information is captured using a standard form instead of traditional voice dictation. Studies show that in some instances, electronic discharge summaries are faster than dictating summaries, capture more useful information, and healthcare providers receiving the reports find them easier to read.<sup>758, 759</sup>

Track discharge dictation delays and feed data back to family doctors or healthcare providers.

**Set standards in hospital for discharge summary dictation delays.** For physicians who habitually delay discharge summaries, revoke hospital privileges as a last resort.

**Use fax or secure e-mail instead of regular mail** to send out reports.<sup>760</sup> Ideally, **transmit this information electronically** from hospitals to electronic medical records in primary care offices. Use of a web-based standardized communication system to transfer information on patients' visits to emergency departments to family doctors makes transfer of information more timely and complete.<sup>761</sup>

Set standards and measure timeliness of information transfer between in-hospital provider and primary care provider. One source recommends that diagnoses, medications, results of procedures, pending tests, follow-up arrangements and suggested next steps be communicated to primary care providers on the day of patient discharge, and detailed discharge summary be received by primary care physician within one week of discharge.<sup>762</sup>

#### Issue: Transfer to rehabilitation following stroke.

Not enough spaces in rehabilitation facilities and outpatient rehabilitation options to accommodate stroke patients. **Ensure the right capacity exists for stroke rehabilitation care.** This consists of specialized stroke rehabilitation units as well as outpatient rehabilitation programs.<sup>763</sup>

#### What is Ontario doing?

The Ontario Association of Community Care Access Centres (OACCAC) is in the process of implementing a survey to home care clients that will help track how well care is integrated from discharge to return back to home.

## 9.1 Unhealthy behaviour

The basis of good health is healthier behaviour.<sup>767</sup> That includes avoiding smoking and heavy drinking, maintaining good physical activity, preventing obesity and enjoying a healthy diet that includes a variety of fruits and vegetables. Adopting a healthy lifestyle is essential to avoiding chronic diseases later in life.<sup>768</sup> Improvements in health also require supportive environments, policies and resources that enable people to achieve their full health potential.<sup>769</sup>

What do Ontarians want?	What if it doesn't happen?	Who benefits most?	
No smoking or exposure to second-hand smoke.	Smoking raises the risk of cancer, lung disease, heart attack, stroke, emphysema, pregnancy complications and other conditions. <sup>770</sup> It kills more than 13,000 people in Ontario each year <sup>771</sup> and is the number one cause of premature mortality. Smoking also accounts for at least 500,000 hospital days per year. <sup>772</sup> Annual costs of smoking include \$1.9 billion on healthcare and \$5.8 billion in lost productivity <sup>773</sup> .	Ontario's 13 million residents.	
No heavy alcohol consumption.	Heavy alcohol use raises the risk of cirrhosis of the liver, pancreatitis and chronic gastritis (irritation and bleeding of the stomach), <sup>774</sup> as well as mouth, throat, esophogeal, colorectal and breast cancer. <sup>775</sup> It is also associated with a higher risk of injuries and violent behaviour. <sup>776</sup>		
Everyone at a healthy weight.	Obesity raises the risk of heart disease, stroke, diabetes, several kinds of cancer (including breast, colorectal, esophogeal, pancreatic, endometrial and kidney), <sup>777</sup> arthritis of the knee and many other conditions. <sup>778</sup> Ontario spends nearly \$1.6 billion each year on obesity: \$647 million in direct and \$905 million in indirect costs. <sup>779, 780</sup>		
Everyone more physically active.	Physical inactivity raises the risk of cancer, $^{781}$ obesity, heart disease, diabetes, osteoporosis and sleep disturbances. $^{782}$ Ontario spends \$1.8 billion each year on costs related to physical inactivity. $^{783, 784}$		
Everyone eating enough fruits and vegetables.	Eating fewer than five servings of fruits and vegetables every day raises the risk of heart disease, stroke, obesity, $^{\it 785}$ and stomach, esophogeal, lung and colorectal cancer. $^{\it 786}$		



Data source:

<sup>+</sup> Canadian Community Health Survey, population represents respondents ages 12 and over, 2009, calculated by Institute for Clinical Evaluative Sciences.

<sup>†</sup> Defined as having five or more drinks on one occasion, at least once within the year.



Indicator	<b>Value</b> <sup>*</sup>	Time trends & comparisons	Bottom line
Percentage of the population who are obese <sup>††</sup>	18%	100 100 0 2009 0 0 0 0 0 0 0 0 0 0 0 0 0	In 2009, 18% of Ontarians reported being obese. Because in- formation is self-reported, the true rate is likely higher. Obesity has gotten slightly worse over the past eight years, and there is huge room to improve. Ontario could aspire to match obesity rates of 10% to 11% in Sweden, Norway, Italy, the Netherlands and France. <sup>794</sup> Obesity is higher among those without high school education (23%) and living in rural areas (22%; see sec- tion 10.2). While efforts targeted to these groups could help, the problem is so widespread that approaches directed to the entire population are also needed.
Percentage of the population who are physically inactive	49%	100 100 50 0 2001 0 0 0 0 0 0 0 0 0 0 0 0 0	Half of Ontarians do not get enough physical activity. Because this information is self-reported, the true rate of physical inactivity is likely even higher. There has been no significant change over the past eight years and there is major room for improvement. Ontario is similar to the national average (48%), but much worse than B.C. (40%), and short of the target of 45% it declared in 2005. <sup>795</sup> Rates of physical inactivity are worse among those with lower education (70%) and low in- come (61%; see section 10.2). Efforts targeted towards these groups, as well as to the entire population are needed.
Percentage of the population with inadequate fruit and vegetable intake	56%	100 50 0 2001 50 0 0 2009 BETTER	56% of Ontarians do not consume enough servings of fruits or vegetables every day (i.e. five or more). There has been no major improvement in this indicator since 2003. Ontario's results are similar to the national average but worse than Quebec's (46%). <sup>796</sup> There is room for improvement.

Data source: \*Canadian Community Health Survey, population represents respondents ages 12 and over, 2009, calculated by Institute for Clinical Evaluative Sciences. \*Defined as body mass index of 30 or more.

# 9.1

## **Root Cause**

Harmful substances, particularly tobacco, are highly addictive.

Poor health habits may be a social norm.

#### **Ideas for Improvement**

Encourage the use of **nicotine replacement therapies (NRTs)** such as nicotine gum, sprays, patches or lozenges.<sup>797</sup> Consider increasing their access to all people regardless of drug coverage plans.

**"De-normalize" unhealthy behaviours.**<sup>798</sup> Smoking bans in public places are effective in reducing smoking.<sup>799, 800</sup> Similarly, consider **limiting unhealthy food choices** in school and workplace cafeterias<sup>801</sup> and remove junk food vending machines from schools.<sup>802</sup>

**Encourage healthy work environments.** Employers can offer physical activity programs, workshops, classes and other resources at the workplace.<sup>803</sup> Employers can also serve fruits and vegetables at meetings instead of pastries and desserts.

**Promote patient self-management**, <sup>804, 805, 806</sup> preferably through a counsellor with certified training in these techniques. <sup>807, 808, 809</sup> Patients learn about their condition and are coached into setting their own reasonable goals for improvement (e.g., "I'll start by losing two pounds in the next three weeks") that fit with their lifestyle (e.g., "I'll have green tea instead of a double-double at my bridge game"). Then, they build gradually on each improvement.

**Develop targeted outreach activities** to low socioeconomic groups that help address their underlying challenges in maintaining healthy lifestyles. See section 10.2 for details.

Providers can refer people to smoking cessation programs, smoking hotlines or healthy eating hotlines for more information (see page 101).

Give **"exercise prescriptions"**,<sup>810</sup> which tell patients precisely what frequency, intensity, type and duration of exercise they should follow, given information about their current fitness level.

**Dispel the myth that healthy eating is expensive.** Promote low-cost healthy foods such as eggs, milk, potatoes, whole grain breads and pasta, oats, beans, apples, bananas, broccoli, spinach, watermelon and squash.<sup>811</sup>

Display calorie and sodium content on menus in restaurants and school and workplace cafeterias, to help consumers make informed choices.<sup>812</sup>

**Simplify routines.** Create written instructions in plain language or simple checklists (e.g., a shopping list of healthy foods, walking program) for patients to follow.

**Create health promotion materials** including posters, pamphlets, videos and other educational materials that use simple wording and are available in multiple languages to meet patient needs.<sup>813, 814</sup> Post materials in healthcare settings, community centres, libraries and other public places where vulnerable populations meet.

**Enable healthy environments and active lifestyles** through community planning and development. Ensure neighbourhoods are safe and pedestrian-friendly, and that people from all income levels have access to recreational facilities.<sup>815, 816</sup> Keep physical activity and recreation an integral part of the school curriculum<sup>817</sup> (see section 10.2).

Motivation issues and/or low self-esteem may exist. People may not be motivated to change, or may not feel as if they have the power to do so.

**Poverty and/or low education** contribute to unhealthy behaviours. Individuals struggling with daily survival may not have the time or energy to maintain a healthy lifestyle.

Knowledge and awareness gaps exist. While most people know unhealthy behaviours should be avoided, they may not be aware of how different choices can affect their health, or the specific steps they need to follow.

Unhealthy food choices are aggressively marketed.

Healthy choices may not be accessible.

HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

#### **Root Cause**

Healthcare providers are challenged to find time to discuss health behaviours.<sup>818</sup>

#### **Ideas for Improvement**

Use decision aids, flow sheets and computerized reminder systems<sup>819, 820</sup> to track health promotion activities.

**Involve all members of the healthcare team** (e.g. have nurses and health promotion educators offer lifestyle counselling.<sup>821</sup>

Try the **"Ask, Advise, Assist, Arrange" protocol** that nurses or other practitioners can use at every health encounter to counsel on smoking. This takes only three minutes to do.<sup>822</sup>

**Ensure provider reimbursement systems encourage** behavioural risk counselling and other prevention activities.

- EatRight Ontario (www.ontario.ca/EatRight, 1-877-510-5102) is a free telephone and e-mail service offering information about food and nutrition, meal planning advice, healthy eating tips and recipes.<sup>823</sup> The Canadian Cancer Society's www.smokershelpline.ca (1-877-513-5333) provides a similar service for smoking.<sup>824</sup>
- Promoting Life-skills for Aboriginal Youth (PLAY) is a pilot program developed by Right To Play, in partnership with the province and other organizations, to help Aboriginal youth improve their health, self-esteem and leadership skills through participation in sport and play activities.<sup>825</sup>
- In 2009, Ontario launched an after-school program to provide children in at-risk communities with nutrition and physical activity programs.<sup>826</sup>
- Beginning in spring 2011, participating family health teams (FHTs) will provide over-the-counter NRTs combined with counselling at no charge to
  patients who want to quit smoking.<sup>827</sup>

#### Maternal and infant health 9.2

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
All steps are taken to avoid preventable infant deaths (e.g. avoid sudden infant death syndrome, injuries).	More infant deaths.	Ontario's approximately 138,000 women who gave birth last year and their families. <sup>831</sup>
Babies born with a healthy weight.	Low birth weight raises the risk of death at birth <sup>832</sup> and at all stages of life. <sup>833</sup> It may also result in learning difficulties, <sup>834</sup> high blood pressure, heart disease, <sup>835</sup> diabetes, <sup>836,837</sup> asthma and hearing and vision problems <sup>838</sup> later in life.	
No smoking by pregnant women.	When pregnant women smoke cigarettes there is an increased risk of pregnancy complications and serious adverse fetal outcomes, including low birth weight, stillbirth, spontaneous abortion, decreased fetal growth, premature birth, placental abruption, and sudden infant death syndrome (SIDS). <sup>839, 840</sup>	
At least six months of breastfeeding after birth.	When mothers do not breastfeed for at least six months after their baby's birth, it may result in less bonding between mother and infant, <sup>841</sup> more infections and aller- gies <sup>842</sup> and possibly a greater risk of diabetes later in life. <sup>843</sup> In addition, breastfeed- ing mothers experience less ovarian cancer, breast cancer and osteoporosis. <sup>844</sup>	

Indicator	Value	Time trends & comparisons	Bottom line
Infant mortality rate <sup>†</sup> (per 1,000 infants)	5.2*	10 5 0 2000 2007 BETTER	Infant mortality has been steady in Ontario over the last several years at 5.2 deaths per 1,000. This rate is higher than the rate of 4.0 in B.C. <sup>845</sup> and 2.5 to 2.7 in Japan, Norway, Sweden and Finland. <sup>846</sup> There is room to improve.
Percentage of babies with low birth weight	6.2%**	better	In Ontario, close to one in 17 babies are born with a low birth weight, weighing less than 5.5 pounds. In 2007, Ontario tied with Nunavut for the third- highest rate of low birth weight babies in Canada, behind Alberta (6.5%) and Yukon (6.4%). This rate has increased by one-tenth over the past seven years. This may be due to an increase in maternal age, the use of assisted reproductive technology, and multiple births. <sup>847, 848</sup>
Percentage of women who smoked during their pregnancy	12%***	100 50 0 2000 2007 BETTER	About one in eight women smoke during their preg- nancy. No change has been seen in the past three years. There is substantial variation within the prov- ince (with Central LHIN having the lowest rate at 4%). This suggests that there is room for improvement.
Percentage of mothers breastfeeding: —— Initiation —— Exclusively for six months	88% <sup>§</sup> 23%	100 100 50 0 2005 100 100 100 100 100 100 100	Almost nine in 10 new mothers initiate breastfeeding after birth. However, many women stop breastfeed- ing too soon. Only one in four continue to breastfeed their babies exclusively for six months after birth, as recommended by the World Health Organization. <sup>849</sup> Breastfeeding rates have decreased in the past year, and Ontario lags behind the national average and BC's rate (34%). <sup>850</sup> There is room to improve.

Data sources:

Data sources:
 "Statistics Canada, 2008, CANSIM table 102-0504. Infant mortality rate is the number of deaths of children less than one year of age, per 1,000 live births.
 "Statistics Canada, CANSIM table 102-4005. Low birth weight (less than 2,500 grams) and borderline viable birth weight-adjusted low birth weight (500 to less than 2,500 grams), by sex, Canada, provinces and territories, annual.
 "Niday Perinatal Database, provided by BORN Ontario, FY 2009/10.
 Statistics Canada, 2009, CANSIM table 105-0501. † includes infant deaths occurring at one year of age or younger.

#### 9.2 Maternal and infant health



#### **Root Cause**

Issue: Infant mortality.851

Parents are unaware of, or do not take steps to avoid sudden infant death syndrome (SIDS) or preventable injuries.

## **Ideas for Improvement**

**Provide public education for parents** on use of restraints in equipment such as strollers and high chairs; proper use of car seats; and how to baby-proof the home. For SIDS, tips include sleeping on the infant's back, eliminating second-hand smoke and clutter in cribs and avoiding overdressing.<sup>852</sup> Public education can be provided by public health units, primary care and prenatal classes.

**Clinical aids** such as the Rourke Baby Record<sup>853</sup> remind healthcare providers to ask about and record key information at each visit, including safety issues.

#### See below.

**Ensure good pre-natal care** to reduce complications such as Group B Strep infections or sexually transmitted diseases passed on to infant. Identify and address problems with addictions, poor nutrition or risky health behaviours.

Screen for congenital anomalies in pregnancy. If the pregnancy is continued, the information may help anticipate problems at delivery.

**Consider targeted outreach programs,** with education materials on safety tailored to these groups. Providers can seek out individuals at risk to encourage them to attend pre-natal care visits, or call them if they have missed appointments.

Complications at pregnancy.

Low birth weight.

Congenital anomalies.

**Health inequities.** Infant mortality is higher in aboriginal communities<sup>854, 855, 856</sup> and mothers with low education<sup>857</sup> or income.<sup>858, 859</sup>

#### Issue: Low birth weight.

**More pre-term babies being born,** due to increased use of fertility treatments, particularly by older women, that lead to multiple births.<sup>860</sup>

Poor maternal health, or poorly controlled chronic diseases like diabetes.

Smoking and drug addictions.

**Poor diet,** which is more common in women with low income or education level (see Section 10.2).

Issue: Low rates of breastfeeding.

**People are unaware of recommendations** for exclusive breastfeeding to six months (instead of four, as previously recommended).<sup>864</sup>

New mothers may struggle or become discouraged with breastfeeding.<sup>865</sup>

Lack of public acceptance of breastfeeding in public places and the workplace.

Consider educating the public about the risks of delayed childbirth to help women and families make informed decisions. Finding ways to support women to have their children earlier without adversely affecting their careers could reduce the need for fertility treatment.

**Ensure universal access to primary care services** (see section 2.2). Offer prenatal care programs targeting the specific needs of their populations, as many Community Health Centres have done.<sup>861, 862</sup>

Ask about addictions and offer counselling.

Provide **nutrition outreach programs** that offer dietary education and supplements to disadvantaged pregnant women. The Canada Prenatal Nutrition Program (CPNP) funds such programs.<sup>863</sup>

**Ensure educational materials have been updated** to reflect the new recommendation of six months. Consider embedding **reminders in electronic medical records** to healthcare providers to continue advocating breastfeeding to six months.

**Provide access to lactation consultants, clinics or home visits** to help mothers with proper latching technique and suggest remedies for complications like sore nipples or mastitis.<sup>866, 867</sup> **Spread the word about breastfeeding support groups** (e.g., La Leche League).<sup>868</sup>

Encourage public places (e.g. shopping malls) to provide private spaces for breastfeeding, and employers to offer breaks and quiet places for women to pump breast milk.<sup>869</sup>

- Midwives attended more than 10% of births in Ontario in FY 2009/10, assisting about 14,400 women.<sup>870</sup>
- In partnership with the Breastfeeding Committee for Canada, the Ontario Breastfeeding Committee is implementing the World Health Organization/ UNICEF Baby-friendly Hospital Initiative (BFHI), helping hospitals and community health services achieve the baby-friendly designation.<sup>871</sup> The BFHI includes a course for maternity staff and tools for self-appraisal, monitoring and reassessment.<sup>872</sup>

#### 9.3 **Sexual health**

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
A reduction in the incidence of sexually transmitted infections, including gonorrhea, syphilis and HIV.	Left untreated, gonorrhea can lead to pelvic inflammatory disease (PID) in women, which can lead to chronic pelvic pain and scarring of the fallopian tubes, infertility and increased risk of ectopic pregnancies. <sup>875</sup> Infections can lead to male infertility. <sup>876</sup> Syphilis starts as an open sore and may be followed by rashes, fever, and muscle and joint pain. Later on, syphilis can damage the brain, blood vessels, heart and bones, and eventually lead to death. <sup>877</sup> HIV is a chronic infection impairing the immune system and can lead to infec- tions, <sup>878</sup> cancers, <sup>879</sup> dementia, <sup>880</sup> other major physical impairments, and AIDS. <sup>881</sup>	All Ontarians benefit (including those who are sexually active) from the reduction in incidence as the burden of these infections on the healthcare system is reduced.
A reduction in teen pregnancies.	For teens, pregnancy is associated with an increased risk of anemia, high blood pressure, eclampsia and depression. <sup>882</sup> Teen mothers are more likely to drop out of school, be on social assistance and live in poverty. <sup>883</sup> In addition, teen pregnancies have an increased risk of low birth weight and pre-term birth. For babies, this can lead to a higher risk of death, developmental problems, learning difficulties, hearing and visual impairments and chronic respiratory problems. <sup>884</sup> When they grow up, the children of teen mothers are also at greater risk of becoming teen parents themselves, perpetuating the cycle of teen pregnancy. <sup>885</sup>	Ontario's 409,000 females aged 15 to 19.886

Indicator	Value	Time trends & comparisons	Bottom line
Gonorrhea rate per 100,000 people	27*	40 20 0 2004 BETTER BETTER	In 2009 there were approximately 3,500 confirmed cases of gonorrhea. The majority of these cases were among those aged 15 to 34, although the peak age for men starts later than women. No major change has been seen in the past five years. Improvement is necessary.
Infectious syphilis rate per 100,000 people	6.0*	40 <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>40</sup> <sup>4</sup>	There were nearly 800 confirmed cases of infectious syphilis in 2009. More than 95% of the cases were men, and of those more than half were aged 25 to 44. This rate has increased over the past five years and improvement is necessary.
HIV incidence per 100,000 people	7.8**	10 10 10 10 10 10 10 10 10 10	Slightly more than 1,000 Ontarians were diagnosed with HIV last year. There has been an 18% decrease in HIV incidence over the last five years in Ontario. According to the UN, declines in HIV incidence have also occurred around the world. <sup>887</sup> The reasons for this are not clear; one possibility is that at-risk populations are taking greater precautions (e.g., safe sex). Although these improvements are welcome, there are still huge opportunities to reduce HIV incidence further.

Data sources: \*Public Health Division, MOHLTC, 2009. \*\*Public Health Laboratory Toronto, Ontario Agency for Health Protection and Promotion, 2009.<sup>888</sup>



#### **Root Cause**

Lack of knowledge about sexual health. People may lack information, skills and attitudes necessary to make decisions that promote and maintain their sexual health and prevent unintended pregnancies and sexually transmitted infections.

People, particularly young people, may feel invulnerable and may underestimate the risk<sup>889</sup> of sexually transmitted infections and unwanted pregnancy — the "It'll never happen to me" attitude.

Lack of communication between partners. People may feel uncomfortable, embarrassed or fearful of talking openly about sex with their partners.

The unique sexual health education needs of specific "hard-to-reach" groups are not being met.

Behavioural/psycho-social factors may predispose people to engage in risky sexual behaviours. These include lack of confidence and low self-worth, past abuse, abusive relationships and mental illness.

Lack of access to contraception. If young people cannot afford contraception or do not know where to get it without embarrassment, they may forego contraception.

#### Unknown spread of infection in the

**community.** Infections that remain undetected will be spread further. Infections could also remain undetected if persons diagnosed with infections are reluctant to tell healthcare providers who their sexual partners have been. Without that information, public health officials can't track down partners to have them tested.

#### **Ideas for Improvement**

**Provide access to comprehensive, relevant and accurate sex education** in an age-appropriate, culturally sensitive manner that is respectful of individual sexual diversity, abilities and choices.<sup>890</sup> This education has historically occurred in schools and community settings with positive results.<sup>891</sup> Also consider public awareness campaigns (e.g., radio, television, billboards) and online information, which potentially offers greater anonymity and more current, interactive information than pamphlets (see the Society of Obstetricians and Gynaecologists of Canada's "Sexuality and U" site).<sup>892</sup>

**Prepare parents to talk openly with their children about sexual health.**<sup>893</sup> Specific tips include talking to their children about sex early and often, always knowing where their teen is, knowing their teen's friends and families, encourage group activities over frequent dating, strongly discouraging their teen from dating someone who is more than two years older, knowing what their teen reads, watches or listens to, and be aware of inappropriate sexual messages in popular media.<sup>894</sup>

**Strengthen skills in decision-making and assertive communication.** This involves increasing awareness of the benefits of taking action to promote sexual health and reducing negative outcomes. Involve individuals in the decision-making process so their values, needs and concerns are integrated in the effort to avoid being pressured into unwanted sexual activity.<sup>895</sup>

**Ensure that educational material and programs address the needs of "hard-to-reach" groups** such as new immigrants, First Nations, Inuit and Métis communities, or individuals who have experienced sexual coercion or abuse.<sup>896</sup> Encourage coordination and collaboration among federal, provincial and local agencies providing these services. Address the sexual health education needs of lesbian, gay, bisexual, trans-identified, two-spirited or queer people.<sup>897</sup>

**Provide programs to address low self-esteem and depression,** including psychotherapy, counselling and activities to help individuals connect with their family, school, community activities or volunteer work.

Offer prevention programs for illicit drug use, which is associated with risky sexual behaviour.898

**Promote strategies to prevent dating violence and sexual abuse.** Dating violence can lead to unwanted sexual activity, which can lead to sexually transmitted infections and unplanned pregnancy. Strategies include identifying those at risk for sexual violence (e.g., history of abuse in the family, low self-esteem) and educating teens that abusive behaviour should not be considered the norm.

**Ensure access to contraception,** especially for people who may not have regular access to primary care. Some health clinics provide free contraception (e.g., birth control pills) to those who cannot afford it and/or those who may not feel comfortable accessing it because of cultural reasons.

Ensure young women have regular pelvic exams to screen for sexually transmitted diseases along with Pap tests (see section 9.4 for specific ideas).

**Deliver treatment and follow-up for people with sexually transmitted infections and their sexual partners** to reduce further spread of infection.<sup>899</sup> Strategies to encourage people to name sexual partners include emphasizing the importance of contact tracing, respecting any wishes for anonymity, and listening to and addressing any concerns about possible violence or retribution if reporting of partners takes place.

- MOHLTC funds more than 90 programs and services across the province to deliver HIV/AIDS prevention, education and support to people at highest risk or those affected by HIV/AIDS. Anonymous HIV testing, point-of-care HIV testing and prenatal HIV testing are available in communities across Ontario. Seventeen HIV clinics across the province provide multi-disciplinary care for people living with HIV/AIDS, and two housing programs — Casey House and Fife House — offer supportive housing and hospice care.<sup>900</sup>
- The AIDS & Sexual Health Info Line provides province-wide information, counselling and referral for individuals who are seeking help.<sup>901</sup>
- MOHLTC provides 75% funding to boards of health across Ontario to provide necessary public health programming to prevent and control sexually transmitted infections and promote healthy sexuality for priority populations, cases and contacts. Under the Ontario Public Health Standards, the board is required to provide clinical services to priority populations, contraceptives, pregnancy tests, comprehensive pregnancy and post-abortion counsellings, free harm-reduction supplies and other services related to sexually transmitted infections and blood-borne infections.<sup>902</sup>

#### **Preventive measures** 9.4

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Flu vaccinations for everyone, especially the elderly.	Without protection from flu vaccinations, more Ontarians will be infected with influenza, which can lead to hospitalization and deaths in a proportion of ill people, <sup>903</sup> and Ontario will incur increased healthcare costs.	Ontario's 13 million residents, especially the elderly and people with chronic diseases.
Screening for breast, cervical and colorectal cancers (mam- mography, Pap test and fecal occult blood test, respectively) for everyone eligible.	Without cancer screening, more Ontarians will experience premature death <sup>904, 905</sup> and suffering caused by the treatment of advanced cancers, and Ontario will incur increased healthcare costs and the economic burden of lost productivity.	Ontario's residents who are at risk for breast cancer (women aged 50 to 69), cervical cancer (women to age 69) and colorectal cancer (men and women aged 50 to 74). Within a lifetime, one in nine women get breast cancer <sup>906</sup> and one in 15 men and women get colon cancer. <sup>907</sup>
Screening for osteoporosis for everyone eligible. <sup>908</sup>	Without osteoporosis screening, more Ontarians will experience fractures that may cause disability, pneumonia, death, hospitalization and/or admission to long-term care (LTC) homes, and Ontario will incur increased healthcare costs. <sup>909, 910</sup>	Ontario's women over age 55 and elderly men with certain risk factors.

Indicator	Value	Time trends & comparisons	Bottom line
Percentage of the population aged 65 and over who reported having a flu shot in the year prior to the survey	72%*	90 100 45 0 2001 45 0 2009 BETTER BETTER	In 2009, 72% of the population reported getting a flu shot in the past year. This represents a decrease from 2005, when Ontario recorded its best performance (78%). Ontario is tied with Nova Scotia for having the best results in Canada, <sup>911</sup> and has rates similar to the better performers among OECD countries (UK, Korea and Mexico). <sup>912</sup> Places with the best results are Chile (82%) <sup>913</sup> and one region in Nova Scotia (80%) <sup>914</sup> . There is still room for improvement.
Percentage of Ontario women aged 50 to 69 who had a mam- mogram within a two-year period	67%**	100 100 0 2005 100 0 100 0 100 0 100 0 100	Approximately two-thirds of women aged 50 to 69 had a mammogram in the past two years. This has increased by a tenth over the past seven years. Nearly three-quarters of adult women had a Pap test in the previous three years. There has been minimal change in the last seven years. There is room for improvement in both areas. Ideally, nearly
Percentage of Ontario women aged 20 to 69 who had a Pap test within a three- year period	73%***	100 100 50 0 2001/03 100 BETTER BETTER	all eligible women should have these tests.
Percentage of people aged 50 to 74 who reported having a fecal occult blood test (FOBT) within a two- year period	35%*	100 50 0 2005 100 BETTER BETTER	More than one in three people aged 50 to 74 received screening for colorectal cancer with an FOBT test in 2009. In the last four years, these rates have improved from 21% to 35%. This change is likely due to Ontario's Colon Cancer Check program, which was implemented in 2008. <sup>915</sup> Ontario has seen a steady increase in this indicator since this time and is progressing towards its target of 40% by 2011. <sup>916</sup>

Data sources:

<sup>1</sup> Canadian Community Health Survey, 2009, calculated by Institute for Clinical Evaluative Sciences (ICES). Self-reported rates tend to overestimate actual rates; therefore, the true rates may be lower. <sup>1</sup> Ontario Breast Screening Program, Ontario Cancer Registry (OCR), Ontario Health Insurance Plan (OHIP), Registered Persons Database (RPD), 2008–2009, calculated by ICES, provided by Cancer Care Ontario (CCO).

"Cytobase, OCR, OHIP, RPD, National Ambulatory Care Reporting System Database, Discharge Abstract Database, 2007–2009, calculated by ICES, provided by CCO.
#### 9.4 Preventive measures





Data sources:

<sup>§</sup> OHIP, Statistics Canada Population Files, RPD, FY 2009/10, calculated by ICES.

### **Root Cause**

People are unaware of, or forget when, they need screening.

People are unaware of the importance of prevention activities, or overestimate risks (e.g. flu vaccinations).

**Ethno-cultural barriers.**<sup>923</sup> Different cultures may associate examination of sexual organs with **shame or embarrassment**. Some cultures may have **myths about screening** (e.g. Pap screening is not necessary after childbirth).<sup>924</sup> **Language barriers** may make it difficult to explain the purpose of the test. Some cultures separate men and women outside of family.<sup>925</sup>

Low-income and education is associated with lower rates for prevention activities.

Patients avoid having preventive procedures because they are uncomfortable.

Healthcare providers have competing demands on their time and may be too busy or may forget to do preventive screening measures.

People do not have access to primary care, where many preventive services are given.

### **Ideas for Improvement**

**Develop provincial registries** to remind people about routine prevention activities. Currently, the Ontario Breast Cancer Screening Program<sup>917</sup> and ColonCancerCheck<sup>918</sup> send written reminders to people who are due for screening. Research suggests such reminders are well-received by patients.<sup>919, 920</sup> These programs could be expanded to other preventive activities.

**Launch public awareness campaigns to encourage screening.**<sup>921</sup> Use different media (pamphlets, posters, videos, and advertisements) to deliver the message. Use easy-to-understand language tailored to the target population.<sup>922</sup>

Always give patients the choice of a male or female healthcare provider for doing a preventive health test.

Prepare **culturally sensitive educational materials in different languages.** Specific ideas include using credible voices from the target community to deliver the message; addressing myths directly; and acknowledging feelings of embarrassment while promoting the benefits of the test.

Consider targeted outreach programs. See section 10.1 for details.

Follow **best practices for reducing discomfort of procedures. Explain each step of the process**<sup>926</sup> before and again during the procedure to reduce anxiety. For Pap tests, ensure the speculum is at an appropriate temperature and use padded stirrups or no-stirrup techniques.<sup>927</sup>

**Use electronic medical records (EMRs) to generate reminders** to call the patient in when he or she is due for a screening test.<sup>928</sup> Use EMR data to **provide primary care practices** with **feedback** on how well they are doing on preventive screening and vaccination.<sup>929</sup>

Instead of doctors, have other health professionals do preventive screening.

Improve access to primary care (see section 2.2)

**Provide access to vaccinations outside primary care offices,** such as vaccination clinics in public health units, workplaces, or public places like shopping malls. Give priority to high-risk people.

**Bring screening to patients.** For example, the Ontario Breast Cancer Screening Program has a van that serves small communities in northern Ontario.<sup>930</sup>

### What is Ontario doing?

- Cancer Care Ontario's Ontario Cancer Plan 2011–2015 indicates that by 2015 Ontario will have an integrated cancer screening strategy for breast, cervical and colorectal cancer, supported by a single information management/information technology system.<sup>931</sup>
- MOHLTC offers vaccination information fact sheets in 22 languages other than English and French.<sup>932</sup>

# 9.5 Deaths and harm that could be avoided by prevention

Many deaths and much harm can be avoided through healthier lifestyles,<sup>933</sup> early detection of cancer,<sup>934</sup> improved mental health status<sup>935</sup> and injury prevention activities.<sup>936</sup> By maximizing these opportunities for prevention, Ontario can have a healthier population and reduce healthcare costs.

What do Ontarians want?	What if it doesn't happen?	Who benefits most?
Diseases related to unhealthy habits such as smoking (including lung cancer and heart attacks) minimized.	Unless Ontario addresses preventable diseases and injuries, more Ontarians will experience	Ontario's 13 million residents.
Preventable injuries (including traffic accidents, falls, sports injuries and worker injuries) avoided. <sup>937</sup>	disability, death, lost time from work and hospi- talization, and the province will incur increased healthcare costs.	
Deaths from cancer where early detection is possible (including breast and colorectal cancer) minimized.		
Suicides and intentional self-harm minimized through community awareness, early recognition of warning signs and access to mental health services and social supports.	Suicides and intentional self-harm have a devas- tating impact that extends beyond the individual to the family and community. In addition, one suicide could trigger others. <sup>938</sup>	Ontario's 13 million residents, especially those experiencing depression, <sup>939</sup> schizophrenia <sup>940</sup> and substance abuse. <sup>941</sup>

Long Constants	Malara	The second of the second second	Dettern Pro-
Indicator	Value	Time trends & comparisons	Bottom line
Lung cancer incidence per 100,000 people:	51* 60 44	100 2 50 3 50 4 50 5	In 2007, approximately one in 2,000 Ontarians developed lung cancer. Incidence rates among men have declined dramatically in the last three decades. Among women, there was a gradual increase from 1982 to 1998 but rates have been stable since then. The overall decrease in lung cancer is likely due to declines in smoking (see section 9.1), but there is still major room for improvement as smoking has not yet been eliminated.
Relationship between lung cancer incidence and smoking rates		$u_{ij} \frac{1}{2} \frac{1}{$	There is a strong relationship between smoking and lung cancer. For every 5% increase in the percentage of people who smoke within a LHIN, there is an associated increase in the rate of lung cancer incidence of 10 per 100,000 people within the same LHIN.
<ul> <li>Breast cancer mortality rate per 100,000 females</li> <li>Colorectal can- cer mortality rate per 100,000 people</li> </ul>	22* 19*	40 20 0 1982 40 6 6 6 1982 2007 6 6 1982 2007 1982 1982 1982 1982 1982 1982 1982 1982 1982 1982 1982 1985 1975	In 2007, there were 1,951 deaths due to breast cancer in Ontario. In Ontario, the rate of female breast cancer mortal- ity has decreased over the past two decades, likely due to better treatments. <sup>942, 943</sup> In 2007, there were more than 3,100 deaths due to colorectal cancer in Ontario. In the past 25 years, the rate of colorectal cancer mortality has decreased by 33%. Screening has the potential to improve survival and further reduce mortality rates, so there is still room for improvement.

Data sources: <sup>°</sup> Cancer Care Ontario, 2007.

9.5 Deaths and harm that could be avoided by prevention



Indicator	Value	Time trends & comparisons	Bottom line
Acute myocar infarction (AM incidence per 100,000 peo aged 20 and o	l) ple	1,000 <sup>B</sup> / <sub>2</sub> 500 0 <sub>2002/03</sub> BETTER 0 <sub>2002/10</sub>	Approximately 21,000 Ontarians aged 20 and older were diagnosed with a heart attack last year. The incidence of heart attack has decreased by 23% over the past seven years, and decreases in smoking likely contributed to this trend. There is still room for improvement; lowering Ontario's AMI rate to what has already been achieved in Toronto Cen- tral LHIN (149) would create enormous benefits.
Relationship b AMI and smok		21 poster ungription parallel poster 100,000 population	The relationship between heart attacks and smoking is strong. For every 5% increase in the percentage of people who smoke within a LHIN, there is an associated increase in the rate of the heart attacks of 50 per 100,000 people within the same LHIN.
Rate of intenti self-harm (suid per 100,000 ——— Overall ——— Males ——— Females	cide) people: 8.0*** 12	<sup>15.0</sup> <sup>2000</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>15.0</sup> <sup>1</sup>	In 2007, approximately 1,000 Ontarians committed suicide. Suicide rates are three times higher among men than women. The tracking of suicide is poor, and suicides may be under- reported. While Ontario has the second lowest suicide rate in Canada (the Yukon's rate being 7.1) <sup>944</sup> , there has been no change in the suicide rate in the past seven years. There is room for improvement in reporting and preventing suicide.
Rate of emerg department vi intentional sel per 100,000 aged 12 and o	sits for f-harm people	<sup>250</sup> <sup>B</sup> / <sub>E</sub> 125 <sup>0</sup> / <sub>2002/03</sub> BETTER <sup>0</sup> / <sub>2009/10</sub>	There was just under one emergency department visit for intentional self-harm per 1,000 Ontarians aged 12 and older in FY2009/10. Even though there has been a 24% decrease in this indicator in the last seven years, there is still room to improve.
Rate of injury-re	lated ncy nent er 0 people lated izations	10,000 <sup>a</sup> / <sub>d</sub> 5,000 0 2002/03	In FY 2009/10, there were about nine emergency department visits for injury per 100 Ontarians, and about four hospitalizations for injury per 1,000 Ontarians. There has been a minor decrease in these rates over the past five years. There is still room for improvement.

Data sources:

<sup>1</sup>Discharge Abstract Database (DAD), Registered Persons Database (RPD), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES).
 <sup>11</sup>Statistics Canada, CANSIM table 102-0552, Deaths and mortality rate, by selected grouped causes and sex, Canada, provinces and territories, annual.
 <sup>8</sup>National Ambulatory Care Reporting System Database (NACRS), RPD, FY 2009/10, calculated by ICES.
 <sup>8</sup>DAD, NACRS, RPD, FY 2009/10, calculated by ICES.

# 9.5

For strategies to avoid deaths or injuries related to unhealthy behaviours (e.g., lung cancer and heart attacks), see section 9.1. For strategies to avoid deaths related to early detection (e.g., breast cancer), see section 9.4.

### **Root Cause**

#### Issue: Suicide or intentional self-harm.

People may not seek help because of stigma of mental illness. They fear being ostracized or losing their job if others know they have a mental illness.

**Warning signs go unnoticed.** Family, friends and colleagues may not recognize mental health issues until it is too late.

#### People have difficulty accessing mental

**healthcare.** They may not know where to seek care, wait too long to get into a program, or get sent to wrong programs before finding the right one.

**Unemployment, poverty or lack of housing** may create stresses that lead to suicide.

**Social isolation** can lead to loneliness and suicide. **Weak communities** without a sense of hope can lead to despair.

**"Copycat suicides."** Cluster suicides have been noted in adolescents<sup>956</sup> and First Nations communities.

### **Ideas for Improvement**

**Public anti-stigma campaigns** can inform the public that mental health issues are far more common than they realize and can counter common stereotypes (e.g., that mental health sufferers are dangerous or irresponsible<sup>945</sup>). One grassroots campaign in the US lobbies local broadcasters and advertisers to stop using these negative stereotypes.<sup>946</sup>

**Develop screening tools to identify risk factors, warning signs and at-risk behaviours.** High risk individuals can then be referred to counselling or other services. Some school-based programs have trained staff to identify students at risk.<sup>947</sup> Similar tools exist for the military.<sup>948</sup> Consider other tools for parents, family members, friends, co-workers, and healthcare providers.<sup>949</sup>

#### Ensure access to a primary care provider (see section 2.2.)

#### Publicize suicide hotlines and crisis services.

Consider **centralized intake programs** or a "one-stop-shop" referral system for all mental health and addictions services within a region. Such programs help match the individual with the most appropriate service (e.g., psychotherapy, counselling, group therapy, peer support groups or substance abuse programs) and eliminate the need for individuals to make multiple phone calls to different programs. This approach successfully reduced wait times for services in Thunder Bay.<sup>950</sup>

**Ensure there are programs designed for vulnerable populations** and the capacity for these programs matches the demand. People at increased risk of suicide include youth,<sup>951</sup> the elderly, victims of abuse, prison inmates, sexual minorities and those who have previously attempted suicide.<sup>952</sup> Aboriginal communities are at particularly high risk.<sup>953</sup>

**Ensure equitable access to social services and case managers.** These services can help the individual find safe, affordable housing, assistance with employment or training, or social assistance.

**Build strong and supportive communities.** There is growing evidence that "community engagement" may help reduce suicides.<sup>954</sup> Examples in Inuit communities include establishment of anti-suicide youth associations, community suicide prevention walks, and youths kayaking to peers in other communities to deliver a "Live Life" message.<sup>955</sup>

Encourage responsible media coverage; avoid sensationalizing suicides in the media to prevent copycat incidents.<sup>957</sup>

**Ensure community intervention after a suicide occurs.** Such interventions may include counselling for community members or use of traditional healers in First Nations communities.<sup>958</sup>

# HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE

#### **Ideas for Improvement Root Cause** Issue: Injuries. Sports injuries. Support sports injury prevention programs. For example, Think First is a national non-profit organization that works with public health units and community sports and recreation organizations to help children and youth understand how to prevent common and serious injuries associated with hockey, cycling, soccer, diving, skiing and snowboarding.959 Consider mandating use of safety equipment. Falls among the elderly. See section 4.5 and 4.6 for strategies to reduce the risk of falls in LTC homes, home care and the community. People injured on the job. See section 7.3 for change ideas related to healthy work environments. Traffic-related injuries. **Promote safe driving** through awareness campaigns and public policy. Enforce or strengthen existing traffic safety laws that target drinking and driving<sup>960</sup>, using cell phones while driving and improper use of car seats for infants and children. Better municipal planning can decrease traffic-related injuries, by building roads with traffic-calming features in areas with heavier pedestrian traffic.961 Help healthcare providers identify people at risk of intimate partner violence. While there is Assaults. debate among experts on how frequently this should be done,<sup>962</sup> healthcare workers may ask brief questions during routine physicals and consider using certain assessment tools for people at higher risk.<sup>963, 964, 965, 966</sup> Ensure healthcare providers are aware of and refer individuals to community supports when violence has been detected. Ensure safety for workers at risk of assault (e.g. cab drivers, delivery persons, healthcare workers). This includes safety training, access to panic buttons, or special equipment (e.g. physical barriers in cabs).

Accidental poisonings.

Promote child safety during primary care visits and during vaccination of infants (see section 9.2).967

### What is Ontario doing?

• Cancer Care Ontario's Ontario Cancer Plan 2011–2015 aims, by 2015, to develop an integrated cancer screening strategy for breast, cervical and colorectal cancer, supported by a single information management/information technology system, and to provide primary care providers with reports, tools, mentorship and supports to enhance their screening performance.<sup>968</sup>

### **10.1** Primary care — access and effectiveness

People should be able to access the healthcare services they need without any difference based on their income, education level, age, sex, location, or whether they are immigrants or born in Canada. This section examines whether healthcare in Ontario is equitable, with a focus on access to primary care, appropriate chronic disease monitoring, unhealthy behaviour, preventive measures, and diseases that could be avoided with a population health focus.

Who benefits most?

Ontario's 13 million

residents.

#### What do Ontarians want? What if it doesn't happen?

No barriers accessing high-quality care because of income, education level, age, sex, urban or rural residence, or whether someone is an immigrant or born in Canada. All Ontarians living healthy lifestyles, regardless of who they are. Health can deteriorate when people who are disadvantaged in society do not get the services they need or engage in unhealthy behaviours. A vicious circle may start as declining health leads to lower income or under-employment and these people may become more disadvantaged and reliant on social assistance. This is bad for the individual and also for family members and dependents. Employers may be affected, too, as worsening workforce health leads to more sick time and/or staff turnover.<sup>969</sup>



Legend: Low = 1st income quintile; Med = 3rd income quintile; High = 5th income quintile. HS = high school graduate; <HS = less than high school graduate; PSE = at least some post-secondary education. F = female; M = male. Urb = urban; Rur = rural. Can Born = Canadian-born. 0-9 = recent immigrant in Canada for nine years or less; 10+ = immigrant in Canada for 10 years or more. Low Imm = low immigrant population area; High Imm = high immigrant population area. T for this indicator, income is not measured directly, but inferred from the average income in one's immediate neighbourhood corresponding to the postal code. A high immigrant population area was defined as one where the neighbourhood, or census dissemination area (DA) surrounding one's postal code, showed the immigrant population at more than 50% of residents, as reported to Census Canada.

Data sources: \*Primary Care Access Survey, FY 2009/10, provided by Institute for Clinical Evaluative Sciences (ICES).

\*\* Ontario Diabetes Database, Ontario Health Insurance Plan, Registered Persons Database, FY 2009/10, calculated by ICES.

# **10.2** Unhealthy behaviour

HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE



Legend: Low = 1st income quintile; Med = 3rd income quintile; High = 5th income quintile. HS = high school graduate; <HS = less than high school graduate; PSE = at least some post-secondary education. F = female; M = male. Urb = urban; Rur = rural. Can Born = Canadian-born. 0-9 = recent immigrant in Canada for nine years or less; 10+ = immigrant in Canada for 10 years or more. Data source: \*Canadian Community Health Survey, 2009, calculated by Institute for Clinical Evaluative Sciences.

### **10.3** Preventive measures



<sup>§</sup>Legend: Low = 1st income quintile; Med = 3rd income quintile; High = 5th income quintile. HS = high school graduate; <HS = less than high school graduate; PSE = at least some post-secondary education. F = female; M = male. Urb = urban; Rur = rural. Can Born = Canadian-born. 0-9 = recent immigrant in Canada for nine years or less; 10+ = immigrant in Canada for 10 years or more.

<sup>1</sup>For these indicators, income is not measured directly, but inferred from the average income in one's immediate neighbourhood corresponding to the postal code. A high immigrant population area was defined as one where the neighbourhood, or census dissemination area (DA) surrounding one's postal code, showed the immigrant population at more than 50% of residents, as reported to Census Canada.

<sup>++</sup>Data suppressed for this category because of small sample size.

Data sources:

\*Ontario Breast Screening Program, Ontario Cancer Registry (OCR), Ontario Health Insurance Plan (OHIP), Registered Persons Database (RPD), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES).

\*\* OCR, OHIP, RPD, National Ambulatory Care Reporting System Database, Discharge Abstract Database, FY 2009/10, calculated by ICES.

\*\*\*Canadian Community Health Survey, 2009, calculated by ICES. Self-reported rates tend to overestimate actual rates; therefore, the true rates may be lower.

# **10.4** Diseases that could be avoided with a population health focus

# HOSPITAL LONG-TERM CARE HOME CARE PRIMARY CARE



Legend: Low = 1st income quintile; Med = 3rd income quintile; High = 5th income quintile. F = female; M = male. Urb = urban; Rur = rural. Low Imm = low immigrant population area; High Imm = high immigrant population area.

<sup>1</sup>For these indicators, income is not measured directly, but inferred from the average income in one's immediate neighbourhood corresponding to the postal code. A high immigrant population area was defined as one where the neighbourhood, or census dissemination area (DA) surrounding one's postal code, showed the immigrant population at more than 50% of residents, as reported to Census Canada.

Data sources:

\*Discharge Abstract Database, National Ambulatory Care Reporting System Database (NACRS), Registered Persons Database (RPD), FY 2009/10, calculated by Institute for Clinical Evaluative Sciences (ICES).

\*\*NACRS, RPD, FY 2009/10, calculated by ICES. Because of data limitations, it was not possible to measure education differences at the same time as income for this set of indicators.

# 10.4

### **Root Cause**

**Low-income persons face cost barriers**. They cannot afford gym memberships or sports programs, have difficulty paying out-of-pocket costs (e.g. transportation), and may believe that healthy eating is too expensive.

#### Day-to-day survival is the main concern.

Peoples' health concerns may be outweighed by priorities such as finding food, shelter, paying bills or escaping abusive situations.

Low-income persons may not access primary care and health promotion programs, because they are unaware of them, or are distracted by day-to-day survival.

**Poverty generates chronic stress**,<sup>977</sup> and people may cope through unhealthy but pleasurable activities like smoking.<sup>978</sup>

Low education or literacy level may lead to low level of knowledge about diet or healthy living.

People may lack skills, motivation or confidence to change health habits.

Disadvantaged persons may live in unhealthy neighbourhoods where people do not feel safe walking about to get exercise, or cannot get to supermarkets easily.

There is a local culture of unhealthy habits. Smoking or dangerous activities may be considered "normal"<sup>989</sup> within a community.

### **Ideas for Improvement**

Promote low-cost healthy foods (see section 9.1 for list).

Consider offering transportation and daycare for health promotion programs, or bring the program to the individual.

Encourage access to low- or no-cost sports and recreation programs<sup>970</sup> and keep physical education in the school curriculum.<sup>971</sup>

**Promote low-cost access to nutritious foods**, such as the Good Food Box, a volunteer-run food distribution system operating in communities throughout the province.<sup>972</sup>

Consider offering free nicotine replacement products to low-income persons.

**Ensure timely access to social workers and case management**. Arrange supportive housing, employment counselling, skills development, or shelter for those suffering from abuse.

**Design outreach programs to make participation easy**. Bring primary care centres and health promotion activities deep into the communities being served — at community centres, malls, and wherever else people naturally congregate.<sup>973</sup> Develop models of care which **house primary care and social services in the same place**,<sup>974</sup> as many community health centres have done.<sup>975, 976</sup>

Consider offering **stress management therapies for low-income persons**. While evidence in the scientific literature is limited, methods that have been tried include meditation,<sup>979</sup> exercise relation and creative activities.<sup>980</sup>

**Simplify and tailor learning materials**. Ensure material uses graphics for those with low literacy, simple English or local slang, or the languages spoken by targeted communities. Keep instructions simple and step-by-step.<sup>981</sup>

Healthcare providers can provide or refer for **counselling on healthy diet or lifestyle**. **Group sessions** allow people with similar life challenges to support each other in their learning.

Consider distributing **healthy, low-cost shopping lists** and offering **supermarket tours**,<sup>982</sup> where individuals can learn where to find low cost, healthy foods, identify unhealthy foods, search for bargains, and read food labels. Consider **cooking classes**<sup>983</sup> **for underprivileged persons** on healthy cooking.

Promote **patient self-management**,<sup>984, 985, 986, 987</sup> where patients learn about their condition and are coached into setting their own reasonable goals for improvement that fit with their lifestyle (see section 9.1).

**Make neighbourhoods safe**. Increase foot patrols of police or security personnel, or organize group walking or physical activities. Work with municipal officials to ensure safe, well-lit paths for walking.

**Design communities for healthy living**, by zoning that encourages walking to shops and supermarkets. Such neighbourhoods tend to have better health outcomes.<sup>988</sup>

**Organize community-wide healthy living events** (e.g. community walks or sporting events). Identify **healthy role models** for disadvantaged children. **Identify "positive deviants"** — individuals or groups who have good health who live within unhealthy communities; find out their secret to success and spread it widely.<sup>990</sup>

10.4 Diseases that could be avoided with a population health focus



### **Root Cause**

Women in certain cultures may be reluctant to undergo mammography and Pap tests (see section 9.4).

**Some rural activities are dangerous**. Farming<sup>991</sup> and rural recreational activities like all-terrain vehicles (ATVs)<sup>992</sup> have high injury rates.

### **Ideas for Improvement**

**Develop culturally sensitive learning materials** in the language of the target audience, which address myths about screening, emphasize the importance of the test, and use credible spokespersons from the culture (see section 9.4).

**Identify common farming accidents** (tractor rollovers, gas poisoning, and power takeoff entanglements) and promote specific measures to prevent them.<sup>993</sup> **Encourage use of safety gear** (e.g. helmets) and proper training when using recreational vehicles like ATVs.

### What is Ontario doing?

MOHLTC has developed a Health Equity Impact Assessment Tool (HEIA) to advance health equity integration in healthcare policy, planning
and decision-making.<sup>994</sup> HEIA maps the unintended *potential* health impacts of a policy, program or project on specific population groups
(e.g., Aboriginal, francophone, gender, income, race, and geography) and prompts the assessor to adjust the initiative to mitigate negative
impacts and maximize positive impacts on impacted populations. MOHLTC began implementation in 2009 and HEIA pilots have been conducted
within MOHLTC and in three of the 14 LHINs.

# 11 LHIN analyses

In this chapter, we present data on differences between each LHIN and the provincial average, for selected indicators where data were available. In the first set of tables, we present data for each LHIN, identifying where its performance is better or worse than average. We also include data over time for select indicators\*. In the table at the end of this chapter, we present more detailed results for each indicator and each individual LHIN. Yellow shading shows that the LHIN was significantly better than the provincial average, while blue shading shows the LHIN is worse than average.

Differences were considered significant if they were both statistically significant<sup>\*\*</sup> and clinically significant. We used the following guidelines to define significant differences:

Type of indicator	Guidelines for a clinically significant difference between a LHIN and the provincial average
Wait times	Relative difference of 25%
Rate of a serious adverse outcome	Relative difference of 25%
Percentage adoption of a best practice (process measure, often with a target of 100%)	Absolute difference of 5%***
Patient experience variable (e.g., percentage satisfied with x)	Absolute difference of 5%

Abbreviations used in this chapter are as follows:

- ALC = alternate level of care (in this case, a hospital bed occupied by someone who could be better served in a different setting, such as a long-term care home)
- AMI = acute myocardial infarction (heart attack)
- CHF = congestive heart failure
- COPD = chronic obstructive pulmonary disease (emphysema or chronic bronchitis)
- ED = emergency department
- FOBT = fecal occult blood test
- LHIN = local health integration network
- LTC = long-term care home

The following tables show some areas in which LHINs can improve. Please refer to the table at the very end of this chapter for more details on selected indicators where data was available.

\*\*\* In some instances where the rate is high (e.g. 96%) but the target is clearly 100%, the difference is treated as defect (e.g. 4%) and a relative difference of 25% is considered significant (in this example, a difference greater than plus or minus 1% would be considered significant).

 $<sup>^{\</sup>ast}$  Only changes in LTC waits of greater than 1 week and changes of ALC >1% are commented on.

<sup>\*\*</sup> For some indicators where data were obtained from other parties, confidence intervals were not available, but statistical significance was inferred based on estimates of the sample size and assumptions about the probability distribution of the variable. See the technical appendix to this document at www.hqontario.ca for more details.



ERIE ST. CLAIR LHIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	<ul> <li>Shorter waits for a LTC bed, especially for people placed from the community</li> <li>Fewer ALC hospital bed days</li> <li>Shorter wait times for CT/MRI scans, cancer surgery, cataract surgery and hip and knee replacements (general surgery waits similar to rest of province)</li> <li>More timely home care visits after hospital discharge</li> </ul>
Average results, still room to improve	<ul> <li>ED wait times (but relatively fast transfer to a bed after admission)</li> <li>Access to primary care</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>Avoidable hospitalizations (readmissions higher for some conditions, lower for others)</li> <li>LTC safety and effectiveness (most indicators at provincial average)</li> <li>Home care safety and effectiveness</li> <li>Avoidable ED visits by LTC residents</li> <li>Preventive health screening</li> </ul>
Worse than average results, major room to improve	<ul> <li>Population health - smoking, obesity, physical activity, binge drinking are at provincial average but rates of fruit and vegetable intake and smoking in pregnancy are worse, and lowest rate in Ontario for breastfeeding</li> </ul>

Note: Erie-St. Clair has a relatively low supply of family doctors and specialists.

#### CHANGES OVER TIME:

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	No major change
	Placement from hospital	Worse by 18 days
	Placement from home	Better by 12 days
ALC (from 2008/09 to 2009/10)		No major change

SOUTHWEST LHIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	<ul> <li>Shorter ED wait times</li> <li>Shorter waits for a LTC bed, especially for people placed from the community</li> <li>Fewer ALC hospital bed days</li> <li>Shorter wait times for CT scans, MRI</li> <li>Lower C-section rates</li> </ul>
Average results, still room to improve	<ul> <li>Access to primary care</li> <li>Wait times for most surgeries (except cancer)</li> <li>Avoidable hospitalizations</li> <li>Avoidable ED visits by LTC residents</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>LTC safety and effectiveness (most indicators at provincial average)</li> <li>Home care safety and effectiveness</li> <li>Preventive health screening</li> </ul>
Worse than average results, major room to improve	<ul> <li>Wait times for cancer surgery</li> <li>Population health – smoking and physical inactivity rates are average but worse results for obesity, fruit and vegetable intake, smoking in pregnancy; higher rates of injury</li> </ul>

Note: Southwest LHIN has a relatively high supply of nurses, nurse practitioners and registered practical nurses.

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	No major change
	Placement from hospital	No major change
	Placement from home	No major change
ALC (from 2008/09 to 2009/10)		No major change

WATERLOO-WELLINGTON LHIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	<ul> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>Shorter waits for cancer and general surgery, cataract surgeries and hip replacements</li> <li>Avoidable ED visits by LTC residents</li> </ul>
Average results, still room to improve	<ul> <li>ED waits – a mixed picture, with longer waits to see a physician but shorter waits for those who are admitted</li> <li>ALC hospital bed days</li> <li>Access to primary care</li> <li>Waits for CT/MRI, cancer surgery, knee replacements</li> <li>Home care waits</li> <li>Avoidable hospitalizations (readmissions higher for some conditions, lower for others)</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>C-section rates</li> <li>LTC safety and effectiveness (most indicators at provincial average)</li> <li>Home care safety and effectiveness</li> <li>Population health (smoking, obesity, physical inactivity, diet, binge drinking similar to provincial rates)</li> <li>Preventive health screening</li> </ul>
Worse than average results, major room to improve	Longer waits for a LTC bed, especially for people placed from the community
Note: Waterloo Wallington has a relatively low supply of specialists	

Note: Waterloo-Wellington has a relatively low supply of specialists.

#### CHANGES OVER TIME:

LTC waits (from Apr-Jun 09 to	Overall	Worse by 26 days
2009/10)	Placement from hospital	Worse by 22 days
	Placement from home	Worse by 20 days
ALC (from 2008/09 to 2009/10)		Improved by 2%. (Note that the paradoxical improvement in ALC with worsening LTC waits may be due to different time periods used for comparisons over time).

HAMILTON-NIAGARA-HALDIMAND-BRANT LHIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	Access to primary care
Average results, still room to improve	<ul> <li>Home care waits</li> <li>Avoidable hospitalizations (most indicators at provincial average)</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>LTC safety and effectiveness</li> <li>Home care safety and effectiveness</li> <li>Avoidable ED visits by LTC residents</li> <li>Population health (most indicators at provincial average)</li> <li>Preventive health screening (but rates of fecal occult blood testing are lower)</li> </ul>
Worse than average results, major room to improve	<ul> <li>Longer ED wait times, especially for admitted patients</li> <li>More ALC hospital bed days</li> <li>Longer waits for a LTC bed, especially for people placed from hospital</li> <li>Longer waits for hip and knee replacement and CT/MRI (waits are average for other surgeries, like cancer, general surgery, cataracts)</li> </ul>

LTC waits (from Apr-Jun 09 to	Overall	Worse by 34 days
2009/10)	Placement from hospital	Worse by 28 days
	Placement from home	Worse by 46 days
ALC (from 2008/09 to 2009/10)		Improved by 3%. (Note that the paradoxical improvement in ALC with worsening LTC waits may be due to different time periods used for comparisons over time).

#### 11. LHIN Analyses

CENTRAL WEST LHIN	
Superior results, no room to improve	Rate of statin use after an AMI surpasses 90% target
Better than average results, still room to improve	<ul> <li>Shortest wait for a LTC bed in Ontario (however, lowest rate of people given their first choice of LTC home)</li> <li>Fewer ALC hospital bed days</li> <li>Shorter wait times for, general surgery, hip replacement, CT scans</li> <li>Population health (lower rates of smoking, smoking in pregnancy, binge drinking, inadequate fruit &amp; vegetable intake; lower rates of injury and self-harm; physical inactivity, however, is worse than average)</li> </ul>
Average results, still room to improve	<ul> <li>Access to primary care</li> <li>Waits for MRI, cancer surgery, cataract surgery, knee replacement</li> <li>Home care waits</li> <li>Avoidable hospitalizations (readmissions higher for some conditions, lower for others)</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>LTC safety and effectiveness</li> <li>Avoidable ED visits by LTC residents</li> </ul>
Worse than average results, major room to improve	<ul> <li>Longer ED wait times</li> <li>Higher C-section rate for uncomplicated deliveries</li> <li>Drug safety in LTC (high rates of new starts of antipsychotics, anti-anxiety drugs)</li> <li>Home care safety and effectiveness</li> <li>Preventive health screening – worse results for flu shots, mammography and fecal occult blood testing</li> </ul>

Note: Central West has a relatively low supply of family doctors, specialists and nurses.

#### CHANGES OVER TIME:

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	No major change
	Placement from hospital	No major change
	Placement from home	Better by 27 days
ALC (from 2008/09 to 2009/10)		No major change

MISSISSAUGA-HALTON LHIN	
Superior results, no room to improve	None
Better than average results, still room to improve	<ul> <li>Fewer ALC hospital bed days</li> <li>Shorter waits for a LTC bed, for people placed from hospital</li> <li>Avoidable hospitalizations – low rates of admission for ambulatory care sensitive conditions and lower readmission rates for AMI, diabetes, GI bleed</li> <li>Shorter wait times for hip and knee replacements</li> </ul>
Average results, still room to improve	<ul> <li>ED wait times</li> <li>Access to primary care</li> <li>Waits for surgery and CT/MRI – mostly similar to provincial average, but waits are lower for hip and knee replacement</li> <li>Home care waits</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>LTC safety and effectiveness (most indicators at average)</li> <li>Home care safety and effectiveness (most indicators at average)</li> <li>Avoidable ED visits by LTC residents</li> <li>Preventive health screening (but worse results on flu vaccination)</li> </ul>
Worse than average results, major room to improve	<ul> <li>Population health (most rates at average but lower rate of smoking during pregnancy, breastfeeding right after birth, inadequate fruit and vegetable intake, and intentional self-harm)</li> </ul>

Mississauga-Halton has a relatively low supply of specialists and nurses.

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	No major change
	Placement from hospital	No major change
	Placement from home	Better by 25 days
ALC (from 2008/09 to 2009/10)		Improved by 4%

TORONTO CENTRAL	
Superior results, no room to improve	• None
still room to improve	<ul> <li>Lower rate of ALC hospital bed days</li> <li>LTC safety and effectiveness – average for most indicators but better results on restraint use, worsening depression/ anxiety, and lower rate of new prescriptions for antipsychotics</li> <li>Population health (lowest rate of smoking; lower obesity; higher rates of breastfeeding)</li> <li>Shorter wait times for hip and knee replacements</li> </ul>
to improve	<ul> <li>Wait time for a LTC bed</li> <li>Access to primary care</li> <li>Waits for CT, MRI</li> <li>Waits for general surgery, cataract surgery</li> <li>Home care waits</li> <li>Avoidable hospitalizations – rate of hospitalizations for ambulatory care sensitive conditions similar to provincial average; rates worse than average for COPD, stroke, diabetes readmissions</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>Avoidable ED visits by LTC residents</li> <li>Preventive health screening – most rates at average but mammography screening, fecal occult blood test lower than average</li> </ul>
major room to improve	<ul> <li>Longer ED wait times</li> <li>Longer waits for cancer surgery</li> <li>Home care safety and effectiveness – average for most indicators but worse results for worsening activities of daily living, depression and worsening cognitive function</li> <li>Highest incidence of HIV infection</li> </ul>
Note: Toronto Central has a relatively high supply of	family doctors and nurses, and the highest concentration of specialists in Ontario.

CHANGES		T18.6.5
	UVER	

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	Worse by 20 days
	Placement from hospital	Worse by 10 days
	Placement from home	Worse by 48 days
ALC (from 2008/09 to 2009/10)		No major change

#### 11. LHIN Analyses

CENTRAL LHIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	<ul> <li>Avoidable hospitalizations – lowest rate of admissions for ambulatory care sensitive conditions in Ontario</li> <li>LTC safety and effectiveness (lower restraint use; lower rate of worsening of activities of daily living, pain, or depression / anxiety)</li> <li>Shorter wait times for surgery, including cataract, cancer, hip and knee replacements and general surgeries</li> <li>Population health – lower rates for smoking in pregnancy, obesity, binge drinking; lower rates for self-harm and injuries</li> </ul>
Average results, still room to improve	<ul> <li>ALC hospital bed days</li> <li>Waits for a LTC bed</li> <li>Wait times for CT/MRI</li> <li>Access to primary care</li> <li>Home care waits</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>Home care safety and effectiveness (better than average results for weight loss; worse results for depression)</li> <li>Avoidable ED visits by LTC residents</li> <li>Preventive health screening (average for most indicators but higher rates of Pap screening)</li> </ul>
Worse than average results, major room to improve	Longer ED waits, especially for admitted patients

Note: The Central LHIN has a relatively low supply of specialists and nurses.

#### CHANGES OVER TIME:

LTC waits (from Apr-Jun 09	Overall	Better by 8 days
to 2009/10)	Placement from hospital	Better by 8 days
	Placement from home	Better by 20 days
ALC (from 2008/09 to 2009/10)		Worsened by 2%. (Note that the paradoxical increase in ALC with improvement in LTC waits may be due to different time periods used for comparisons over time).

CENTRAL EAST LHIN		
Superior results, no room to improve	• None	
Better than average results, still room to improve	Shorter wait times for general surgeries	
Average results, still room to improve	<ul> <li>ALC hospital bed days</li> <li>Wait time for a LTC bed is similar to provincial average, but waits are longer than average for people being placed from hospital</li> <li>Access to primary care</li> <li>Waits for CT/MRI and surgery (with shorter waits for general surgery, longer waits for cataract surgery)</li> <li>Home care waits</li> <li>Avoidable hospitalizations (most indicators at provincial average; lower readmission rate for stroke and diabetes)</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>LTC safety and effectiveness</li> <li>Avoidable ED visits by LTC residents</li> <li>Population health (smoking, obesity at provincial average but higher than average rate of physical inactivity)</li> <li>Preventive health screening (but lower rate of fecal occult blood testing)</li> </ul>	
Worse than average results, major room to improve	<ul> <li>Longer ED wait times, especially for admitted patients</li> <li>Relatively low supply of specialists and nurses</li> </ul>	
CHANGES OVER TIME:		
LTC waits (from Apr-Jun 09 to 2009/10)	Overall	Worse by 26 days
	Placement from hospital	Worse by 26 days
	Placement from home	Worse by 8 days
ALC (from 2008/09 to 2009/10)		Worsened by 3%

SOUTHEAST LHIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	<ul> <li>ED wait times</li> <li>Access to primary care</li> <li>Shorter wait times for cataract surgery</li> </ul>
Average results, still room to improve	<ul> <li>ALC hospital bed days</li> <li>Waits for a LTC bed</li> <li>Waits for surgery is at average, but long waits for MRI scans</li> <li>Avoidable hospitalizations (readmissions higher for some conditions, lower for others)</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>LTC safety and effectiveness (most indicators similar to provincial average; worse rates for bladder infections, restraint use, use of new antipsychotics)</li> <li>Home care safety and effectiveness (most indicators similar to provincial average)</li> <li>Avoidable ED visits by LTC residents</li> <li>Supply of family doctors, specialists and nurses</li> <li>Preventive health screening (most indicators similar to provincial average, but with higher rates of Pap tests)</li> </ul>
Worse than average results, major room to improve	<ul> <li>Longer waits for MRI</li> <li>Home care waits</li> <li>Population health (higher rate of smoking but less physical inactivity; higher rates of self-harm, injury-related ED visits)</li> </ul>
Note: The Southeast has a relatively higher supply of nurse	s and nurse practitioners and an average supply of doctors.

Note: The Southeast has a relatively higher supply of nurses and nurse practitioners and an average supply of doctors.

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	No major changes
	Placement from hospital	Better by 23 days
	Placement from home	No major changes
ALC (from 2008/09 to 2009/10)		No major changes

CHAMPLAIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	<ul> <li>Shorter wait times for MRI scans</li> <li>Avoidable hospitalizations (rate of admission for ambulatory care sensitive conditions slightly lower than average; readmissions lower for AMI, asthma, stroke and Gl bleed)</li> </ul>
Average results, still room to improve	<ul> <li>ALC hospital bed days</li> <li>ED wait times</li> <li>Access to primary care</li> <li>Wait times for CT scans, cancer surgery</li> <li>Avoidable ED visits by LTC residents</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>LTC safety and effectiveness (average for most indicators; worse results for restraint use but better (lower) results for new starts of anti-anxiety drugs)</li> <li>Home care safety and effectiveness (average for most indicators)</li> <li>Population health (average results for smoking, obesity; higher rate of binge drinking but lower rate of physical inactivity; higher rate of breastfeeding)</li> <li>Preventive health screening (average for most indicators but better results for Pap screening)</li> </ul>
Worse than average results, major room to improve	<ul> <li>Longer waits for hip and knee replacement, cataract surgery, general surgery</li> <li>Longest wait time for a LTC bed in Ontario, for people placed from the community. (Wait times for people placed from hospital are close to the provincial average)</li> <li>Longer time to home care visit after hospital discharge</li> <li>Second-highest rate of HIV incidence in Ontario (after Toronto Central)</li> </ul>

Note: Champlain has a relatively high family physician and specialist supply.

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	Better by 28 days
	Placement from hospital	Worse by 11 days
	Placement from home	Worse by 13 days
ALC (from 2008/09 to 2009/10)		Worsened by 2%

NORTH SIMCOE MUSKOKA LHIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	Shorter ED waits
Average results, still room to improve	<ul> <li>ALC hospital bed days</li> <li>Access to primary care (% of population without a family doctor similar to average, but very low % of population who can get timely appointment with family doctor)</li> <li>Waits for CT scans, general surgery &amp; cancer surgery</li> <li>Home care waits</li> <li>Avoidable hospitalizations (most indicators at provincial average; readmissions higher for asthma, Gl bleed)</li> <li>Chronic disease management (diabetes complications, AMI survival)</li> <li>Use of right drugs for hospital patients with AMI, CHF</li> <li>C-section rates</li> <li>LTC safety and effectiveness (average for most indicators; higher (worse) rate of new starts of benzodiazepines among residents)</li> <li>Home care safety and effectiveness (average results for most indicators; worse results for decline in cognitive function and unintended weight loss)</li> <li>Average supply of family doctors and nurses; relatively low supply of specialists; relatively high supply of registered practical nurses</li> <li>Preventive health screening (lower physical inactivity but higher smoking in pregnancy)</li> </ul>
Worse than average results, major room to improve	<ul> <li>Longer waits for a LTC bed</li> <li>Longer waits for hip and knee replacement</li> <li>Higher rate of avoidable ED visits by LTC residents, especially for low acuity visits</li> </ul>
CHANGES OVER TIME:	

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	Worse by 23 days
	Placement from hospital	Worse by 14 days
	Placement from home	Worse by 49 days
ALC (from 2008/09 to 2009/10)		No major changes

#### 11. LHIN Analyses

NORTHEAST LHIN	
Superior results, no room to improve	None
Better than average results, still room to improve	<ul><li>Shorter ED waits, especially for admitted patients</li><li>Shorter wait for MRI, general surgery</li></ul>
Average results, still room to improve	<ul> <li>Wait times for CT</li> <li>C-section rates</li> <li>LTC safety and effectiveness</li> <li>Home care safety and effectiveness</li> <li>Avoidable ED visits by LTC residents, rate of avoidable low acuity visits is worse than average</li> <li>Preventive health screening</li> </ul>
Worse than average results, major room to improve	<ul> <li>Longer waits for hip and knee replacement, urgent cancer surgery, cataract surgery</li> <li>More ALC hospital bed days</li> <li>Longer waits for a LTC bed, especially for people placed from hospital (but higher percentage of residents placed in first choice of home)Higher percentage of residents placed into LTC who might not need to be there (low MAPLe score)</li> <li>Access to primary care – lower proportion of population with a family doctor</li> <li>Home care waits</li> <li>Lower rate of use of right drugs for hospital patients with AMI</li> <li>More avoidable hospitalizations (high rates of admission for ambulatory care sensitive conditions, stroke and AMI readmissions)</li> <li>Worse chronic disease management (higher rates of diabetes complications)</li> <li>Population health (higher rates of smoking, obesity, AMI incidence, injuries, intentional self-harm; however, lower rates of physical inactivity)</li> </ul>

The Northeast has an average supply of family doctors, a relatively low supply of specialists and high supply of nurses and nurse practitioners.

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	No major changes
	Placement from hospital	Worse by 21 days
	Placement from home	No major changes
ALC (from 2008/09 to 2009/10)		No major changes

NORTHWEST LHIN	
Superior results, no room to improve	• None
Better than average results, still room to improve	<ul> <li>Shorter ED waits</li> <li>Shorter waits for CT, MRI and knee replacements</li> <li>Shorter time to home care visit after hospital discharge</li> </ul>
Average results, still room to improve	<ul> <li>ALC hospital bed days</li> <li>Waits for hip replacement, general surgery</li> <li>C-section rates</li> <li>Home care safety and effectiveness (most indicators at provincial average)</li> <li>Avoidable ED visits by LTC residents</li> <li>Preventive health screening (but rates of flu vaccination lower)</li> </ul>
Worse than average results, major room to improve	<ul> <li>Lower access to primary care (higher % of population without a family doctor; more difficult to get appointment next day with family doctor)</li> <li>Longer waits for a LTC bed for people placed from the community</li> <li>Higher waits for urgent cancer surgery</li> <li>Lower use of right drugs for hospital patients with AMI, CHF</li> <li>Worse chronic disease management (higher rates of diabetes complications)</li> <li>More avoidable hospitalizations (high rates of hospitalization for ambulatory care sensitive conditions; highest rate of readmissions for CHF; high stroke readmissions)</li> <li>LTC safety and effectiveness (higher rates of unintended weight loss; new starts of benzodiazepines; new pressure ulcers; use of restraints)</li> <li>Population health (higher rates of smoking, smoking in pregnancy, obesity, binge drinking, AMI incidence, injuries, intentional self-harm; however, lower rates of physical inactivity)</li> </ul>

Note: The Northwest has an average supply of family doctors, relatively low supply of specialists and high supply of nurses and nurse practitioners.

LTC waits (from Apr-Jun 09 to 2009/10)	Overall	Worse by 21 days
	Placement from hospital	Worse by 26 days
	Placement from home	Better by 181 days
ALC (from 2008/09 to 2009/10)		No major changes

North West	3.4%	0.8	1.8	11.2	4.2	12%	27%	96%	81%	87%	23	93%	71%	23%	76
North East	3.9%	6.0	1.9	10.6	4.1	14%	34%	88%	53%	81%	30	73%	56%	41%	113
North Simcoe Muskoka	4.1%	1.0	3.5	8. 8.	3.7	5.9%	23%	%96	70%	81%	32	94%	43%	17%	94
nislqmsdЭ	4.6%	1.3	2.9	13.2	5.2	10%	42%	93%	61%	68%	34	84%	60%	35%	104
South East	3.6%	0.9	2.3	9.4	4.1	2.4%	46%	78%	76%	93%	16	61%	20%	22%	100
Central East	5.3%	1.4	4.5	11.8	4.3	5.8%	40%	81%	57%	85%	27	65%	38%	32%	104
Central	5.3%	1.5	6.0	13.2	3.8	6.0%	44%	%96	59%	78%	41	66%	53%	30%	124
Toronto Central	5.3%	1.5	3.5	14.2	4.9	8.0%	48%	95%	70%	74%	35	76%	51%	39%	125
sgussissiM Halton	4.5%	1.3	3.8	10.1	3.7	5.0%	49%	88%	60%	89%	23	%06	74%	13%	141
Central West	5.8%	1.6	3.3	11.7	4.1	7.3%	49%	%26	75%	100%	14	70%	57%	19%	128
Hamilton Niagara Haldimand Brant	5.7%	1.1	4.0	14.9	4.7	3.0%	45%	%06	56%	70%	41	57%	40%	15%	120
Waterloo Waterloo	6.1%	1.5	2.1	10.1	5.3	4.3%	48%	81%	56%	95%	26	78%	27%	62%	61
teaW dtuo2	3.4%	6.0	1.8	9.3	3.8	6.1%	46%	%06	78%	87%	25	72%	51%	38%	70
Erie St. Clair	5.2%	1.2	1.8	0.6	4.5	6.2%	31%	%96	74%	88%	28	86%	78%	44%	20
Ontario	4.7%	1.2	3.0	11.7	4.3	6.4%	43%	%06	64%	81%	29	75%	51%	30%	113
Directionality	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER
Indicator	Emergency Department - Percentage of patients left without being seen, 2009/10	Emergency Department - time to MD assessment by all levels, 2009/10	Emergency Department - time from admission to transfer to bed all levels, 2009/10	$90^{\rm th}$ percentile wait time for high complexity ED patients, Sep 2010	90 <sup>th</sup> percentile wait time for low complexity ED patients, Sep 2010	Percentage of adults without a family doctor, 2009/10	Percentage of adults able to see their doctor on the same day or next day the last time they were sick or needed medical attention, 2009/10	CT Scans Percentage of priority 2 cases done within target, Dec 2010	Percentage of priority 3 cases done within target, Dec 2010	Percentage of priority 4 cases done within target, Dec 2010	90 <sup>th</sup> percentile wait time, Dec 2010	MRI Percentage of priority 2 cases Scans done within target, Dec 2010	Percentage of priority 3 cases done within target, Dec 2010	Percentage of priority 4 cases done within target, Dec 2010	90 <sup>th</sup> percentile wait time, Dec 2010
Attribute/Theme In		departments En as	요	EL EL	EE EE	Accessible Pe 2.2 Access to 20			and Access to Specialists			S			

= better than average = not significantly different from average = worse than average

129

North West	56%	91%	100%	32	Ž	Ž	98%	78	Ž	R	86%	203	Ž	77%	92%	173
North East	50%	67%	89%	47	Z	59%	93%	124	Z	53%	82%	323	Z	61%	63%	427
North Simcoe Muskoka	67%	80%	100%	44	60%	%16	%16	151	N	47%	67%	243	N	45%	80%	228
nislqmsdЭ	71%	%62	98%	49	Z	55%	94%	169	63%	26%	51%	351	Z	34%	81%	226
South East	Z	83%	88%	40	R	98%	%66	93	50%	75%	94%	141	N	74%	79%	160
Central East	56%	85%	%66	43	R	81%	%96	156	R	74%	81%	185	63%	78%	%06	172
Central	81%	92%	%66	34	100%	100%	100%	82	R	88%	100%	140	R	82%	94%	166
Toronto Central	60%	72%	88%	55	Z	%96	100%	112	91%	95%	91%	125	100%	85%	%96	135
Mississauga Halton	71%	85%	%96	54	R	89%	%96	160	R	63%	91%	128	N	75%	94%	138
Central West	R	82%	93%	62	R	R	%66	116	R	R	100%	131	R	86%	%06	180
Hamilton Niagara Haldimand Brant	73%	72%	89%	56	75%	72%	88%	143	73%	62%	74%	235	57%	49%	86%	209
Waterloo Materloo	64%	91%	96%	40	Z	100%	100%	94	Z	80%	100%	97	Z	%69	%06	176
tesW dtuo2	54%	67%	84%	71	29%	%66	%66	72	83%	54%	95%	163	R	44%	86%	211
Erie St. Clair	87%	86%	96%	43	%26	81%	100%	68	N	100%	100%	93	N	89%	94%	126
Ontario	67%	78%	93%	51	%06	83%	98%	122	74%	66%	84%	206	63%	63%	88%	200
Directionality	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER
	Percentage of priority 2 cases done within target, Dec 2010	Percentage of priority 3 cases done within target, Dec 2010	Percentage of priority 4 cases done within target, Dec 2010	90 <sup>th</sup> percentile wait time, Dec 2010	Percentage of priority 2 cases done within target, Dec 2010	Percentage of priority 3 cases done within target, Dec 2010	Percentage of priority 4 cases done within target, Dec 2010	90 <sup>th</sup> percentile wait time, Dec 2010	Percentage of priority 2 cases done within target, Dec 2010	Percentage of priority 3 cases done within target, Dec 2010	Percentage of priority 4 cases done within target, Dec 2010	90th percentile wait time, Dec 2010	Percentage of priority 2 cases done within target, Dec 2010	Percentage of priority 3 cases done within target, Dec 2010	Percentage of priority 4 cases done within target, Dec 2010	90th percentile wait time, Dec 2010
Indicator	Cancer Surgeries				Cataract Surgeries				Hip Replacements				Knee Replacements			
Attribute/Theme	Accessible 2.3 Surgical Wait Times and Access to Specialists															

\* NV= no value for this LHIN = better than average = not significantly different from average = worse than average

Attribute Indicator Directionality Couth West South West Waterloo Wellington Waterloo Wellington Waterloo Waterloo Waterloo Waterloo Wellington Waterloo	Accessible General Surgeries Percentage of priority 2 2.3 Surgical wait times Dec 2010 Dec 2010	Percentage of priority 3 cases done within target, Dec 2010	Percentage of priority 4 bETTER 97% 94% 96% 96% Dec 2010	90 <sup>th</sup> percentile wait time, bETTER 97 101 104 79 104	Accessible Median number of overall DeTTER 113 92 84 182 178 2.4 Access to long- care home place	home care and ment, 2009/10: from acute care hospital BETTER 58 44 47 64 107	from community <b>BETTER</b> 173 119 125 266 215	Percentage of residents who are placed in their first choice of LTC home, 2009/10 34% 43%	Percentage of beta-blocker beta-blocker 79% 79% 79% 76% ized for AMI who.	hospital within 90 days ACE/VARB ACE/VARB 78% 76% 78% 83% 79%	INC. (2003/10) statin 87% 86% 82% 87% 86%	all three drugs = BETTER 59% 57% 68% 57%	Percentage of beta-blocker BETTER 66% 61% 63% 73% 65% ized for CHF who.	within 90 days post-discharge, filled a prescription 75% 75%	DOT: 12003/10) both beta-blocker and ABETTER 50% 45% 48% 59% 50%
Central West Mississauga Halton Toronto Central	100% 91% 78%	100% 88% 83%	%96 %66	60 86 110	47 120 98	28 40 63	77 176 195	30% 36% 43%	80% 76% 80%	83% 82% 77%	93% 88% 88%	66% 61% 62%	71% 64% 68%	71% 73% 69%	51% 50% 51%
Central Central East	100% 83%	98% 94%	%66 %66	80 85	116 114	48 83	194 144	36% 36%	81% 79%	79% 79%	89%	61% 60%	67% 67%	72% 70%	51% 49%
ss£ dtuo2 Champer	63% 88%	89%	96% 93%	99 154	111 209	61 58	148 314	40% 39%	83% 83%	81% 78%	89%	63% 63%	72% 70%	66% 69%	52% 52%
North Simcoe Muskoka North East	95% 79%	83% 94%	100% 98%	101 90	143 133	85 125	223 151	36% 49%	80% 72%	77% 72%	86% 82%	58% 46%	65% 62%	71% 70%	49% 48%
teeW dfrou	81%	91%	93%	107	139	71	229	35%	72%	75%	87%	51%	59%	73%	46%

\* NV= no value for this LHIN = better than average = not significantly different from average = worse than average

South East nislqmsd	28% 29%	16% 13%	55% 52%	60% 60%	67% 65%	47% 46%	4.8 4.3	38 36	9.7 8.8	317 257	3.5 3.1	11 12	0.0 2.3	10 8.1	1.2 1.9	1.6 0.8	8.1 7.1	7.6 6.1
Central East	30%	16%	51%	63%	70%	51%	4.1	35	9.1	254	3.3	10	3.9	7.7	1.9	2.6	8.0	8.3
Central	30%	15%	47%	60%	65%	47%	3.4	36	9.3	181	3.7	11	4.0	5.0	2.9	1.1	7.1	4.5
Toronto Central	30%	17%	46%	60%	64%	47%	4.2	36	10	241	3.2	12	3.4	12	3.5	1.4	8.1	7.6
synssissiM Malfon	28%	13%	49%	60%	66%	48%	3.4	33	8.9	203	3.0	10	4.7	5.8	2.1	1.1	6.7	1.7
Central West	30%	19%	46%	59%	66%	47%	3.8	30	8.8	280	6.3	11	5.9	7.4	2.4	1.2	6.7	2.2
Hamilton Niagara Hamilton Niagara	29%	15%	53%	61%	71%	50%	4.7	35	9.6	325	4.2	11	3.5	8.1	2.5	1.8	7.7	7.9
Waterloo Waterloo	26%	13%	55%	58%	68%	46%	4.4	35	9.8	236	3.6	7.9	1.3	8.0	1.9	2.5	7.5	2.0
South West	22%	10%	54%	58%	67%	45%	4.5	40	10	300	3.9	8.5	4.1	6.4	3.1	2.1	8.2	4.7
Erie St. Clair	25%	13%	54%	58%	69%	46%	4.7	37	9.7	338	5.1	9 13	5.1	7.2	2.9	2.8	8.3	3.9
Ontario	R 28%	R 15%	R 51%	R 60%	R 67%	R 48%	R 4.3	R 35	R 9.1	R 278	R 4.25	R 11.09	R 3.86	R 7.65	R 2.64	R 1.75	R 7.8	R 5.38
Directionality	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER
	per	deliver a r	an eye	_	an ACEI/ARB	both statin and ACEI/ARB	iad any 09/10	after diagno- ), 2008/09	art attack year after	atory care 09/10		Ŀ	asthma	COPD	stroke (medical)	GI bleed	gastrointestinal	diabetes
	ean section	thers who section pe	s who had 2009/10	a statin	an AC	both ACEI	who h n, 200	e year a	00 hei Id one /09	ambul o, 20(	AMI	CHF	ac	0	•/			0
Indicator	Rate of delivering via Caesarean section per 100 deliveries, 2009/10	Rate of low-risk first-time mothers who deliver full-term baby via Caesarean section per 100 deliveries, 2009/10	Percentage diabetes patients who had an eye exam in the past 12 months, 2009/10	Percentage of elderly a statir diabetes patients regularly taking:		both	Rate of people with diabetes who had any serious diabetes complication, 2009/10	Adjusted mortality rate in the year after diagno- sis of congestive heart failure (CHF), 2008/09	Adjusted rate of death per 100 heart attack patients between 30 days and one year after their first heart attack, 2008/09	Rates of hospitalizations for ambulatory care sensitive conditions in Ontario, 2009/10	Rate of readmission to acute care hospi- tal within 30 days of		SB	0				0

												1		
tsəW drod	24%	30%	11%	25%	9.4%	12%	50%	45%	24%	7.9%	50%	17%	5.1%	
North East	23%	34%	14%	26%	12%	6.8%	44%	41%	20%	7.0%	46%	15%	4.6%	
North Simcoe Muskoka	22%	34%	13%	29%	14%	8.0%	53%	43%	23%	8.3%	57%	19%	6.0%	
Champlain	21%	34%	12%	27%	14%	6.5%	50%	43%	24%	8.1%	62%	20%	3.9%	
South East	20%	36%	13%	30%	11%	7.2%	49%	49%	25%	6.4%	54%	20%	4.3%	
Central East	19%	31%	10%	24%	12%	6.9%	49%	46%	24%	9.8%	55%	18%	4.3%	
Central	17%	28%	%6	18%	11%	7.1%	55%	45%	24%	14%	49%	16%	2.8%	
Toronto Central	17%	30%	%6	17%	10%	6.7%	47%	52%	25%	12%	57%	19%	4.8%	
Rississauga Halton	18%	31%	10%	24%	14%	6.3%	54%	51%	23%	7.2%	47%	17%	3.8%	
Central West	19%	31%	10%	21%	12%	7.8%	59%	60%	22%	11%	61%	21%	5.9%	
Hamilton Niagara Handinand Brant	24%	38%	14%	30%	14%	6.3%	52%	48%	24%	8.2%	52%	18%	5.2%	
Waterloo notgnill9W	23%	35%	13%	32%	14%	8.9%	54%	55%	23%	7.6%	52%	18%	5.3%	
teaW dtuo2	25%	36%	14%	30%	15%	6.8%	48%	41%	22%	8.2%	47%	17%	4.2%	
Erie St. Clair	25%	38%	14%	32%	15%	7.3%	47%	45%	24%	6.1%	43%	14%	3.3%	
Ontario	21%	33%	12%	26%	13%	7.1%	50%	46%	24%	%6	51%	17%	4.3%	
Directionality	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	
	worsening bladder control	increasing difficulty carrying out normal, everyday tasks (getting dressed, eating, personal hygiene)	pain that got worse recently	worsening symptoms of depression or anxiety	recent decrease in language, memory and thinking abilities	recent unintended weight loss	bladder function has recently decreased or did not improve	a new problem with normal everyday tasks (getting dressed, eating, personal hygiene) OR an old problem that is not getting better	pain that is not well controlled	serious signs of depression (e.g. profound sadness, withdrawal from normal activities)	recent decline in language, memory and thinking abilities	new problem communicating or understanding others or an existing problem that did not improve over a period of time	recent unintended weight loss	
Indicator							Percentage of long-stay home care clients in 2009/10 with:							
Attribute/Theme	Effective 3.4 Keeping people heatthy in long- term care						3.5 Keeping people healthy in home care							

= better than average = not significantly different from average = worse than average

Torth West	8.2	5 20%	13%	31%	15	6 3.7%	5 11%	6 5.0%	23%	14%	58%	6 1.6%	2%	18%	20%	25	8.1
North East	14	22%	10%	24%	16	2.4%	14%	5.4%	20%	15%	25%	1.2%	13%	28%	31%	20	11.3
North Simcoe Muskoka	13	22%	17%	31%	16	2.9%	15%	6.4%	14%	14%	30%	3.6%	%6	20%	23%	23	11.4
Champlain	8.7	17%	11%	14%	16	2.7%	15%	5.8%	23%	14%	25%	1.4%	10%	16%	20%	21	9.4
sast fruoS	8.9	22%	20%	22%	12	3.0%	15%	6.9%	24%	14%	26%	1.6%	12%	17%	23%	17	9.2
Central East	11.5	19%	12%	29%	15	2.5%	12%	4.3%	18%	13%	23%	1.2%	12%	18%	24%	23	9.1
Central	9.8	18%	15%	25%	15	2.2%	10%	4.5%	12%	13%	23%	1.5%	12%	15%	23%	25	6.2
Toronto Central	10	16%	10%	20%	13	2.8%	10%	4.4%	10%	14%	23%	1.5%	13%	12%	23%	24	6.1
sgussissiM Halton	11	16%	%6	22%	15	2.5%	13%	5.8%	11%	14%	28%	1.7%	12%	%6	22%	25	3.4
Central West	10	18%	21%	32%	15	2.7%	11%	5.1%	%6	14%	26%	1.7%	11%	10%	23%	26	4.4
Hamilton Niagara Haldimand Brant	11	18%	17%	28%	13	3.0%	15%	6.2%	17%	14%	29%	2.1%	13%	21%	22%	17	5.9
Waterloo Materloo Mellngton	9.3	17%	10%	24%	12	3.1%	16%	5.4%	19%	15%	27%	2.0%	10%	18%	20%	14	7.4
tesW dtuo2	10	20%	14%	25%	13	3.1%	15%	5.8%	17%	15%	24%	1.4%	12%	12%	24%	20	9.8
Erie St. Clair	10	21%	17%	21%	13	3.4%	16%	5.2%	22%	14%	27%	1.7%	10%	11%	24%	22	7.9
Ontario	9.6	19%	14%	24%	14.0	2.7%	14%	5.4%	17%	14%	25%	1.5%	12%	16%	23%	21.2	7.8
Directionality	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER
	Adjusted death rate from heart attack per 100 patients within 30 days, 2008/09	Percentage of elderly long-term care residents prescribed a drug that should be avoided in the elderly (Beers list), 2009/10	Incidence of new starts of antipsychotics with- out a clear indication in seniors newly admitted to long-term care homes, 2008/09	Incidence of new starts of benzodiazepines in seniors newly admitted to long-term care homes, 2008/09	Rate of falls among seniors (aged 65+) resulting in an emergency department visit or inpatient hospitalization per 100 resident-years in long- term care homes, 2008/09	a new pressure ulcer (stage 2 or higher)	behaviour that has worsened recently	a recent bladder infection	physical restraint	a fall in the last 30 days	a fall in the last 90 days	a new pressure ulcer (stage 2 or higher)	unexplained injuries, burns, or fracture	Percentage of acute care bed days which are designated as alternative level care (ALC) 2009/10	Percentage of people placed into an LTC home who do not have with high or very high needs for LTC services (or care needs) as measured using the MAPLe scores, 2009/10	Number of non-urgent, avoidable ED visits per 100 LTC residents per year, 2009/10	Rates (per 100 person years) of Low Acuity ED Visits by LTC Residents, 2009/10
Indicator	Adjusted death patients within (	Percentage of ( prescribed a dr elderly (Beers li	Incidence of ne out a clear indic to long-term ca	Incidence of ne in seniors newly homes, 2008/C	Rate of falls am in an emergenc hospitalization r term care home	Percentage of LTC residents in 2009/10	with:				Percentage of long-stay home care clients in	2009/10 with:		Percentage of a are designated 2009/10	Percentage of r who do not hav for LTC services using the MAPL	Number of non- 100 LTC reside	Rates (per 100 Visits by LTC Re
Attribute/Theme	Safe 4.3 Mortality in hospital	Safe 4.4 Drug Safety in long term care			Safe 4.3 Avoiding harm in long term care homes						Safe 4.5 Avoiding harm in community			Efficient 6.2 Right service in right place		Efficient 6.3 Avoidable	ernergency department visits

Attribute/Theme In	Efficient Rat 6.4 Avoiding unneces- sary drugs and tests	Pe ed tre	esourced nan	resources				Integrated Per 8.1 Discharge/ act transitions	ath behaviour							9.2 Maternal and child health Per									
Indicator	te of pre-opera 0 cataract sur, centage of elc hypertension t atment, 2009/		e of pre-opera ) cataract sun	e of pre-opera ) cataract sun	e of pre-opera ) cataract sur	e of pre-opera ) cataract sur	e of pre-opera I cataract surg cataract surg :entage of eld ypertension t iment, 2009/		e of pre-opera I cataract surg catarace of eld ypertension ti iment, 2009/		centage of elc typertension t tment, 2009/	Supply, per 100,000 neonle of					rcentage of st ute care to inp	centage of the above who re	rcentage of no oulation aged gularly expose	Percentage of the population	above who: (2009)			rcentage of m th, 2009	rcentage of we
	Rate of pre-operative chest X-ray testing per 100 cataract surgeries, 2009/10	Percentage of elderly patients with uncomplicated integration treated with diuretic as first-line treatment, 2009/10	family physicians, 2009	specialists, 2009	nurse practitioners, 2010	registered nurses, 2010	registered practical nurses, 2010	Percentage of stroke patients discharged from acute care to inpatient rehabilitation, 2010	Percentage of the general population age 12 and above who report smoking, 2009	Percentage of non-smokers in the general population aged 12+ years who report being regularly exposed to tobacco smoke in any of their home, vehicle or in public places, 2009	have had binge drinking episodes	are obese	are physically inactive	have inadequate fruit and vegetable intake	Percentage of mothers breastfeeding right after birth, 2009	Percentage of women who smoked during their pregnancy. 2009/10									
Directionality	BETTER	BETTER						BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER									
Ontario	3.93	17%	88	66	11	713	231	29%	19%	18%	21%	18%	49%	56%	88%	12%									
Erie St. Clair	3.1	13%	64	61	12	680	260	34%	21%	18%	22%	21%	51%	62%	68%	18%									
teeW thuo2	3.1	20%	84	110	15	929	305	28%	22%	19%	21%	22%	51%	61%	86%	18%									
Waterloo Wellington	2.6	20%	8	60	12	554	233	25%	18%	16%	23%	17%	50%	58%	92%	13%									
Hamilton Niagara Haldimand Brant	3.5	16%	78	102	12	736	262	25%	21%	18%	23%	20%	45%	59%	80%	17%									
Central West	3.7	19%	65	47	c	308	98	22%	14%	16%	14%	17%	60%	48%	%06	6%									
Malton Malton	3.9	15%	75	62	4	474	110	33%	17%	19%	22%	14%	49%	51%	93%	6%									
Toronto Central	5.3	15%	141	298	23	1374	238	28%	13%	11%	22%	13%	51%	58%	95%	5%									
Central	5.7	13%	81	70	ę	437	140	29%	17%	18%	16%	14%	52%	58%	92%	4%									
Central East	4.5	14%	70	60	9	501	195	34%	18%	24%	17%	19%	55%	55%	88%	11%									
South East	3.4	21%	104	112	17	901	374	25%	24%	20%	23%	22%	41%	54%	80%	24%									
nislqmsd)	3.5	22%	114	132	13	859	269	29%	18%	15%	26%	19%	42%	53%	94%	11%									
North Simcoe Muskoka	3.3	17%	88	55	12	682	294	27%	27%	21%	25%	18%	42%	56%	82%	19%									
North East	3.4	19%	95	99	25	927	411	25%	26%	20%	24%	26%	44%	59%	86%	27%									
North West	7.4	22%	108	71	31	1065	465	36%	25%	17%	28%	27%	39%	58%	82%	36%									

135

teaw dtrou	2.1	66%	69%	72%	40%	57%	21	299	224	14913	822
North East	6.5	73%	67%	71%	34%	71%	21	282	156	13579 1	679
North Simcoe Muskoka	1.8	68%	63%	70%	32%	84%	22	242	110	11530	491
nislqmsd)	14	72%	70%	79%	41%	83%	22	194	84	8276	343
South East	3.5	78%	66%	77%	36%	74%	21	216	135	12815	387
Central East	3.5	76%	68%	72%	33%	84%	19	198	06	8486	360
Central	4.5	71%	70%	74%	35%	88%	18	149	49	6571	290
Toronto Central	43	71%	60%	70%	25%	86%	19	150	69	6927	352
Rississauga Halton	5.2	63%	67%	72%	42%	87%	19	164	61	6665	315
Central West	2.3	%69	62%	%69	30%	83%	15	202	57	6446	303
Hamilton Niagara Haldimand Brant	6.8	73%	65%	72%	33%	85%	20	251	107	10097	489
Waterloo Wellington	4.1	73%	66%	74%	40%	82%	19	219	120	8088	413
səW dino2	3.6	74%	67%	72%	35%	73%	22	212	107	12964	548
Erie St. Clair	2.9	77%	%69	68%	40%	71%	21	237	17	9126	444
Ontario	8.6	72%	67%	72%	35%	81%	20	203	88	8,845	404
Directionality	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER	BETTER
Indicator	HIV incidence per 100,000, 2008	Percent of the population aged 65+ reporting having received a flu shot in the past year, 2009	Percentage of women (aged 50-69) who reported having a mammogram in the last two years,2008/09	Percentage of women (aged 20-69) who reported having a Pap test in the last three years,2007/09	Percentage of people (aged 50-74) who reported having a fecal occult blood test in the 2 years prior to the survey, 2009	Rate of women aged 65 years who have had a bone mineral density (BMD) test since turning 55 years of age, 2009/10	Mortality age-standardized rates per 100,000 for colon and rectum, 2003-2007	AMI incidence per 100,000, 2009/10	Rate of ED visits for intentional self-harm per 100,000, 2009/10	Rate of injury-related ED visits per 100,000, 2009/10	Rate of injury-related hospitalizations per 100,000, 2009/10
Attribute/Theme	Focused on Population Helath 9.3 Sexual health	Focused on Population Helath	9.4 rrevenuve measures				Focused on Population Helath 9.5 Dearm that could be avoided by prevention				

= better than average = not significantly different from average = worse than average

11. LHIN Analyses



# 12.1



## **Examples of success** — Reducing ED waits

**Organizations:** Mount Sinai Hospital, a major Toronto teaching hospital; Oakville-Trafalgar Memorial Hospital, a very high volume community hospital; and Georgetown Hospital, a high volume community hospital. The two community hospitals are a part of Halton Healthcare Services.

**Aims and Measures:** increase the percentage of admitted and non-admitted patients treated within the recommended timeframe (four and eight hours respectively) and decrease the 90th percentile amount of time spent in the ED, for admitted patients, high-acuity non-admitted patients and low-acuity non-admitted patients.

#### **Change Ideas:**

Mount Sinai Hospital:

- Set up a Rapid Assessment Zone (RAZ) in the ED in July 2008, where an internal waiting room is used to assess patients who don't require care in a stretcher on an ongoing basis, but do require privacy for their examination or treatment. This waiting room allows patients to effectively "share" a stretcher, which speeds up processing and reduces wait times.
- Establish an "offload nurse" program where a dedicated nurse initiates care for patients who arrive by ambulance.
- Have nurses initiate diagnostic tests and interventions.
- Provide porters to help imaging technologists move patients, extend hours for CT scans, add technologists on some night shifts and work closely with Diagnostic Imaging to reduce wait times for ED patients.
- Establish the new position of Patient Flow Coordinator and support policy and process changes that have improved flow of admitted patients from the ED to in-patient units.

Oakville-Trafalgar Memorial Hospital:

- Establish a Rapid Assessment Zone similar to the above.
- Implement ED Controllers to support overburdened nurses and act as a liaison between the ED and in-patient units.
- Adjust team roles and responsibilities so each professional can focus on one area of ED care; for example, having a Medical Lab Assistant initiate laboratory tests, manage flow or process orders — tasks previously completed by overburdened nurses.
- Host weekly ED Operations meetings or ongoing audits to address progress and barriers.

#### Georgetown Hospital:

• Divide the ED into smaller units, each of which treats people with a similar acuity level, and establish a Rapid Assessment and Fast Track area similar to the Rapid Assessment Zone above.

- Establish seven additional "flex beds" and respond to peaks in demand by adjusting roles and enhancing nursing and clerical hours.
- Change the culture so that patients are thought of as being "pulled" to the floor by inpatient units rather than being "pushed" out of the ED. Inpatient units were given increased decision-making authority to augment staff based on increases in activity.

Positive engagement of front line physicians, nurses and inpatient medicine units was an important success factor for all sites.

#### **Results:**

These three hospitals have achieved major reductions in ED waits, over a time period of just under three years. For low acuity patients, Mount Sinai decreased the 90th percentile time spent in the ED by 2 hours, and Oakville-Trafalgar and Georgetown decreased this time by one hour. For high acuity patients, Mount Sinai dropped the time spent in the ED by 7.1 hours, a decrease of more than a third. Oakville-Trafalgar and Georgetown decreased this time by 2.1 and 1.4 hours respectively.







#### Leading Results in Ontario, October-December 2010:

Other hospitals in Ontario can aim to meet or exceed the results achieved by these organizations:

Hospital	Туре		rcentile of stay	Percentage of patients treated within recommended time frame					
		Low acuity	High acuity	Admitted patients	Non-admitted high acuity patients	Non-admitted low acuity patients			
Mount Sinai	Teaching hospital	5.1 hrs	13.2 hrs	32%	91%	85%			
Oakville-Trafalgar	Very high volume community hospital	3.7 hrs	6.9 hrs	72%	100%	95%			
Georgetown	High volume community hospital	3.7 hrs	5.1 hrs	93%	100%	95%			



### **Examples of success — Primary care**

#### **Organizations:**

- The North York Family Health Team (FHT), a large interdisciplinary team which includes 56 physicians, nurses, nurse practitioners, pharmacists, social workers and dieticians. The FHT serves 54,000 patients across nine locations in North York.
- The Smithville Medical Centre Family FHT, which includes 8 family physicians and nursing staff, and serves the small community of Smithville, just southeast of Hamilton.

#### Aims and measures:

- Reduce wait times for an appointment, as measured by the time to third next available appointment.
- Maintain the total time spent by a patient in the office from arrival to departure ("office cycle time") to under 60 minutes.
- Ensure at least half of the time spent by a patient in the office is actually spent with a provider (the "50% Red Zone Time" target).
- Improve physician continuity so patients see their own provider at least 85% of the time.

#### **Change Ideas:**

Ideas related to scheduling include:

- Monitoring the supply and demand for appointments and working continuously to ensure they are in balance.
- Temporarily creating extra appointment blocks each week to work down the backlog.
- Increasing the interval between visits for stable patients with chronic disease from four times a year to twice a year to make more room for other patients.
- Scheduling appointments early in the day to allow same-day appointments for patients who call in the morning.

Ideas to improve the efficiency include:

- Stocking each patient room with a complete set of identical supplies and equipment so providers have easy access to the tools they need.
- Investing in an electronic medical record (EMR) system to reduce documentation and file retrieval times.

#### Results:

#### Teams with improved average time to third next available appointment



The following represents results for selected physician practices within each FHT:

- Both FHTs improved the average time to the third next available appointment; at Smithville Medical Centre FHT, this wait time dropped to zero days in August 2009.
- Both FHTs surpassed the 50% Red Zone Time target, and maintained an average office cycle time of under 60 minutes.
- Both FHTs exceeded the 85% physician continuity target; in particular, North York FHT has achieved almost 100% since April 2010.

Each FHT is currently in the process of spreading these improvements to all physicians and patients within the practice.

12.3

### **Examples of success — Surgical wait times**

#### Organization: North York General Hospital

**Aims and measures:** Maximize the percentage of patients who receive their surgery within the recommended wait time, for hip and knee replacement, cataract, general and cancer surgery.

#### **Change Ideas:**

- Develop an online booking system to reduce errors and delays due to lost paperwork, illegible writing or incomplete information.
- Provide training to surgeons' administrative assistants on how to complete bookings. This includes on-site visits, phone training, and FAQs (frequently asked questions) and is repeated when there is turnover of these assistants.
- Review submitted cases daily to validate if cases have been entered correctly. If there is a problem, the surgeon's office is notified immediately.
- Continuously follow up with 84 surgeons' offices to identify and address quality issues, such as entering bookings on time, assigning correct priority levels, entering patients' unavailable dates, etc.
- For cancer surgery, have multidisciplinary cancer conferences to allow case review, improve care and ensure the right care is provided at the right time.

- Use wait times funding to help fund a dedicated wait times strategy co-ordinator to develop and track action plans to support programs to meet identified targets and monitor the booking schedule.
- Deploy a comprehensive bed management system that is updated every four hours, to track which beds are available at any moment and establish standard protocols to put the right patient into the right bed. This helps prevent situations where a surgical case is cancelled at the last minute because no beds are available.
- Ensure there is the right capacity to meet demand. Wait Times Funding has been used to increase operating room time and new surgeons and other health professionals have been added to meet increases in demand.
- Have a Surgical Program Director present quarterly reviews to all members of the Surgical Program to celebrate successes, initiate discussions to review barriers and implement action plans to improve results.

#### **Results:**

North York General now ensures that 100% of patients in all priority categories and all types of surgery are served within the recommended timeframe.

Surgery Type		in all priority categories nmended timeframe
	July-September 2009	July-September 2010
General Surgery	100%	100%
Cancer Surgery	100%	100%
Cataract surgery	100%	100%
Hip replacement	100%	100%
Knee replacement	100%	100%



### Examples of success — MRI wait times

**Organizations:** Windsor Regional Hospital (a very large community hospital with two campuses), and St. Joseph's Health Care, London.

Aims and measures: Reduce the 90<sup>th</sup> percentile wait time for MRI scans.

#### **Change Ideas:**

Windsor Regional Hospital applied Lean methods to MRI scans. They conducted value stream maps, identifying each step in the process, waits between steps and error rates at each step. They found major inefficiencies, including incomplete requisitions sent in by ordering physicians. Radiologists need accurate information about the patient's clinical condition in order to select the right protocol (e.g. the areas to be scanned and whether contrast dyes, drugs or procedures need to be administered). If this information is missing, then the test is delayed. Staff made multiple phone calls to correct information and sometimes re-booked patients multiple times. Also, when patients were left not knowing their appointment time due to the above delays, they would call the office for information, which in turn wasted more staff time. Ideas for improvement include:

- Revising the requisition form for ordering an MRI, to clarify the exact information that ordering physicians needed to communicate to the radiologist.
- Tracking which ordering physicians repeatedly sent in incomplete requisitions and having them re-submit — which puts the onus on them to get it right the first time.
- Using an electronic scheduler system to make it easier to track appointment times.
- Informing patients of the time period within which they could expect to receive an appointment time, which reduced unnecessary phone inquiries.
- Ensuring that there was consistent coding of priority levels for scans.
- Booking scans on a first-in first-out basis.

Windsor also uncovered wasted time in planning and performing scans. Ideas included the following:

- Standardizing protocols. This reduced the time radiologists spent on designing the protocols, decreased the use of contrast dye (which is costly and adds extra time), and shortened the average time per patient by decreasing the number of unnecessary scans that were sometimes included in protocols.
- Booking similar body parts on same day, to reduce set-up time to change coils on the machine for doing a different body part.
- Booking more complex cases during the day, with later hours used for priority 4 cases when additional resources are not required.

*St. Joseph's Health Care*, London participated in the provincial MRI process improvement project, and also applied LEAN techniques. It found that it took too long to book the test after receiving a requisition; that the time booked for a scan often didn't match the actual time required, leading to idle scanner time; and wasted staff time due to unnecessary interruptions or looking for supplies. Ideas for improvement include:

- Streamlining the booking and protocol-setting process, which decreased booking time from 13 to 1.5 days.
- Improved scheduling, which better matched the expected time to do a scan with the actual time.
- Better organization of supplies to reduce wasted time searching for them.

St. Joseph's also added a second scanner, which added about 10% extra capacity. (Before, one scanner was used around the clock and weekends; afterwards, two scanners were used 16 hours per weekday).

#### **Results:**

Windsor regional hospital has cut its 90<sup>th</sup> percentile MRI wait times by more than half, and St. Joseph's Health Care has dramatically reduced its waits to less than a third of what they used to be in early 2008.

#### Windsor regional Hospital



#### St Joseph's Health Care, London


12.5



## **Examples of success — Chronic disease management**

**Organizations:** Petawawa Centennial Family Health Team (FHT), which serves 15,000 people, and Timmins FHT, which has 24 family physicians and five nurse practitioners.

**Aims and measures:** Improve management of type 2 diabetes, including improving screening to prevent vascular complications, ensuring appropriate treatment and encouraging self-management. Specific targets are noted in the results table below.

### **Change Ideas:**

Both of these sites implemented a variety of changes related to the redesign of how care is delivered, including:

- an electronic medical record (EMR);
- a diabetes patient registry;
- an electronic diabetes flow sheet to remind providers to follow and document the use of evidence-based practices at each visit;
- a patient follow-up system to ensure that patients are routinely notified of when follow-up appointments need to take place so that no one misses them;

- use of group medical visits, developed from Impact BC's model, where participants with diabetes learn in a group about lifestyle improvements, healthy living, interpretation of lab tests and how to set personal goals; and,
- promotion of chronic disease self-management. At Timmins, the Registered Nurse Program Lead was responsible for championing this approach and documenting patient self-management goals.

Timmins also worked to ensure diabetes patients are booked close to their birthday for an annual physical, during which a comprehensive care plan is developed. Petawawa also reorganized work flow so that lab work would routinely be done ahead of scheduled appointments, not afterwards.

### **Results:**

The following represents results for pilot sites within each FHT. Each FHT is currently in the process of spreading these improvements to all physicians and patients within the practice.

			Petawawa Centennial FHT			Timmins FHT		
Measure	Percentage of diabetes patients with:	Target	Baseline (May 2009)	Most recent value (Octo- ber 2010)	% relative improvement	Baseline (June 2008)	Most recent value (Sep- tember 2010)	% relative improvement
Process	A1C test in past six months	>90%	37%	78%	111%	89%	90%	1.1%
	Retinopathy screening in past 24 months	>90%	27%	28%	6.4%	63%	60%	-5.7%
	Comprehensive foot exam in past 12 months	>90%	11%	40%	281%	85%	73%	-13%
	Microalbuminuria screening in past 12 months	>65%	8.1%	84%	935%	85%	71%	-16%
Outcome	$A1C \le 7$	>60%	39%	51%	30%	56%	63%	12%
	$LDL \le 2.0 \text{ nmol/l}$	>65%	0.0%	39%	∞	39%	60%	52%
	BP ≤ 130/80	>55%	71%	65%	-8.1%	56%	83%	47%
	On ACEI or ARB	>60%	53%	69%	29%	79%	78%	-0.6%
	Patients have at least one self- management goal in 12 months	>70%	16%	34%	111%	69%	85%	23%

# 12.6



# Examples of success — Congestive heart failure readmissions

Organization: Ottawa Heart Institute

Aim and measure: Reduce 30-day readmissions for congestive heart failure

### Change ideas:

Ottawa Heart Institute aimed to improve transitions from acute to primary care and follow-up in the community, by doing the following:

- Provide patients with written discharge checklists and ensuring patients understand instructions for their care.
- Use electronic health records (EHRs) to transfer medical information to the primary care provider and/or the patient.
- Promote and teach patient self-management.
- Employ tele-homecare technology to monitor cardiac patients at home. Patients who meet pre-set criteria (e.g., having complex heart failure, acute arrhythmia or complex cardiac surgery) receive home monitors which can transmit information such as vital signs over the phone.
- Employ interactive voice response technology to monitor symptoms, adherence to treatment and lifestyle issues like smoking cessation. Patients receive automated phone calls at regular intervals with a pre-set series of questions. If the response indicates a worrisome problem, the call is flagged and the patient receives a call back from a member of the care team. If the response indicates a high-risk symptom (e.g. chest pain or shortness of breath), the system holds the patient on the line and connects him or her directly to medical staff.

### **Results:**

Ottawa Heart Institute has seen a 24% reduction in CHF readmission rates from 2008/09 to 2009/10.

### Ottawa Heart Institute — Congestive Heart Failure Readmission Rates 2008/09 - 2009/10



# Examples of success — Ventilator-associated pneumonia, central line infection and c. difficile infection

**Organizations:** Collingwood General and Marine Hospital, a large community hospital.

### Aims and measures:

Reduce:

- Ventilator-associated pneumonia (VAP) rate per 1,000 bed days on a ventilator;
- Central line infection (CLI) rate per 1,000 bed days on a central line; and,
- Rate of hospital-acquired C. difficile infection per 1,000 bed days.

### **Change Ideas:**

- Develop a strong culture of patient safety. In 2009, the hospital launched the "Single Safety System" initiative with a vision, "an inclusive and pervasive culture of safety supported by leadership and owned by all." Activities to nurture this culture include:
  - Creating an infection prevention and control (IP&C) service, with a coordinator who reports to senior management, and "Super Users" and "Unit Representatives" who provide front-line integration and organizational support to ensure concerns can be raised in a timely manner. These peer leaders are trained through IP&C and provide 'role modeling' and just-in-time information and training at the departmental level.
  - o Patient safety walkabouts, where senior leaders interact regularly with front-line staff on safety issues.
  - 'Safety Boards' in each department and monthly open forums called 'Safety Salutes', where information on infection rates and prevention tips are communicated to frontline staff and progress on reducing infections is celebrated.
- Embed, into the hospital's electronic medical record, a VAP screening tool to ensure consistent daily screening for VAP cases and a 'ventilator round sheet' to ensure consistent patient assessment and documentation. These tools help ensure all standard care practices are implemented (i.e. use of subglottic drainage endotracheal tubes, elevation of head of bed, monitoring of cuff pressures, daily sedation vacations and spontaneous breathing trials to ensure patients are extubated in the most timely manner possible).

- Develop standard protocols for eliminating CLI, and build a daily assessment screening tool into the electronic documentation system.
- Implement a Hand Hygiene Campaign, including the installation of additional hand rub stations using staff input for optimal placement to facilitate use according to workflow.
- Ensure early detection of C. Difficile cases, by emphasizing through IP&C the importance of early specimen collection and reducing lab turnaround time through a change in the hospital's external lab partner.
- Improve procedures for isolation. All patients with symptoms of diarrhea are isolated immediately. Isolation room signage was simplified to improve understanding and compliance. A patient/care provider video was implemented for education regarding precautions.
- Improve cleaning procedures. All isolation rooms are cleaned twice a day with Virox. Spray wands were removed from washrooms (as they can inadvertently splash contaminants throughout a room) and disposable toilet brushes were added. For patients using bedpans and commodes, gel liners are used for all cases with diarrhea.

### **Results:**

Collingwood's C. Difficile infection rate has declined steadily over the past two years. The hospital has also had zero cases of CLI since January 2009 and zero cases of VAP since October 2009.





# Examples of success — Alternative level of care

### **Organizations:**

12.8

- London Health Sciences Centre University Hospital site, a major teaching hospital.
- Halton Healthcare Services a very-high-volume community 3 site hospital.
- St. Thomas-Elgin General Hospital a high-volume community hospital.

### Aims and measures:

To reduce the percentage of acute care bed days that are designated as alternative level of care (ALC).

### **Change Ideas:**

London Health Sciences Centre - University Site

- Working with St Joseph's Health Care and SW CCAC to create a transitional care unit at Parkwood Hospital to provide short-term restoration care in an in-patient environment that enables individuals to return to the community.
- Working with CCCACs and home care organizations on initiatives such as:
  - o "Safe at Home", which adds additional service levels, allowing individuals to avoid hospital or be discharged earlier
  - o "Wait at Home", which provides personal care services to enable ALC patients who are waiting for LTC to wait at home

### Halton Healthcare Services

- Adopting the Home First strategy, including enhancing community resources, changing workflow processes, encouraging greater collaboration among healthcare team members, the hospital and community CCAC teams, and actively promoting a shift in culture regarding the care and flow of elderly patients.
- Introducing a Joint Discharge Operations Group, comprising the manager of social work and discharge planning, the hospital's chief operating officer and the community care access centre (CCAC); the group reviews every ALC patient and potential ALC patients, initiates early referrals to the CCAC, discusses options for discharge and ensures that each patient has a clear discharge plan.

### St. Thomas-Elgin General Hospital

- Partnering with organizations such as CCACs on quality improvement activities aimed at improving transitions from acute care to subsequent care destinations, including the "Home at Last" and "Wait at Home" programs, and the FLO Collaborative, which reduced and sustained a one-day reduction in length of stay in the acute medical unit
- Implementing the Medworxx utilization management system, which monitors each patient daily to evaluate the level of care requirements (acute care, rehabilitation, complex care, or home independently or with community supports)
- Using a bed optimization system or electronic bed board to obtain real-time data on bed availability throughout the organization
- Participating in the 4<sup>th</sup> wave of the Ministry of Health, Emergency Department Process Improvement Program (ED-PIP) to improve patient flow through the ED and reduce the wait time for admission to acute care
- Participating in the Rapid Emergency Assessment for Community Transition project to divert patients who are medically stable from hospital's emergency department, preventing hospital admissions and facilitating the patient's safe return to home or redirection to alternative settings

### **Results:**

All three hospitals have shown reduction in the percentage of acute care bed days that are designated as ALC. In particularly, Halton Healthcare Services Corporation — Oakville site and St. Thomas-Elgin General hospital had a relative improvement of over 50%.

### Percentage of acute care bed days that are designated as ALC



### Endnotes

- Derlet RW, Richards JR. Overcrowding in the nation's emergency departments: complex causes and disturbing effects. Annals of Emergency Medicine. 1999;35(1):63–68. Chan BTB, Schull NJ, Schultz SE. Emergency department services in Ontario. Institute for Clinical Evolution of Contents. 2007
- Evaluative Sciences, Toronto, 2001. National Ambulatory Care Reporting System Database, FY 2009/2010, calculated by Institute for Clinical Evaluative Sciences.

- National Aniobatory Cate Reporting System Database, PT 2009/2010, Calculated by institute for Clinical Evaluative Sciences. MOHLTC-LHIN performance agreement, local health system performance indicators, technical information. August 13, 2010. Johnson M, et al. Patients who leave the emergency department without being seen. J Emerg Nurs. 2009;35:105–108, Polevoi, SK, et al. Factors associated with patients who leave without being seen. Acad Emerg Med. 2005;12(3). Beveridge R, et al. Canadian Emergency Department Triage and Acuity Scale: implementation and guidelines. CJEM. 1999;(suppl 1):S2–28. These guidelines suggest that all patients should be seen within the following targets: resuscitation, immediate; emergent, 15 minutes; urgent, 30 minutes; semi-urgent, 60 minutes; non-urgent, 120 minutes. The following data (not published on the graph) for median times for 2009 were: emergent, 54 minutes; urgent, 54 minutes, semiurgent, 72 minutes, non-urgent, 72 minutes. For the first three categories, the results are worse than the guidelines, and this is the basis for the statement that wait times are too high. For non-urgent, the median is better than the target, but this target is meant to apply to all patients, not just to those at the 50th percentile. The figures in this report are median wait times, meaning that 50% of patients have wait times higher than the figures reported. Thus, the fact that the median non-urgent wait time is slightly below the target of 120 minutes means that almost half of these patients are waiting longer than desirable.
- desirable. OMonitor: 2010 Report on Ontario's Health System. Ontario Health Quality Council, Toronto,

- Official Constraints and the set of the set

- Forster AJ, Stiell I, Wells G, et al. I. he effect of hospital occupancy on emergency department length of stay and patient disposition. *Acad. Emerg Med.* 2003;10:127133.
   McKinney M, Watching the big board to reduce overcrowding. Electronic bed tracking systems can improve patient throughput, but staff buy-in is a must. Hosp Health Netw. October 2009;33(10):48–50.
   Vermeulen MJ, Ray JG, Bell C, Cayen B, Stukel TA, Schull MJ. Disequilibrium between admitted and discharged hospitalized patients affects emergency department length of stay. Annals of Emergency Medicine. 2009;54(6):794-804.
   Litvak E. Optimizing patient flow by managing its variability. In: JCAHO, from front office to front line: essential issues for health care leaders. Joint Commission Resources, Inc., Oakbrook Herace, IL, 2005. www.saludygestion.com/archives/B Joint%20Commission%20Resources%20Article%20 From%200Hec%20Tont%20Clores%200%20Forth%20Line.pdf: accessed January 14, 2011.
   Bond K, Ospina MB, Bitz S, Friesen C, Innes G, Yoon P, Curry G, Holroyd B, Rowe B. Emergency departments, Ottawa. CADTH. 2006. www.cadth.ca/index.phy/en/hta/reports-publications/ search/publication/621: accessed January 15, 2011.
   Det Ja D, Cantor J. Emergency department utilization and capacity. Research Synthesis Report. 2009;17. www.mic.org/ines/research/072109policysynthesis1 7.emergencyutilization.pdf: accessed January 16, 2011.
   Guarisco JS. Using computer technology in the automation of clinical and operating systems in emergency wedicine. The Ochsner Journal. 2001;3:57–69.
   Improving the efficiency O hospitalbased emergency care. Chapter 4 in: Hospitalbased emergency department torowding. Annals of Emergency Medicine. 2003;4:2(2): 173–180.
   Altinay E, A, Ardal S, Woodward GL, Schull MJ, Variation in emergency duptimeti visits for conditions they resonach/O72109policysynthesis17.emergencyutization.pdf; accessed January 16, 2011.
   Altin BR, M

- Heimler R, et al. Hospital readmission and morbidity following early newborn discharge. Clinical Pediatrics. 1998;37:609–616.
  BORN Ontario, FY 2009/2010.
  MOHLTC. Health Care Connect. Jan 25, 2011.
  www.health.gov.on.c.?en/ms/healthcareconnect/public/default.aspx; accessed March 7, 2011.
  Pickin M, O'Cathain A, Sampson FC, Dixon S. Evaluation of Advanced Access in the National Primary Care Collaborative. British Journal of General Practice 2004;54:334-340.
  Scottish Primary Care Collaborative (SPCC). Health Care Connect. Dec 2, 2008.
  www.scotland.gov.uk/Topics/Health/NHS Scotland/Delivery-Improvement/1835/1860; accessed March 22, 2011.
  Alberta AIM. AIM (Access Improvement Measures) measurement overview. 2009.
  www.albertaaim.ca/documents/learning\_session\_materials/20091120\_Measurement\_Overview.pdf; accessed February 11, 2011.
  Valenti WM, Bookhardt-Murray LJ. Advanced access scheduling boosts quality, productivity and revenue. Drug Benefit Trends. 2004;161(0):510, 513–514.
  Coutts J. Accelerating excellence report: engaging physicians to improve quality. Healthcare Q. 2010;13(3):23–25.
  Institute for Healthcare Improvement. Measure and understand supply and demand.

- <sup>9</sup> Valenti WM, Bookhardt-Murray LJ. Advanced access scheduling boosts quality, productivity and revenue. Drug Benefi Trends. 2004;1610:510, 513–514.
   <sup>9</sup> Cotts J. Accelerating excellence report: engaging physicians to improve quality. Healthcare Q. 2010;13(3):23–25.
   <sup>9</sup> Institute for Healthcare Improvement. Measure and understand supply and demand. www.ihi.org/HI/Topics/OfficePractices/Access/Andegs/ MeasureandUnderstandSupplyandDemand.htm; accessed January 15, 2010.
   <sup>9</sup> Ontario Health Ouality Council. Improvement Guide: Module 1 Access. 2009. www.ohgc.ca/ dpfs/access.pdf; accessed January 15, 2011.
   <sup>9</sup> Ontario Health Ouality Council. Success study: primary care at Athens District Family Health Team. In: Quality Monitor. 2010 Report on Ontario's Health System. www.ohgc.ca/dfs/2010\_report. = english.pdf: accessed January 20, 2011.
   <sup>9</sup> Ontario Health Ouality Council. Success / Connect-L-Ontario -+Phase-HI-Projects. htm; accessed Jonuary 20, 2011.
   <sup>9</sup> Ornario Health Ouality Coutber 28, 2010.
   <sup>9</sup> Willis D. Making every minute count: tools to improve office efficiency. Family Practice Management. April 2005. www.internetgroup.ca/clientmet\_new/docs/Making%20Every%20 Minute%20Count.pdf: accessed January 20, 2011.
   <sup>9</sup> Willis D. Making every minute count: tools to improve office efficiency. Family Practice Management. April 2005. www.internetgroup.ca/clientmet\_new/docs/Making%20Every%20 Minute%20Every%20 Minute%20Count.pdf: accessed January 20, 2011.
   <sup>9</sup> Wang SJ. Middleton B, Prosser LA, Bardon CG, Spurr CD, Carchidi PJ, Kittler AF, Goldszer RC, Farichil DG, Sussman AJ, Kuperman GJ, Bates DW. A cost-benefit analysis of electronic medical records in primary care. Am J Med. 2003;114(5):397–403.
   <sup>1</sup> Langley J, Beasley C. Health information rechnology for improving quality of care in primary care settings. Prepared by the Institute for Healthcare Improvemary Cares sto primary care.

- March 22, 2011. Canadian Health Services Research Foundation. Picking Up the Pace. http://www.chsrf.ca/Libraries/Picking\_up\_the\_pace\_files/Mary\_Fleming.sflb.ashx; accessed March 22, 2011. MOHLTC. OMA and MOHLTC Agreement. www.health.gov.on.ca/english/providers/physiciansa/ docs/oma\_agreement.pdf; accessed March 22, 2011. MOHLTC. Creating More Opportunities for Nurses. www.health.gov.on.ca/en/news/release/2010/ may/nr\_20100511.aspx; accessed March 22, 2011. MOHLTC. HealthForceOntario Northern and Rural Recruitment and Retention Initiative Guidelines. Mown bealth gov\_on ca/english/providers/forgare.

- MOHLTC. HealthForceOntario Northern and Rural Recruitment and Retention Initiative Guidelines. www.health.gov.on.ca/english/providers/program/uap/guidelines/nrrri\_guidelines.html; accessed March 22, 2011. MOHLTC. Ontario Wait Times. Nov 1, 2010. www.health.gov.on.ca/en/public/programs/ waittimes; accessed November 1, 2010. Chan BTB and Austin PC. Patient, physician and community factors affecting referrals to specialists in Ontario. Canada: a population-based, multi-level modeling approach. Medical Care. 2003;41(4):500–511. The medical Care. The second Statement of th

- 2003;41(4):500–511. Williams JI, Llewellyn-Thomas H, Arshlnoff R, Young N, Naylor D. The burden of waiting for hip and knee replacements in Ontario. Journal of Evaluation in Clinical Practice. 1997;3(1):54–68. Coyte PC, Wright JG, Hawker GA, Bombardier C, Dittus RS, Paul JE, Freund DE, Elsa H. Waiting times for knee-replacement surgery in the United States and Ontario. The New England Journal of Medicine. October 20, 1994;331:1068–1071. Hodge W, Horsley T, Albiani D, Baryla J, et al. The consequences of waiting for cataract surgery: a systematic review. CMAJ. 2007;176(9):1285–1290.

- 69
- FY 2009/10 calculated by Institute for Clinical Evaluative Sciences. Morgan CD, Sykora K, Naylor CD, the Steering Committee of the Cardiac Care Network of Ontario. Analysis of deaths while waiting for cardiac surgery among 29,293 consecutive patients in Ontario, Canada. Heart. 1998;79:345–349. Cancer Care Ontario and Wait Times Information System, calculated by OHQC for FY 2009/2010.
- 71
- Cancer Care Ontario and Wait Times Information System, calculated by OHQC for FY 2009/2010. Cancer Care Ontario. www.cancercare.on.ca/ocs/csurv/stats. Jensena AR, Nellemanne HM, Overgaarda J. Tumor progression in waiting time for radiotherapy in head and neck cancer. Radiotherapy and Oncology. 2007;84:5–10. FY 2009/10 calculated by Institute for Clinical Evaluative Sciences. Based on data from the Institute for Clinical Evaluative Sciences, in 2002/03, there were 7,861 CT and 1,516 MRI scans per 100,000 adults; by 2009/10, there were 13,619 CT and 4,383 MRI scans per 100,000 adults. Wright CJ, Chambers GK, Robens-Paradise Y. Evaluation of indications for and outcomes of elective surgery. CMAJ. 2002;167:461–466. Lehnert, Bree. Analysis of appropriateness of outpatient CT and MRI referred from primary care clinics at an academic medical center: how critical is the need for improved decision support? 72
- 74

- Lehnert, Bree. Analysis of appropriateness of outpatient CT and MRI referred from primary care clinics at an academic medical center: how critical is the need for improved decision support? J Am Coll Radiol. 2010;7:192–197. Brenner DJ, Hall EJ. Computed tomography: an increasing source of radiation exposure. NEJM. 2007;357(22):2277–2284. Tsai PF, Tak S. Disease-specific pain measures for osteoarthritis of the knee or hip. Geriatric Nursing. 2003;24(2):106–109. WOMAC 3.1 Index: Knee and Hip Osteoarthritis Index. 2010. www.womac.org/womac/index.htm; accessed January 15, 2011. Cross WW III, Saleh KJ, Wilt TJ, Kane RL. Agreement about indications for total knee arthroplasty. Clin Orthop Relat Res. 2006;446:34–39. Quintana JM, et al. Appropriateness of total hip joint replacement. Int. J. Qual. Health Care. 2005;17:315–321.
- 77 78
- 79
- 81
- Amesbury EC, Grossberg AL, Hong DM, Miller KM. Functional visual outcomes of cataract surgery in patients with 20/20 or better preoperative visual acuity. J Cataract Refract Surg. 2009;35:1505–1508. 82
- 83
- 84
- 85
- 87
- 89
- 90
- 92
- 93
- 94

- 97
- 99
- 100
- 102 103

- 2005;17:315–321.
   Arnesbury EC, Grossberg AL, Hong DM, Miller KM. Functional visual outcomes of cataract surgery in patients with 20/20 or better preoperative visual acuity. J Cataract Refract Surg. 2009;35:1050–1508.
   Steinberg EP, Tielsch JM, Schein OD, Javitt JC, Sharkey P, Cassard SD, Legro MW, Diener-West M, Bass EB, Damiano AM, Steinwachs DM, Sommer A. The VF-14: an index of functional impairment in patients with cataract. Arch Ophthalmol. 1994;112:630–638.
   Lehnert, Bree, Analysis of appropriateness of outpatient CT and RMI referred from primary care clinics at an academic medical center: how critical is the need for improved decision support? J Am Coll Radiol. 2010;7:192–197.
   MRJCT decision support tool for referring physicians. 2009. www.mrictdecisionsupporthol.ca/ OEBI/IndicationSearch/Help.Introductions accessed January 15, 2011.
   Noseworthy TW, McGurran JJ, Hadorn DC and the steering committee of the Western Canada Waiting List Project. Lists for cataract surgery: validation of the Western Canada Waiting List Project. Lists for cataract surgery: validation of the Western Canada Waiting List Project. Lists for cataract surgery: validation of the Western Canada Waiting List Project. Lists for cataract and pain and knee replacement. Obesity. December 2010;18(12):2362–2366.
   ML Brandeau, F Sainfort, WP Pierskalla. Operations research and health care. Kluwer Academic Publishers, Norwell MA, 2004.
   Champlan Local Health Integration Network. Central intake project for dignostic imaging. June 2008. www.champlainlini.on.ca/Page.aspx?id=1700; accessed October 28, 2010.
   Community of Chamm-Kent. Connect Ontario: implement an online scheduling system. June 2009. www.chamment.ec.docunct-Hand-Andimithtation/municipal-Health Integrations (Accessed January 15, 2011.
   Stern KS. Improving queue management through operational research and system redesign. cpm.org/documents/43276.en.pdf; accessed Ja

- available choice. If that is not their first of second choice, they can request to be transferred when a bed becomes available there.
   CCAC, 2009, www.ccac-ont.ca.
   Based on data from the Ministry of Health and Long-term Care, the number of people admitted to LTC from hospital decreased from 1,445 in January-March 2009 to 1,176. The number of people admitted to LTC from their home increased from 2,101 to 2,430 in the same time period.
   CHQI. Home First: Maximizing use of LHIN investments while creating better outcomes for seniors and reducing ALC. www.chqi.ca/Assets/FLO%20Spread/FLO%20Spread%20-%20Ql%20Tools/ HomeFirstMHCCACHaltonHealthcare.pdf; accessed 9 March 2011.
   MOHLTC. Refusals of Offers of Long-Stay Admission to LTC Homes. Sep, 2006. www.health.gov.on.ca/english/providers/pub/manuals/ccac/cspm\_sec\_12/12-11.html; accessed March 1, 2011.
   Canadian Home Care Association. Home care: meeting the needs of an aging population. 2008. www.choinecare.ca/media.php?mid=1914; accessed January 16, 2011.
   Chapel INL, Reid RC, Dow E. Respite reconsidered: a typology of meanings based on the caregiver's point of view. Journal of Aging Studies. 2001;15(2):201–216.
   Health Transition Fund, Health Canada. Final report of the national evaluation of the cost-effectiveness of home care. 2002. www.health.gov.on.clop. 2002.
   Ministry of Health and Long-Term Care. Assisted Living Services for High Risk Seniors Policy, 2011.
   Momitor: 2009 Report on Ontario's Health System. Ontario Health Quality Council, Toronto, 2009; page 31.
   Ministry of Health and Long-Term Care. Sussed Living Services for High Risk Seniors Policy, 2011.
   Momitor: 2009 Report on Ontario's Health System. Ontario Health Quality Council, Toronto, 2009; page 31.
- 2009; page 31.
   <sup>115</sup> MOHLTC. Senior's Care: Supportive Housing. Nov 23, 2010. www.health.gov.on.ca/english/ public/program/ttc/13\_housing.html; accessed January 11, 2011.

- <sup>116</sup> MOHLTC. Seniors' Care: Comparing Residential Care Options. Nov 23, 2010. http://www.health.gov.on.ca/english/public/program/ttc/16\_options.html; accessed January 11, 2011.
   <sup>117</sup> Ministry of Health and Long-Term Care. Senior's care: supportive housing related housing options. www.health.gov.on.ca/english/public/program/ttc/13\_housing.html#6; accessed
- January 11, 2011. <sup>118</sup> Korteben P. Rehabilitation for hospital-associated deconditioning. Am J Phys Med Rehabil. 2009 Jan;88(1):66-77.
- <sup>2009</sup> Jan;86(1):56-77.
   <sup>119</sup> Regional Geriatrics Program of Ontario. Senior Friendly Hospital Strategy. Available at: http:// rgps.on.ca/key\_elements\_senior\_friendly; accessed March 21, 2011.
   <sup>120</sup> Hirdes JP, Poss JW, Curtin-Felegdi N. The Method for Assigning Priority Levels (MAPLe): a new decision-support system for allocating home care resources. BMC Medicine. 2008;6:9.
   <sup>121</sup> South West LHIN. Aging at Home Strategy. www.southwestihin.on.ca/Page.aspx?id=3174; accessed November 20, 2010.
   <sup>222</sup> Politic for allocating for allocating individual with second page.aspx?id=3174;

- <sup>222</sup> Building a better system: caring for older individuals with aggressive behavioural issues. 2007. www.southwestlhin.on.ca/uploadedFiles/Public\_Community/Aging\_at\_Home/Building%20a%20 better%20system%20%20EP%20report%20final%20draft%20March%207%202007.pdf;
- better%20system%20%20ÉP%20repori%20final%20draft%20March%207%202007.pdf;
   accessed November 20, 2010.
   Toronto Central LHIN. Aging at Home. www.torontocentrallhin.on.ca/uploadedFiles/Public\_Community/Aging\_at\_Home,/FINAL%20Call%20for%20proposals%20FINAL%2004-NOV-09.pdf;
   accessed March 9, 2011.
   Ontario's local health integration network Aging at Home Strategy. www.lhins.on.ca/page.aspx?id=880; accessed September 28, 2010.
   Toronto Central LHIN. Aging at Home. www.torontocentrallhin.on.ca/Page.aspx?id=888; accessed September 28, 2010.
   Toronto Central LHIN. Aging at Home. www.torontocentrallhin.on.ca/Page.aspx?id=2838; accessed January 11, 2011.
   Ontario Provincial Government. Modernizing Long-Term Care Beds in Ontario. Feb 5, 2010. http://news.ontario.ca/mohltc/en/2010/02/modernizing-long-term-care-beds-in-ontario.html; accessed March 1, 2011.
   MOHITC. Increasing Access to Modern Long-Term Care Homes. Feb 3, 2010. www.bealth.org.

- accessed March 1, 2011.
   7 MOHLTC. Increasing Access to Modern Long-Term Care Homes. Feb 3, 2010. www.health.gov. on.ca/en/news/release/2010/feb/nr\_20100203.aspx; accessed September 28, 2010.
   MOHLTC. Increasing Access to Modern Long-Term Care Homes. Feb 3, 2010. www.health.gov. on.ca/en/news/release/2010/feb/nr\_2010203.aspx; accessed September 28, 2010.
   MOHLTC. Increasing Access to Modern Long-Term Care Homes. Feb 3, 2010. www.health.gov. on.ca/en/news/release/2010/feb/nr\_2010203.aspx; accessed September 28, 2010.
   MOHLTC. Ontario Strengthens Home Care Services. Dec 15, 2008. www.health.gov.on.ca/english/ media/news\_releases/archives/nr\_08/dec/nr\_20081215\_2.html; accessed October 18, 2010.
   Tracey J. Zelmer J. Merid M, Boruvka A. Variation in heart attack mortality in Canada. Healthcare Quarterly. 2006;9(4):23-25.
   Formardel N, et al. B blockdo. after proceeding infanction; extenditic pricing and moto procession.

- Quarteriy. 2006;9(4):23–25.
   <sup>131</sup> Freemantle N, et al. B blockade after myocardial infarction: systematic review and meta regression analysis. BMJ. 1999;318:1730–1736.
   <sup>132</sup> Piehn JF, et al. Reduction of stroke incidence after myocardial infarction with pravastatin: CARE study. Circulation. 1999;99:216–223.
   <sup>133</sup> The Heart Outcomes Prevention Evaluation Study Investigation. Effects of an angiotensin-converting enzyme inhibitor, ramipril, on cardiovascular events in highrisk patients. 2000;342:145–153.
   <sup>134</sup> Pfeffer MA, McMurray JJ, Velazquez EJ, Rouleau JL et al. Valsartan, captopril, or both in myocardial infarction complicated by heart failure, left ventricular dysfunction, or both. N Engl J Med. 2003;349(20):1893-906.
   <sup>135</sup> Vebene L, Cripther E J, Mandaoi M, Moineddin P, Llochur PE. Targacet analysis of soute presential.

- 2003;349(20):1893-906.
   <sup>135</sup> Vrbova L, Crighton EJ, Mamdani M, Moineddin R, Upshur RE. Temporal analysis of acute myocardial infarction in Ontario, Canada. Can J Cardiol. 2005;21(10):841-845.
   <sup>136</sup> Garg R, Yusuf S. Overview of randomized trials of angiotensin-converting enzyme inhibitors on mortality and morbidity in patients with heart failure. Collaborative group on ACE inhibitor trials. JAMA. 1995;273:1450–1456.
   <sup>137</sup> Packer M, Coats AJ, Fowler MB, et al; Carvedilol Prospective Randomized Cumulative Survival Study Group. Effect of carvedilol on survival in severe chronic heart failure. N Engl J Med 2001;344:1651-8.
   <sup>38</sup> Studio gurrent these with entertial definition gaspare for EOW to COW of heart failure.
- 2001;344:1651-8.
   <sup>138</sup> Studies suggest those with systolic dysfunction account for 50% to 60% of heart failure patients. See these two studies: Bursi F, Weston SA, Redfield MM, et al. Systolic and diastolic heart failure in the community. JAMA 2006;296:2209-16. Diller PM, Smucker DR, David B, Graham RJ. Congestive Heart Failure Due to Diastolic or Systolic Dysfunction Frequency and Patient Characteristics in an Ambulatory Setting. Arch Fam Med. 1999;8:414-420.
   <sup>139</sup> Arnold JMO, Liu P, Demers C, et al. Canadian Cardiovascular Society consensus conference recommendations on heart failure, 2006: diagnosis and management. Can J Cardiol. 2006;22(1):23–45. The guidelines point out that there are studies suggesting better outcomes for diastolic dysfunction patients who take these drugs but more data to confirm this benefit are needed before their use is routinely recommended. The guidelines specify that these drugs shuld be "considered", especially if there are other good reasons to prescribe them (e.g. for hypertension or diabetes). Boohzelly Stanction patients who take these drugs but more data to confirm this benefit are needed before their use is routinely recommended. The guidelines specify that these drugs should be "considered", especially if there are other good reasons to prescribe them (e.g. for hypertension or diabetes).
   JMO Arnold, JG Howlett, P Dorian, et al. Canadian Cardiovascular Society Consensus Conference recommendations on heart failure update 2007: Prevention, management during intercurrent illness or acute decompensation, and use of biomarkers. Can J Cardiol 2007;23(1):21:45.
   Registered Persons Database, Discharge Abstract Database, FY 2009/2010, calculated by Institute for Clinical Evaluative Sciences.
   Sacco RL, Adams R, Albers G, Alberts MJ, et al. Guidelines for prevention of stroke in patients with ischemic stroke or transient ischemic attack. Stroke. 2006;37:577.
   Johansen HL, Wielgoz AT, Nguyen K, Fry RN. Incidence, comorbidity, case fatality and readmission of hospitalized stroke patients in Canada. Can J Cardiol. 2006;22(1):65–71. According to this source, the incidence of new ischemic stroke is 12 per 10,000. Given that Ontario's population is 13 million, there are approximately 16,000 new ischemic stroke is torke prever.
   Adams HP Jr, del Zopo G, Alberts MJ, Batt DL, et al. Guidelines for the early management of adults with ischemic. Stroke. Stroke. 2007;38(5):655–711.
   Friksen, H., A. R. Sæther, H. L. Løver, S. Vangen, R. Hjetland, H. Lundmark, and P. Aavitsland. Infections after Caesarean Sections. Tidskr Mr Na Legeforen 2009;129(7-26): 618-22.
   Hansen, A. K., K. Wisborg, N. Uldbjerg, and T. B. Henriksen. Risk of Respiratory Morbidity in Term Infants Delivered by Elective Caesarean Section: Cohort Study. BMJ 2008;336(7635): 85-87.
   BORN Ontario, FY 2009/10.
   Tu JV, Khalid L, Donovan LR, Ko DT, et al. Indicators of quality of care for patients with acute myocardial infarction. GMAJ. 200

- Jany et 2010,1711,20-33. Tu JV, Donovan LR, Lee DS, Wang JT, Austin PC, Alter DA, Ko DT. Effectiveness of public report cards for improving the quality of cardiac care the EFFECT study: a randomized trial. JAMA. 2009;302(21):2330–2337.

- <sup>158</sup> Jamtvedt G, Young JM, Kristoffersen DT, O'Brien MA, Oxman AD. Audit and feedback: effects on professional practice and health care outcomes. Cochrane Database of Systematic Reviews. 2006, Issue 2. Art. No. CD000259.
- 2006, Issue 2. Art. No. CD000259.
   <sup>159</sup> Wong JD, Bajcar JM, Wong GG, Alibhai SM, Huh JH, Cesta A, Pond GR, Fernandes OA. Medication reconciliation at hospital discharge: evaluating discrepancies. Ann Pharmacother. October 2008;42(10):1373–1379.
   <sup>160</sup> Vira T, Colquhoun M, Etchells E. Reconcilable differences: correcting medication errors at hospital admission and discharge. Qual Saf Health Care. April 2006;15(2):122–126.
   <sup>161</sup> Cua YM, Kripalani S. Medication use in the transition from hospital to home. Ann Acad Med Singapore. February 2008;37(2):136.
   <sup>162</sup> Clipter B. Anderson SE. Delifering K Lacut Recomputed and the transition from hospital to home. Ann Acad Med Singapore. February 2008;37(2):136.

- <sup>161</sup> Cua YM, Kripalani S. Medication use in the transition form hospital to further and the second structure of the second structure of
- January 17, 2011. <sup>173</sup> Hill MD, Barber PA, Demchuk AM, Sevick RJ, Newcommon NJ, Green T, Buchan AM. Building a "brain attack" team to administer thrombolytic therapy for acute ischemic stroke. CMAJ.

- 2010. strokeBestpractices.ca/wp.content/uploads/2010/12/2010\_BP\_ENG.pdf; accessed "January 17, 2011.
   17 Hill MD, Barber PA, Demchuk AM, Sevick RJ, Newcommon NJ, Green T, Buchan AM. Building a "brain attack" team to administer thrombolytic therapy for acute ischemic stroke. CMAJ. May 30, 2000;162(11):1589–1593.
   17 Finger C. Caesarean section agining acceptance. CMAJ. March 2, 2004;170(5):775.
   17 Liu S, Liston RM, Joseph KS, Heaman M, Sauve R, Kramer MS, Maternal Health Study Group of the Canadian Perinatal Survellance System. Maternal mortality and severe morbidity associated with lowrisk planned cesarean delivery versus planned vaginal delivery at term. CMAJ. February 13, 2007;176(4):455–60.
   17 Canadian Institute for Health Information. Giving birth in Canada: the costs. 2006. secure.chi.ca/chiweb/products/Costs. Report 0.6. Eng. pdf; accessed January 10, 2011.
   18 New Jersey Maternity Care Worst to First 2010. New Jersey Hospitals. 2010. www.cmpa-acpm.ca/cmapad04/docs/membership/fees/2011.ca/esarean section secsof March 4, 2011.
   19 Ontario Women's Health Council. Caesarean section best practices project. 2002. www.ccnpa-acpm.ca/cmapad04/docs/membership/fees/2011ca/e.pdf; accessed March 4, 2011.
   19 Ontario Medical Protective Association Fee Schedule for 2011. Available at: www.cmpa-acpm.ca/cmapad04/docs/membership/fees/2011ca/e.pdf; accessed March 4, 2011.
   19 Myers SA, Gleicher N. A successful program to lower cesarean-section rates. N Engl J Med. 1988 Dec 3:110(23):1511-6.
   19 Ark JS, Robinson JN, Norwitz ER, Rotational forceps: Should these procedures be abandoned? Seminars in Perinatology 2003: 27(1):112-120.
   19 SOGC Chiccial practice guideline: gradieng if? Simulation in the training of obstetricians and grynaecologists. Curr Opin Obstet Gynaecol. 2005 Dcc;17(6):557-61.
   19 Ark JS, Robinson JN, Norwitz ER, Rotational forceps: Should these procedures be aban
- Litzelman, et al. Reduction of lower extremity clinical abnormalities in patients with non-insulin-dependent diabetes mellitus: a randomized, controlled trial. Ann Intern Med. 1993;119(1):36–41. McCabe, et al. Evaluation of a diabetic foot screening and protection programme. Diabetes Med. 1998;15(1):80–84. 200
- 201
- 203
- 1998;15(1):80-84. Registered Persons Database, Discharge Abstract Database, Ontario Diabetes Database, FY 2009/2010, calculated by Institute for Clinical Evaluative Sciences. Registered Persons Database, Discharge Abstract Database, FY 2009/2010, calculated by Institute for Clinical Evaluative Sciences. Ministry of Health and Long-Term Care. Preventing and managing chronic disease: Ontario's framework. May 2007;1–50. Canadian Diabetes Association. Management. 2011. http://www.diabetes.ca/diabetes-and-you/ living/management/; accessed March 4, 2011 204

- <sup>205</sup> Public Health Agency of Canada. How Can I Reduce my Risk of Developing Heart Disease and Having a Heart Attack? Jan 29, 2009. http://www.phac-aspc.gc.ca/cd-mc/cvd-mcv/ hd\_reduce-md\_reduire-eng.php; accessed March 4, 2011
   <sup>206</sup> The Heart Outcomes Prevention Evaluation Study Investigation. Effects of an angiotensin-convert-ing-enzyme inhibitor, ramipril, on cardiovascular events in high-risk patients. 2000; 342: 145-153.
   <sup>207</sup> Stratton, IM et al. Association of glycaemia with macrovascular and microvascular complications of type 2 diabetes (UKPDS 35): prospective observational study. BMJ. 2000; 321: 405-411.
   <sup>208</sup> Adler, Al et al. Association of systolic blood pressure with microvascular and macrovascular complications of type 2 diabetes (UKPDS 36): prospective observational study. BMJ. 2000; 321: 412-419.
   <sup>209</sup> Pyorala, K. Pedersen IR et al. Cholesterol lowering with simvastatin improves prognosis

- complications of type 2 diabetés (UKPDS 36): prospective observational study. BMJ. 2000; 321: 412-419.
   <sup>209</sup> Pyorala, K, Pedersen TR et al. Cholesterol lowering with simvastatin improves prognosis of diabetic patients with coronary heart disease. A subgroup analysis of the Scandinavian Simvastatin Survivals Study (4s). Diabetes Care; 1997:201 (4).
   <sup>210</sup> Lorig KR, et al. Evidence suggesting that a chronic disease self-management program can improve health status while reducing hospitalization. Medical Care. 1999;73(1):5–14.
   <sup>211</sup> Cholesterol Treatment Trialists (CTI) Collaborators, Kearney PM, Blackwell L, Collins R, Keech A, Simes J, Peto R, Armitage J, Baigent C. Effcacy of cholesterol-lowering therapy in 18,686 people with diabetes in 14 randomised trials of statins: a meta-analysis. Lancet. January 12, 2008;371(9607):117–125.
   <sup>212</sup> Canadian Diabetes Association Clinical Practice Guidelines Expert Committee. Canadian Diabetes Association. Clinical practice guidelines for the prevention and management of diabetes in Canada. Can J Diabetes. 2008;32(suppl 1):S109. The Canadian Diabetes Association's guidelines say diabetes. Findings from the National Health and Nutrition Examination Survey, 1999 to 2002. J Gen Intern Med. April 2006;21(4):315–319. Canadian guidelines suggest that all diabetes patients thigh risk of cardiovascular disease, high blood pressure, or protein in their urine should be on an ACE/ARB (preceding reference, page S105, S116, S130). This study suggests that 92% to 100% of diabetes patients file and rol Organizational Factors. Ann Fam Med. 2009 July; 7(4): 309–318. doi: 10.1370/afm.982.
   <sup>215</sup> Project for an Ontario Women's Health Evidence-Based Report. Cardiovascular Disease. www.powerstudy.ca/the-power-report/the-power-report-claradioascular-disease; accessed March 2, 2011.
   <sup>216</sup> Davis D, Goldman J, Palda VA. Canadian Medical Association handbook on clinical practice guidelines. Chawa: Canadian Medical

- www.powerstudy.ca/thepower-report/the-power-report-volume-1/cardiovascular-disease; accessed March 2, 2011.
   <sup>216</sup> Davis D, Goldman J, Palda VA. Canadian Medical Association handbook on clinical practice guidelines. Ottawa: Canadian Medical Association, 2007. www.cma.ca/handbook.pdf; accessed Januar J, 2011.
   <sup>217</sup> Grimshaw JM, Thomas RE, MacLennan G, et al. Effectiveness and efficiency of guideline dissemination and implementation strategies. Health Technol Assess. 2004;8(6).
   <sup>218</sup> Dobbins M, Hanna SE, Ciliska D, Manske S, Cameron R, Mercer SL, O'Mara L, DeCorby K, Robeson P. A randomized controlled trial evaluating the impact of knowledge translation and exchange strategies. Implement Sci. September 23, 2009;4:61.
   <sup>219</sup> Grol R and Wensing M. What drives change? Barriers to and incentives for achieving evidence-based practice. Med J Aust. 2004;180(suppl 6):S57–S60.
   <sup>220</sup> Patasi B, Conway JR. Enhancing diabetes care in family practice: a flow sheet. Can Fam Physician. September 2008;54(9):1237–1238.
   <sup>221</sup> Canadian Diabetes Association. Sample Diabetes Patient Care Flow Sheet for Adults. 2008. www.diabets.ca/documents/for-professionals/Clinical-flow-sheet.pdf
   <sup>222</sup> Sequist TD, Gandhi TK, Karson AS, et al. A randomized trial of electronic clinical reminders to improve quality of care for diabetes and coronary artery disease. J Am Med Inform Assoc. July/August 2005;12(4):431–437.
   <sup>224</sup> Weber V, Bloom F, Pierdon S, Wood C. Employing the electronic health record to improve diabetes: care an untifaceted intervention in an integrated delivery system. J Gen Intern Med. 2008;23(4):379–382.
   <sup>225</sup> Hicks LS, Sequist TD, Ayanian JZ, Shaykevich S, Fairchild DG, Orav J, Bates DW. Impact of computating device on blood preserve management and cortex is a robusting device on blood preserve management and cortex is a robusting device on blood preserve management and cortex is a robusting device on blood

- 2008;23(4):379–382.
   <sup>225</sup> Hicks LS, Sequist TD, Ayanian JZ, Shaykevich S, Fairchild DG, Orav J, Bates DW. Impact of computerized decision support on blood pressure management and control: a randomized controlled trial. J Gen Int Med. 2008;23(4):429–441.
   <sup>226</sup> Hicks LS, Sequist TD, Ayanian JZ, Shaykevich S, Fairchild DG, Orav J, Bates DW. Impact of computerized decision support on blood pressure management and control: a randomized ontrolled trial. J Gen Int Med. 2008;23(4):429–441.
   <sup>227</sup> Jamtvedt G, Young JM, Kristoffersen DT, O'Brien MA, Oxman AD. Audit and feedback: effects on professional practice and health care outcomes. Cochrane Database of Systematic Reviews. 2006;2: 04+ Me. C0000250.
- On professional practice and realitic care outcomes. Growing in contrasts of systematic relations of systematic relations of the systematic relations of the
- 2008;23(4):379–382. Flodgren G, Parmelli E, Doumit G, Gattellari M, O'Brien MA, Grimshaw J, Eccles MP. Local opinion
- <sup>230</sup> Flodgrein G, Parmelin E, Dournit G, Gattellari M, O'Brien MA, Grintshav J, Eccles MP. Local opinio leaders: effects on professional practice and health care outcomes. Cochrane Database of Systematic Reviews. 2007;1. Art. No. CD000125.
   <sup>231</sup> Majumdar SR, Tsuyuki RT, McAlister FA. Impact of opinion leader-endorsed evidence summaries on the quality of prescribing for patients with cardiovascular disease: a randomized controlled trial. Am Heart J. 2007;153:22.e1–8.
   <sup>232</sup> Davis D, Goldman J, Palda VA. Canadian Medical Association handbook on clinical practice

- bite Qdman J, Palda VA. Canadian Medical Association backse. a randomized continued trial. Am Heart J. 2007;153:22.e1-8.
   Davis D, Goldman J, Palda VA. Canadian Medical Association handbook on clinical practice guidelines. Ottawa: Canadian Medical Association, 2007. www.cma.ca/handbook.pdf; accessed January 1, 2011.
   McAlister FA, Stewart S, Ferrua S, McMurray JJV. Multidisciplinary strategies for the management of heart failure patients at high risk for admission: a systematic review of randomized trials. J Am Coll Cardiol. 2004;44:810–819.
   Health Council of Canada. Teams in action: Ontario. 2009. www.healthcouncilcanada.ca/docs/ rpts/2009/TIA\_Ontario.pdf; accessed October 4, 2010.
   Liddy C, Dusseault JJ, Dahrouge S, Hogg W, Lemelin J, Humbert J. Telehomecare for patients with multiple chronic illnesses: pilot study. Can Fam Physician. January 2008;54(1):58–65. Erratum in: Can Fam Physician. March 2008;54(3):351.
   Tran K, Polisena J, Coyle D, Coyle K, Kluge EH W, Cimon K, McGill S, Noorani H, Palmer K, Scott R. Home telehealth for chronic disease management. Technology report number 113. Ottawa: Canadian Agency for Drugs and Technologies in Health. 2008. www.cadth.ca/media/pdf/ 00475\_Home\_Telehealth\_to\_e.pdf; accessed January 17, 2011.
   Vaard Hu, Drzazen HH. Clinical assessment of heart failure: utility of symptoms, signs, and daily weights. Heart Fail Clin. April 2009;5(2):149–160.
   Schillinger D, Piette J, Grumbach K, et al. Closing the loop: physician communication with diabetic patients who have low health literacy. Arch Intern Med. 2003;163:83–90.
   Williams MW, Davis TC, Parker RM, Weiss BD. The role of health literacy in patient-physician communication with diabetic patients who have low health literacy. Arch Intern Med. 2003;163:83–90.
   Williams MW, Davis TC, Parker RM, Weiss BD. The role of health literacy in patient-physician communication fame Med. 2002;34:333–349.
   Fl

- <sup>244</sup> Comprehensive Vascular Disease Prevention and Management Initiative. About CVDPMI. 2011.
   <sup>245</sup> MOHLTC. Expanding Access To Affordable Drugs. Apr 7, 2010. http://www.health.gov.on.ca/en/news/release/2010/apr/nr\_20100407.aspx; accessed March 24, 2011.
   <sup>246</sup> Heart and Stroke Foundation. My Blood Pressure Action Plan. 2011.
   <sup>247</sup> Heart and Stroke Foundation. My Blood Pressure Action Plan. 2011.
   <sup>248</sup> Heart and Stroke Foundation. My Blood Pressure Action Plan. 2011.
   <sup>247</sup> Heart and Stroke Foundation. My Blood Pressure Action Plan. 2011.
   <sup>248</sup> Heart and Stroke Foundation of Ontario. Hypertension Management Initiative. 2011.
   <sup>247</sup> Heart and Stroke Foundation of Ontario. Hypertension Management Initiative. 2011.
   <sup>248</sup> Forster AJ, Murff HJ, Peterson JF, Gandhi TK, Bates DW. The incidence and severity of adverse events affecting patients after discharge from the hospital. Ann Intern Med. 2003;138:161–167.
   <sup>249</sup> Forster AJ, Murff HJ, Peterson JF, Gandhi TK, Bates DW. Adverse drug events occurring following hospital discharge. J Gen Intern Med. 2002;28:737–742.
   <sup>250</sup> Lorig KR, et al. Evidence suggesting that a chronic discarge summaries: not reaching follow-up physicians. C Fam Phys. 2002;28:737–742.
   <sup>251</sup> Lorig KR, et al. Evidence suggesting that a chronic disease self-management program can improve health status while reducing hospitalization. Medical Care. 1999;73(1):5–14.
   <sup>252</sup> Registered Persons Database, Discharge Abstract Database, FY 2009/2010, calculated by Institute for Clinical Evaluative Sciences.
   <sup>254</sup> Weiner LL Appresite IL Appresite negative readment.

- Institute for Clinical Evaluative Sciences 253
- Weinat JJ. Hospital readmissions under the spotlight. Journal of Healthcare Managemen 2010;55(4):252–264. 254
- Wijeysundera H, Nallamothu B, Krumholz H, Tu J, Ko D. Meta-analysis: effects of percutaneous coronary intervention versus medical therapy on angina relief. Ann Intern Med. 2010;152:370–379.
   TW Haywood, HM Kravitz, LS Grossman, JL Cavanaugh Jr, JM Davis and DA Lewis. Predicting the "revolving door" phenomenon among patients with schizophrenic, schizoaffective, and affective disorders. Am J Psychiatry 1995; 152:856-861.
   Project for an Ontario Women's Health Evidence-Based Report. Depression, www.powerstudy.ca/
- 257
- 259
- Project for an Ontario Women's Health Evidence-Based Report. Depression. www.powerstudy.ca/ the-power-report/the-power-report-volume-1/depression; accessed November 5, 2010.
   Damiani G, Pinnarelli L, Colosimo SC, Almiento R, Sicuro L, Galasso R, Sommella L, Ricciardi W. The effectiveness of computerized clinical guidelines in the process of care: a systematic review. BMC Health Services Research. 2010;10:2.
   Roy CL, Kachalia A, Woolf S, Burdick E, Karson A, Gandhi TK. Hospital readmissions: physician awareness and communication practices. J Gen Intern Med. March 2009;24(3):374–380.
   Jamtvetd G, Young JM, Kristoffersen DT, O'Brien MA, Oxman AD. Audit and feedback: effects on professional practice and health care outcomes. Cochrane Database of Systematic Reviews. 2006;2. Art. No. CD000259.
   Jamtvetd G, Young JM, Kristoffersen DT, O'Brien MA, Oxman AD. Audit and feedback: effects on professional practice and health care outcomes. Cochrane Database of Systematic Reviews. 2006;2. Art. No. CD000259.
- 261

- On protessional practice and nearth care outcomes. Contraine Database of Systematic Reviews. 2006;2. Art. No. CD000259.
   Jamtvedt G, Young JM, Kristoffersen DT, O'Brien MA, Oxman AD. Audit and feedback: effects on professional practice and health care outcomes. Cochrane Database of Systematic Reviews. 2006;2. Art. No. CD000259.
   Yan Walraven C, Irfan A, Dhalla CB, Etchells E, Stiell IG, Zarnke K, Austin PC, Forster AJ. Derivation and validation of an index to predict early death or unplanned readmission after discharge from hospital to the community. CMAJ. April 6, 2010;182(6).
   Kripalani S, Lefevre F, Phillips CO, Williams MV, Basaviah P, Baker DW. Deficits in communication and information transfer between hospital-based and primary care physicians: implications for patient safety and continuity of care. JAMA. February 28, 2007;297(8):831–41.
   Bell CM, Schnipper JL, Auerbach AD, Kaboli PJ, Wetterneck TB, Gonzales DV, Arora VM, Zhang JX, Mettzer DO. Association of communication between hospital-based physicians and primary care providers with patient outcomes. J Gen Intern Med. March 2009;24(3):381–386.
   Kergoat MU, Latour J, Juijen I, Plante MA, Lebel P, Mainville D, Bolduc A, Buckland JA. A discharge summary adapted to the frail elderly to ensure transfer of relevant information from the hospital to community settings: a model. BMC Geriatr. September 23, 2010;10(1):69.
   Balaban RB, Weissman JS, Samuel PA, Woolhandler S. Redefining and redesigning hospital discharge to enhance patient care: a randomized controlled study. J Gen Intern Med. August 2008;23(8):1228–1233.
   Otugh A, Williams KW, Grigsby J, Coleman EA. Better transitions: improving comprehension of discharge instructions. Front Health Serv Manage. Spring 2009;25(3):11–32.
   Vansuch M, Naessens JM, Stroebel RJ, Huddleston JM, Williams AR. Effect of discharge Instructions on readmission of hospital bised patients with heart failure: col al of the
- 273 274

- 277

- <sup>7</sup> Törottö Central Community Care Access Centre. In the Know for our Community Partners. 2010. www.ccac-ont.ca/Upload/toronto/General/Community-Summer-Final.pdf; accessed March 10, 2011.
   <sup>8</sup> Torotto Central LHIN. Improving Outcomes and Reducing Hospital Readmissions with Virtual Ward. Dec 9, 2010. www.torontocentralhin.on.ca/uploadedEfiles/Home\_Page/News\_Rooms/LHINf0%20Minute%20%20Virtual%20Ward.pdf; accessed March 1, 2011.
   <sup>9</sup> Canadian Health Services Research Foundation. Picking Up the Pace. Nov 1, 2010. http://www.chsrf.ca/Libraries/Picking\_up\_the\_pace\_files/Marg\_Fleming.sflb.ashx; accessed Dec 8, 2010.
   <sup>9</sup> Canadian Physiotherapy Association. How Physiotherapy Helps. 2011. www.physiotherapy.ca/public.asp/WCE=C=4/IK=23453[RefreshT=222436][RefreshS=LeftNav]RefreshD=2224360; accessed October 18, 2010.
   <sup>9</sup> Canadian Mental Health Care for Elderly. www.caring-for-aging-parents.com/health.acre-for-elderly.html; accessed October 19, 2010. www.ontario.cms.ca/docs/policy/cmha\_ontario.issues for older adults. full\_report\_2010.9.pdf; accessed October 19, 2010.
   <sup>9</sup> Ganadian DJ, Rodan LH, Sahlas DJ, Lee L, Murray BJ, Ween JE, Perry JR, Chenkin J, Morrison LJ, Beck S, Black SE. A citywide prehospital protocol increases access to storke thrombolysis in Toronto. Stroke. December 2009;401(2):3841–3844.
   <sup>9</sup> Fink HA, Taylor BC, Tacklind JW, Rutks IR, Wilt TJ. Treatment interventions in nursing home residents with urinary incontinence: a systematic review of randomized trials. Mayo Clin Proc. 2008;83(12):1332–1343. 280
- 282
- 283
- 285

- - Pinquart M, Duberstein PR, Lyness JM. Treatments for later-life depressive conditions: a meta-analytic comparison of pharmacotherapy and psychotherapy. Am J Psychiatry. September 2006;163(9):1493–1501.
  - Yang Y. How does functional disability affect depressive symptoms in late life? The role of perceived social support and psychological resources. J Health Soc Behav. December 2006;47(4):355–372. 319

  - perceived social support and psychological resources. J Health Soc Behav. December 2006;47(4):355–372.
     <sup>320</sup> Alexopoulos GS. Depression in the elderly. Lancet. June 2005;4–10;365(9475):1961–1970.
     <sup>321</sup> Yang Y. How does functional disability affect depressive symptoms in late life? The role of perceived social support and psychological resources. J Health Soc Behav. December 2006;47(4):355–372.
     <sup>322</sup> Beyer JL. Managing depression in geriatric populations. Ann Clin Psychiatry. October-December 2007;19(4):221–238.
     <sup>323</sup> Yang Y. How does functional disability affect depressive symptoms in late life? The role of perceived social support and psychological resources. J Health Soc Behav. December 2006;47(4):355–372.
     <sup>324</sup> Yang Y. How does functional disability affect depressive symptoms in late life? The role of perceived social support and psychological resources. J Health Soc Behav. December 2006;47(4):355–372.
     <sup>324</sup> Yang Y. How does functional disability affect depressive symptoms in late life? The role of perceived social support and psychological resources. J Health Soc Behav. December 2006;47(4):355–372.
     <sup>326</sup> Residents First. 2011. www.residentsfirst.ca/home; accessed December 8, 2010.
     <sup>327</sup> MolULTC. Compliance transform/new\_process.html; accessed December 8, 2010.
     <sup>328</sup> An adul tong-stay client is defined as an adult who requires more than 60 uninterrupted days of service through a CCAC, or a client who requires admission to an LTC home.
     <sup>329</sup> Ontario Home Care Association. Home Care in 2010 Essential for an Aging Population. Sep 2010. www.homecareontario.ca/public/docs/publications/position%20apers/2010/homecare-essential-for-an-aging-population. Off: accessed February 2011.
     <sup>330</sup> Bos JT, Frijters DH, Wagner C, Carpenter GI, Finne-Soveri H, Topinkova E, Garms-Homolová V, Henrard JC, Jonsson PV, Sarbya L, Ljunggren G, Schroll M, Gambassi G, Bernabei R. Var

# Registered Nurses' Association of Ontario. Promoting continence: using prompted voiding: a nursing best practice guideline. Sociation of Ontario. Continence/constipation workshop for RNs in long-term care: a facilitator's guide. May 2007. www.mao.org/Storage/3U/2510\_Rev\_LTCCC.pdf: accessed January 17, 2011. Public Health Agency of Canada. Evidence-based best practices for the prevention of fals: false intermediate care sets and prevention of fals: false intermediate care sets and anary 17, 2011. Public Health Agency of Canada. Evidence-based best practices for the prevention of fals: cacessed January 17, 2011. Batten H, Maki BE. Assistive devices for balance and mobility. benefits, demands, and adverse consequences. Arch Phys Med Rehabil. 2005;66:134–145. Ministry of Health and Long-term Care. RAMDS 2: 0.11C homes — practice requirements. *Byth Rev Natl Biol*, 2007;0:2003;26:134–145. Ministry of Health and Long-term Care. RAMDS 2: 0.11C homes — practice requirements. *Drift Sciences* 40:199;1:D0004294. Nursing Homo-Dailihylitic/biolic/corgostrenatic Review and meta-analysis of infong-term care. Cochrane Database of Systematic Reviews and meta-analysis of antomics for the prevention of falls in older pacing and the substance of Systematic Review and thest. J Pain Sermet BA, Pain 1: orgonitively impaired older people with severe dementia: a systematic review and nurse from eptaleints. J Pain Sermet BA, Pain 1: orgonitively impaired nurse from eptales. J Pain Sermet BA, Review J Pain Symptom Marag. 2006;31(2):170–192. ZvavAahalen KM, Hamer SH, AbuSad HH, Berger MFT, Pain In eldery people with severe dementia: a systematic review of behavioural pain assessment fools. BMC Genatics, 2006;63. The maragement of throne pain notice prevision. KSin Parina Contome Pain Older People with severe dementia: a systematic review of persistent pain in older persons. Revin an idder persons. Review Pain an idder pers

- <sup>332</sup> Mahieu LM, et al. Additional hospital stay and charges due to hospital-acquired infections in a neonatal intensive care unit. J Hosp Infection. 2001;47(3):223–229.
   <sup>333</sup> Discharge Abstract Database, FY 2009/10, calculated by Institute for Clinical Evaluative Sciences.
   <sup>334</sup> Parienti JJ, et al. Hand-rubbing with an aqueous alcoholic solution vs. traditional surgical hand-scrubbing and 30-day surgical site infection rates: a randomized equivalence study. JAMA. 2001;288:722–727.

- <sup>333</sup> Discharge Abstract Database, PY 2009/10, calculated by Institute for Clinical Evaluative Sciences.
   <sup>344</sup> Parienti JJ, et al. Handrubbing with an aqueous alcoholic solution vs. traditional surgical hand-scrubbing and 30-day surgical site infection rates: a randomized equivalence study. JAMA. 2001;288:722–727.
   <sup>345</sup> Gardam MA, Lemieux C, Reason P, van Dijk M, Goel V. Healthcare-associated infections as patient safety indicators. Healthcare Papers, 2009;93(8):2-4.
   <sup>345</sup> Control of Clostridum difficile Infection (CDI) Outbreaks in Hospitals: A Guide for Hospital and Health Unit Staff. Available at: www.health.gov.on.ca/english/media/news\_releases: Improving Patient Safety In Hospitals. Ministry of Health and Long-term Care, Toronto, September 26, 2008. Available at: www.health.gov.on.ca/english/media/news\_releases: Archives/m. (DS/sept/nr 2008)926. html; accessed March 31. 2011.
   <sup>346</sup> Ontanio's rate appears to be betfer than Quebec's rate of 0.6 per 1000 patient-days. [Donnies de surveillance des infections dans les centres hospitaliers au Oubec. Bulletin on 2.2, juillet 2010. Comparisons should be made with caution due to differences in how cases are counted. June 2010. according to www.hpa.org.uk/hpr/infections/hcaintin taccessed November 2, 2010. Comparisons should be made with caution due to differences in how cases are counted.
   <sup>340</sup> Uaalty Healthcare Network. Suffer Healthcare New? 2007. Nww.qhn.ca/about/saferhealthcare.html; accessed March 28, 2011.
   <sup>344</sup> Heyland DK, Cook DJ, Griffin L, Keenan SP, Brun-Buisson C. The attributable cortical mortality of central line-associated loodstream infections in intersive care departments in Argentina: A prospective, matched analysis. Am J Inter Control. 2003 Dec;31(8):475-80.
   <sup>345</sup> Rosentha UO, Guzma S, Migone O, Cnrich CJ. The attributable cost.length of hospital stay, and mortality O central line-associated intertors. Institute for Healthcare Network. Spr.2725
- <sup>1996</sup>;276:322–328.
   <sup>354</sup> Pronovost PJ, Angus DC, Dorman T, Robinson KA, Dremsizov TT, Young TL. Physician staffing patterns and clinical outcomes in critically ill patients: a systematic review. JAMA. November 6, 2002;288(17):2151–2162.
   <sup>355</sup> Kahn JM, Brake H, Steinberg KP. Intensivist physician staffing and the process of care in academic medical centres. Qual Saf Health Care. October 2007;16(5):329–333.
   <sup>356</sup> Geubbels EL, Bakker HG, Houtman P, van Noort-Klaassen MA, Pelk MS, Sassen TM, Wille JC. Promoting quality through surveillance of surgical site infections: five prevention success stories. Am J Infect Control. November 2004;32(7):424–430.
   <sup>357</sup> Zoutman DE, Ford BD. The relationship between hospital infection surveillance and control activities and antibiotic-resistant pathogen rates. Am J Infect Control. February 2005;33(1):1–5.
   <sup>358</sup> MOHITC. Fact sheet: hand hyeine compliance reporting, www.health.gov.on.ca/patient safet/v
- MOHLTC: Fact sheet: hand hygiene compliance reporting. www.health.gov.on.ca/patient\_safety/ pro/hh/pro\_resource/hh\_reporting\_fs.pdf; accessed January 24, 2011.
   Damschroder LJ, Banaszak-Holl J, Kowalski CP, Forman J, Saint S, Krein SL. The role of the manual state of the state o

- MOHLI C. Fact sheet: hand hygiene compliance reporting. www.health.gov.on.ca/patient\_safety/ pro/hh/pro\_resource/hh reporting fs.pdf; accessed January 24, 2011.
   Damschroder LJ, Banaszak-Holl J, Kowalski CP, Forman J, Saint S, Krein SL. The role of the "champion" in infection prevention: results from a multisite qualitative study. Qual Saf Health Care. 2009;18:434-440.
   S Million Lives Campaign. Getting started kit: prevent central line infections how-to guide. Cambridge, MA: Institute for Healthcare Improvement, 2008. www.ihi.org/HI/Topics/CriticalCare/ IntensiveCare/Changes/ImplementtheCentralLineBundle.htm; accessed January 22, 2010.
   Roehr B. Suspension of privileges improves physician adherence to hand hygiene. Medicaape Medical News. 2007. www.medscape.com/viewarticle/5644005; accessed February 10, 2010.
   Pittet D, Panesar SS, Wilson K, Longtin Y, Morris T, Allan Y, Storr J, Cleary K, Donaldson L. Involving the patient to ask about hospital hand hygiene: a National Patient Safety Agency feasibility study. J Hosp Infect. January 12, 2011.
   McGuckin M, Storr J, Longtin Y, Allegranzi B, Pittet D. Patient empowerment and multimodal hand hygiene promotion: a wirwin strategy. Am J Med Qual. January/February 2011;26(1):10–17.
   De Wandel D, Maes L, Labeau S, Vereecken C, Blot S. Behavioral determinants of hand hygiene compliance in intensive care units. Am J Crit Care. May 2010;19(3):230–239.
   MOHLIC. F act sheet: hand hygiene compliance reporting. www.health.gov.on.ca/patient\_safety/ pro/hh/pro\_resource/hh\_reporting fs.pdf; accessed January 24, 2011.
   De Wandel D, Maes L, Labeau S, Vereecken C, Blot S. Behavioral determinants of hand hygiene compliance in intensive care units. Am J Crit Care. May 2010;19(3):230–239.
   Boyce JM, Pittet D. Guideline for hand hygiene in health-care settings: recommendations of the Healthcare Infection Control Practices Advisory Committee Tanscripts: Standing Commi

- <sup>374</sup> Baker GR, Norton PG, Flintoft V, Blais R, Brown A, Cox J, Etchells E, Ghali WA, Hébert P, Majumdar SR, O'Beirne M, Palacios-Derflingher L, Reid RJ, Sheps S, Tamblyn R. The Canadian Adverse Events Study: the incidence of adverse events among hospital patients in Canada. Canadian Medical Association Journal. May 25, 2004;170(11):1678–1686.
   <sup>375</sup> Wrong Diagnosis. Types of Medical Mistakes. 26 Jan, 2011. www.wrongdiagnosis.com/mistakes/types.htm; accessed September 2010.
   <sup>376</sup> Elder NC, Vonder Meulen M, Cassedy A. The identification of medical errors by family physicians during outpatient visits. Ann Fam Med. 2004;2:125–129.
   <sup>377</sup> Geerts WH, Pineo GF, Heit JA, et al. Prevention of venous thromboembolism. The Seventh ACCP Conference on Antithrombotic and Thrombolytic Therapy. Chest. 2004;126:338S–400S. The rate of DVT or pulmonary embolism, if blood thinners are not used, is 10% to 40% after general surgery and 40% to 60% after hip surgery.
   <sup>378</sup> Heit JA, Silverstein MD, Mohr DN, et al. Predictors of survival after deep vein thrombosis and pulmonary embolism: a population-based cohort study. Archives of Internal Medicine.
- pulmonary embolism: a population-based cohort study. Archives of Internal Medicini 1999;159:445–453.

- and 40% to outwaiter inplosingery.
   Heit JA, Silverstein MD, Mohr DN, et al. Predictors of survival after deep vein thrombosis and pulmonary embolism: a population-based cohort study. Archives of Internal Medicine. 1999;159:445–453.
   Center for Outcomes Research University of Massachusetts Medical School. Best Practices: Preventing Deep Vein Thrombosis and Pulmonary Embolism. www.outcomes-umassmed.org/DVT/ best\_practice; accessed January 30, 2011.
   Prandoni P, Le, Iansing AW, Cogo A, et al. The long-term clinical course of acute deep venous thrombosis. Annals Intern Med. 1996;125:1–7.
   Zhan C, Miller MR. Excess length of stay, charges and mortality attributable to medical injuries during hospitalization. JAMA. 2003;290:1868–1874.
   Discharge Abstract Database, FY 2009/10, Canadian Institute for Health Information.
   Government of Ontario. intelliHealth. https://www.intellihealth.moh.gov.on.ca/frontpage; accessed November 9, 2010.
   Haynes A, Weiser TG, Berry WR, Lipsitz SR, Breizat AS, Dellinger EP, Herbosa T, Joseph S, Kibatala PL, Lapitan MCM, Merry AF, Moorthy K, Reznick RK, Taylor B, Gawande AA, for the Safe Surgery Saves Lives Study Group. A surgical safety checklist to reduce morbidity and mortality in a global population. N Engl J Med. 2009;360:491–4599.
   Serious Reportable Events (SREs): Transparency & Accountability are Critical to Reducing Medical Errors. National Quality Forum. October 1, 2000;360:491–459.
   Discharge Abstract Database, FY 2009/10, calculated by Institute for Clinical Evaluative Sciences.
   Savitz LA, Jones CB, Bernard S, Quality indicators sensitive to nurse staffing in acute care settings. In: Advances in Patient Safety: New Directions and Alternative Approaches. Volumes 1-4, AHRQ Publication Nos. 08-0034 (1-4). July 2008.
   Brennan TA, Leape LL, Laird NM, Hebert L, Localio AR, Lawthers AG, et al. Incidence of adverse events and negligence in

- 2006;54(9):1414–1418. Courtney B, Ruppman J, Cooper H. Save our skin: initiative cuts pressure ulcer incidence in half. Nurs Manage. 2006;37(4):36–45. Medical Advisory Secretariat, MOHLTC. Prevention and management of chronic pressure ulcers. Ontario Health Technology Assessment Series. 2009;9:2–3. www.health.gov.on.ca/english/ providers/program/ohtac/tech/recommend/rec\_pup\_20091020.pdf; accessed January 24, 2011. Duncan KD. Preventing pressure ulcers: the goal is zero. J Comm J Qual Pat Saf. 2007;33(10):605–610. Benn J, Koutantji M, Wallace L, Spurgeon P, Rejman M, Healey A, Vincent C. Feedback from incident reporting: information and action to improve patient safety. Qual Saf Health Care. 2009;18:11–21. Locklek SW. Barger LK Avas NT. Rothschild IM. Craisler CA. Landrigen CP. Hanard Work Heart
- <sup>400</sup> Lockley SW, Barger LK, Ayas NT, Rothschild JM, Czeisler CA, Landrigan CP, Harvard Work Hours, Health and Safety Group. Effects of health care provider work hours and sleep deprivation on safety and performance. Jt Comm J Qual Patient Saf. November 2007;33(11 suppl):7–18.
- <sup>401</sup> Wallace DE, Lemaire JB, Ghali WA. Physician wellness: a missing quality indicator. Lancet. 2009;374:1714–1721.
   <sup>402</sup> Lemaire JB, Ghali WA. Physician wellness: a missing quality indicator. Lancet. 2009;374:1714–1721.
   <sup>402</sup> Lemaire JB, Wallace JE, Dinsmore K, Lewin AM, Ghali WA, Roberts D. Physician nutrition and cognition during work hours: effect of a nutrition-based intervention. BMC Health Services Research. 2010;10:241.
   <sup>403</sup> Kane RL, Shamliyan TA, Meeller C, Duval S, Wilt TJ. The association of registered nurse staffing laudic and active trained motion to incompton with used motion explosion.

- Research. 2010;10:241.
   Kane RL, Shamilyan TA, Mueller C, Duval S, Wilt TJ. The association of registered nurse staffing levels and patient outcomes: systematic review and meta-analysis. Med Care. December 2007;45(12):1195–1204.
   National Health Service Institute for Innovation and Improvement. The productive ward: releasing time to care learning impact and review. Coventry, England, 2010. www.institute.nhs.uk/images// documents/PWard/PW%20exec%20summary.pdf; accessed January 24, 2011.
   Rapid Impact Assessment of The Productive Ward: Releasing time to care<sup>10</sup>. National Health Service, London, January 2011. www.institute.nhs.uk/images//documents/Quality\_and\_value/productiveseries/Rapid%20Impact%20Assessment%20full%20feprot%20FINAL.pdf; accessed March 28, 2011.
   MOHLTC. Excellent Care for All. Oct 12, 2010. www.health.gov.on.ca/en/ms/ecfa/public/default.aspx; accessed on March 29, 2011.
   Niagara Health System. Safer Healthcare Now: Prevention of Surgical Site Infection. www.niagarahealth.on.ca/quality-and-performance/patientsafety/safer-healthcare-now. html#surgicalinfection; accessed November 2, 2010.
   Hagnes AB, et al. A surgical safety checklist to reduce morbidity and mortality in a global population. N Engl J Med. 2009(5):360:491–499.
   LifeExtension. Blood clot prevention: batting a dangerous condition. www.lef.org/protocols/heart\_circulatory/blood\_clot\_01.htm; consulté le 5 novembre 2010.
   Haynes AB, et al. A surgical safety checklist to reduce morbidity and mortality in a global population. N Engl J Med. 2009(5):360:491–499.
   LifeExtension. Blood clot prevention: batting a dangerous condition. www.lef.org/protocols/heart\_circulatory/blood\_clot\_01.htm; consulté le 5 novembre 2010.
   Haynes AB, et al. A surgical safety checklist to reduce morbidity and mortality in a global population. N Engl J Med. 2009(5):360:491–499.
   LifeExtension. Blood clot prevention: batting

- 2003;348(25):2526–2534.
   212 Discharge Abstract Database, FY 2009/2010, calculated by Institute for Clinical Evaluative Sciences.
   <sup>413</sup> Black D, Lewis M, Monaghan B, Trypuc J. System change in healthcare: the Ontario Stroke Strategy. Hospital Quarterly. 2002;6(4):44–47.
   <sup>414</sup> Damiani G, Pinnarelli L, Colosimo SC, Almiento R, Sicuro L, Galasso R, Sommella L, Ricciardi W. The effectiveness of computerized clinical guidelines in the process of care: a systematic review. BMC Health Serv Res. January 4, 2010;10:2.
   <sup>415</sup> Robb E, Jarman B, Suntharalingam G, Higgens C, Tennant R, Elcock K. Using care bundles to reduce in-hospital mortality: quantitative survey. BMJ. 2010;340:c1234.

- Barochia AV, Cui X, Vitberg D, Suffredini AF, O'Grady NP, Banks SM, Minneci P, Kern SJ, Danner RL, Natanson C, Eichacker PQ, Bundled care for septic shock: an analysis of clinical trails. Crit Care Med 2010;38(2):669-74.
   Chamberlain DJ, Wills EM, Bersten AB. The severe sepsis bundles as processes of care: A meta-analysis. Aust Crit Care. 2011 Feb 14 (epub ahead of print).
   Chandan Patient Stately Institute. Safer Healthcare Now Interundoms. www.saferhealthcarenow.ca/
   Kileen SD, O'Sulliura MJ, Coffey JC, Krwwn WO, Redmont PP. Provide volume and outcomes for oncological procedures. Br J Surg: April 2005;92(4):339-402.
   Halm EA, Lee C, Chassin MR. Is volume radued b. Jochams A. Hospital volume and patient outcomes in pulmonary embolism. CMAJ. January J. 2008;178(1):27-33.
   Young EJ, Holt PJ, Poloniecki JD, Loftus IM, Thompson MM. Meta-analysis and systematic review of the relationship between surgeon annual caseload and mortality for elective open abdominal acritic are systematic: review. of the relationship between surgeon annual caseload and mortality for elective open abdominal acritic are systematic: review. Med Care. October 2003;46(6):1287-1284.
   Hott PJ, Poloniecki JD, Loftus IM, Thompson MM. Meta-analysis and systematic review of the relationship between surgeon annual caseload and mortality for elective open abdominal acritic are: a systematic review. Med Care. October 2003;41(10):1129-1141.
   Lindsay MP, Gubit CG, Bayley M, Hill MD, Davies Schinkel C, Singh S, Phillips S. Canadian best practice: a systematic review. Med Care. October 2003;41(10):1129-1141.
   Giadstone DJ, Rodan LH, Safilas DJ, Lee L, Murray BJ, Ween JE, Perry JR, Chenkin J, Morrson LJ, Beck S, Back W, Berkhork P, Calol J, 2010 Laste JD. Debal of the Canadian Stroke Brance as systematic review. Med Care. October 2003;41(10):1129-1141.
   Giadstone DJ, Rodan LH, Safilas DJ, Lee L, Murray BJ, Ween JE, Perry JR, Chenkin J, M

- <sup>442</sup> Safer Healthcare Novel www.saferhealthcarenow.ca/EN/news/nodeUpdates/ontarioNode/ Documents/Safer%20Healthcare%20Now!%20in%20Ontario.pdf; accessed March 29, 2011.
   <sup>443</sup> Prevent Stroke. Prevent A Stroke. 2011. www.preventstroke.ca/en/page.php?Section=Prevent&id=37;
- 444
- accessed March 28, 2011. MERCK. Aging and Drugs. Feb 2009. www.merck.com/mmhe/sec26/ch321/ch321a.html; accessed October 19, 2010.
- Gurwitz JH, et al. The incidence of adverse drug events in two large academic long-term care facilities. Am J Med. 2005;118(3):251–258. Fick DM, Cooper JW, Wade WE, Waller JL, Maclean JR, Beers MH. Potentially inappropriate
- medications for the elderly according to the revised Beers criteria. Arch Int Med. 2003;163:2716–2724.

- <sup>1440</sup> Fick Div, Cooper JW, Wade WE, Waller JL, Maclean JK, Beers MH. Potentially inappropriate medications for the elderly according to the revised Beers criteria. Arch Int Med. 2003;163:2716–2724.
   <sup>1447</sup> MERCK. Aging and Drugs. Feb 2009. www.merck.com/mmhe/sec26/ch321/ch321a.html; accessed October 19, 2010.
   <sup>1458</sup> Fick DM, Cooper JW, Wade WE, Waller JL, Maclean JR, Beers MH. Potentially inappropriate medications for the elderly according to the revised Beers criteria. Arch Int Med. 2003;163:2716–2724.
   <sup>1469</sup> Zhan C, Sangl J, Bierman AS, Miller MR, Friedman B, Wickizer SW, Meyer GS. Potentially inappropriate medication use in the community-dwelling elderly: findings from the 1996 Medical Expenditure Panel Survey. JAMA. 2001;286(22):2823–2829.
   <sup>1469</sup> Gladstone DJ, Rodan LH, Sahlas DJ, Lee L, Murray BJ, Ween JE, Perry JR, Chenkin J, Morrison LJ, Beck S, Black SE. A citywide prehospital protocol increases access to stroke thrombolysis in Toronto. Stroke. December 2009;40(12):3841–3844.
   <sup>1451</sup> Woolcott JC, Richardson KJ, Wiens MO, Patel B, Marin J, Khan KM, Marra CA. Meta-analysis of the impact of 9 medication classes on falls in elderly persons. Arch Intern Med. 2009;169(21):1952-60.
   <sup>1452</sup> Zec RF, Burkett NR. Non-pharmacological and pharmacological antipsychotic drugs and the risk of sudden cardiac death. N Engl J Med. 2009;360(3):225–235.
   <sup>1453</sup> Zec RF, Burkett NR. Non-pharmacological and pharmacological treatment of the cognitive and behavioral symptoms of Alzheimer disease. NeuroRehabilitation. 2008;23(5):425–438.
   <sup>1454</sup> Magner L, Rust TB. Safety in long-term care settings: broadening the patientsafetyinstitute.ca/English/research/commissionedResearch/SafetyinIstifut C. 208. www.patiensafetyinstitute.ca/English/research/commissionedResearch/SafetyinIstifut C. 2008. dvi; accessed January 21, 2011.
   <sup>1455</sup> Stanley MA, Wilson NL, Novy DM, Rhoades HM, Wagener PD, Greisinger AJ, Cully JA

- <sup>460</sup> Chen Y. Unexplained variation across US nursing homes in antipsychotic prescribing rates. Archives of Internal Medicine. 2010;170(1):89–95.
   <sup>461</sup> Marcum ZA, Handler SM, Wright R, Hanlon JT. Interventions to improve suboptimal prescribing in nursing homes: a marative review. Am J Geriatr Pharmacother. 2010;8(3):183–200.
   <sup>462</sup> Rochon PA, Field TS, Bates DW, Lee M, Gavendo L, Erramuspe-Mainard J, Judge J, Gurvitz JH. Clinical application of a computerized system for physician order entry with clinical decision support to prevent adverse drug events in long-term care. OMAJ. January 3, 2006;174(1):52–54.
   <sup>463</sup> Wagner L, Rust TB. Safety in long-term care settings: broadening the patient safety agenda to include long-term care services. Canadian Patient Safety Institute, 2008. www.patientsafetyinsti-tute.ca/English/research/commissionedResearch/SafetyinLongTermCareSettings/Documents/ Reports/11C%20paper%20-%20Safety%20In%20LTC%20Settings%20-%202008.pdf; accessed January 24, 2011
   <sup>464</sup> Davis DA, Taylor-Vaisey A. Translating guidelines into practice. a systematic review of theoretic concepts, practical experience and research evidence in the adoption of clinical practice guidelines. CMAJ. 1997;157(4):408–416.
   <sup>464</sup> Marcum ZA, Handler SM, Wright R, Hanlon JT. Interventions to improve suboptimal prescribing in nursing homes: a narrative review. Am J Geriatr Pharmacother. 2010;8(3):183–200.
   <sup>465</sup> Ontario Long-term Care Homes Act 2007, Regulation 79/10. MOHLTC, Toronto, sections 117 and 120. www.elaws.gov.on.ca/html/sourc/regs/english/2010/elaws\_src\_regs\_r10079\_e. htm#BK139; accessed March 31, 2011.
   <sup>464</sup> Kaur X, Mitchell G, Wright R, Hanlon JT. Interventions to improve suboptimal prescribing in the elderly: a systematic review. Am J Geriatr Pharmacother. 2010;8(3):183–200.
   <sup>465</sup> Choi NG, Willie RJ, Ransom S. Risk factors and intervention programs for depression in nursing home ensidents: twe view. Am

- <sup>170</sup> Cellue for inelativities (Value) improvement. Releasing time to Care.
   <sup>171</sup> www.chdi.ca/Currenthitiatives/Releasing TimetoCare.aspx; accessed January 27, 2011.
   <sup>172</sup> Institute for Medication Practices Canada. Medication Safety Initiatives in Ontario.
   <sup>174</sup> www.ismp-canada.org/download/MedicationSafetyInitiativesInOntario.pdf; accessed on March 30, 2011.
   <sup>174</sup> MOHLTC. MedisCheck Program Expands. Nov 8, 2010. www.health.gov.on.ca/en/news/ release/2010/nov/m\_20101108.aspx; accessed on March 30, 2011.
   <sup>175</sup> Government of Ontario. Ontario Regulation 79/10 made under the Long-Term Care Homes Act, 2007. March 10, 2010. www.elaws.gov.on.ca/html/source/regs/english/2010/elaws\_src\_regs\_r10079\_e.htm; accessed on March 30, 2011.
   <sup>176</sup> Government of Contario Moves to Reduce Narcotics Abuse. Nov 29, 2010. www.health.gov.on.ca/en/ news/release/2010/nov/m\_20101129.aspx; accessed on March 30, 2011.
   <sup>177</sup> Organisation for Economic Co-operation and Development. Conceptual framework and definition of long-term care expenditure. 2008. www.oecd.org/dataoecd/24/58/40760216.pdf; accessed 0 Otwens PL, Russo A, Spector W, Mutter R. Emergency department visits for injurious falls among the elderly, 2006. Healthcare Cost & Utilization Project. October 2009:1–20. www.hcup-us.ahrq.gov/reports/statbriefs/sb80.pdf; accessed October 20, 2010.
   <sup>179</sup> Gladstone DJ, Rodan LH, Sahlas DJ, Lee L, Murray BJ, Ween JE, Perry JR, Chenkin J, Morrison LJ, Beck S, Black SE. A citywide prehospital protocol increases access to stroke thrombolysis in Toronto. Stroke. December 2009:40(12):3841–3844.
   <sup>184</sup> Takhashi PY, Kiemele LJ, Jones JP. Wound Care for elderly patients: advances and clinical applications for practicing physicians. Mayo Clin Proc. 2004; 79:260–267.
   <sup>184</sup> Rubenstein LZ, et al. Falls in the nursing home. Annals Intern Med. 1994;121:442–451.
   <sup>184</sup> WebMD. Understanding Bladder Infections

- <sup>147</sup> an actue care setting. J Adv Nurs. 2007 Nov50(4):427-35.
   <sup>148</sup> Falls prevention and management. Registered Nurses Association of Ontario, Toronto.
   <sup>1487</sup> www.ltctoolkit.mao.ca/resources/falls.prevention-and-management?page=1; accessed March 31, 2011.
   <sup>1487</sup> Institute for Healthcare Improvement. Preventing Pressure Ulcers Turn Clock Tool.
   <sup>1487</sup> www.ltctoolkit.mao.ca/resources/falls.preventing Pressure Ulcers Turn Clock Tool.
   <sup>1487</sup> www.ltctoolkit.mao.ca/resources/falls.preventing Pressure Ulcers Turn Clock Tool.
   <sup>1487</sup> www.ltctoolkit.mao.ca/resources/falls.preventing Pressure Ulcers Turn Clock Tool.
   <sup>1489</sup> Wealth Quality Council of Saskatchewan. Releasing time to care.
   <sup>1490</sup> Centre for Healthcare Quality Improvement. Releasing time to care.
   <sup>1490</sup> Centre for Healthcare Quality Improvement. Releasing time to care.
   <sup>1490</sup> Centre for Healthcare Quality Improvement. Releasing time to care.
   <sup>1490</sup> CurrentInitiatives/Releasing Time toCare.aspx; accessed January 27, 2011.
   <sup>1490</sup> Evans D, Wood J, Lambert L. Patient injury and physical restraint devices: a systematic review.
   <sup>1401</sup> Advis. 2003;41(3):274-282.
   <sup>1491</sup> Ontario Health Quality Council. Success study: reducing falls in long-term care. In: Quality Monitor: 2010 Report on Ontario's Health System. 2010:112. www.ohqc.ca/pdfs/2010\_report\_-english.pdf; accessed January 21, 2011.
   <sup>1492</sup> American Geriatrics Society (AGS), Foundation for Health in Aging. AGS clinical practice guideline: pharmacological management of persistent pain in older persons. 2008. www.americangeriatrics.org/ health%5fcare%5fprofessionals/clinical%5fpractice/clinical%5fguidelines%5frecommendations/2009; accessed January 17, 2011.
   <sup>1493</sup> Fick DM, Cooper JW, Wade WE, et al. Updating the Beers criteria for potentially inappropriate medication use in older adults: results of

- medication use in older adults: results of a US consensus panel of experts. Arch Intern Med. 2003;163:2716–2724.
   Fick DM, Cooper JW, Wade WE, et al. Updating the Beers criteria for potentially inappropriate medication use in older adults: results of a US consensus panel of experts. Arch Intern Med. 2003;163:2716–2724.
   Weber V, White A, McIlvried R. An electronic medical record (EMR)-based intervention to reduce polypharmacy and falls in an ambulatory rural elderly population. J Gen Intern Med. 2007;23(4):399–404.
   Public Health Agency of Canada. Evidence-based best practices for the prevention of falls. Report on Seniors' Falls in Canada. 2005. www.phac-aspc.gc.ca/seniors-aines/publications/pro/injury-blessure/falls-chutes/chap4-eng.php#foot105; accessed January 17, 2011.
   Bateni H, Maki BE. Assistive devices for balance and mobility: benefits, demands, and adverse consequences. Arch Phys Med Rehabil. 2005;86:134–145.
   Gillespie LD, Robertson MC, Gillespie WJ, Lamb SE, Gates S, Cumming RG, Rowe BH. Interventions for preventing falls in older people living in the community. Cochrane Database of Systematic Reviews. 2009;2.
   Healey F, Oliver D, Mine A, Connelly JB. The effect of bedrails on falls and injury: a systematic review of clinical studies. Age Ageing. July 2008;37(4):368–378.
   McCarthy R, Adedokun CW, Moody Fairchild R. Preventing falls in the elderly long term care facilities. Journal of Nursing. Www.ripournal.org/journal.of\_nursing/preventing\_falls\_in\_the\_elderly.long term\_care\_facilities.htm; accessed January 23, 2011.
   Burgio L, Allen-Burge R, Roth D, Bourgeois M, Dijkstra K, Gerstie J, et al. Come talk with me: improving communication between nursing assistants and nursing home residents during care routines. The Gerontologist. 2001;41:449–460.
   Canadian Agency for Drugs and Technologies in Health. Programs to manage aggressive behaviour in long-term care patients: a review of cl

- <sup>503</sup> Woods DL, Dimond M. The effect of therapeutic touch on agitated behavior and cortisol in persons with Alzheimer's disease. Biological Research for Nursing. 2002;4(2):104–114.
   <sup>504</sup> www.residentsfirst.ca/leadership. Accessed March 7, 2011.
   <sup>505</sup> www.health.gov.or.ca/english/public/program/ttc/compliance\_transform/new\_process.html; accessed March 7, 2011.
   <sup>506</sup> Workhealth.gov.or.ca/english/public/program/ttc/compliance\_transform/new\_process.html; accessed March 7, 2011.
   <sup>507</sup> North Simcce Muskoka LHIN. Behavioural Support Systems Project. www.nsmlhin.on.ca/Page. aspXid=4984&chmesl=e2f22c9a\_72\_280\_4984\_6; accessed February 2, 2011.
   <sup>508</sup> Ontario Home Care Association. Home Care in 2010 Essential for an Aging Population. Sep 2010. www.homecareontario.ca/public/docs/publications/position%20papers/2010/home-care-essentialfor-an-aging-population.pdf; accessed February 2011.
   <sup>509</sup> Prevention of falls and fall-related injuries in community-dwelling seniors. Ontario Health Technology Assessment Series. 2008;8(2).
   <sup>510</sup> Cooper C, Balamurali TB, Selwood A, Livingston G. A systematic review of intervention studies about anxiety in caregivers of people with dementia. International Journal of Geriatric Psychiatry. 2007;22:131–188.
   <sup>511</sup> Selwood A, Cooper C, Owns C, Blanchard M, Livingston G. What would help me stop abusing? The family carer's perspective. International Journal of Psychogeriatrics. 2009;21(2):309–313.
   <sup>513</sup> Ommunity Care Access Centre. Integrated Client Care Project. www.ccac-ont.ca/Content.aspx? EnterpriseID=15&LanguageID=1&MenuID=1054; accessed March 30, 2011.
   <sup>514</sup> MOHLTC. Ontario Strengthens Home Care Services. Dec 15, 2008. www.health.gov.on.ca/english/media/news releases/archives/m-08/dec/m-20081215\_2.html; accessed March 30, 2011.
   <sup>515</sup> Mannin R et al. Impact of star performance ratings in English acute hospital trusts. J Health Serv Res Poli

- 519
- Parchman ML, Zeber JE, Palmer RF. Participatory Decision Making, Patient Activation, Medication Adherence, and Intermediate Clinical Outcomes in Type 2 Diabetes: A STARNet Study. Ann Fam 520
- Med 2010; 8(5): 410–7. Kehlet H. Acute pain control and accelerated postoperative surgical recovery. Surgical Clinics of North America. 1999;79(2):431-443.
- <sup>1010</sup> (80): 410–7.
   <sup>1021</sup> Kehlet H. Acute pain control and accelerated postoperative surgical recovery. Surgical Clinics of North America. 1999;79(2):431-443.
   <sup>1021</sup> Carr EC. Impact of postoperative pain on patient experience and recovery. Prof Nurse. 2001;17(1):37-40.
   <sup>1022</sup> The Hospital Compare site of the US Department of Health and human services provides detailed comparisons between hospitals on patient experience and other outcomes. Best performers on the indicator for "definitely recommend to family or friends" include Massachusetts General (88%), Brigham and Women's (87%) and Mayo Clinic St. Mary's Hospital (84%). www.hospitalcompare.hhs.gov; accessed March 31, 2011.
   <sup>1023</sup> Spaite DW, et al. Rapid process redesign in a university-based emergency department: decreasing waiting time intervals and improving patient satisfaction. Annals of Emergency Medicine. February 2002;39(2):168–177.
   <sup>1024</sup> Taylor C, Benger JR. Patient satisfaction in emergency medicine. Emerg Med J 2004;21:528–532.
   <sup>1025</sup> Hedges JR, Trout A, Magnusson AR. Satisfied Patients Exiting the Emergency Department (SPEED) Study. Acad Emerg Med. 2009;54:360-367.
   <sup>1026</sup> Ontario Hospital Association. Hospital Report 2007. http://www.oha.com/KnowledgeCentre/Library/HospitalReports/Documents/Hospital%20Reports/2020Department%202
   <sup>1027</sup> Department%20Care.pdf; accessed January 30, 2011.
   <sup>1038</sup> Taylor C, Benger JR. Patient satisfaction in emergency medicine. Emerg Med J 2004;21:528–532.
   <sup>1049</sup> Mayer TA, Cates RJ, Mastorovich MJ, Royalty DL. Emergency department patient satisfaction: customer service training improves patient satisfaction and ratings of physician and nurse skill. J Healtht Manag. 1988 Sep-Oct:43(5):427-40.
   <sup>1034</sup> At SickKids; Irrom the President Winter 2011. http://www.sickkids.ca/AboutSickKids/pres-ceo/atsickKids/index.html; accessed March 3, 2011.
   <sup>1034</sup> Merboth MK

- <sup>534</sup> Rupp T, Delaney KA. Inadequate Analgesia in Emergency Medicine. Ann Emerg Med. 2004;43:494-503.
   <sup>535</sup> Decosterd I, Hugi O, Tamches E, Blanc C, Mouhsine E, Givel JC et al. Oligoanalgesia in the Emergency Department: Short-Term Beneficial Effects of an Education Program on Acute Pain. Ann Emerg Med. 2007;50:462-471.
   <sup>536</sup> Institute for Clinical Systems Improvement. Health Care Guideline: Assessment and Management of Acute Pain. Sixth Edition, March 2008. www.icsi.org/pain\_acute/pain\_acute\_assessment\_and\_management.of\_3.html; accessed February 3, 2011.
   <sup>537</sup> Taylor DM, Cameron PA. Discharge instructions for emergency department patients: what should we provide? J Accid Emerg Med. Aurch 2000; 17(2):86-90.
   <sup>538</sup> Yu KT, Green RA. Critical aspects of emergency department documentation and communication. Emerg Med Clin North Am. 2009;27(4):641-654, ix.
   <sup>540</sup> Flores, G. The impact of medical interpreter services on the quality of health care: a systematic review. Med Care Res Rev. 2005. 62(3): 255-299.
   <sup>541</sup> Schillinger D, Piette J, Grumbach K, et al. Closing the loop. Physician communication with diabetic patients who have low health literacy. Arch Intern Med. 2003;163:83-90.
   <sup>542</sup> Afialo M, Lang E, Léger R, Xue X, Colaccne A, Soucy N, Vandal A, Boivin JF, Unger B. Impact of a standardized communication system on continuity of care between family physicians and the emergency department. CEM. 2007 Mar;9(2):72-78-86.
   <sup>543</sup> Marra G. Katz, Terry A. Jacobson, Emir Veledar, and Sunil Kripalani. Patient Literacy and Question-asking Behavior During the Medical Encounter: A Mixed-methods Analysis. J Gen Intern Med. 2007; 7:210.
   <sup>544</sup> MOHLITC. Legislation: The Excellent Care for All Act, 2010. June 23, 2010. www.health.gov.on.ca/ endoridation system or contal Acting Patient.
- <sup>544</sup> MOHLTC, Legislation: The Excellent Care for All Act, 2010. June 23, 2010. www.health.gov.on.ca/ en/legislation/excellent\_care; accessed March 31, 2011.
   <sup>545</sup> Quality Improvement Plan Guidance Document. Ministry of Health and Long-term Care, January 2011. www.health.gov.on.ca/en/ms/ecfa/pro/updates/qualityimprove/qip\_guide.pdf; accessed March 31, 2011.
   <sup>546</sup> Street R, et al. How does communication heal? Pathways linking clinician-patient communication to health outcomes. Patient Education and Counseling. 2009;74:295–301.
   <sup>547</sup> Narayan KM, Gregg EW, FagotCampagna A, Gary TL, et al. Relationship between quality of diabetes care and patient satisfaction. J Natl Med Assoc. Jan 2003;95(1):64–70.
   <sup>548</sup> Battersby M, Von Korff M, Schaefer J, Davis C, Ludman E, Greene SM, Parkerton M, Wagner EH. Twelve evidence-based principles for implementing self-management support in primary care. Jt Comm J Qual Patient Sat. 2010;36(12):561-70.
   <sup>549</sup> Wang SJ, Middleton B, Prosser LA, Bardon CG, Spurr CD, Carchidi PJ, Kittler AF, Goldszer RC, Fairchild DG, Sussman AJ, Kuperman GJ, Bates DW. A Cost-Benefit Analysis of Electronic Medical Records in Primary Care. Am J Med. 2003;114(5):397-403.

- Langley J, Beasley C. Health Information Technology for Improving Quality of Care in Primary Care Settings. Prepared by the Institute for Healthcare Improvement for the National Opinion Research Center under contract No. 290-04-0016. AHRQ Publication No. 07-0079-EF. Rockville, Maryland: Agency for Healthcare Research and Quality; July 2007. http://www.ihi.org/HI/Topics/OfficePractices/Access/Literature/Health1TorImprovingQualityofPrimaryCare.htm; accessed January 30, 2011.
   Willis D. Making Every Minute Count: Tools to Improve Office Efficiency. Family Practice Management April 2005. http://www.internetgroup.ca/clientnet\_new/docs/Making%20Every%20 Minute%20Count.pdf; accessed January 30, 2011.
   Schillinger D, Piette J, Grumbach K, et al. Closing the loop. Physician communication with diabetic patients who have low health literacy. Arch Intern Med. 2003;163:83-90.
   Afilalo M, Lang E, Léger R, Xue X, Colacone A, Soucy N, Vandal A, Boivin JF,Unger B. Impact of a standardized communication system on continuity of care between family physicians and the emergency department. CLM. 2007 Mar;9(2):79-86.
   Beach MC, Sugarman J, Johnson RL, Arbelaez JJ, Duggan PS, Cooper LA. Do patients treated with dignity report higher satisfaction, adherence and receipt of preventive care? Ann Fam Med. 2005;3:331-338.
   Shaw WS, Zaia A, Pransky G, Winters T, Patterson WB. Perceptions of provider communication and patient satisfaction for treatment of acute low back pain. J Occup Environ Med. 2005;47:1036–1043.
   Government of Ontario. More Nurse Practitioner Clinics Coming To Ontario. Aug 23, 2010. news.ontario.ca/mohltc/en/2010/08/more-nurse-practitioner-clinics.coming.to-ontario.html; accessed April 1, 2011.
   PricewaterhouseCoopers' Health Research Institute. You get what you pay for: a global look at balancing demand, quality, and efficiency in healthcare navment reform. 2008. Wavent were communication and patient getwaterhouseCoopers' Heal

- news.ontario.ca/nonic.ya/cont/2010/08/inforeIntrsepractuoner-clinics-conting-oronatrio.html;
   accessed April 1, 2011.
   PricewaterhouseCoopers' Health Research Institute. You get what you pay for: a global look at balancing demand, quality, and efficiency in healthcare payment reform. 2008. www.pwc.com/us/en/healthcare/publications/you-get-what-you-pay-for.jhtml; accessed December 31, 2010.
   Canadian Institute for Health Information. 2008. The cost of acute care hospital stays by medical condition in Canada: 2004–2005. 2008. https://secure.chi.ca/estore/productSeries.
   htm?pc=PCC398; accessed December 31, 2010.
   Brown R. The promise of care coordination: models that decrease hospitalizations and improve outcomes for medicare beneficiaries with chronic illnesses. Report commissioned by the National Coolition on Canada: 10.006/166/1822–1828.
   Coleman EA, Parry C, Chalmers S, Min SJ. The care transitions intervention: results of a randomized controlled trial. Arc (Int Med. 2006;166:1822–1828.
   Lorig KR, Ritter P, Stewart AL, Sobel DS, Brown BW, Bandura A, Gonzalez VM, Laurent DD, Holman HR. Chronic disease self-management program: 2year health status and health care utilization outcomes. Med Care. 2001;39(11):1217–1223.
   Naylor MD, Brooten DA, Campbell RL, Maislin G, McCauley KM, Schwartz JS. Transitional care of older adults hospitalized with heart failure: a randomized controlled trial. J Am Geriatrics Soc. 2004;52:675–684.

- 2004;52:675-684

- olfer adults hospitalized with heart failure: a randomized controlled trial. J Am Geriatrics Soc. 2004;52:675–684.
   <sup>564</sup> Wheeler J. Can a disease self-management program reduce health care costs? The case of older women with heart disease. Med Care. 2003;41:706–715.
   <sup>565</sup> Ansari Z, Laditka JN, Laditka SB. Access to health care and hospitalization for ambulatory care sensitive conditions. Med Care. Res Rev. 2006;63:719.
   <sup>566</sup> Glazier R, Moineddin R, Agha M, Zagorski B, Hall R, Manuel D, Sibley L, Kopp A. The impact of not having a primary care physician among people with chronic conditions: ICES investigative report. Toronto: Institute for Clinical Evaluative Sciences, 2008. www.ices.on.ca/webpage.cfm?site\_id=1&Rorg\_id=31&Rorg\_id=0.32&Rorg\_id=0.32&Rorg\_id=0.31&Rorg\_id=0.32&Rorg\_id=0.31&Rorg\_id=0.32&Ro

- 2009;5(1):54–65.
   573 You JJ, Purdy I, Rothwell DM, Przybysz R, Fang J, Laupacis A. Indications for and results of outpatient computed tomography and magnetic resonance imaging in Ontario: a population-based study. Canadian Association of Radiologists Journal. 2008;59(3):135–143.
   574 Quirtana JM, et al. Healthrelated quality of life and appropriateness of knee or hip joint replacement. Arch. Intern. Med. 2006;166:220–226.
   575 Escobar A, et al. Development of explicit criteria for total knee replacement. Int. J. Technol. Assess. Health Care. 2003;19:57–70.
   576 Quintana JM, et al. Appropriateness of total hip joint replacement. Int. J. Qual. Health Care. 2003;19:57–70.
   576 Quintana JM, et al. Appropriateness of total hip joint replacement. Int. J. Qual. Health Care. 2005;17:315–321

- Quintana JM, et al. Appropriateness of total hip joint replacement. Int. J. Qual. Health Care. 2005;17:315–321.
  Connor-Spady BL, Samnugasunderam S, Courtright P, Mildon A, McGurran JJ, Noseworthy TW, Steering Committee of the Western Canada Waiting List Project. The prioritization of patients on waiting lists for cataract surgery: validation of the Western Canada Waiting List Project cataract priority criteria tool. Ophthal Epidem. 2005;12:81–90.
  Stewart BA, Fernandes S, Rodriguez-Huertas E, Landzberg M. A preliminary look at duplicate testing associated with lack of electronic health record interoperability for transferred patients. JAMIA. 2010;17:341–344.
  Eber MR, Laxminarayan R, Perencevich EN, Malani A. Clinical and economic outcomes attributable to health care-associated sepsis and pneumonia. Arch Intern Med. 2010;17(4):347–353.
  Hwang RW, Hendon JH. The business case for patient safety. Clin Orthop Relat Res. April 2007;457:21–34. 577
- 580

- 582

- 592
- <sup>b</sup> Hwang RW, Hendon JH. The business case for patient safety. Clin Orthop Relat Res. April 2007;457:21-34.
   <sup>b</sup> Brilli RJ, Sparling KW, Lake MR, Butcher J, Myers SS, Clark MD, Helpling A, Stutler ME. The business case for preventing ventilator-associated pneumonia in pediatric intensive care unit patients. Jt Comm J Qual Patient Saf. November 2008;34(11):629–638.
   <sup>c</sup> Xakellis GC Jr, Frantz RA, Lewis A, Harvey P. Cost-effectiveness of an intensive pressure ulcer prevention protocol in long-term care. Adv Wound Care. January/February 1998;11(1):22–29.
   <sup>e</sup> Wong H, Wu RC, Tomlinson G, et al. How much do operational processes affect hospital inpatient discharge rates? J Public Health (Oxf). 2009;31:546–53.
   <sup>e</sup> Dainty P, Elizabeth J. Timely discharge of older patients from hospital: improving the process. Clin Med. 2009;9:311–14.
   <sup>e</sup> http://www.health.gov.on.ca/en/ms/ecfa/pro/ecfa\_pbp.aspx; accessed March 17, 2011.
   <sup>e</sup> Ontario has a standard definition for ALC, which is used in this report. See www.cancercare.on.ca/ cms/one.aspx?objectId=43214&contextId=1377; accessed November 1, 2010.
   <sup>e</sup> Hirdes JP, Poss JW, Curtin-Telegdi N. The method for assigning priority levels (MAPLe): a new decision-support system for allocating home care resources. BMC Medicine. 2008;6:9. www. biomedcentral.com/1741-7015/6/9; accessed December 9, 2010.
   <sup>e</sup> Creditor MC. Hazards of hospitalization of the elderly. Ann Intern Med. 1993;118:219–223.
   <sup>e</sup> Glick MR, Serrell NA, Gillick LS. Adverse consequences of hospitalization in the elderly. Soc Sci Med. 1982;16:1033–1038.
   <sup>e</sup> Hirsch CH, Sommers L, Olsen A, Mullen L, Winograd CH. The natural history of functional morbidity in hospitalized older patients. J Am Geriatr Soc. 1990;38:1296–1303.
   <sup>e</sup> Mickey LJ, Becker PM, Saltz CC, Feussner JR, Cohen HJ. Effect of a geriatric consultation team on functional status of elderly hospital Association, Fe

- Ontario Association of Community Care Access Centres. www.ccac-ont.ca; accessed December 9, 2010.
   Fried LP, Tangen CM, Walston J, Newman AB, Hirsch C, Gottdiener J, Seeman T, Tracy R, Kop WJ, Burke G, McBurnie MA, for the Cardiovascular Health Study Collaborative Research Group. Frailty in older adults: evidence for a phenotype. J Gerontol A Biol Sci Med. 2001;56A(3):M146–M156.
   Rothman MD, Leo-Summers L, Gill TM. Prognostic significance of potential frailty criteria. J Am Ger Soc. 2008;56(12):2211–2216.
   Freiheit EA, Hogan DB, Eliasziw M, Meekes MF, Ghali WA, Partlo LA, Maxwell CJ. Development of a frailty index for patients with coronary artery disease. J Am Ger Soc. 2001;58(8):1526–1531.
   Community Care Access Centre. The geriatric system navigation program. www.ccac-ont.ca/Upload/mt/General/GSN4Prochure.pdf; accessed January 14, 2011.
   MOHLTC Health System Performance and Accountability Division. Transitional care program framework. November 2009. www.ontla.on.ca/library/repository/mon/24003/300343.pdf; accessed November 17, 2010.
   Champlain LHIN. Health care options in the Champlain Region. 2010. www.champlainlhin.on.ca/Page.aspx?id=1212; accessed November 17, 2010.
   Boyd CM, Landefeld CS, Counsell SR, Palmer RM, Fortinsky RH, Kresevic D, Burant C, Covinsky KE. Recovery of activities of daily living in older adults after hospitalization for acute medical illness. J Am Ger Soc. 2008;56(12):2171–2179.
   Hirdes JP, Poss JW, Curtin-Telegdi N. The method for assigning priority levels (MAPLe): a new decision-support system for allocating home care resources. BMC Medicine. 2008;6:9. www.biomedcentral.com/1741-7015/6/9; accessed December 9, 2010.
   Starr-Hemburrow L, Parks JM, Bisaillon S. Home first: reducing ALC and achieving better outcomes for seniors through inter-organizational collaboration. Healthc Q. 2011;14(11:70–76.
   MOHLTC. Senior's care: supportive housing. 2010. www.heal

- 611
- December 9, 2010. Chan BT, Ovens HJ. Frequent users of emergency departments: do they also use family physicians' services? Can Fam Physician. October 2002;48:1654–1660. Calculated by Institute for Clinical Evaluative Sciences from Registered Persons Database, National Ambulatory Care Reporting System Database. Most recent value is for FY 2009/2010. Fisher ES. Building a medical neighborhood for the medical home. New England Journal of Medicine. 2008;359(12):1202–1205. Kayser-Jones JS, Wiener CL, Barbaccia JC. Factors contributing to the hospitalization of nursing home residents. The Gerontologist. 1989;29(4):502–510. MOHLTC, Health Data Branch. Long-term care home system report, as of May 31, 2010. World Health Organization. Guide to mental and neurological health in primary care: dementia.
- World Health Organization. Guide to mental and neurological health in primary care: dem www.mentalneurologicalprimarycare.org/page\_view.asp?c=16&did=1903&fc=005050; accessed September 2010. dementia
- Hernsi L. Living with dementia. Postgraduate Medical Journal. October 1982;58:610–617. Boushy D, Dubinsky I. Primary care physician and patient factors that result in patients seeking emergency care in a hospital setting: the patient's perspective. J Emerg Med. May/June 1999;17(3):405–412. 617
- 618
- 619 620

- 622
- seeking emergency care in a hospital setting: the patient's perspective. J Emerg Med. May June 1999;17(3):405–412.
   Hogenbirk JC, Pong RW, Lemieux SK. Impact of telephone triage on medical service use: implications for rural and remote areas. J Agric Saf Health. May 2005;11(2):229–237.
   Hogenbirk JC, Pong RW. An audit of the appropriateness of teletriage nursing advice. Telemed J E Health. Spring 2004;10(1):53–60.
   Dent RL. The effect of telephone nurse triage on the appropriate use of the emergency department. Nurs Clin N Am. 2010;45(1):65–69.
   Boushy D, Dubinsky I. Primary care physician and patient factors that result in patients seeking emergency care in a hospital setting: the patient's perspective. J Emerg Med. May/June 1999;17(3):405–412.
   Howard M, Goertzen J, Kaczorowski J, Hutchison B, Morris K, Thabane L, Levine M, Papaioannou A. Emergency department and walkin clinic use in models of primary care practice with different after-hours accessibility in Ontario. Healthcare Policy. August 2008;4(1):73–88.
   Howard M, Goertzen J, Kaczorowski J, Hutchison B, Morris K, Thabane L, Levine M, Papaioannou A. Emergency department and walkin clinic use in models of primary care practice with different after-hours accessibility in Ontario. Healthcare Policy. August 2008;4(1):73–88.
   Health Council of Canada. Teams in action: Ontrio. 2009. www.healthcouncilcanda.ca/docs/ rpts/2009/TIA\_Ontario.pdf; accessed October 4, 2010.
   Ontario Long Term Care Association. Long-term care homes joining Ontario Telemedicine Network. www.oltac.com/xaiom/DailyNews/2009/September/September10.html; accessed October 3, 2010.
   UHN Nursing: a blog for nurses. uhnnursing.wordpress.com/2010/03/29/mobile-emergency-nurses-making-acute-care-house-calls-to-long-term-care-residents; accessed October 3, 2010.
   Central East LHIN. NPSTAT Nurse Practitioners Supporting Teams Averting Transfers. www.centraleastlin.on.
- 625
- 626

- 629 630
- 631
- December 6, 2010. www.centraleasthin.on.ca/uploadedFiles/Home\_Page/Report\_and\_Publications/AAH\_Year\_Three\_ WEBSITE\_POSTING.pdf; accessed March 1, 2010. www.health.gov.on.ca/transformation/fht/guides/fht\_thas.pdf; accessed December 8, 2010. Lira R, Nascimento MA, Moreira-Filho DC, Kara-José N, Arieta C. Are routine preoperative medical tests needed with cataract surgery? Rev Panam Salud Publica/Pan Am J Public Health. 2001;10(1):21,27 633 2001:10(1):13-17. 634
- 635
- 637
- 2001;10(1):13-17.
   Fretheim A, Aaserud M, Oxman AD. The potential savings of using thiazides as the first choice antihypertensive drug: costminimisation analysis. BMC Health Services Research. 2003;3:18.
   Yuan H, et al. Elimination of preoperative testing in ambulatory surgery. Can J Anaesth. 2006;53:264-266.
   Schein OD, et al. The value of routine pre-operative medical testing before cataract surgery. NEJM. 2000;34(2):168-175.
   Imasogie N, et al. Elimination of routine testing in patients undergoing cataract surgery allows for substantial savings in laboratory costs: a brief report. Can J Anesth. 2003;50:246-248.
   In Ortario, FY 2009/10, Institute for Clinical Evaluative Sciences.
   Ontario Guidelines Advisory Committee. Summary of recommended guideline hypertension: pharmacologic management. Reference #241. September 2005. www.gacguidelines.ca/site/GAC Guidelines/assets/pdf/HYPE05-Pharmacologic\_Management.pdf; accessed November 1, 2010.
- Conductinesy assets your first cost manufactory of using the pretion of the pretion

- <sup>645</sup> Vartanians VM, Sistrom CL, Weilburg JB, Rosenthal DI, Thrall JH. Increasing the appropriateness of outpatient imaging: effects of a barrier to ordering low-yield examinations. Radiology. 2010 Jun;255(3):842-9.
- 2010 Jun;250(3):842-9. CMS announces selection of demonstration participants in medicare imaging demonstration project. Centres for Medicare and Medicaid, Washington, February 2, 2011. Available at: www. cms.gov/apps/media/press/release.asp?Counter=3896&intNumPerPage=10&checkDate=&chec kKey=&srchType=1&numDays=3500&srchOpt=0&srchData=&keywordType=All&chkNewsType=1 %2C+2%2C+3%2C+4%2C+5&intPage=&showAll=&pYear=&year=&desc=&cboOrder=date; accessed March 14\_2011

- %2C+2%2C+3%2C+4%2C+5&intPage=&showAll=&pYear=&year=&desc=&cboOrder=date; accessed March 14, 2011.
   <sup>647</sup> van Walraven C, Goel V, Chan B. Effect of population-based interventions on laboratory utilization: a time-series analysis. JAMA. 1998 Dec 16;280(23):2028-33.
   <sup>648</sup> Jamtvedt G, Young JM, Kristoffersen DT, et al. Audit and feedback: effects on professional practice and health care outcomes. Cochrane Review. In: The Cochrane Library. 2006;2. onlinelibrary.wiley. com/o/cochrane/clsysrev/articles/CD000259/frame.html; accessed January 14, 2011.
   <sup>649</sup> Doumit G, Gattellari M, Grimshaw J, et al. Local opinion leaders: effects on professional practice and health care outcomes. Cochrane review). 2009. http://onlinelibrary.wiley.com/o/cochrane/ clsysrev/articles/CD000125/frame.html. Accessed January 14, 2011.
   <sup>650</sup> Majumdar SR, Tsuyuki RT, McAlister FA. Impact of opinion leader-endorsed evidence summaries on the quality of prescribing for patients with cardiovascular disease: a randomized controlled trial.
- the quality of prescribing for patients with cardiovascular disease: a randomized controlled trial. Am Heart J 2007;153:22.e1–8. Spurling GK, Mansfield PR, Montgomery BD, Lexchin J, Doust J, et al. Information from
- Spuring GK, Marisheid FK, Montgörnery BD, Lexcrin J, Doust J, et al. Information from pharmaceutical companies and the quality, quantity, and cost of physicians' prescribing: a systematic review. PLoS Med. 2010;7(10). Kondro W. Academic drug detailing: an evidence-based alternative. CMAJ. 2007;176(4):429–431. Simon SR, Majumdar SR, Prosser, LA, et al. Group versus individual academic detailing to improve the use of antihypertensive medications in primary care: a cluster-randomized controlled trial. American Journal of Medicine. 2005;118:521–528. 653

- the use of antihypertensive medications in primary care: a cluster-randomized controlled trial.
   American Journal of Medicine. 2005;118:521–528.
   Davis DA, Taylor-Vaisey A. Translating guidelines into practice. a systematic review of theoretic concepts, practical experience and research evidence in the adoption of clinical practice guidelines. CMAJ. 1997;157(4):408–416.
   Marcum ZA, Handler SM, Wright R, Hanlon JT. Interventions to improve suboptimal prescribing in nursing homes: a narrative review. Am J Geriatr Pharmacother. 2010;8(3):183–200.
   Guidelines Advisory Committee. Annual Report 2005–2006. www.gacguidelines.ca/site/GAC\_Guidelines/assets/pdf/Website\_GAC05\_06Annual\_Report.l.pdf; accessed January 10, 2011.
   www.health.gov.on.ca/english/providers/program/ohip/bulletins/4000/bul4515.pdf; accessed December 9, 2010.
   National Health Expenditure Trends, 1975 to 2010, page 57, 111, and 139-141, October 2010.
   Etriksson D, Bjornberg A. Euro-Canada Health Consumer Index, 2009. Policy series 61. Frontier Centre for Public Policy & Health Consumer Powerhouse. ISSN 1491-78. www.healthpowerhouse.com/files/09-05-25 Euro-Canada&20Health%20Consumer%20Index%202009%20FINAL[1].pdf, accessed October 29, 2010.
   OECDHealth Data 2008; WHO World Health Survey 2006, personal interviews from www.healthpowerhouse.com/files/sg\_indicators\_2008/5.5.%20Mammograph%20reach.pdf, accessed March 14, 2011.
   GEGDEHealth Data 2001.
   Global Health Devatory Database http://apps.who.int/ghodata/, accessed March 14, 2011.
   Global Health Inforway. www2.inforwations.2008/JOB.Sc. %20Mammograph%20reach.pdf, accessed March 14, 2011.
   Global Health Inforway. The second the structure of Public Policy Conversion from www.healthpowerhouse.com/files/sg\_indicators\_2008/5.5.%20Mammograph%20reach.pdf, accessed March 14, 2011.
   GOECDHealth Data 2008/2011.
   Global Health Inforway. w

- Canada Health Infoway. www2.inforoute.ca/Documents/Telehealth%20Workshop\_ Converging%20Silos%20Report\_EN.pdf; accessed May 10, 2010. Henkin RE, Harolds JA. Health information technology and the electronic medical record. Clin Nucl Med. 2010;35:788–789. 666
- Henkin RE, Harolds JA. Health information technology and the electronic medical record. Clin Nucl Med. 2010;35:788–789.
   Pederson L, Leonard K. Measuring information technology investment among Canadian academic health science centres. Electronic Healthcare. 2005;3(3):94–102. eHealth health system facts and figures from Ontario Hospital Association. www.healthystemfacts.com/client/oha/hsf\_lp4w\_Ind\_webstation.nsf/page/ehealth; accessed October 29, 2010.
   Canada Health Infoway. www.infoway.inforoute.ca; accessed October 29, 2010.
   Canada Health Infoway. userid=2&control=SetCommunityPage&cached=true&parentiame=Community Page&parentid=0&in\_hi. userid=2&control=SetCommunityPage&cached=true&parentiame=Community Page&parentid=0&in\_hi. userid=2&control=SetCommunityPage&cached=true&parentiame=Community Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentiame=Community Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentiame=Community Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentiame=Community\_Page&cached=true&parentiame=Community\_Page&cached=true&parentiame=Community\_Page&cached=true&parentiame=Community\_Page&cached=true&parentiame=Community\_Page&cached=true&parentiame=Community\_Page&cached=true&parentiame=Community\_Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentine=community\_Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentname=Community\_Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentname=Community\_Page&cached=true&parentname=Community\_Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentname=Community\_Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentname=Community\_Page&parentid=0&in\_hi. userid=2&control=SetCommunity\_Page&cached=true&parentname=Community\_Page&cached=true&parentname=Community\_Page&cached=true&parentname=Community\_Page&cached=true&parentname=Comm

- <sup>1</sup> Final Sigurd M, Bates DW, Brighan and Wolfners Prospital CPC working and provide the analysis of the system. J Am Med Inform Assoc. 2006;13(3):365–367.
   <sup>677</sup> Barlow S, Johnson J, Steck J. The economic effect of implementing an EMR in an outpatient clinical setting. Journal of Healthcare Information Management. Winter 2004;18(1). www.himssehra.org/docs/caseStudies/Allscripts\_JHIM\_Central#20Utah%20Utah%20Case%20 Study-pdf; accessed February 4, 2011.
   <sup>678</sup> Walker J, Pan E, Johnston D, Adler-Milstein J, Bates DW, Middleton B. The value of health care information exchange and interoperability. Health Aff: 2005;web-exclusive suppl:WS-10-W5-18. content.healthaffairs.org/content/early/2005/01/19/hlthaff.w5.10.long; accessed January 10, 2011.
   <sup>679</sup> Work K. Information technology for the treatment of diabetes: improving outcomes and controlling cost. Journal of Managed Care Pharmacy. 2008;14(2)SA suppl:S12–S17. www.amc.org/data/jmcc/JMCPSupp\_March%2008 S12-S17.pdf; accessed January 10, 2011.
   <sup>680</sup> Hook JM, Grant E, Samarth A. Health information technology and health information exchange implementation in rural and underserved areas: findings from the AHRQ health IT portfolio. Rockville, MD: Agency for Health care Research and Quality, February 2010. health:ahrq.gov/portal/server.pt/gateway/PTARGS\_0\_11699\_911225\_0\_0\_0\_18/H1%20Health%20Info%20rural.pdf; accessed January 10, 2011.
   <sup>681</sup> Moidudin A, Moore J. The underserved and health information technology: issues and opportunities. Paper prepared for the Office of the Assistant Secretary for Planning and Evaluation and U.S. Department of Health and Human Services. November 2008. aspe.hths.gov/sp/ report5/2009/underserved/report.pdf; accessed January 10, 2011.
   <sup>682</sup> Moidudin A, Moore J. The underserved and health information technology: issues and opportunities. Paper prepared for the Office of the Assistant Secretary for Planning and Evaluation and U.S. Department of Health and H

- Joos D, Chen Q, Jirjis J, Johnson KB. An electronic medical record in primary care: impact on satisfaction, work efficiency and clinic processes. AMIA Annu Symp Proc. 2006;394–398. www. ncbi.Imm.nli.gov/pmc/articles/PMC1839545; accessed February 4, 2011.
   Lorenzi NM, Kouroubali A, Detmer DE, Bloomrosen M. How to successfully select and implement electronic health records (EHR) in small ambulatory practice settings. BMC Medical Informatics and Decision Making. 2009;9(15).
   OntarioMD. Transition Support Program. www.ontariomd.ca/portal/server.pt?space=Community Page&ached=true&Qarentame=CommunityPage&parentid=O&in.hi.userid=2&control=SetComm units&CommunityID=554&PageID=0; accessed February 13, 2011.
   Dawes M, Chan D. Knowing we practise good medicine: implementing the electronic medical record in family practice. Can Fam Physician. January 2010;56(1):15–16. www.cfp.ca/cgi/content/full/56/1/15; accessed February 4, 2011.
   Shoniregun CA, Dube K, Mtenzi F. Introduction to e-healthcare information security. Advances in Information Security. 2010;53:1–27.
   Sequist TD, Cullen T, Hays H, Taualii MM, Simon SR, Bates DW. Implementation and use of an electronic health record within the India Health Service. J Am Med Inform Assoc. 2007;14:191–197.

- 692
- electronic health record within the Indian Health Service. J Am Med Inform Assoc. 2007;14:191–197. Torda P, Han ES, Scholle SH. Easing the adoption and use of electronic health records in small practices. Health Affairs. 2010;29(4):668–675. mhcc.maryland.gov/electronichealth/ easing\_the\_adoption\_and\_use\_fina\_0410.odf; accessed January 10, 2011. Lawrence D. Can you hear me now? Voice recognition for the EMR has made big strides, and many say meaningful use requirements will accelerate adoption. Healthcare Informatics. 2009;26. www. healthcare-informatics.com/ME2/dirmod.asp?sid=&nm=&type=Publishing&mod=Publications%3A %3AArticle&mid=873A7027421841978F188E895F87F791&tier=4&id=1C09651233894BEABB DP9PEE01470DP7: accessed hanvary 10, 2011 693
- %3AArticle&mid=8F3A7027421841978F18BE895F87F7918tier=4&id=1C09651233894BEABB DD8DEF01470DD7; accessed January 10, 2011. Fung CH, Tsai JS, Lulejian A, Glassman P, Patterson E, Doebbeling BN, Asch SM. An evaluation of the Veterans Health Administration's clinical reminders system: a national survey of generalists. J Gen Int Med. 2008;23(4):392–398. Canada Health Infoway. Standards Collaborative Guide. 2010. www.infoway-inforoute.ca/flash/ langen/scguide/docs/StandardsCatalogue\_en.pdf; accessed May 10, 2010. Alberta Netcare. 2006. www.albertanetcare.ca/RXBrochure.pdf; accessed January 10, 2011. PACS in the UK: roundup of progress to date. E-Health Insider. www.e-health-insider.com/Features/ item.cfm?docId=198; accessed January 10, 2011. Canada Health Infoway. www.infoway-inforoute.ca/about-infoway/news/news-releases/637; accessed May 10, 2010. 694
- 695
- 697
- 698

- item.cfm?docld=198; accessed January 10, 2011.
   Canada Health Inforway, improving outcomes: focus on workplace issues. Curr Opin Crit Care. 2006;12(5):395–398.
   Rossberg JI, Melle I, Opjordsmoen S, Friis S. The relationship between staff members' working conditions and patients' perceptions of the treatment environment. International Journal of Social Psychiatry. 2008;54(5):437–446.
   There were 708,700 healthcare and social assistance workers in Ontario in 2009. There were 7,175,100 people employed in Ontario in 2009. Statistics Canada, Canada Year Book, 2010.
   Statistics Canada. Work Absence, 2009. Catalogue no. 71-211-X.
   Health and Safety Statistics for Ontario, Workplace Safety and Insurance Board. Available at: www. wsib.on.ca/en/community/WSIB/230/ArticleDetail/24338?ygnextoid=2609e35c819d7210VgnVC MI 00000449c710aRCRD. Accessed March 10, 2011.
   Institute for Work & Health. Workers' compensation and the business cycle. Toronto: Institute for Work and Health, March 2009. www.inh.on.ca/briefings/business-cycle; accessed March 11, 2011.
   Statistics Canada. Work Absence, 2009. Catalogue no. 71-211-X.
   ActNowBC. Creating a healthy workplace environment: workbook and toolkit. www.actnowbc.ca/ media/Workbook.pdf; accessed January 11, 2011.
   Feletto M, Graze W. A back injury prevention guide for health care providers. www.dir.ca.gov/dosh/dosh\_publications/backin\_Jdf; accessed January 11, 2011.
   Occupational Safety and Health Administration. Guidelines for preventing workplace violence for health care and social service workers. Publications J3148-01R. Washington: Occupational Safety and Health Administration. Guidelines for preventing workplace violence for health care and social service workers. Publications/SoSpecial/2026/itom%20-%2020 Violence\*200m/2024. Www.osha.gov/Publications/OSHA148/osha3148.html; accessed January 11, 2011.
   Ontario Hospital Association. Health a

- Dement JM, Pompeii LA, Østbye T, Epling C, Lipscomb HJ, James T, Jacobs MJ, Jackson G, Thomann W. An integrated comprehensive occupational surveillance system for health care workers. American J Indus Med. 2004;45(6):528–538.
   ActNowBC. Creating a healthy workplace environment: workbook and toolkit. www.actnowbc.ca/ media/Workbook.pdf; accessed January 11, 2011.
   Gaba DM, Howard SK. Fatigue among clinicians and the safety of patients. N Engl J Med. 2002;347:1249–1255. www.neimorg/doi/ful/10.1056/NEJMs3020846; accessed February 4, 2011.
   Lemaire JB, Wallace JE, Dinsmore K, Lewin AM, Ghali WA, Roberts D. Physician nutrition and cognition during work hours: effect of a nutrition based intervention. BMC Health Services Research. 2010;10:241. www.biomedcentral.com/1472-6963/10/241; accessed February 4, 2011.
   Ngan K, Drebit S, Siow S, Yu S, Keen D, Alamgir H. Risks and causes of musculoskeletal injuries among health care workers. Occup Med. May 2010. occmed.oxfordjournals.org/content/ early/2010/05/16/occmed.kqq052.full; accessed January 11, 2010
   Rosenthal K. Keeping I.V. therapy safe with needleless systems. Nursing. October 2003;33:16–20.
   Boiselle PM, Levine D, Horwich PJ, Barbaras L, Siegal D, Shillue K, Affeln D. Repetitive stress symptoms in radiology: prevalence and response to ergonomic interventions. J Am Coll Radiol. August 2008;5(8):919-922.
   Nielsen K, Trinkoff A. Applying ergonomics to nurse computer workstations: review and recommendations. Comput Inform Nurs. May/June 2003;21(3):150–157.
   Lockley SW, Barger LK, Ayas NT, Rothschild JM, Czeisler CA, Landrigan CP, The Harvard Work Hours Health and Safety Group. Effects of health care provider work hours and sleep deprivation on safety and performance. Joint Commission Journal on Quality and Patient Safety. November 2007;31(2)(Suppl 1):7–18.
   Canadian Centre for Occupational Health and Safety. Rotational shiftwork. www.cosh.ca/oshanswe

- <sup>728</sup> www.health.gov.on.ca/en/ms/ecfa/public/default.aspx; accessed December 14, 2010.

- <sup>729</sup> www.oha.com/Services/HealthyWorkEnvironments/QualityHealthCareWorkplaceAwards/Pages/ Default.aspx; accessed December 14, 2010.
   <sup>730</sup> www.healthforceontario.ca/upload/en/media/health%20sector%20labour%20market%20 policy%20branch%20%20june%2017th%20symposium.pdf; accessed December 14, 2010.
   <sup>731</sup> www.ofa.gov.on.ca/en/franco-map.html; accessed October 29, 2010.
   <sup>732</sup> news.ontario.ca/mohltc/en/2010/05/creating-more-opportunities-for-nurses.html; accessed December 14, 2010.
   <sup>733</sup> www.health.gov.on.ca/en/franco-map.html; accessed 2010/nov/mr, 2010/1109 oppy: accessed December 14, 2010.

- 733
- www.health.gov.on.ca/en/news/release/2010/nov/nr\_20101108.aspx; accessed December 14, 2010. www.health.gov.on.ca/en/news/release/2010/may/nr\_20100511.aspx; accessed December 14, 2010. www.health.gov.on.ca/english/public/program/filhs/health\_planning\_entities.html; accessed February 15, 2011. 735
- http://ths.mcmaster.ca/main/news/news\_2010/physician\_assistant\_graduation\_2010.html; accessed March 14, 2011.
- www.healthforceontario.ca/upload/en/work/paprojectupdatesummer2010.pdf; accessed
- Www.neatmorceontario.ca/upload/en/work/paprojectupdatesummer2010.pdr; accessed March 14, 2011.
   http://www.health.gov.on.ca/en/news/release/2010/nov/nr\_20101119.aspx; accessed March 14, 2011.
   Kripalani S, et al. Deficits in communication and information transfer between hospital-based and primary care physicians: implications for patient safety and continuity of care. JAMA. 2007;297(8):831–841.
   Deficits and the safety and continuity of care. JAMA.
- 2007;297(8):831–841. Nelson EA, Maruish ME, Axler JL, Effects of discharge planning and compliance with outpatient appointments on readmission rates. Psychiatr Serv. July 2000;51(7):885–889. Chan BTB, Schull MJ, Schultz SE. Atlas Reports: emergency services in Ontario from 1993 to 2000. Toronto: Institute for Clinical Evaluative Sciences; 2001. Discharge Abstract Database (DAD), FY 2008/09, calculated by Institute for Clinical Evaluative 740

- Chan BTB, Schull MJ, Schultz SE. Atlas Reports: emergency services in Ontario from 1993 to 2000. Toronto: Institute for Clinical Evaluative Sciences: 2001.
   Discharge Abstract Database (DAD), FY 2008/09, calculated by Institute for Clinical Evaluative Sciences.
   Nelson EA, Maruish ME, Axler JL. Effects of discharge planning and compliance with outpatient appointments on readmission rates. Psychiatr Serv. July 2000;51(7):885–889.
   DAD, Ontario Mental Health Reporting System, calculated by Institute for Clinical Evaluative Sciences, FY 2009/10.
   Duncan PW, Zorowitz R, Bates B, Choi JY, Glasberg JJ, et al. Management of Adult Stroke Rehabilitation Care: A Clinical Practice Guideline. Stroke. 2005;36:e100–e143.
   Cfu DX, Stewart DG. Factors affecting functional outcome after stroke: a critical review of rehabilitation inter rehoritons. Arch Phys Med Rehabil. 1999;80(5 suppl 1):S35–S39.
   Ontario Broke System Stroke Evaluation Office, National Rehab System Database, 2005/06.
   Target recommended by the Heart and Stroke Foundation, Teasel R, et al. Evidence-based review of stroke rehabilitation. Heart and Stroke Foundation, 12003.
   www.powerstudy.ca/thepower-report/the-power-report-volume-1/depression; accessed November 5, 2010.
   Taylor DM, Carmeron PA. Discharge instructions for emergency department patients: what should we provide? J Accid Emerg Med. March 2009;17(1):86–90.
   Yu KT, Green RA. Critical aspects of emergency department documentation and communication. Emerg Med Clin North Am. 2009;27(4):641–654, ix.
   Sudore RL, Schillinger D. Interventions to Improve Care for Patients with Limited Health Literacy. J Clin Outcomes Manag. January 1, 2009;16(1):20–29.
   Schillinger D, Piette J, Grumbach K, et al. Closing the loop. Physician communication with diabetic patients who have low health literacy. Arch Intern Med. 2003;163:83–90.
   Williams MV, Davis T
- May 2005;7(3):150–101.
   <sup>761</sup> Afilalo M, Lang E, Léger R, Xue X, Colacone A, Soucy N, Vandal A, Boivin JF, Unger B. Impact of a standardized communication system on continuity of care between family physicians and the emergency department. CLEM. March 2007;9(2):79–86.
   <sup>762</sup> Kripalani S, Jackson AT, Schnipper JL, Coleman EA. Promoting effective transitions of care at hospital discharge: a review of key issues for hospitalists. J Hosp Med. September 2007;2(5):314–323.

- <sup>176</sup> Teasell RW, Foley NC, Salter KL, Jutai JW. A blueprint for transforming stroke rehabilitation care in Canada: the case for change. Arch Phys Med Rehabil. March 2008;89(3):575–578.
   <sup>176</sup> Taylor SE, Sirois FM. Health psychology. Canadian ed. McGraw-Hill Ryerson Limited, 2009.
   <sup>176</sup> Taylor SE, Sirois FM. Health psychology. Canadian ed. McGraw-Hill Ryerson Limited, 2009.
   <sup>176</sup> Ottawa Charter for Health Promotion. World Health Organization, Geneva, 1986.
   <sup>170</sup> WW.who.int/hpr/NPH/docs/ottawa.charter.hp.pdf; accessed April 1, 2011.
   <sup>170</sup> U.S. Department of Health and Human Services. The health consequences of smoking:
   a report of the surgeon general. Centers for Disease Control and Prevention, 2004.
   <sup>170</sup> Www.cdc.gov/tobacco/data\_statistics/sgr/2004; accessed November 2009.
   <sup>171</sup> Ministry of Health Promotion and Sport. Creating a SmokeFree Ontario. Jan 2010.
   <sup>172</sup> Wuidence to Guide Action: Comprehensive Tobacco Control in Ontario", April 15, 2010, available at www.oahpp.ca/services/evidence-to-guide-action-ctc-in-ontario.html#downloads; accessed February 8, 2011.
   <sup>174</sup> National Institute on Alcohol Abuse and Alcoholism (NAAA). Health risks and benefits of alcohol

- February 8, 2011.
   National Institute on Alcohol Abuse and Alcoholism (NIAAA). Health risks and benefits of alcohol consumption. Alcohol Research & Health. 2000;24(1). pubs.niaaa.nih.gov/publications/arh24-1/05-11.pdf; accessed April 6, 2011.
   World Cancer Research Fund/American Institute for Cancer Research (AICR). Food, nutrition, physical activity and the prevention of cancer: a global perspective. AICR, Washington, DC, 2007.
   Rodgers H, et al. Alcohol and stroke: a case control study of drinking habits past and present. Stroke. 1993;24(10):1473–1477. Biyik, I. and Ergene O. Alcohol and acute myocardial infarction. J Intern Med Res 2007;35(1):46–51.
   World Cancer Research Fund/American Institute for Cancer Research (AICR). Food, nutrition, physical activity and the prevention of cancer: a global perspective. AICR, Washington, DC, 2007.
   National Institutes of Health (NIH). Clinical guidelines on the identification, evaluation, and treatment of overweight and obesity in adults: the evidence report. NIH, Bethesda, Maryland, September 1998. NIH publication no. 98-4083. www.nhlbi.nih.gov/guidelines/obesity/ob\_gdins.pdf; accessed November 2009. November 2009.
- November 2009.
   79 Katzmazyk PT, and Janssen I. The economic costs associated with physical inactivity and obesity in Ontario. 2001. Submitted to the Leisure Information Network. Kingston, Ontario.
   Rodkin A, Ding HK, Scale S. Obesity: an overview of current landscape and prevention-related activities in Ontario. Prepared for the Public Health Agency of Canada. April 30, 2009:1–76.
   Holmes MD, Chen WY, Hankinson SE, Willett WC. Physical activity's impact on the association of fat and fiber intake with survival after breast cancer. Am J Epidemiol. November 15, 2009:174(10):1256.
- 009;170(10):1250-1256
- Warburton DE. et al. Health benefits of physical activity: the evidence. CMAJ. 2006;174(6):801-809.

- <sup>783</sup> Katzmarzyk PT, Janssen I. The economic costs associated with physical inactivity and obesity in Ontario. 2001. Submitted to the Leisure Information Network. Kingston, Ontario.
   <sup>784</sup> Ministry of Health Promotion. http://www.healthunit.com/tFIKZWRrw&W192bukZ2rOsrmUZCNMyIT1dw ATiejHoulQawmqdiy34Mu2nNNoVoU4StspCNE2bjpV99/UGPL\_D046ZGyKndKZWrmccFNGhMao1)// articles/PDF/13650.pdf; accessed December 13, 2010.
   <sup>785</sup> Hall M, Moore S, Harper SB, Lynch JW. Global variability in fruit and vegetable consumption. Am J Prev Med. 2009;36(5):402–409, e5.
   <sup>786</sup> Lock J, et al. The global strategy on diet. Bull World Health Organization. 2005;83(2):100–108.
   <sup>787</sup> Www.hc.sc.gc.ca/hc.ps/tobac-tabac/legislation/federal/tobac-tabac-eng.php. The 1997 Federal Tobacco. Act limits on sales to minors. advertising and sponsorshin of sports events (hoased in Chaster).

- Tobacco Act limits on sales to minors, advertising and sponsorship of sports events (phased in

- implications for the global strategy on diet. Bull World Health Organization. 2005;83(2):100–108.
   www.hcs.gc.cz.Arc.brychoac-tabac/regisitaton/federal/tobac-tabac-eng.php. The 1997 Federal Tobacco Act limits on sales to minors, advertising and sponsorship of sports events (phased in over 5 years).
   www.hcs.gc.cz.Arc.brychubi/chubi/chubi/chubacco/act.html. Ontario's 1994 Tobacco Control Act bans selling tobacco to those under 19 and restricts smoking in some public places.
   www.hcs.gc.cz.Arc.brychubs/tobac-tabac/rc/index-eng.php. A Regulation to Federal Tobacco Act passed in 2000 mandtaded graphic warnings on tobacco products.
   Www.hcs.gc.cc.a/hc.ps/pubs/tobac-tabac/rc/index-eng.php. A Regulation to Federal Tobacco Act passed in 2000 mandtaded graphic warnings on tobacco products.
   Yatistics Canada, 2009, Canadian Community Health Survey, CANSIM table 105-0501.
   1999 Toorto City Counci plasses harmonized No Smoking Bylaw No. 441:1999.
   Statistics Canada, 2009, Canadian Community Health Survey, CANSIM table 105-0501.
   OECD Health Data 2010. www.oedc Decrember 13, 2011.
   Ministry of Health Promotion and Sport. ACTIVE/2010: Ontario's Sport and Physical Activity Strategy. August 2005. ISBN 0-7794-88733. http://www.mhp.gov.on.ca/en/active-living/about/active/strategy. August 2009. Canadian Community Health Survey, CANSIM table 105-0501.
   Stad LF, Perera R, Bullen C, Mant D, Lancaster T. Nicotine replacement therapy for smoking essation. Cochrare Database of Systematic Reviews 2008, Issue 1. Art. No.: CD000146. DOI: 10.102/14/651858.CD00146, Feub3.
   Schutz ASH, Nowatzki J, Dunn DA, Griffith LJ. Effects of socialization in the household on youth susceptibility is smoking: a secondary analysis of the 2004/06 Canadian Youth Smoking Survey. Chronic Diseases in Canada. 2010;30(3): 71–77.
   Ontario Tobacco Research Unit. The Tobacco Control Chroin Charia and Beyond. Mon
- Federal AC, Latalize GN, Shapiro S, Overend T. Improving aerouic miless in order audits. effects of a physician-based exercise counseling and prescription program. Can Fam Physician. May 2010;56(5):e191–200. The Low-Cost, Moderate-Cost, and Liberal Food Plans, 2007 (CNPP-20). U.S. Department of Agriculture, Center for Nutrition Policy and Nutrition, Washington. www.cnpp.usda.gov/ Publications/FoodPlans/MiscPubs/FoodPlans2007AdminReport.pdf; accessed April 1, 2011. Ontario Medical Association. Ontario's Doctors Call for Calorie Labelling on Fast Food and Cetebric Manuel April 2000. usua one or 40 AdminRepole.pdf 811

- of Agriculture, Center for Nutrition Policy and Nutrition, Washington. www.cnpp.usda.gov/ Publications/FoodPlans/MiscPubs/FoodPlans2007AdminReport.pdf: accessed April 1, 2011.
   Ontario Medical Association. Ontario's Doctors Call for Calorie Labelling on Fast Food and Cafeteria Menus. April 2009. www.oma.org/Mediaroom/PressReleases/Pages/ CallforCalorieLabelling.aspx; accessed January 17, 2010.
   Schillinger D, Piette J, Grumbach K, et al. Closing the loop. Physician communication with diabetic patients who have low health literacy. Arch Intern Med. 2003;163:83–90.
   Flores, G. The impact of medical interpreter services on the quality of health care: a systematic review. Med Care Res Rev. 2005;62(3):255–299.
   Ministry of Health Promotion and Sport. ACTWE2010: Ontario's Sport and Physical Activity Strategy. 2005. www.mhp.gov.on.ca/en/active-living/about/active2010-strategy-e.pdf; accessed February 9, 2011.
   Glazier RH, Booth GL, Gozdyra P, Creatore MI, Tynan M, editors. Neighbourhood Environments and Resources for Healthy Living—A Focus on Diabetes in Toronto: ICES Atlas. Toronto: Institute for Clinical Evaluative Sciences; 2007. www.ices.on.ca/webgae.cfm?site.id=1 & Rorg\_id=67&morg\_ id=0&grec\_id=0&item id=4406&type=atlas; accessed January 17, 2011.
   Strong WB, Malina RM, Blimkie CJ, Daniels SR, Dishman RK, Gutin B, Hergenroeder AC, Must A, Nixon PA, Pivarnik JM, Rowland T, Trost S, Trudeau F. Evidence based physical activity for schoolage youth. J Pediatr. June 2005;146(6):732–737.
   Searajft R. Realistic approaches to counseling in the office setting. Am Fam Physician. February 15, 2009;79(4):277–284.
   Grimshaw JM, Thomas RE, MacLennan G, et al. Effectiveness and efficiency of guideline dissemination and implementation strategies. Health Technol Assess. 2004;8(6). www.hta.ac.uk/ execsumm/summ806.htm; accessed January 17, 2011.
   Sequist D, Gandhi TK, Karson AS. A randomized trial of electronic cl

- <sup>829</sup> Zero to Three. Brain Development. www.zerotothree.org/child-development/brain-development; accessed September 2010.
   <sup>830</sup> Net Industries. Prenatal Development Prenatal Environmental Influences. http://social.jrank.org/pages/515/Prenatal-DevelopmentPrenatal-Environmental-Influences.html; accessed September 2010.
   <sup>830</sup> BORN Ontario, FY 2009/10.
   <sup>832</sup> Association of Public Health Epidemiologists in Ontario, 2005.
   <sup>833</sup> Das UG, Sysyn GD. Abnormal fetal growth: intrauterine growth retardation, small for gestational age, large for gestational age. Paediatric Clinics of North America. 2004;51(3):639–654.
   <sup>834</sup> O'Keeffe MJ, O'Callaghan M, Williams GM, Najman JM, Bor W. Learning, cognitive, and attentional problems in adolescents born small for gestational age. Padiatrics. 2003;112:301–307. Lundgren EM, Cnattingius S, Jonsson B, Tuvemo T. Intellectual and psychological performance in males born small for gestational age at term in a regional cohort study. Paediatrics. 2001;108:111–115.
   <sup>835</sup> Barker DJ, Winter PD, Osmond C, Margetts B, Simmonds SJ. Weight in infancy and death from ischaemic heart disease. Lancet. 1989;(2):577–580.
   <sup>836</sup> Hales CN, Barker DJ, Clark PM, Cox LJ, Fall C, Osmond C, Winter PD. Fetal and infant growth and impaired glucose tolerance at age 64. BMJ. 1991;303:1019–1022.
   <sup>837</sup> Barker DJ. Fetal nutrition and cardiovascular disease in later life. British Medical Bulletin. 1997;53(1):96–108.
   <sup>836</sup> Cole C, Haaadorn L, Kim C, et al. Criteria for determining disability in infants and children: low birth

- Impaired glucose tolerance at age 64. BMJ. 1991;303:1019–1022.
   Barker DJ. Fetai nutrition and cardiovascular disease in later life. British Medical Bulletin. 1997;53(1):96–108.
   Cole C, Hagadorn J, Kim C, et al. Criteria for determining disability in infants and children: low birth weight: Evidence report/technology assessment no. 70. Agency for Healthcare Research and Quality. Rockville, Maryland, 2002. Matsuo, H. The health consequences of low birth weight: literature review and critique. UCL working paper no. 23. l'Université catholique de Louvain, Louvain-la-Neuvem, Belgium, 2005.
   Health Canada. Smoking and Your Body: Pregnancy. Nov 22, 2007. www.hc-sc.gc.ca/hc-ps/ tobac-tabac/body-corps/pregros-eng.php; accessed November 16, 2010.
   Abel EL. Smoking during pregnancy: a review of effects on growth and development of offspring. Human Biology. 1980;52(4):593–625.
   College of Family Physicians of Canada. Infant feeding policy statement 2004. College of Family Physicians of Canada, Toronto, 2004. www.cfp.cca/local/files/Communications/Health%20Policy/ Final\_O4Infant\_Feeding\_Policy\_Statement.pdf; accessed Docember 2009.
   Canadian Paediatric Society, Dieticians of Canada. Health Canada. Benefits of breast-feeding to infants in Canada. In: Nutrition for healthy term infants. Minister of Public Works and Government Services Canada, Ottawa, 2005.
   Owen CG, Martin RM, Whincup PH, Smith GD, Cook DG. Does breast-feeding influence risk of type 2 diabetes in later life? A quantitative analysis of published evidence. American Journal of Clinical Nutrition, 2006;84(5):1043–1054.
   Newcomb PA, et al. Lactation and a reduced risk of premenopausal breast cancer. New England

- Valuetes in Jater Iner, Audantizative analysis of published evidence. American Journal of Clinical Nutrition. 2006;84(5):1043–1054.
   Newcomb PA, et al. Lactation and a reduced risk of premenopausal breast cancer. New England Journal of Medicine. 1994;330(2):81–87. Byers T, et al. Lactation and breast cancer: evidence for a negative association in premenopausal women. American Journal of Epidemiology. 1985;121:664–674. Siskind V, et al. Breast cancer and breastfeeding: results from an Australian case-control study. American Journal of Epidemiology. 1989;130:229–236. Brock, KE. Sexual, reproductive, and contraceptive risk factors for carcinoma-in-situ of the uterine cervix in Sydney. Medical Journal of Australia. 1989. Schneider, AP. Risk factor for ovarian cancer. New England Journal of Medicine. 1987.
   Infant mortality rates by provinces and territories. Statistics Canada. www40.statcan.gc.ca/101/cst01/health21a-eng.htm; accessed April 1, 2011.
   OECD Health Data 2010. www.oecd.org/document/16/0,3343
   GECD Health Data 2010. www.oecd.org/document/16/0,3343
   Canadian Institute for Health Information. Keeping Canada's Future Healthy. Summer 2007. www. cihi.ca/ClHi-extportal/pdf/internet/NLETTER 01JUL2007. PDF EN; accessed November 2009.
   Best Start. Reflecting on the Trend: Pregnancy After Age 52. 2007. www.beststart.org/resources/ rep.health/pdf/sp.pregnancy\_age35.pdf; accessed November 2009.
   World Health Organization. Breastfeeding. www.who.int/topics/breastfeeding/en; accessed November 2009.
   Infant Mortality Health Survey, 2009.
   Infant Mortality Bates for 10 Leading Causes of Infant Death Linited States. 2005

- November 2009.
   <sup>850</sup> Canadian Community Health Survey, 2009.
   <sup>851</sup> Infant Mortality Rates for 10 Leading Causes of Infant Death United States, 2005.
   MMWR, October 26, 2007 / 56(42);1115. Centers for Disease Control, Washington.
   www.cdc.gov/mmwr/preview/mmwrhtml/mm5642a8.htm; accessed April 1, 2011.
   <sup>852</sup> American Academy of Pediatrics Task Force on Sudden Infant Death Syndrome. The changing concept of sudden infant death syndrome: diagnostic coding shifts, controversies regarding the sleeping environment, and new variables to consider in reducing risk. Pediatrics.
   <sup>853</sup> Rourke Baby Record: Evidenced-based infant/child health maintenance guide. 2006. www rourkehabvrecord ca: accessed April 1 2011

- <sup>853</sup> Rourke Baby Record: Evidenced-based infant/child health maintenance guide. 2006. www.rourkebabyrecord.ca; accessed April 1, 2011.
   <sup>854</sup> Smylie J, Adomako P. Indigenous Children's Health Report: Health Assessment in Action. Toronto: Centre for Research on Inner City Health; 2009. www.stmichaelshospital.com/crich/indigenous\_ childrens. health\_report.php; accessed January 17, 2011.
   <sup>855</sup> Office of the Provincial Health Officer. The Health and Well-being of the Aboriginal Population in British Columbia: Interim Update. Victoria: Government of British Columbia; 2007. www.health.gov. bc.ca/library/publications/year/2007/Aboriginal\_Population\_Interim\_report\_Final.pdf; accessed January 17, 2011.
   <sup>856</sup> Hearman MI, Blanchard JF, Gupton AL, Moffatt ME, Currie RF. Risk factors for spontaneous preterm birth among Aboriginal and non-Aboriginal women in Manitoba. Paediatr Perinat Epidemiol. May 2005;19(3):181–193.
   <sup>857</sup> Lup ZC, Wilkins B, Kergner MS, Effect of neighbourbood income and maternal education on birth

- <sup>863</sup> Public Health Agency of Canada. Childhood and Adolescence. Feb 16, 2011. www.phac.aspc.gc.ca/ dca-dea/programs-mes/cpnp\_maineng.php; accessed January 17, 2011.
   <sup>864</sup> Registered Nurses' Association of Ontario. Breastfeeding Best Practice Guidelines for Nurses. Toronto: Registered Nurses' Association of Ontario; 2007. www.rnao.org/Page. asp?PageID=924&ContentID=795; accessed January 17, 2011.
   <sup>865</sup> McInnes RJ, Chambers JA. Supporting breastfeeding mothers: qualitative synthesis. J Adv Nurs. May 2008;62(4):407-427.
   <sup>866</sup> Britton C, McCormick FM, Renfrew MJ, Wade A, King SE. Support for breastfeeding mothers. Cochrane Database of Systematic Reviews 2007, Issue 1. Art. No.: CD001141. DOI: 10.1002/14651858.CD001141.pub3.
   <sup>867</sup> Chung M, Raman G, Trikalinos T, Lau J, Ip S. Interventions in primary care to promote breastfeeding: an evidence review for the U.S. Preventive Services Task Force. Ann Intern Med. 2008;149(8):565–582.
   <sup>868</sup> Ministry of Health and Long-Term Care. Public Health Unit locations. www.health.gov.on.ca/english/
- 2008;149(8):565–582. Ministry of Health and Long-Term Care. Public Health Unit locations. www.health.gov.on.ca/english/ public/contact/phu/phuloc\_mn.html; accessed November 7, 2010. Ontario Human Rights Commission. Policy on Discrimination Because of Pregnancy and Breastfeeding. 2008. www.ohrc.on.ca/en/resources/Policies/PolicyPregBreastfeedEN/pdf; accessed January 17, 2011. College of Midwives of Ontario. Becoming Registered. 2007. www.cmo.on.ca/ITM.php; accessed January 17, 2011.

- <sup>871</sup> Ontario Breastfeeding Committee. About Us. 2010. www.breastfeedingontario.org/about.html; accessed January 17, 2011.
   <sup>872</sup> World Health Organization. Nutrition: Baby-friendly Hospital Initiative. 2011. www.who.int/nutrition/ topics/bhi/en/index.html; accessed January 17, 2011.
   <sup>873</sup> Centre for Disease Control and Prevention. Sexual Health. www.cdc.gov/sexualhealth; accessed November 15, 2010.
   <sup>874</sup> World Health Organization, Sexual Health. www.who.int/topics/sexual\_health/en; accessed September 2010.
   <sup>875</sup> Centre for Disease Control and Prevention. Sexual Health. respective of Diseases of the Disease of Diseases of the Disease of Diseases of Disea

- September 2010.
   Centre for Disease Control and Prevention. Sexually Transmitted Diseases: Gonorrhea.
   Mar 16, 2011. www.cdc.gov/std/gonorrhea/default.htm; accessed April 6, 2011.
   Hook EW III and Handsfield HH. Gonococcal infections in the adult.htm: K. Holmes, P. Sparling, P. Markh et al (eds). Sexually Transmitted Diseases, 3rd Edition. New York: McGraw-Hill, 1999, 451-466
- Centre for Disease Control and Prevention. Sexually Transmitted Diseases: Syphilis. Apr 1, 2011. www.cdc.gov/std/syphilis/default.htm; accessed April 6, 2011. World Health Organization, HIV/AIDS. www.who.int/topics/hiv\_aids/en; accessed September 878
- 2010 Schulz TF. Cancer and viral infections in immunocompromised individuals. Int J Cancer. October 15, 2009;125(8):1755–1763. 879
- 880
- 883
- 2009;125(8):1755–1763.
   Lipton SA. AIDS:related dementia and calcium homeostasis. Ann N Y Acad Sci. December 15, 1994;747:205–224.
   Becker MH, Joseph JG. AIDS and behavioral change to reduce risk: a review. American Journal of Public Health. April 1988;78(4):394–410.
   Paranjothy S, Broughton H, Adappa R, Fone D. Teenage pregnancy: who suffers? Arch Dis Child. 2009;94(3):239–245.
   Kearney MS, Levine PB. Socio-economic disadvantage and early childbearing. National Bureau of Economic Research, Cambridge, Massachusetts, 2007. www.nber.org/papers/w13436.pdf; accessed November 2009.
   Chen XK, Wen SW, Fleming N, Demissie K, Rhoads GG, Walker M. Teenage pregnancy and adverse birth outcomes: a large oppulation-based retrospective cohort study. Int Journal Eoidemiology.
- 884 birth outcomes: a large population-based retrospective cohort study. Int Journal Epidemiology. April 2007;36(2):368–373. Wellings, K. Causes and consequences of teenage pregnancy. In: Baker P. Guthrie K. Hutchinson
- 885

- April 2007;36(2):368–373.
   Wellings, K. Causes and consequences of teenage pregnancy. In: Baker P, Guthrie K, Hutchinson C, Kane R, Wellings K, eds. Teenage pregnancy and reproductive health. Royal College of Obstetricians and Gynaecologists, London, 2007;70.
   Statistics Canada's 2005 population estimate. www.statcan.gc.ca/pub/91-520-x/00105/4095252-eng.htm, accessed November 2009.
   Trends in HIV incidence and prevalence: natural course of the epidemic or results of behavioural change? World Health Organization, Geneva 1999. www.who.int/hiv/strategic/surveillance/en/unaids. 99.12.pdf; accessed April 1, 2011.
   Ontario HIV Epidemiologic Monitoring Unit. HIV Update. Feb 27, 2009. www.phs.utoronto.ca/ohemu/HIVupdate.html, accessed March 14, 2011.
   Mill JE, Jackson RC, Worthington CA, Archibald CP, Wong T, Myers T, Prentice T, Sommerfeldt S. HIV testing and care in Canadian Aborginal youth: a community based mixed methods study. BMC Infect Dis. 2008 Oct 7;8:132.
   Public Health Agency of Canada. Canadian Guidelines for Sexual Health Education. 2008. www. publichealth.gc.ca/sti; accessed January 17, 2011.
   Yamada J, DiCenso A, Feldman L, Cormillott P, Wade K, Wignall R, Thomas H. A systematic review of the effectiveness of primary prevention provent sexually transmitted diseases in adolescents. Effective Public Health Practice Project, Ontario Ministry of Health, Region of Hamilton-Wentworth, Social and Public Health Scies Division. 1999. http://old.hamilton.ca/phcs/ephp/Research/Full-Review/98-99/Adolescent-STDPrevention-review.pdf; accessed January 17, 2011.
   Society of Obstetricians and Gynaecologists of Canada. www.sexualityandu.ca/; accessed January 17, 2011.
   Tops for Parents to Help Their Children Avoid Teen Pregnancy, National Campaign to Prevent Feen and Unplanned Pregnancy, Washington DC, 2007. www.thenationalcampaign.org/resources/10Tips.aspx; accessed April 1, 2011.
   Topis fo

- www.publichealth.gc.ca/sti; accessed January 17, 2011.
   Public Health Agency of Canada. Canadian Guidelines for Sexual Health Education. 2008.
   www.publichealth.gc.ca/sti; accessed January 17, 2011.
   Tapert SF, Aarons GA, Sedlar GR, Brown SA. Adolescent Substance Use and Sexual Risk-Taking Behavior. J Adolesc Health. 2001;28:181–189.
   Public Health Agency of Canada. Primary care and sexually transmitted infections. January 2010.
   www.publichealth.gc.ca/sti; accessed January 17, 2011.
   Tapert SF, Aarons GA, Sedlar GR, Brown SA. Adolescent Substance Use and Sexual Risk-Taking Behavior. J Adolesc Health. 2001;28:181–189.
   Public Health Agency of Canada. Primary care and sexually transmitted infections. January 2010.
   www.pubac-aspc.gc.ca/stdmts/sti-its/guide-lignesdir-eng.php: accessed November 7, 2010.
   MOHLTC. Ontario's response to HIV/AIDS. Feb 23, 2011. www.health.gov.on.ca/english/public/program/hivaids/general/ontario\_response.html; accessed February 14, 2011.
   MOHLTC. Sexual Health and sexually transmitted infections prevention and control protocols. www.health.gov.on.ca/english/public/program/hivaids/general/ontars/pregram/pubhealth/oph\_standards/ophs/progstds/protocols/sexual.health\_sti.pdf; accessed February 14, 2011.
   MOHLTC. Sexual Health and sexually transmitted infections prevention and control protocols. www.health.gov.on.ca/english/providers/program/pubhealth/oph\_standards/ophs/progstds/protocols/sexual.health\_sti.pdf; accessed February 14, 2011.
   Mohdel JS, et al. Reducing mortality from colorectal cancer by screening for fecal occult blood. NELM. 1993;328(19):1365–1371.
   Humphrey LL, et al. Breast cancer screening: a summary of the evidence for the US Preventive Services Task Force. Annals of Internal Medicine. 2002;137(5, Part 1):347–360.
   Canadian Breast Cancer Foundation, Breast cancer in Canada. www.cbcf.org/breast

- September 2010.
   Brown JP, Josse RG. 2002 clinical guidelines for the diagnosis and management of osteoporosis in Canada. CMAJ. 2002;167(90100).
   Law MR, Wald NJ, Meade TW. Strategies for prevention of osteoporosis and hip fracture. British Medical Journal. August 24, 1991;303:453–459.
   Chrischilles E, Shireman T, Wallace R. Costs and health effects of osteoporotic fractures. Bone. 1994;15(4):377–386.
   Statistics Canada CANSIM table 105-05011.
   OECD Health Data 2010. www.oecd.org/document/16/0,3343,en\_2649\_34631\_2085200\_ 1\_1\_1\_1,00.html; accessed April 1, 2011. Results for Chile represent an average over the past nine years.
- nine years.
   <sup>914</sup> Statistics Canada CANSIM table 105-05011. Zone 6, Nova Scotia has the best results (80% averaged over 2008 and 2009).
   <sup>915</sup> In January 2007, the Ontario government, in collaboration with Cancer Care Ontario, introduced ColonCancerCheck the first population-based provincial colorectal cancer screening program in Canada. The goal of ColonCancerCheck is to decrease mortality from colorectal cancer through early detection and treatment. See coloncancercheck.ca/docs/factsheets/fact\_sheet\_English.pdf for more information; accessed December 13, 2010.

- <sup>916</sup> Cancer Care Ontario. Ontario Cancer Plan 2008–2011. www.ontariocancerplan.on.ca; accessed
- Gancer Care Ontario. Ontario Concert. and Land Program. www.cancercare.on.ca/pcs/screening/ March 3, 2009.
   Cancer Care Ontario. Ontario Breast Screening Program. www.cancercare.on.ca/pcs/screening/ breastscreening/DBSP/; accessed January 17, 2011.
   Ministry of Health and Long-Term Care. ColonCancerCheck. coloncancercheck.ca; accessed Invest, 17, 2011.
- 919
- Ministry of realin and Long-term Care. ColonCancerCriteck. colonCancerCriteck.ca; accessed January 17, 2011. Karwalajtys T, Kaczorowski J, Lohfeld L, Laryea S, Anderson K, Roder S, Sebaldt RJ. Acceptability of reminder letters for Papanicolaou tests: a survey of women from 23 Family Health Networks in Ontario. J Obstet Gynaecol Can. October 2007;29(10):829–834. Kaczorowski J, Karwalajtys T, Lohfeld L, Laryea S, Anderson K, Roder S, Sebaldt RJ. Women's
- Kaczorowski J, Karwalajtys I, Lohfeld L, Laryea S, Anderson K, Roder S, Sebaldt NJ. Women's views on reminder letters for screening mammography: Mixed methods study of women from 23 family health networks. Can Fam Physician. June 2009;55(6):622–623. e1–4. Austoker J, Bankhead C, Forbes LJ, Atkins L, Martin F, Robb K, Wardle J, Ramirez AJ. Interventions to promote cancer awareness and early presentation: systematic review. Br J Cancer. December 3, 2009;101(Suppl 2):S31–S39. Jacobson TA, Thomas DM, Morton FJ, Offrutt G, Shevlin J, Ray S. Use of a low-literacy patient education tool to enhance pneumococcal vaccination rates. A randomized controlled trial. JAMA. 1999;282(7):646–650.

- <sup>1999</sup>/282(1):646–650.
   <sup>2923</sup> Amarkwah E, Ngwakongnwi E, Hude Q. Why many visible minority women in Canada do not participate in cervical cancer screening. Ethnicity & Health. 2009;14:4,337–349.
   <sup>2924</sup> Sent L, Ballem P, Paluck E, Velland L, Vogel AM. The Asian Women's Health Clinic: addressing cultural barriers to preventive health care. CMAJ. 1998;159(4):350-4.
   <sup>2955</sup> http://www.mwlusa.org/topics/sexuality/sexuality.html; accessed April 11, 2011.
   <sup>2966</sup> Miller D, Livingstone V, Herbison GP. Interventions for relieving the pain and discomfort of functional participation and the participation of the participat
- screening mammography. Cochrane Database of Systematic Reviews 2008, Issue 1. Art. No.: CD002942. D0I: 10.1002/14651858.CD002942.pub2 Seehusen DA, Johnson DR, Earwood JS, Sethuraman SN, Cornali J, Gillespie K, Doria M, Farnell E 4th,
- Seenusen DA, Jonnson DA, Earwood JS, Seninarinan SN, Cornain J, Ginespie A, Dora Mi, Farniel E 411, Lanham J. Improving women's experience during speculum examinations at routine gynaecological visits: randomised clinical trial. BMJ. 2006;333(7560):171. Epub 2006 Jun 27. Weber V, Bloom F, Pierdon S, Wood C. Employing the Electronic Health Record to Improve DiabetesCare: A Multifaceted Intervention in an Integrated Delivery System. J Gen Intern Med. 2008;23(4):379–382.
- Weber V, Bloom F, Pierdon S, Wood C. Employing the Electronic Health Record to Improve Diabetes Care: A Multifaceted Intervention in an Integrated Delivery System. J Gen Intern Med. 2008;23(4):379–382. 929
- 930
- 932
- 2008;23(4):379–382. Cancer Care Ontario. Ontario Breast Screening Program: Mobile Coach Visits. www.cancercare. on.ca/cms/One.aspx?portalld=1377&pageld=9486#mcoach; accessed January 17, 2011. Cancer Care Ontario. Ontario Cancer Plan 2011-2015. ocp.cancercare.on.ca; accessed December 23, 2010. Ministry of Health and Long-Term Care. Immunization. www.health.gov.on.ca/english/providers/ pub/pub\_menus/pub\_immun.html; accessed September 26, 2010. Taylor SE, Sirois FM. Health psychology. Canadian ed. McGraw-Hill Ryerson Limited, 2009. Etzioni R, et al. The case for early detection. Nature Reviews Cancer. April 2003;3:243–252. Herrman H. The need for mental health promotion. Australian and New Zealand Journal of Psychiatry. 2001;35:709–715. European Commission's Directorate for Public Health and Risk Assessment. http://ec.europa.eu 935
- r sycniatry. 2001;35:/09–/15. European Commission's Directorate for Public Health and Risk Assessment. http://ec.europa.eu/ health/ph\_determinants/environment/IPP/pp\_en.htm; accessed September 2010. Ministry of Health Promotion. Ontario's injury prevention strategy: working together for a safer, healthier Ontario. 2007. ISBM: 978-4249-4237-3. www.mhp.gov.on.ca/en/prevention/ injury-prevention/strategy.pdf; accessed November 15, 2010. Stack S. Media coverage as a risk factor in suicide. J Epidemiol Community Health. 2003;57(4):238-240.
- 937
- 938

- Injury-prevention/strategy.pdf; accessed November 19, 2010.
   Stack S. Media coverage as a risk factor in suicide. J Epidemiol Community Health. 2003;57(4):238–240.
   Canadian Mental Health Association, suicide statistics. www.ontario.cmha.ca/fact\_sheets. asp?cID=3965; accessed September 2010.
   Pompili M, Amador XF, Girardi P, Harkavy-Friedman J, et al. Suicide risk in schizophrenia: learning from the past to change the future. An Gen Psychiatry. March 16, 2007;16):10.
   Beautrais AL. Risk factors for suicide and attempted suicide among young people. Australian and New Zealand Journal of Psychiatry. 2000;34(3):420–436.
   Berry DA, Cronin KA, Plevritis SK, Fryback DG, Clarke L, Zelen M, Mandelblatt JS, Yakovlev AY, Habberna JDF, Feuer EJ, for the Cancer Intervention and Surveilance Modeling Network (CISNET) Collaborators. Effect of screening and adjuvant therapy on mortality from breast cancer. New England Journal of Medicine. 2005;353:1784–1792.
   Cancer Intervention and Surveilance Modeling Network (CISNET) Breast Cancer Collaborators. The Impact of Mammography and Adjuvant Therapy on U.S. Breast Cancer Mortality (1975–2000): Collective Results from the Cancer Intervention and Surveilance Modeling Network. Journal of the National Cancer Institute. 2006;36:1–126.
   Statistics Canada, CANSIM Table 102-0552.
   Corrigan PW, Lurie BD, Goldman HH, Slopen N, Medasani K, Phelan S. How adolescents perceive the stigma of mental ilness and alcohol abuse. Psychiatr Serv. 2005 May;56(5):544-50.
   The Mental Health Association. AntrStigma Campaign. www.mhasuffolk.org/documents/ antistigma.php; accessed April 8, 2011.
   National Centre for Tipury Prevention and Control. Youth Suicide Prevention Programs: A Resource Guide. Atlanta: Centers for Disease Control and Prevention; 1992. wonder.cdc. gov/wonder/ prevegid/p0000024/P0000024.aspfthead000200000000000000000; accessed January 17, 2011.
   Ca

- Prevention Strategy. 2004. www.casp.acps.ca/Publications/BlueprintFINAL.pdf; accessed January 17, 2011. MacMillan HL, MacMillan AB, Offord DR, Dingle JL. Aboriginal health. CMAJ. December 1, 1996;155(11):1569–1578. www.ncbi.nlm.nih.gov/pmc/articles/pmid/8956834/?tool=pubmed; accessed January 17, 2011. Ramey HL, Busseri MA, Khanna N, Hamilton YN, Ottawa YN, Rose-Krasnor L. Youth engagement and suicide risk: testing a mediated model in a Canadian community sample. J Youth Adolesc. March 2010;39(3):243–258. Kral M, Wiebe PK, Nisbet K, Dallas C, Okalik L, Enuaraq N, Cinotta J. Canadian Inuit community engagement in suicide prevention. Int J Circumpolar Health. June 2009;68(3):292–308. www.ijch. fi/show\_abstract.php?abstract\_id=813; accessed January 17, 2011. Hazell P. Adolescent suicide clusters: evidence, mechanisms and prevention. Aust N Z J Psychiatry. 1993;27(4):653-65. 954

- Nepon J, Fotti S, Katz LY, Sareen J. Media guidelines for reporting suicide. Canadian Psychiatric Association Policy Paper. 2009. publications.cpa-apc.org/media.php?mid=733&xwm=true; accessed January 17, 2011. 957

- <sup>958</sup> Health Canada. Acting on What We Know: Preventing Youth Suicide in First Nations. www.hcsc.gc.ca/fniah-spnia/pubs/promotion/\_suicide/prev\_youth-jeunes/index-eng.php; accessed January 17, 2011.
   <sup>959</sup> ThinkFirst Foundation of Canada. www.thinkfirst.ca; accessed January 17, 2011.
   <sup>960</sup> Asbridge M, Mann RE, Flam-Zalcman R, Stoduto G. The criminalization of impaired driving in Canada: assessing the deterrent impact of Canada's first per se law. J Stud Alcohol. July 2004;65(4):450–459.
   <sup>961</sup> Ontario College of Family Physicians. The Health Impacts of Urban Sprawl. Volume Two: Road Injuries and Fatalities. 2005. www.ocfp.on.ca/local/files/Urban%20Sprawl/UrbanSprawlRoadInjuries.pdf; accessed January 17, 2011.
   <sup>962</sup> MacMillan HL, Wathen CN, Jamieson E, Boyle MH, Shannon HS, Ford-Gilboe M, Worster A, Lent B, Coben JH, Campbell JC, McNutt LA; McMaster Violence Against Women Research Group. Screening for intimate partner violence in health care settings: a randomized trial. JAMA. August 5, 2009;302(5):493–501.
   <sup>963</sup> MacMillan HL, Wathen CN, Jamieson E, Boyle M, McNutt LA, Worster A, Lent B, Webb M; McMaster Violence Against Women Research Group. Approaches to screening for intimate partner violence in health care settings: a randomized trial. JAMA. August 2, 2006;296(5):530–536.
   <sup>964</sup> Snider C, Webster D, O'Sullivan CS, Campbel J. Intimate partner violence: development of a brief risk assessment for the emergency department. Acad Emerg Med. November 2009;16(11):1208–1216.
   <sup>965</sup> Jack SM, Jamieson E, Jamillan HL. The feasibility of screening for intimate partner violence during postpartum home visits. Can J Nurs Res. June 2008;40(2):150–170.
   <sup>966</sup> MacKina energency departments. CJEM. July 2008;10(4):325–328.
   <sup>976</sup> Gardner HG, Americaa Academy of Pediatrics Committee on Injury, Violence, and Poison Prevention. Office based counseling for unintentional injury prevention. Pediatrics. January 20

- January 2007;119(1):202–206. Cancer Care Ontario. Ontario Cancer Plan 2011-2015. ocp.cancercare.on.ca; accessed December 23, 2010. Center for Prevention and Health Services. Eliminating Racial and Ethnic Health Disparities: A Business Case Update for Employers. February 2009. www.minorityhealth.hhs.gov/Assets/pdf/ checked/1/Eliminating\_Racial\_Ethnic\_Health\_Disparities\_A\_Business\_Case\_Update\_for\_ Employers.pdf; accessed December 10, 2010. Ministry of Health Promotion and Sport. ACTIVE2010: Ontario's Sport and Physical Activity Strategy. 2005. www.mhp.gov.on.ca/en/active-living/about/active2010-strategy-e.pdf; accessed January 12, 2011. Strong WB, et al. Evidence-based physical activity for school-age youth. Journal of Pediatrics. 2005;146:6:732–737.
- 970
- 971
- 2005;146:6:732–737. FoodShare. The Good Food Box. www.foodshare.ca; accessed January 12, 2011. The power to reduce health disparities: voices from REACH communities. U.S. Department of Health and Human Services, Centers for Disease Control and Prevention, Attanta, 2007. www.cdc.gov/react/pdf/voices\_101007.pdf; accessed January 12, 2011. Rachlis M. Operationalizing Health Equity: How Ontario's Health Services Can Contribute To Reducing Health Disparities. Wellesley Institute. 2008. wellesleyinstitute.com/files/ Operationalizing HealthEquity.pdf; accessed January 12, 2011. Rachlis M. Operationalizing HealthEquity: How Ontario's Health Services Can Contribute To Reducing Health Disparities. Wellesley Institute. 2008. wellesleyinstitute.com/files/ Operationalizing HealthEquity.pdf; accessed January 12, 2011. 973

- <sup>976</sup> Unison Health and Community Services. unisonhcs.org; accessed January 17, 2011.
   <sup>977</sup> Orpana HM, Lemyre L, Gravel R. Income and psychological distress: The role of the social environment. Statistics Canada. www.statcan.gc.ca/pub/82-003-x/2009001/article/10772-eng.htm;
- environment. Statistics Canada. www.statcan.gc.ca/pub/82-003-x/2009001/article/10772-eng.htm; accessed January 12, 2011. Krueger PM, Chang WU, Being poor and coping with stress: health behaviors and the risk of death. Am J Public Health. May 2008;98(5):889–896. www.ajph.org/cgi/pmidlookup?view=long&p mid=18382003; accessed January 12, 2011. Abercrombie PD, Zamora A, Korn AP. Lessons learned: providing a mindfulness-based stress reduction program for low-income multiethnic women with abnormal pap smears. Holist Nurs Pract. 2007 Jan-Feb;21(1):26-34. Crouch RB. A community-based stress management programme for an impoverished population in South Africa. Occup Ther Int. 2008;15(2):71-86. Canadian Public Health Association. Health Literacy Interventions. 2006. www.cpha.ca/uploads/ nortals/hI/interventions e.odf: accessed January 12, 2011.

- 981

- <sup>981</sup> Canadian Public Health Association. Health Literacy Interventions. 2006. www.cpha.ca/uploads/ portals/h1/interventions.e.pdf; accessed January 12, 2011.
   <sup>981</sup> van Assema P, Brug J, Glanz K, Dolders M, Mudde A. Nationwide implementation of guided supermarket tours in The Netherlands: a dissemination study. Health Educ Res. 1998 Dec;13(4):557-66.
   <sup>983</sup> Brown BJ, Hermann JR. Cooking classes increase fruit and vegetable intake and food safety behaviors in youth and adults.J Nutr Educ Behav. 2005;37(2):104-5.
   <sup>984</sup> Coleman MT, Newton KS. Supporting self-management in patients with chronic illness. Am Fam Physician. October 15, 2005;72(8):1503–1510. www.aafp.org/link\_out2pmid=16273817; accessed January 17, 2011.
   <sup>985</sup> Bodnheimer T, Lorig K, Holman H, Grumbach K. Patient Self-management of Chronic Disease in Primary Care. JAMA. 2002;288(19):2469–2475.
   <sup>984</sup> Lorig K, Sobel D, Stewart AL, Brown BW, Andurap AB, Ritter P, Gonzalez VM, Laurent DD, Holma HR. Evidence Suggesting That a Chronic Disease Self-Management Toorgarm Can Improve Health Status while Reducing Hospitalization: A Randomized Trial. Med Care. 1999;37(1):5–14.
   <sup>987</sup> Coleman MT, Newton KS. Supporting self-management in patients with chronic illness. 987
- Health Status while Reducing Hospitalization: A Randomized Trial. Med Care. 1999;37(1):5–14. Coleman MT, Newton KS. Supporting self-management in patients with chronic illness. Am Fam Physician. October 15, 2005;72(8):1503–1510. Glazier RH, Booth GL, Gozdyra P, Creatore MI, Tynan M, editors. Neighbourhood Environments and Resources for Healthy Living—A Focus on Diabetes in Toronto: ICES Atlas. Toronto: Institute for Clinical Evaluative Sciences; 2007. www.ices.on.ca/webpage.cfm?site\_id=1&org\_id=67&morg\_ id=0&gsec\_id=0&item\_id=4406&type=atlas; accessed January 12, 2011. Stead M, MacAskill S, MacKintosh AM, Reece J, Eadie D. "It's as if you're locked in": qualitative explanations for area effects on smoking in disadvantaged communities. Health Place. December 2001;7(4):333–343.

- December 2001;7(4):333–343.
   The Canadian PD Project. Website of the Canadian Positive Deviance (PD) Project. Dec 23, 2010.
   www.positivedeviance.ca; accessed April 9, 2011.
   Canadian Agricultural Injury Surveillance Program. Agricultural Fatalities and Hospitalizations in Ontario 1990-2004. http://cairsbac.ca/ontull.pdf; accessed April 9, 2011.
   Canadian Institute for Health Information. All-Terrain Vehicle (ATV-related Trauma Hospitalizations in Ontario, 2001-2002. http://secure.cihi.ca/cihiweb/en/downloads/bl\_otr\_ATV\_hospitalizations\_dec2003.e.pdf; accessed April 9, 2011.
   Farm Safety Association. Farm Accident Rescue. Sept 1985. www.farmsafety.ca/factsheets/ tips-e/farm\_acidenten.PDF; accessed April 9, 2011.
   Toronto Central LHIN. Health Equity. www.torontocentrallhin.on.ca/Page.aspx?id=4396; accessed March 7, 2011.

# 14 Acknowledgements

Development of this report was led by a scientific team from Health Quality Ontario (HQO) including Ben Chan, Rebecca Comrie, Michelle Rey, Imtiaz Daniel and Geoff Anderson, and a project management/epidemiology/communications team from HQO that included Maggie Chen, Katherine McLaughlin, Ryan Emond, Ja Young Kim, Sisi Wang, Brad Kim, Laura Corbett, Wilson Kwong, Susan Brien, Céline St-Louis, Colin Longhurst, Ivan Langrish, Suzanne Dugard and Nilam Kassam. Analyses were also provided by Astrid Guttmann, Chad Leaver and Jun Guan from the Institute for Clinical Evaluative Sciences (ICES).

Health Quality Ontario acknowledges and thanks the many dedicated individuals who contributed to this report, including:

- Senior Research and Methodological Advisors, Geoff Anderson and Jeff Poss.
- HQO's Performance Measurement Advisory Board, a group of research and measurement experts from around the province who provided advice on the selection of indicators: Imtiaz Daniel (chair), Arlene Bierman, Geoff Anderson, Sten Ardal, Suzanne Dionne, Faith Donald, Alan Forster, Bob Gardner, Gillian Hawker, Cam Mustard, Raymond Pong, Walter Rosser, Carol Sawka, Katya Duvalko, Kaveh Shojania, Sam Shortt, Eugene Wen and Michael Wolfson.
- HQO's Performance Measurement Peer Review Panel, a group of research and measurement experts from around the province who provided advice on all quantitative research and analysis: Geoff Anderson, Helen Angus, Sten Ardel, Chaim Bell, Jennifer Bennie, Arlene Bierman, Patti Cochrane, Suzanne Dionne, Faith Donald, Katya Duvalko, Kevin Empey, Alan Forster, Michael Gardam, Bob Gardner, Amir Ginzburg, Andrea Gruneir, Astrid Guttman, Christey Hackney, Simon Hagens, Gillian Hawker, Sherrie Hertz, Luidmila Husak, Jon Irish, Anthony Jonker, Maureen Kelly, Kori Kingsbury, Paul Kurdyak, Hussein Lalani, Elizabeth Lin, Ian McKillop, Charlotte Moore, Cam Mustard, Adam Nagler, Howard Ovens, Raymond Pong, Jeff Poss, Dan Purdham, Paula Rochon, Walter Rosser, Carol Sawka, Lloy Schindeler, Michael Schull, Baiju Shah, Kaveh Shojania, Sam Shortt, Kevin Smith, Ann Sprague, Terry Sullivan, Beth Thesis, Jack Tu, Tamara Wallington, Eugene Wen, Walter Wodchis and Vandad Yousefi.
- The following organizations, which provided reviews or data for the report: BORN Ontario, Canadian Institute for Health Information, Cancer Care Ontario, College of Nurses of Ontario, College of Physicians and Surgeons of Ontario, Commonwealth Fund, HIMSS Analytics, NRC-Picker, Ontario Hospital Association, OntarioMD, Ontario Physician Human Resources Data Centre, Ontario Public Health Laboratories, Organisation for Economic Co-operation and Development, Statistics Canada and the Workplace Safety and Insurance Board.
- Leaders and staff from the following organizations who provided information for the examples of success: Collingwood General and Marine, Georgetown Hospital (Halton Healthcare Services), Halton Healthcare Services Corporation — Oakville, London Health Sciences Centre — University Hospital Site, Mount Sinai Hospital, North York FHT, North York General Hospital, Oakville-Trafalgar Memorial Hospital (Halton Healthcare Services), Ottawa Heart Institute, Petawawa Centennial FHT, Smithville Medical Centre FHT, St. Joseph's Health Care — London, St. Thomas — Elgin General Hospital, Timmins FHT and Windsor Regional Hospital.
- Staff at multiple divisions and branches of the Ministry of Health and Long-Term Care for supplying data and background information and verifying facts.
- Alison MacAlpine, and Kelly Lang, Jess Rogers and Tupper Bean from the Centre for Effective Practice who were instrumental in writing the report.

# **15** Members of Health Quality Ontario

### Board

Lyn McLeod (Newmarket)	Chair, Health Quality Ontario Interim Chair, Governance and Nominations Chair, Management and Resources Committee				
Gilbert Sharpe (Toronto)	Member, Management and Resources Committee				
Richard Alvarez (Toronto)					
Andy Molino (Ottawa)	Chair, Audit & Resources Committee Member, Management and Resources Committee				
Dr. Arlene Bierman (Toronto)	Member, Performance Measurement Advisory Board				
Dr. André Hurtubise (New Liskeard)	Member, Governance and Nominations				
Faith Donald (Toronto)	Member, Performance Measurement Advisory Board Member, Audit & Resources Committee				
Bob Gardner (Toronto)	Member, Performance Measurement Advisory Board				
Biographies are posted at www.hqontario.ca/en/governance.php.					

Management

Dr. Ben Chan	President & Chief Executive Officer
Nizar Ladak	Vice President & Chief Operating Officer
Eileen Patterson	Vice President, Quality Improvement
Harpreet Bassi	Director, Strategic Projects
Céline St-Louis	Director, Communications

### Mandate

The functions of Health Quality Ontario are:

- (a) to monitor and report to the people of Ontario on,
  - (i) access to publicly funded health services,
  - (ii) health human resources in publicly funded health services,
  - (iii) consumer and population health status, and
  - (iv) health system outcomes;
- (b) to support continuous quality improvement;
- (c) to promote healthcare that is supported by the best available scientific evidence by,
  - (i) making recommendations to healthcare organizations and other entities on standards of care in the health system, based on or respecting clinical practice guidelines and protocols, and
  - (ii) making recommendations, based on evidence and with consideration of the recommendations in subclause (i), to the Minister concerning the Government of Ontario's provision of funding for healthcare services and medical devices.

**HEALTH QUALITY ONTARIO** 

130 Bloor Street West, Suite 702 Toronto, ON M5S 1N5

Telephone: 416.323.6868 Toll-free: 1.866.623.6868 Fax: 416.323.9261 Email: info@hqontario.ca www.hqontario.ca

ISSN 1926-2191 (Print) ISBN 978-1-4435-5554-8 (Print, 2011 ed.) ISSN 1925-7015 (Online) ISBN 978-1-4435-5555-5 (HTML, 2011 ed.) ISSN 1925-7015 (Online) ISBN 978-1-4435-5556-2 (PDF, 2011 ed.) ©Queen's Printer for Ontario 2011

